Huntsman International LLC

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

		Form	10-Q			
(Mark One)						
	QUARTERLY REPORT P EXCHANGE ACT OF 193		CTION 13 OR 1	15(d) OF THE SEC	URITIES	
	Fo	or the quarterly period	ended March 31,	2010		
		OR	1			
	TRANSITION REPORT P EXCHANGE ACT OF 193		CTION 13 OR 1	5(d) OF THE SEC	URITIES	
	For the trans	ition period from	to			
Commission File Number 001-32427		ant as Specified in its Charter, ess and Telephone Number	Principal Office	State of Incorpora Delay		
333-85141	Salt Lake City, Utah 84108 (801) 584-5700 Huntsman International LLC 500 Huntsman Way Salt Lake City, Utah 84108 (801) 584-5700			Delav	ware 87-0€	530358
Exchange A	e by check mark whether the regiset of 1934 during the preceding 12 subject to such filing requirement	2 months (or for such sho				
Huntsman C Huntsman Ir	orporation aternational LLC				YES ☒ YES ☒	NO □ NO □
Interactive D	e by check mark whether the regis lata File required to be submitted a period that the registrant was requ	and posted pursuant to R	ule 405 of Regulat			
Huntsman C Huntsman Ir	orporation tternational LLC				YES □ YES □	NO □ NO □
	e by check mark whether the regis mpany. See definition of "accelera					
Huntsman Corpo Huntsman Intern		Large accelerated filer ⊠ Large accelerated filer □	Accelerated filer □ Accelerated filer □	Non-accelerated filer □ Non-accelerated filer ☑	Smaller reporting co	
Indicat	e by check mark whether the regis	strant is a shell company	(as defined in Rule	12b-2 of the Exchange	Act).	
Huntsman C	orporation				YES □	NO 🗷

YES □

NO 🗷

On April 27, 2010, 239,059,793 shares of common stock of Huntsman Corporation were outstanding and 2,728 units of membership interests of Huntsman International LLC were outstanding. There is no established trading market for Huntsman International LLC's units of membership interests. All of Huntsman International LLC's units of membership interests are held by Huntsman Corporation.

This Quarterly Report on Form 10-Q presents information for two registrants: Huntsman Corporation and Huntsman International LLC. Huntsman International LLC is a wholly owned subsidiary of Huntsman Corporation and is the principal operating company of Huntsman Corporation. The information reflected in this Quarterly Report on Form 10-Q is equally applicable to both Huntsman Corporation and Huntsman International LLC, except where otherwise indicated. Huntsman International LLC meets the conditions set forth in General Instructions H(1)(a) and (b) of Form 10-Q and, to the extent applicable, is therefore filing this form with a reduced disclosure format.

HUNTSMAN CORPORATION AND SUBSIDIARIES HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES QUARTERLY REPORT ON FORM 10-Q FOR THE QUARTERLY PERIOD ENDED MARCH 31, 2010

TABLE OF CONTENTS

		Page
PART I	FINANCIAL INFORMATION	<u>3</u>
<u>ITEM 1.</u>	Financial Statements:	3
	Huntsman Corporation and Subsidiaries:	3
	Condensed Consolidated Balance Sheets (Unaudited)	3 3 3 3
	Condensed Consolidated Statements of Operations and Comprehensive Loss	
	(Unaudited)	<u>4</u>
	Condensed Consolidated Statements of Cash Flows (Unaudited)	4 6 8 9
	Condensed Consolidated Statements of Equity (Unaudited)	<u>8</u>
	Huntsman International LLC and Subsidiaries:	<u>9</u>
	Condensed Consolidated Balance Sheets (Unaudited)	<u>9</u>
	Condensed Consolidated Statements of Operations and Comprehensive Loss	
	(Unaudited)	<u>10</u>
	Condensed Consolidated Statements of Cash Flows (Unaudited)	<u>11</u>
	Condensed Consolidated Statements of Equity (Unaudited)	<u>13</u>
	Huntsman Corporation and Subsidiaries and Huntsman International LLC and Subsidiaries:	<u>14</u>
	Notes to Condensed Consolidated Financial Statements (Unaudited)	<u>14</u>
ITEM 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	69
ITEM 3.	Quantitative and Qualitative Disclosures About Market Risk	<u>87</u>
ITEM 4.	Controls and Procedures	88
PART II	OTHER INFORMATION	89
ITEM 1.	Legal Proceedings	<u>89</u>
ITEM 1A.	Risk Factors	11 13 14 14 69 87 88 89 89
ITEM 2.	Unregistered Sales of Equity Securities and Use of Proceeds	<u>89</u>
<u>ITEM 6.</u>	<u>Exhibits</u>	<u>90</u>

PART I. FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS

HUNTSMAN CORPORATION AND SUBSIDIARIES

CONDENSED CONSOLIDATED BALANCE SHEETS (UNAUDITED)

(In Millions, Except Share and Per Share Amounts)

		arch 31, 2010	Dec	cember 31, 2009
ASSETS				
Current assets:				
Cash and cash equivalents	\$	1,110	\$	1,745
Restricted cash		8		5
Accounts and notes receivable (net of allowance for doubtful accounts of \$55 and \$56, respectively), (\$538				
and nil pledged as collateral, respectively)		1,371		1,018
Accounts receivable from affiliates		12		1
Inventories		1,248		1,184
Prepaid expenses		36		42
Deferred income taxes		36		36
Other current assets		132		109
Total current assets		3,953		4,140
Property, plant and equipment, net		3,373		3,516
Investment in unconsolidated affiliates		252		250
Intangible assets, net		120		125
Goodwill		94		94
Deferred income taxes		154		138
Notes receivable from affiliates		7		8
Other noncurrent assets		434		355
	_		_	
Total assets	\$	8,387	\$	8,626
LIABILITIES AND EQUITY				
Current liabilities:				
Accounts payable	\$	875	\$	730
Accounts payable to affiliates		23		25
Accrued liabilities		564		623
Deferred income taxes		2		2
Current portion of debt		316		431
Total current liabilities		1,780		1.811
Long-term debt		3,859		3,781
Notes payable to affiliates		5,059		5,761
Deferred income taxes		289		289
Other noncurrent liabilities		816		875
	_	010	_	073
Total liabilities		6,749		6,761
Commitments and contingencies (Notes 13 and 14)		0,749		0,701
Equity				
Huntsman Corporation stockholders' equity:				
Common stock \$0.01 par value, 1,200,000,000 shares authorized, 239,066,356 and 237,225,258 issued and				
236,323,524 and 234,081,490 outstanding in 2010 and 2009, respectively		2		2
Additional paid-in capital		3,181		3,155
Unearned stock-based compensation				,
Accumulated deficit		(18)		(11) (1,015)
Accumulated deficit Accumulated other comprehensive loss		(1,217)		
	_	(331)	_	(287)
Total Huntsman Corporation stockholders' equity		1,617		1,844
Noncontrolling interests in subsidiaries		21		21
Total equity		1,638		1,865
Total liabilities and equity	\$	8,387	\$	8,626
Total nationales and equity	φ	0,50/	φ	0,020

CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE LOSS (UNAUDITED)

(In Millions, Except Per Share Amounts)

	Three months ended			ths
	March 31,			1.
		2010		2009
Revenues:				
Trade sales, services and fees, net	\$	2,049	\$	1,673
Related party sales		45		7
Total revenues		2,094		1,680
Cost of goods sold		1,813		1,531
Gross profit		281		149
Operating expenses:				
Selling, general and administrative		218		194
Research and development		36		36
Other operating expense (income)		2		(8)
Restructuring, impairment and plant closing costs		3		14
Total expenses		259		236
Operating income (loss)		22		(87)
Interest expense, net		(61)		(55)
Loss on accounts receivable securitization program		_		(4)
Equity in income of investment in unconsolidated affiliates		1		1
Loss on early extinguishment of debt		(155)		—
Expenses associated with the Terminated Merger and related litigation		_		(7)
Loss from continuing operations before income taxes		(193)		(152)
Income tax benefit (expense)		34		(138)
Loss from continuing operations		(159)		(290)
Loss from discontinued operations, net of tax		(13)		(4)
Net loss		(172)		(294)
Net loss attributable to noncontrolling interests		_		4
Net loss attributable to Huntsman Corporation	\$	(172)	\$	(290)
Net loss	\$	(172)	\$	(294)
Other comprehensive loss		(44)		(82)
Comprehensive loss		(216)		(376)
Comprehensive loss attributable to noncontrolling interests		_		4
Comprehensive loss attributable to Huntsman Corporation	\$	(216)	\$	(372)

(continued)

CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE LOSS (UNAUDITED) (Continued)

(In Millions, Except Per Share Amounts)

	Three months ended March 31,
	2010 2009
Basic loss per share:	
Loss from continuing operations attributable to Huntsman Corporation common stockholders	\$ (0.68) \$ (1.22)
Loss from discontinued operations attributable to Huntsman Corporation common stockholders, net of tax	(0.05) (0.02)
Net loss attributable to Huntsman Corporation common stockholders	\$ (0.73) \$ (1.24)
Weighted average shares	234.8 233.7
Diluted loss per share:	
Loss from continuing operations attributable to Huntsman Corporation common stockholders	\$ (0.68) \$ (1.22)
Loss from discontinued operations attributable to Huntsman Corporation common stockholders, net of tax	(0.05) (0.02)
Net loss attributable to Huntsman Corporation common stockholders	\$ (0.73) \$ (1.24)
Weighted average shares	234.8 233.7
Amounts attributable to Huntsman Corporation common stockholders:	
Loss from continuing operations	\$ (159) \$ (286)
Loss from discontinued operations, net of tax	(13) (4)
Net loss	\$ (172) \$ (290)
Dividends per share	\$ 0.10 \$ 0.10

CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED)

(Dollars in Millions)

	Three months ended March 31,	
Operating Activities:	2010	2009
Net loss	\$ (172)	\$ (294)
Adjustments to reconcile net loss to net cash used in operating activities:	Ψ (1/2)	Ψ (2)+)
Equity in income of investment in unconsolidated affiliates	(1)	(1)
Depreciation and amortization	98	126
Provision for losses on accounts receivable	3	4
Loss (gain) on disposal of businesses/assets, net	1	(1)
Loss on early extinguishment of debt	155	_
Noncash interest expense	10	2
Noncash restructuring, impairment and plant closing costs	_	4
Deferred income taxes	(20)	131
Net unrealized (gain) loss on foreign currency transactions	(3)	2
Stock-based compensation	7	5
Other, net	(2)	(1)
Changes in operating assets and liabilities:		
Accounts and notes receivable	(114)	57
Accounts receivable from A/R Programs (Note 7)	(254)	_
Inventories	(96)	114
Prepaid expenses	6	7
Other current assets	(25)	1
Other noncurrent assets	(77)	_
Accounts payable	153	(113)
Accrued liabilities	(56)	(71)
Other noncurrent liabilities	(21)	(10)
Net cash used in operating activities	(408)	(38)
Investing Activities:		
Capital expenditures	(37)	(61)
Proceeds from sale of businesses/assets, net of adjustments		(1)
Investment in unconsolidated affiliates, net of cash received	(3)	
Change in restricted cash	(3)	
Net cash used in investing activities	(43)	(62)

(continued)

CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED) (Continued)

(Dollars in Millions)

	Three m end Marcl	ed h 31,
Financing Activities	2010	2009
Financing Activities: Net borrowings (repayments) under revolving loan facilities	\$ 3	\$ (8)
Revolving loan facility from A/R Programs (Note 7)	254	\$ (6)
Net repayments on overdraft facilities	(1)	(1)
Repayments of short-term debt	(44)	(73)
Borrowings on short-term debt	55	47
Repayments of long-term debt	(611)	(17)
Proceeds from issuance of long-term debt	375	
Repayments of notes payable	(21)	(23)
Borrowings on notes payable		14
Debt issuance costs paid	(16)	_
Call premiums related to early extinguishment of debt	(153)	
Dividends paid to common stockholders	(24)	
Repurchase and cancellation of stock awards	(6)	_
Proceeds from issuance of common stock	2	—
Excess tax benefit related to stock-based compensation	4	_
Other, net	(1)	
Net cash used in financing activities	(184)	
Effect of exchange rate changes on cash		(3)
Decrease in cash and cash equivalents	(635)	(187)
Cash and cash equivalents at beginning of period	1,745	657
Cash and cash equivalents at end of period	\$ 1,110	\$ 470
Supplemental cash flow information:		
Cash paid for interest	\$ 43	\$ 38
Cash paid for income taxes	8	17

During the three months ended March 31, 2010 and 2009, the amount of capital expenditures in accounts payable decreased by \$11 million and \$24 million, respectively. The value of share awards that vested during the three months ended March 31, 2010 and 2009 was \$18 million and \$11 million, respectively.

CONDENSED CONSOLIDATED STATEMENTS OF EQUITY (UNAUDITED)

(Dollars in Millions)

	Huntsman Corporation Stockholders							
	Common Stock Shares Amount		•		Accumulated other Accumulated comprehensive deficit loss		Noncontrolling interests in subsidiaries	Total equity
Balance, January 1,								
2010 Net loss	234,081,490	\$ 2	\$ 3,133	\$ (11)			\$ 21	\$1,865
Other	_		_		(172)	_	_	(172)
comprehensive loss	_	_	_	_	_	(44)	_	(44)
Issuance of nonvested			4.0	(10)				
stock awards			10	(10)				_
Vesting of stock awards	1,803,817	_	9	_	_	_	_	9
Recognition of stock-based compensation	_		1	3	_	_	_	4
Repurchase and cancellation of			-	Ü				
stock awards	(425,001)	_	_	_	(6)	_	_	(6)
Stock options exercised	863,218	_	2	_	_	_	_	2
Excess tax benefit related to stock-based compensation	_	_	4	_	_	_	_	4
Dividends declared on common stock	_	_	_	_	(24)	_	_	(24)
Balance, March 31, 2010	236,323,524	\$ 2	\$ 3,181	\$ (18)	\$ (1,217)	\$ (331)	\$ 21	\$ 1,638

	Huntsman Corporation Stockholders							
	Common St		Additional paid-in	Unearned stock-based	Accumulated	Accumulated other comprehensive	Noncontrolling interests in	Total
D 1	Shares	Amount	capital	compensation	deficit	loss	subsidiaries	equity
Balance,								
January 1,	222 552 515	Φ 2	Φ 2 1 4 1	Φ (12)	Φ (1.0 2.1)	Φ (400)	Ф 22	Ф 1 (22
2009	233,553,515	\$ 2	\$ 3,141	\$ (13)				\$ 1,632
Net loss					(290)	_	(4)	(294)
Other								
comprehensive								
loss	_	_	_	_	_	(82)	_	(82)
Issuance of								
nonvested								
stock awards		_	8	(8)	_	_		
Vesting of stock								
awards	538,225	_	_	_	_	_	_	_
Recognition of								
stock-based								
compensation	_	_	2	3	_	_		5
Repurchase and								
cancellation of								
stock awards	(132,635)	· —	_	_	_	_	_	_
Dividends								
declared on								
common stock	_		_	_	(23)	_	_	(23)
Balance,								

March 31, 2009 233,959,105 \$ 2 \$ 3,151 \$ (18)\$ (1,344)\$ (571)\$ 18 \$1,238

CONDENSED CONSOLIDATED BALANCE SHEETS (UNAUDITED)

(Dollars in Millions)

	March 31, 2010		· · · · · · · · · · · · · · · · · · ·	
ASSETS				
Current assets:				
Cash and cash equivalents	\$	551	\$	919
Restricted cash		8		5
Accounts and notes receivable (net of allowance for doubtful accounts of \$55 and \$56, respectively), (\$538				
and nil pledged as collateral, respectively)		1,371		1,018
Accounts receivable from affiliates		76		32
Inventories		1,248		1,184
Prepaid expenses		34		42
Deferred income taxes		34		33
Other current assets		122		109
Total current assets		3,444		3,342
Property, plant and equipment, net		3,220		3,357
Investment in unconsolidated affiliates		252		250
Intangible assets, net		123		129
Goodwill		94		94
Deferred income taxes		175		158
Notes receivable from affiliates		7		8
Other noncurrent assets		434		355
Total assets	\$	7,749	\$	7,693
LIABILITIES AND EQUITY				
Current liabilities:				
Accounts payable	\$	873	\$	715
Accounts payable to affiliates		22		41
Accrued liabilities		562		613
Deferred income taxes		2		2
Note payable to affiliate		100		25
Current portion of debt		316		195
Total current liabilities		1,875		1,591
Long-term debt		3,859		3,781
Notes payable to affiliates		330		530
Deferred income taxes		82		79
Other noncurrent liabilities		815		865
Total liabilities		6,961		6,846
Commitments and contingencies (Notes 13 and 14)		0,901		0,040
Equity				
Huntsman International LLC members' equity:				
Members' equity, 2,728 units issued and outstanding		3.031		3,021
Accumulated deficit		(1,873)		(1,847)
Accumulated other comprehensive loss		(391)		(348)
Total Huntsman International LLC members' equity	_	767		826
Noncontrolling interests in subsidiaries		21		826
Total equity	_	788		847
	\$		e.	
Total liabilities and equity	2	7,749	\$	7,693

CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE LOSS (UNAUDITED)

(Dollars in Millions)

		Three months ended March 31,		
	2010		2009	
Revenues:				
Trade sales, services and fees, net	\$ 2,04		5 1,673	
Related party sales	4	5	7	
Total revenues	2,09	1	1,680	
Cost of goods sold	1,80	3	1,526	
Gross profit	28	5	154	
Operating expenses:				
Selling, general and administrative	21	5	189	
Research and development	3	5	36	
Other operating income	(7)	(8)	
Restructuring, impairment and plant closing costs		3	14	
Total expenses	24	7	231	
Operating income (loss)	3)	(77)	
Interest expense, net	(6	5)	(55)	
Loss on accounts receivable securitization program	_	_	(4)	
Equity in income of investment in unconsolidated affiliates		1	1	
Loss on early extinguishment of debt	(9)	_	
Loss from continuing operations before income taxes	(3	5)	(135)	
Income tax benefit (expense)	2	2	(146)	
Loss from continuing operations	(1	3)	(281)	
Loss from discontinued operations, net of tax	(1	3)	(4)	
Net loss	(2	5)	(285)	
Net loss attributable to noncontrolling interests	_	_	4	
Net loss attributable to Huntsman International LLC	\$ (2	5) \$	(281)	
Net loss	\$ (2	5) \$	(285)	
Other comprehensive loss	(4	3)	(79)	
Compush ansiya logg	(6	9)	(364)	
Comprehensive loss Comprehensive loss attributable to noncontrolling interests	_		4	
Comprehensive loss attributable to Huntsman International LLC	\$ (6	9) §		
Comprehensive loss attributable to munisman international LEC	\$ (0	<i>)</i>	(300)	

CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED)

(Dollars in Millions)

	Three months ended		
	Marc		
	2010	2009	
Operating Activities:			
Net loss	\$ (26)	\$ (285)	
Adjustments to reconcile net loss to net cash used in operating activities:			
Equity in income of investment in unconsolidated affiliates	(1)	(1)	
Depreciation and amortization	92	120	
Provision for losses on accounts receivable	3	4	
Loss (gain) on disposal of businesses/assets, net	1	(1)	
Loss on early extinguishment of debt	9	_	
Noncash interest expense	14	5	
Noncash restructuring, impairment and plant closing costs	_	4	
Deferred income taxes	(18)	143	
Net unrealized (gain) loss on foreign currency transactions	(3)	2	
Noncash compensation	6	1	
Changes in operating assets and liabilities:			
Accounts and notes receivable	(114)	57	
Accounts receivable from A/R Programs (Note 7)	(254)	_	
Inventories	(96)	114	
Prepaid expenses	8	10	
Other current assets	(14)	1	
Other noncurrent assets	(77)	_	
Accounts payable	161	(128)	
Accrued liabilities	(47)	(39)	
Other noncurrent liabilities	(20)	(10)	
Net cash used in operating activities	(376)	(3)	
Investing Activities:			
Capital expenditures	(37)	(61)	
Proceeds from sale of businesses/assets, net of adjustments	_	(1)	
Investment in unconsolidated affiliates, net of cash received	(3)		
Change in restricted cash	(3)	_	
(Increase) decrease in receivable from affiliate	(50)	7	
Net cash used in investing activities	(93)	(55)	

(continued)

CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED) (Continued)

(Dollars in Millions)

		e months nded
		rch 31,
	2010	2009
Financing Activities:		
Net borrowings (repayments) under revolving loan facilities	\$ 3	+ (-)
Revolving loan facility from A/R Programs (Note 7)	254	
Net repayments on overdraft facilities	(1	, ()
Repayments of short-term debt	(44	, ()
Borrowings on short-term debt	55	
Repayments of long-term debt	(375	, , ,
Proceeds from issuance of long-term debt	375	
Repayments of notes payable to affiliate	(125	,
Proceeds from notes payable to affiliate	_	- 529
Repayments of notes payable	(21	, ,
Borrowings on notes payable	_	- 11
Debt issuance costs paid	(16	,
Call premiums related to early extinguishment of debt	(7	
Dividends paid to parent	_	- (23)
Excess tax benefit related to stock-based compensation	4	
Other, net	(1	.) 1
Net cash provided by financing activities	101	444
Effect of exchange rate changes on cash	_	(3)
(Decrease) increase in cash and cash equivalents	(368	383
Cash and cash equivalents at beginning of period	919	87
Cash and cash equivalents at end of period	\$ 551	\$ 470
Supplemental cash flow information:		
Cash paid for interest	\$ 34	
Cash paid for income taxes	ϵ	5 17

During the three months ended March 31, 2010 and 2009, the amount of capital expenditures in accounts payable decreased by \$11 million and \$24 million, respectively. During the three months ended March 31, 2010 and 2009, Huntsman Corporation contributed \$6 million and \$1 million, respectively, related to stock-based compensation.

CONDENSED CONSOLIDATED STATEMENTS OF EQUITY (UNAUDITED)

(Dollars in Millions)

		Huntsman In					
	Membe Units	rs' equity Amount	A	ccumulated deficit	Accumulated other comprehensive loss	Noncontrolling interests in subsidiaries	Total quity
Balance, January 1, 2010	2,728	\$ 3,021	\$	(1,847)	\$ (348)	\$ 21	\$ 847
Net loss	_	_		(26)	_		(26)
Other comprehensive loss	_	_		_	(43)	_	(43)
Contribution from parent, net of distributions	_	6		_	_	_	6
Excess tax benefit related to stock-based compensation	_	4		_	_	_	4
Balance, March 31, 2010	2,728	\$ 3,031	\$	(1,873)	\$ (391)	\$ 21	\$ 788

		Huntsman Ir				
	Membe	rs' equity Amount	Accumulated deficit	Accumulated other comprehensive loss	Noncontrolling interests in subsidiaries	Total equity
Balance, January 1, 2009	2,728	\$ 2,865	\$ (1,414)	\$ (554)	\$ 22	\$ 919
Net loss	_	_	(281)	_	(4)	(285)
Other comprehensive loss	_	_	_	(79)	_	(79)
Contribution from parent, net of distributions	_	1	_	_	_	1
Dividends paid to parent	_	_	(23)	_	_	(23)
Balance, March 31, 2009	2,728	\$ 2,866	\$ (1,718)	\$ (633)	\$ 18	\$ 533

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

1. GENERAL

CERTAIN DEFINITIONS

For convenience in this report, the terms "Company," "our," "us" or "we" may be used to refer to Huntsman Corporation and, unless the context otherwise requires, its subsidiaries and predecessors. Any references to our "Company," "we," "us" or "our" as of a date prior to October 19, 2004 (the date of our Company's formation) are to Huntsman Holdings, LLC and its subsidiaries (including their respective predecessors). In this report, "Huntsman International" refers to Huntsman International LLC (our 100% owned subsidiary) and, unless the context otherwise requires, its subsidiaries; "HPS" refers to Huntsman Polyurethanes Shanghai Ltd. (our consolidated splitting joint venture with Shanghai Chlor-Alkali Chemical Company, Ltd); and "SLIC" refers to Shanghai Liengheng Isocyanate Company (our unconsolidated manufacturing joint venture with BASF AG and three Chinese chemical companies).

In this report, we may use, without definition, the common names of competitors or other industry participants. We may also use the common names or abbreviations for certain chemicals or products.

INTERIM FINANCIAL STATEMENTS

Our interim condensed consolidated financial statements (unaudited) and Huntsman International's interim condensed consolidated financial statements (unaudited) were prepared in accordance with accounting principles generally accepted in the United States of America ("GAAP" or "U.S. GAAP") and in management's opinion, reflect all adjustments, consisting only of normal recurring adjustments, necessary for a fair presentation of results of operations, financial position and cash flows for the periods presented. Results for interim periods are not necessarily indicative of those to be expected for the full year. These condensed consolidated financial statements (unaudited) should be read in conjunction with the audited consolidated financial statements and notes to consolidated financial statements included in the Annual Report on Form 10-K for the year ended December 31, 2009 for our Company and Huntsman International.

DESCRIPTION OF BUSINESS

We are a global manufacturer of differentiated organic chemical products and of inorganic chemical products. Our products comprise a broad range of chemicals and formulations, which we market globally to a diversified group of consumer and industrial customers. Our products are used in a wide range of applications, including those in the adhesives, aerospace, automotive, construction products, durable and non-durable consumer products, electronics, medical, packaging, paints and coatings, power generation, refining, synthetic fiber, textile chemicals and dye industries. We are a leading global producer in many of our key product lines, including MDI, amines, surfactants, maleic anhydride, epoxy-based polymer formulations, textile chemicals, dyes and titanium dioxide.

We operate in five segments: Polyurethanes, Performance Products, Advanced Materials, Textile Effects and Pigments. Our Polyurethanes, Performance Products, Advanced Materials and Textile Effects segments produce differentiated organic chemical products and our Pigments segment produces inorganic chemical products. We ceased operation of our Australian styrenics business during the first quarter of 2010 and report the results of that business as discontinued operations. See "Note 19. Discontinued Operations."

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

1. GENERAL (Continued)

COMPANY

Our Company, a Delaware corporation, was formed in 2004 to hold the Huntsman businesses. Jon M. Huntsman founded the predecessor to our Company in the early 1970s as a small packaging company. Since then, we have grown through a series of acquisitions and now own a global portfolio of businesses.

Currently, we operate all of our businesses through Huntsman International, our 100% owned subsidiary. Huntsman International is a Delaware limited liability company and was formed in 1999.

HUNTSMAN CORPORATION AND HUNTSMAN INTERNATIONAL FINANCIAL STATEMENTS

Except where otherwise indicated, these notes relate to the consolidated financial statements for both our Company and Huntsman International. The differences between our financial statements and Huntsman International's financial statements relate primarily to the following:

- purchase accounting recorded at our Company for the 2003 step-acquisition of Huntsman International Holdings LLC, the former parent company of Huntsman International that was merged into Huntsman International in 2005;
- the different capital structures;
- a note payable from Huntsman International to us;
- income (expenses) associated with our terminated merger with a subsidiary of Hexion (the "Terminated Merger" or the "Hexion Merger");
- the \$250 million 7% convertible notes due 2018 (the "Convertible Notes") issued in connection with our December 14, 2008 settlement agreement with Apollo, Hexion and certain of their affiliates (the "Apollo Settlement Agreement")(see "Note 17. Income (Expenses) Associated with the Terminated Merger and Related Litigation"), which we repurchased on January 11, 2010; and
- the results of our settlement agreement (the "Texas Bank Litigation Settlement Agreement") in connection with our litigation (the "Texas Bank Litigation") against affiliates of Credit Suisse Securities (USA) LLC and Deutsche Bank Inc. (collectively, the "Banks") (see "Note 17. Income (Expenses) Associated with the Terminated Merger and Related Litigation").

PRINCIPLES OF CONSOLIDATION

Our condensed consolidated financial statements (unaudited) and Huntsman International's condensed consolidated financial statements (unaudited) include the accounts of our wholly-owned and majority-owned and controlled subsidiaries and any variable interest entities for which we are the primary beneficiary. All intercompany accounts and transactions have been eliminated.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

1. GENERAL (Continued)

VARIABLE INTEREST ENTITIES

We evaluate our investments and transactions to identify variable interest entities for which we are the primary beneficiary. We hold a variable interest in two joint ventures for which we are the primary beneficiary.

One joint venture manufactures products for our Polyurethanes segment. The structure of the joint venture is such that the total equity investment at risk is not sufficient to permit the joint venture to finance its activities without additional financial support. By virtue of the operating agreement with this joint venture, we purchase a majority of the output, absorb a majority of the operating costs and provide a majority of the additional funding.

The other joint venture manufactures products for our Pigments segment. In this joint venture we supply all the raw materials through a fixed cost supply contract, operate the manufacturing facility and market the products of the joint venture to third party customers. Through a fixed price raw materials supply contract with the joint venture we are exposed to the risk related to the fluctuation of raw material pricing.

As the primary beneficiary of these two variable interest entities, the joint ventures' assets, liabilities and results of operations are included in our condensed consolidated financial statements (unaudited). The following table summarizes the carrying amount of the two joint ventures' assets and liabilities included in our condensed consolidated balance sheet (unaudited), before intercompany eliminations, as of March 31, 2010 (dollars in millions):

Current assets	\$ 91
Property, plant and equipment, net	16
Other noncurrent assets	32
Deferred income taxes	38
Total assets	\$ 177
Current liabilities	\$ 85
Long-term debt	4
Other noncurrent liabilities	93
Total liabilities	\$ 182

USE OF ESTIMATES

The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

1. GENERAL (Continued)

RECLASSIFICATIONS

Certain amounts in the consolidated financial statements for prior periods have been reclassified to conform with the current presentation. In connection with the closure and abandonment of our Australian styrenics operations in the first quarter of 2010, we have treated this business as discontinued operations beginning in the first quarter of 2010. All relevant information for prior periods has been restated to reflect this change.

During the first quarter of 2010, we began reporting our last-in, first-out ("LIFO") inventory valuation reserve charges as part of Corporate and other, which was previously reported in our Performance Products segment. All segment information for prior periods has been restated to reflect this change.

RECENT DEVELOPMENTS

Repurchase of Convertible Notes

On January 11, 2010, we repurchased the entire \$250 million principal amount of our outstanding Convertible Notes for approximately \$382 million from Apollo and its affiliates. The Convertible Notes were issued to Apollo in December 2008. The Convertible Notes, which would have matured on December 23, 2018, bore interest at the rate of 7% per year and were convertible into approximately 31.8 million shares of our common stock at any time by the holders. As a result of the repurchase of the Convertible Notes, we recorded a loss on early extinguishment of debt in the first quarter of 2010 of \$146 million.

Senior Subordinated Note Offering

On March 17, 2010 we completed a \$350 million offering of 8.625% senior subordinated notes due 2020 (the "2020 Subordinated Notes"). We used the net proceeds to redeem approximately €184 million (approximately \$247 million) senior subordinated notes due 2013 and approximately €59 million (approximately \$79 million) senior subordinated notes due 2015. In connection with the redemption of these notes, we recorded a loss on early extinguishment of debt in the first quarter of 2010 of \$9 million. In connection with these redemptions and the issuance of the 2020 Subordinated Notes, Huntsman International entered into \$350 million notional amount of five-year euro/dollar cross-currency interest rate contracts. In accordance with these swap transactions, the interest payment on the 2020 Subordinated Notes will be denominated in euros at an effective rate of approximately 8.41% for the next five years.

Amendment of Senior Credit Facilities

On March 9, 2010, Huntsman International entered into a Fifth Amendment to Credit Agreement (the "Amendment") with JPMorgan Chase Bank, N.A. and the other financial institutions party thereto, which amended certain terms of Huntsman International's existing senior secured credit facilities (the "Senior Credit Facilities"). Among other things, the Amendment (i) replaces Deutsche Bank AG New York Branch as administrative agent, collateral agent and U.K. security trustee with JPMorgan Chase Bank, N.A. as administrative agent and collateral agent and JPMorgan Chase Bank, N.A. or an affiliate

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

1. GENERAL (Continued)

thereof as U.K. security trustee, (ii) extends the stated maturity of the revolving facility to March 9, 2014, and (iii) limits the aggregate amount of the revolving commitments allowable under the revolving facility to an amount up to \$300 million, including \$225 million currently committed by the lenders. The Amendment increased the applicable LIBOR margin range on the revolving loans by 1.75% to 3.50% per annum and increased the unused commitment fee percentage to a range of 0.50% to 0.75%.

Accounting for our Accounts Receivable Securitization Program

On January 1, 2010, in connection with the adoption of Accounting Standards Update ("ASU") No. 2009-16, we have accounted for sales of accounts receivable under our accounts receivable securitization programs as secured borrowings. See "Note 2. Recently Issued Accounting Pronouncements" and "Note 7. Debt—Accounts Receivable Securitization."

2. RECENTLY ISSUED ACCOUNTING PRONOUNCEMENTS

Accounting Pronouncements Adopted During 2010:

In February 2010, the Financial Accounting Standards Board ("FASB") issued and we adopted ASU No. 2010-09, *Subsequent Events (Topic 855)—Amendments to Certain Recognition and Disclosure Requirements*. This ASU provides a definition of the term "SEC filer" and removes the requirement for entities that are SEC filers to disclose the date through which subsequent events have been evaluated. We evaluate subsequent events through the date the financial statements are issued.

In January 2010, the FASB issued ASU No. 2010-06, Fair Value Measurements and Disclosures (Topic 820)—Improving Disclosures about Fair Value Measurements. This ASU clarifies existing disclosure requirements to provide a greater level of disaggregated information and to provide more information regarding valuation techniques and inputs to fair value measurements. It requires additional disclosure related to transfers between the three levels of fair value measurement, as well as information about purchases, sales, issuances, and settlements in the roll forward of activity for Level 3 measurements. The enhanced disclosures required by this ASU are effective for interim and annual reporting periods beginning after December 15, 2009, except for the disclosures about purchases, sales, issuances, and settlements in the roll forward of activity for Level 3 measurements, which is effective for interim and annual reporting periods beginning after December 15, 2010. See "Note 9. Fair Value."

Effective January 1, 2010, we adopted ASU No. 2009-17, Consolidations (Topic 810)—Improvements to Financial Reporting by Enterprises Involved with Variable Interest Entities, which codified Statement of Financial Accounting Standards ("SFAS") No. 167, Amendments to FASB Interpretation No. 46(R). This statement amends FASB Interpretation No. ("FIN") 46(R), Consolidation of Variable Interest Entities, to replace the quantitative-based risks and rewards calculation for determining which enterprise has a controlling financial interest in a variable interest entity with a qualitative approach. This new approach focuses on identifying which enterprise has the power to direct the activities of a variable interest entity that most significantly impact the entity's economic performance and (1) the obligation to absorb losses of the entity or (2) the right to receive benefits from the entity. It also requires ongoing assessments of whether an enterprise is the primary beneficiary of a variable interest entity, and it requires additional disclosures about an enterprise's involvement in variable interest entities. The adoption of this

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

2. RECENTLY ISSUED ACCOUNTING PRONOUNCEMENTS (Continued)

statement did not have a significant impact on our condensed consolidated financial statements (unaudited). See "Note 1. General—Variable Interest Entities."

Effective January 1, 2010, we adopted ASU No. 2009-16, *Transfers and Servicing (Topic 860)—Accounting for Transfers of Financial Assets*, which codified SFAS No. 166, *Accounting for Transfers of Financial Assets—an amendment of FASB Statement No. 140*. This statement removes the concept of a qualifying special-purpose entity ("QSPE") from SFAS No. 140 and removes the exception from applying FIN 46(R) to QSPEs. SFAS No. 166 modifies the derecognition provisions in SFAS No. 140 and requires that a transferor recognize and initially measure at fair value all assets obtained (including a transferor's beneficial interest) and liabilities incurred as a result of a transfer of financial assets accounted for as a sale. It also requires additional disclosures regarding the transferor's continuing involvement with transferred financial assets and the related risks retained. Upon adoption of this statement, sales of accounts receivable under our accounts receivable securitization programs no longer qualified for derecognition and were accounted for as secured borrowings beginning in January 2010. See "Note 7. Debt—Accounts Receivable Securitization." Prior to the adoption of this statement, receivables transferred under our U.S. and European accounts receivable securitization programs (the "U.S. A/R Program," the "EU A/R Program" and collectively the "A/R Programs") qualified as sales.

Accounting Pronouncements Pending Adoption in Future Periods:

In October 2009, the FASB issued ASU No. 2009-13, *Revenue Recognition (Topic 605): Multiple-Deliverable Revenue Arrangements*—a consensus of the FASB Emerging Issues Task Force. This ASU provides amendments to the criteria for separating consideration in multiple-deliverable arrangements. The amendments in this ASU replace the term "fair value" in the revenue allocation guidance with "selling price" to clarify that the allocation of revenue is based on entity-specific assumptions rather than assumptions of a marketplace participant, and they establish a selling price hierarchy for determining the selling price of a deliverable. The amendments in this ASU will eliminate the residual method of allocation and require that arrangement consideration be allocated at the inception of the arrangement to all deliverables using the relative selling price method, and they significantly expand the required disclosures related to multiple-deliverable revenue arrangements. The amendments in this ASU will be effective prospectively for revenue arrangements entered into or materially modified in fiscal years beginning after June 15, 2010. We are evaluating this ASU to determine its impact on our condensed consolidated financial statements (unaudited).

3. BUSINESS COMBINATIONS

BARODA ACQUISITION

On June 23, 2009, we announced the acquisition of the Baroda Division ("Baroda") of Metrochem Industries Limited ("MCIL"), a manufacturing facility for the production of intermediates and specialty dyes for textiles, located in Baroda, India. Baroda had been a significant supplier to our Textile Effects division and this acquisition strengthens the Textile Effects division's competitiveness and supports its development in Asia. We initially entered into an agreement to acquire Baroda on June 29, 2007. The initial agreement provided either party with the right to terminate the agreement if a transaction was not consummated by April 30, 2008. On February 6, 2009, we entered into a non-binding agreement in

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

3. BUSINESS COMBINATIONS (Continued)

principle with MCIL under which the purchase price was revised to be approximately \$35 million (U.S. dollar equivalents), which included receivables existing on the closing date due to MCIL from our affiliates, which were also settled at acquisition. Payment of the acquisition cost was phased in various tranches. The first tranche of \$7 million was paid during 2008; additional tranches were paid during 2009; and a final payment of \$2 million was made upon completion of the audit of net working capital acquired in the first quarter of 2010. In addition, \$5 million of accounts payable by us to MCIL were forgiven in connection with this acquisition.

We have accounted for the Baroda acquisition using the acquisition method. As such, we analyzed the fair value of tangible and intangible assets acquired and liabilities assumed. The preliminary allocation of acquisition cost to the assets acquired and liabilities assumed is summarized as follows (dollars in millions):

Acquisition cost:	
Cash payments made in 2008	\$ 7
Cash payments made in 2009	31
Cash payments made in 2010	2
Forgiveness of amounts payable from us to MCIL	(5)
Total acquisition cost	\$ 35
Fair value of assets acquired and liabilities assumed:	
Accounts receivable	\$ 2
Inventories	3
Other current assets	2
Property, plant and equipment	31
Intangible assets	4
Deferred tax asset	1
Accounts payable	(3)
Short-term debt	(3)
Deferred tax liability	(2)
Total fair value of net assets acquired	\$ 35

The acquisition cost allocation is preliminary pending finalization of the determination of the fair value of intangible assets and deferred taxes. For purposes of this preliminary allocation of fair value, we have assigned any excess of acquisition cost of historical carrying values to amortizable intangible assets and no amounts have been allocated to goodwill. We expect that it is reasonably possible that changes to this allocation could occur.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

4. INVENTORIES

Inventories are stated at the lower of cost or market, with cost determined using LIFO, first-in first-out, and average costs methods for different components of inventory. Inventories consisted of the following (dollars in millions):

	arch 31, 2010	Dec	cember 31, 2009
Raw materials and supplies	\$ 269	\$	240
Work in progress	75		77
Finished goods	965		917
Total	 1,309		1,234
LIFO reserves	(61)		(50)
Net	\$ 1,248	\$	1,184

For each of March 31, 2010 and December 31, 2009, approximately 10% of inventories were recorded using the LIFO cost method.

In the normal course of operations, we at times exchange raw materials and finished goods with other companies for the purpose of reducing transportation costs. The net non-monetary open exchange positions are valued at cost. The amounts included in inventory under non-monetary open exchange agreements receivable by us as of March 31, 2010 and December 31, 2009 were \$4 million and \$3 million, respectively. Other open exchanges are settled in cash and result in a net deferred profit margin. The amounts under these open exchange agreements receivable or payable by us at both March 31, 2010 and December 31, 2009 were nil.

5. INVESTMENT IN UNCONSOLIDATED AFFILIATES

During the year ended December 31, 2008, we contributed \$44 million as our 50% equity contribution to the Arabian Amines Company, our ethyleneamines manufacturing joint venture in Jubail, Saudi Arabia. This joint venture's funding requirements will be satisfied through a combination of debt and equity, with the equity already provided on a 50/50 basis by us and Zamil Group. The joint venture obtained various loan commitments in the aggregate amount of approximately \$195 million in U.S. dollar equivalents, of which \$177 million, including bridge loans, was drawn as of March 31, 2010. The plant will have approximate annual capacity of 60 million pounds with production expected in early second quarter of 2010. We will purchase and sell all of the production from this joint venture. We have provided certain guarantees of approximately \$14 million for these commitments, which will terminate upon completion of the project and satisfaction of certain conditions. We have estimated that the fair value of these guarantees was nil as of the closing date of this transaction and, accordingly, no amounts have been recorded. This joint venture is currently accounted for under the equity method during its development stage.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

6. RESTRUCTURING, IMPAIRMENT AND PLANT CLOSING COSTS

As of March 31, 2010 and December 31, 2009, accrued restructuring costs by type of cost and initiative consisted of the following (dollars in millions):

	Workforce reductions(1)				Non-cancelable lease costs		Other restructuring costs		To	otal(2)
Accrued liabilities as of January 1,										
2010	\$	50	\$	1	\$	3	\$	21	\$	75
2010 charges for 2009 initiatives		1		_		_		1		2
2010 charges for 2010 initiatives		4		_		_		_		4
Reversal of reserves no longer										
required		(3)				_		_		(3)
2010 payments for 2006 initiatives		(1)		_		_		_		(1)
2010 payments for 2008 initiatives		(2)				_		_		(2)
2010 payments for 2009 initiatives		(5)		_		_		(2)		(7)
Net activity of discontinued										
operations		(23)				_		6		(17)
Foreign currency effect on reserve balance		(2)		_		_		_		(2)
Accrued liabilities as of March 31, 2010	\$	19	\$	1	\$	3	\$	26	\$	49

⁽¹⁾ The total workforce reduction reserves of \$19 million relate to the termination of 300 positions, of which 254 positions had not been terminated as of March 31, 2010.

(2) Accrued liabilities by initiatives were as follows (dollars in millions):

	ch 31, 010	mber 31, 009
2005 initiatives & prior	\$ 3	\$ 3
2006 initiatives	4	5
2008 initiatives	5	7
2009 initiatives	33	60
2010 initiatives	4	_
Total	\$ 49	\$ 75

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

6. RESTRUCTURING, IMPAIRMENT AND PLANT CLOSING COSTS (Continued)

Details with respect to our reserves for restructuring, impairment and plant closing costs are provided below by segment and initiative (dollars in millions):

	Polyureth	anes	Performa Produc		Advano Materi		Tex Effe		Pigme	nts	Discontin Operatio		Corporate other	e &	To	tal
Accrued liabilities as																
of January 1, 2010	\$	2	\$	_	\$	7	\$	17	\$	11	\$	34	\$	4	\$	75
2010 charges for 2009 initiatives		_		_		_		_		1		_		1		2
2010 charges for 2010 initiatives		_		_		_		_		_		_		4		4
Reversal of reserves no longer required						(2)								(1)		(3)
2010 payments for		_		_		(2)						<u> </u>		(1)		
2006 initiatives 2010 payments for		_		_		_		(1)		_		_		_		(1)
2008 initiatives		(1)		_		_		(1)		_		_		_		(2)
2010 payments for 2009 initiatives		_		_		(1)		(2)		(3)		_		(1)		(7)
Net activity of discontinued operations		_		_		_		_		_		(17)		_		(17)
Foreign currency effect on reserve balance		_		_		_		(1)		(1)		_		_		(2)
Accrued liabilities as of March 31, 2010	\$	1	\$	_	\$	4	\$	12	\$	8	\$	17	\$	7	\$	49
Current portion of restructuring																
reserves	\$	1	\$	_	\$	3	\$	12	\$	5	\$	17	\$	7	\$	45
Long-term portion of restructuring																
reserves		_		_		1		_		3		_		_		4
Estimated additional future charges for current restructuring projects																
Estimated additional charges within one																
year	\$	_	\$	_	\$	1	\$	_	\$	7	\$	_	\$	1	\$	9
Estimated additional charges beyond																
one year		_		_		_		_		3		_		_		3

Details with respect to cash and non-cash restructuring charges by initiative are provided below (dollars in millions):

	er	e months nded n 31, 2010
Cash charges:		
2010 charges for 2009 initiatives	\$	2
2010 charges for 2010 initiatives		4
Reversal of reserves no longer required		(3)
Total 2010 Restructuring, Impairment and Plant Closing Costs	\$	3

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

6. RESTRUCTURING, IMPAIRMENT AND PLANT CLOSING COSTS (Continued)

	e	e months nded n 31, 2009
Cash charges:		
2009 charges for 2008 initiatives	\$	2
2009 charges for 2009 initiatives		9
Reversal of reserves no longer required		(1)
Non-cash charges		4
Total 2009 Restructuring, Impairment and Plant Closing Costs	\$	14

During the three months ended March 31, 2010, our Advanced Materials segment reversed accruals of \$2 million primarily related to workforce reductions in connection with a reorganization towards a regional management structure.

During the three months ended March 31, 2010, our Pigments segment recorded charges of \$1 million primarily related to the closure of our Grimsby plant. We expect to incur additional charges of \$10 million through December 31, 2011, primarily related to the closure of our Grimsby, U.K. plant.

During the three months ended March 31, 2010, we recorded net charges of \$4 million in Corporate and other related to workforce reductions in connection with a reorganization and regional consolidation of our transactional accounting activities.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

7. DEBT

Outstanding debt consisted of the following (dollars in millions):

Huntsman Corporation

	March 31, 2010			2009
Senior Credit Facilities:				
Term loans	\$	1,950	\$	1,968
Amounts outstanding under A/R programs		242		_
Senior notes		438		434
Subordinated notes		1,268		1,294
Australian credit facilities		38		34
HPS (China) debt		179		163
Convertible notes		_		236
Other		60		83
Total debt—excluding debt to affiliates	\$	4,175	\$	4,212
Current portion of debt	\$	316	\$	431
Long-term portion		3,859		3,781
Total debt—excluding debt to affiliates	\$	4,175	\$	4,212
Total debt—excluding debt to affiliates	\$	4,175	\$	4,212
Notes payable to affiliates-noncurrent		5		5
Total debt	\$	4,180	\$	4,217

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

7. DEBT (Continued)

Huntsman International

	March 31, 2010		December 31, 2009	
Senior Credit Facilities:				
Term loans	\$	1,950	\$	1,968
Amounts outstanding under A/R programs		242		_
Senior notes		438		434
Subordinated notes		1,268		1,294
Australian credit facilities		38		34
HPS (China) debt		179		163
Other		60		83
Total debt—excluding debt to affiliates	\$	4,175	\$	3,976
Current portion of debt	\$	316	\$	195
Long-term portion		3,859		3,781
Total debt—excluding debt to affiliates	\$	4,175	\$	3,976
Total debt—excluding debt to affiliates	\$	4,175	\$	3,976
Notes payable to affiliates-current		100		25
Notes payable to affiliates-noncurrent		330		530
Total debt	\$	4,605	\$	4,531

DIRECT AND SUBSIDIARY DEBT

Huntsman Corporation's direct debt and guarantee obligations consist of the following: our guarantees of certain debt of HPS (our Chinese MDI joint venture); our guarantee of certain debt of the Arabian Amines Company (our ethyleneamines manufacturing joint venture in Jubail, Saudi Arabia); certain indebtedness incurred from time to time to finance certain insurance premiums; and our guarantee of certain obligations of Huntsman International in its capacity as a contributor and servicer guarantor under the U.S. A/R Program.

Substantially all of our other debt, including the facilities described below, has been incurred by our subsidiaries (primarily Huntsman International); such subsidiary debt is nonrecourse to us and we have no contractual obligation to fund our subsidiaries' respective operations.

TRANSACTIONS AFFECTING OUR DEBT

Senior Credit Facilities

As of March 31, 2010, our Senior Credit Facilities consisted of (i) our \$225 million revolving facility ("Revolving Facility"); (ii) our \$1,509 million term loan B facility ("Term Loan B"); and (iii) our \$495 million (\$441 million carrying value) term loan C facility ("Term Loan C" and, collectively with Term Loan B, the "Dollar Term Loans"). As of March 31, 2010, we had no borrowings outstanding under our Revolving Facility, and we had approximately \$40 million (U.S. dollar equivalents) of letters of credit and bank guarantees issued and outstanding under our Revolving

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

7. DEBT (Continued)

Facility. All of our Senior Credit Facilities are obligations of Huntsman International and are not direct obligations of Huntsman Corporation.

On March 9, 2010, Huntsman International entered into the Amendment with JPMorgan Chase Bank, N.A., as successor administrative agent and collateral agent, and the other financial institutions party thereto, which amended certain terms of Huntsman International's existing Senior Credit Facilities. Among other things, the Amendment (i) replaces Deutsche Bank AG New York Branch as administrative agent, collateral agent and U.K. security trustee with JPMorgan Chase Bank, N.A. as administrative agent and collateral agent and JPMorgan Chase Bank, N.A. or an affiliate thereof as U.K. security trustee, (ii) extends the stated maturity of the Revolving Facility to March 9, 2014 and provides for optional extensions of such stated maturity date from time to time with the consent of the lenders and subject to certain specified conditions and exceptions, (iii) limits the aggregate amount of the revolving commitments allowable under the Revolving Facility to an amount up to \$300 million, including \$225 million currently obtained from the lenders, (iv) terminated the 2009 Waiver and (v) reduces the maximum letter of credit sublimit to \$75 million (not including existing letters of credit issued by Deutsche Bank AG New York Branch) and the maximum swing line sublimit to \$25 million.

Additionally, the Amendment increases the applicable LIBOR margin range on the revolving loans by 1.75% to 3.50% per annum and increases the unused commitment fee percentage to a range of 0.50% to 0.75%. The Amendment also amends the mandatory prepayment provisions of the Senior Credit Facilities to permit the reinvestment of certain insurance and condemnation proceeds.

Our Senior Credit Facilities are subject to a single financial covenant (the "Leverage Covenant") which applies only to the Revolving Facility and is tested at the Huntsman International level. The Leverage Covenant is applicable only if borrowings, letters of credit or guarantees are outstanding under the Revolving Facility (cash collateralized letters of credit or guarantees are not deemed outstanding). Following the termination of the 2009 Waiver, the Leverage Covenant is a net senior secured leverage ratio covenant, with a maximum ratio of senior secured debt to EBITDA (as defined in the applicable agreement) of no more than 3.75 to 1.

At the present time, borrowings under the Revolving Facility, Term Loan B and Term Loan C bear interest at LIBOR plus 3.50%, LIBOR plus 1.75% and LIBOR plus 2.25%, respectively. However, the applicable interest rate of Term Loan B is subject to a reduction to LIBOR plus 1.5% upon achieving certain secured leverage ratio thresholds. The Revolving Facility matures on March 9, 2014 (subject to optional extensions from time to time with the consent of the lenders and subject to certain specified conditions and exceptions), Term Loan B matures in 2014 and Term Loan C matures in 2016; provided, however, that the maturities of the Revolving Facility and the Dollar Term Loans will accelerate if we do not repay or refinance all but \$100 million of Huntsman International's outstanding debt securities on or before three months prior to the maturity dates of such debt securities.

During the three months ended March 31, 2010, we paid the annual scheduled repayment of \$16 million on our Term Loan B and \$5 million on our Term Loan C. On April 26, 2010, we prepaid \$124 million of Term Loan B and \$40 million of Term Loan C with the proceeds from excess cash flow.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

7. DEBT (Continued)

Accounts Receivable Securitization

Under our A/R Programs, we grant an undivided interest in certain of our trade receivables to bankruptcy-remote special purpose entities ("SPEs"). This undivided interest serves as security for the issuance of debt. The A/R Programs provide for financing through a conduit program (in both U.S. dollars and euros). Receivables transferred under the A/R Programs qualified as sales through December 31, 2009. Upon adoption of new accounting guidance in 2010, sales of accounts receivable under our A/R Programs no longer meet the criteria for derecognition. Accordingly, the amounts outstanding under our new A/R Programs are accounted for as secured borrowings as of January 1, 2010. See "Note 2. Summary of Significant Accounting Policies."

As of March 31, 2010, the A/R Programs had \$242 million in U.S. dollar equivalents in loans outstanding (consisting of \$55 million and €139 million (approximately \$187 million)). As of March 31, 2010, \$538 million of accounts receivable were pledged as collateral under the A/R Programs. As of December 31, 2009, the A/R Programs had \$254 million in U.S. dollar equivalents in loans outstanding (consisting of \$55 million and €139 million (approximately \$199 million)).

U.S. ACCOUNTS RECEIVABLE SECURITIZATION PROGRAM

The maximum funding availability under the U.S. A/R Program is \$250 million, which is divided between two facilities: a \$125 million three-year facility with a maturity of October 2012 and a \$125 million two-year facility, with a maturity of October 2011. The amount of actual availability under the U.S. A/R Program is subject to change based on the level of eligible receivables sold. Availability is further subject to changes in the credit ratings of Huntsman International's customers, customer concentration levels, and certain characteristics of the accounts receivable being transferred. The yield on the three-year facility is based on the LIBOR rate (as defined in the applicable agreement) plus a margin rate of 3.75% per annum and, in the case of the two-year facility, if funded by commercial paper ("CP") by the lender, the CP Rate (as defined in the applicable agreement) plus a margin rate of 3.50% per annum. In addition, the U.S. SPE is obligated to pay commitment fees to the lenders based on the amount of each lender's commitment.

The U.S. A/R Program contains various customary affirmative and negative covenants and also contains customary default and termination provisions, which provide for acceleration of amounts owed under the U.S. A/R Program upon the occurrence of certain specified events, including, but not limited to, failure by the U.S. SPE to pay interest and other amounts due, defaults on certain indebtedness, certain judgments, change in control, certain events negatively affecting the overall credit quality of transferred accounts receivable, bankruptcy and insolvency events, and failure of Huntsman International to maintain a minimum liquidity level of \$400 million (the "Liquidity Requirement"). We guarantee certain obligations of Huntsman International in its capacity as contributor and servicer guarantor under the U.S. A/R Program.

EUROPEAN ACCOUNTS RECEIVABLE SECURITIZATION PROGRAM

The maximum funding availability under the EU A/R Program is €225 million (approximately \$302 million) with a maturity of October 2011. The amount of actual availability under the EU A/R Program is subject to change based on the level of eligible receivables sold. Availability is further subject to changes in the credit ratings of the originators' customers and country, customer

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

7. DEBT (Continued)

concentration levels, and certain characteristics of the accounts receivable being transferred. The yield is based on GBP LIBOR, USD LIBOR or EURIBOR (each as defined in the applicable agreement) plus a margin rate of 3.75% per annum if funded by commercial paper by the lender. In addition, the EU SPE is obligated to pay a commitment fee to the lender based on the amount of the lender's commitment.

The EU A/R Program contains various customary affirmative and negative covenants and also contains customary default and termination provisions, which provide for acceleration of amounts owed under the EU A/R Program upon the occurrence of certain specified events, including, but not limited to, failure by the EU SPE to pay interest and other amounts due, defaults on certain indebtedness, certain judgments, change in control, certain events negatively affecting the overall credit quality of transferred accounts receivable and bankruptcy and insolvency events and a cross acceleration provision tied to the Liquidity Requirement.

2020 Subordinated Notes

On March 17, 2010, Huntsman International completed the \$350 million offering of the 2020 Subordinated Notes. The 2020 Subordinated Notes bear interest at 8.625% per annum and are due March 15, 2020. We used the net proceeds of \$343 million to redeem a portion of our euro-denominated senior subordinated notes due 2013 (€184 million (approximately \$247 million)) and a portion of our euro-denominated senior subordinated notes due 2015 (€59 million (approximately \$79 million)). See "—Redemption of Notes" below.

At any time prior to March 15, 2013, Huntsman International may redeem up to 40% of the aggregate principal amount of the 2020 Subordinated Notes with the net cash proceeds of certain equity offerings. Huntsman International may redeem the 2020 Subordinated Notes in whole or in part prior to March 15, 2015 at a price equal to 100% of the principal amount thereof plus a "make-whole" premium. The 2020 Subordinated Notes are redeemable on or after March 15, 2015 at 104.3125%, declining ratably to par on or after March 15, 2018.

Interest is payable on the 2020 Subordinated Notes semiannually on March 15 and September 15 of each year. The 2020 Subordinated Notes are general unsecured senior subordinated obligations of Huntsman International and are guaranteed on a general unsecured senior subordinated basis by our subsidiary guarantors (the "Subsidiary Guarantors"). The Indenture governing the 2020 Subordinated Notes contains covenants relating to, among other things, the incurrence of additional indebtedness; the payment of dividends and the payment of certain other restricted payments; transactions with affiliates; creating dividend or other payment restrictions affecting restricted subsidiaries; the merger or consolidation with any other person or any sale, assignment, transfer, lease, conveyance or other disposition of all or substantially all of its assets; or the adoption of a plan of liquidation.

Redemption of Notes

On March 17, 2010, Huntsman International repaid €184 million (approximately \$247 million) of its 6.875% senior subordinated notes due 2013. The amount paid to redeem the notes, excluding accrued interest, was €189 million (approximately \$254 million), which included principal of €184 million (approximately \$247 million) and premium of €5 million (approximately \$7 million). As of

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

7. DEBT (Continued)

March 31, 2010, the 6.875% senior subordinated notes due 2013 have a remaining balance of €216 million (approximately \$290 million).

On March 17, 2010, Huntsman International repaid \in 59 million (approximately \$79 million) of its 7.5% senior subordinated notes due 2015. Amount paid to redeem the notes, excluding accrued interest, was \in 59 million (approximately \$79 million). As of March 31, 2010, the 7.5% senior subordinated notes due 2015 have a remaining balance of \in 76 million (approximately \$102 million).

For information regarding the redemption of our Convertible Notes, see "Note 1. General—Recent Developments—Repurchase of Convertible Notes."

Other Debt

On April 1, 2010, our \$25 million European overdraft facility was terminated. This facility was a demand facility used for the working capital needs of our European subsidiaries. As of both March 31, 2010 and December 31, 2009, we had nil U.S. dollar equivalents in borrowings outstanding under this European overdraft facility. We also maintain other foreign overdraft facilities used for working capital needs

HPS obtained secured loans for the construction of its MDI production facility. This debt consists of various committed loans, including both U.S. dollar and RMB term loans and RMB working capital loans. During the three months ended March 31, 2010, HPS refinanced 130 million in RMB (approximately \$19 million) in working capital loans that were scheduled to be repaid during the quarter. The loans were refinanced for three years at the same interest rate of 90% of the Peoples Bank of China rate, which was 4.9% as of March 31, 2010.

Our Australian subsidiaries have credit facilities with a scheduled maturity of May 2010. We are in the process of refinancing a portion of these facilities with the existing lender. The remaining portion will be repaid in connection with the shutdown of the Australian styrenics operations. See "Note 19. Discontinued Operations."

Intercompany Note

Under an existing promissory note (the "Intercompany Note"), we have provided financing to Huntsman International. As of March 31, 2010, the outstanding total balance of the Intercompany Note was \$425 million. Under the agreements governing our Senior Credit Facilities, Huntsman International cannot repay amounts under the Intercompany Note if there are any outstanding revolving loans, swing line loans or outstanding letters of credit that are not cash collateralized, unless, before and after giving effect to such payment on a pro forma basis, Huntsman International is in compliance with the Leverage Covenant. Huntsman International is currently in compliance with Leverage Covenant, and during the three months ended March 31, 2010, repaid \$125 million under the Intercompany Note. As of March 31, 2010, and in accordance with the limitation contained in the agreements governing the Senior Credit Facilities as described above, Huntsman International would be permitted to repay up to approximately \$425 million on the Intercompany Note.

The Intercompany Note is unsecured and \$100 million of the outstanding amount is classified as current as of March 31, 2010 on the accompanying condensed consolidated balance sheets (unaudited). As of March 31, 2010, under the terms of the Intercompany Note, Huntsman International promises to

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

7. DEBT (Continued)

pay us interest on the unpaid principal amount at a rate per annum based on the previous monthly average borrowing rate obtained under our U.S. A/R Program, less 10 basis points (provided that the rate shall not exceed an amount that is 25 basis points less than the monthly average borrowing rate obtained for the U.S. LIBOR-based borrowings under our Revolving Facility). Subject to the conditions of the Senior Credit Facilities, with our consent, the principal and accrued interest outstanding under the Intercompany Note may also be forgiven, capitalized or satisfied with any alternate form of consideration.

COMPLIANCE WITH COVENANTS

Our management believes that we are in compliance with the covenants contained in the agreements governing our material debt instruments, including our Senior Credit Facilities, our A/R Programs and the indentures governing our notes.

Our Senior Credit Facilities are subject to a single financial covenant, the Leverage Covenant, which applies only to the Revolving Facility and is tested at the Huntsman International level. The Leverage Covenant is applicable only if borrowings, letters of credit or guarantees are outstanding under the Revolving Facility (cash collateralized letters of credit or guarantees are not deemed outstanding). Following the termination of the 2009 Waiver, the Leverage Covenant is a net senior secured leverage ratio covenant, with a maximum ratio of senior secured debt to EBITDA (as defined in the applicable agreement) of no more than 3.75 to 1.

If in the future Huntsman International is not able to meet the secured leverage ratio under the Leverage Covenant, unless Huntsman International obtains an amendment or waiver (as to which we can provide no assurance), then, for so long as Huntsman International did not meet such secured leverage ratio, we would not have access to the liquidity otherwise available under our Revolving Facility. If Huntsman International fails to meet this secured leverage ratio at a time when we have loans or letters of credit outstanding under the Revolving Facility, Huntsman International would be in default under our Senior Credit Facilities, and, unless we obtain a waiver or forbearance with respect to such default (as to which we can provide no assurance), we could be required to pay off the balance of our Senior Credit Facilities in full and would not have further access to such facilities.

Our A/R Programs consist of two facilities: Our \$250 million U.S. A/R Program and our €225 million (approximately \$302 million) EU A/R Program. The agreements governing our U.S. A/R Program and the agreements governing our EU A/R Program also contain certain financial covenants. Any material failure to meet the applicable A/R Program's covenants in the future could lead to an event of default under the A/R Programs, which could require us to cease our use of such facilities. Under these circumstances, unless any default was remedied or waived, we would likely lose the ability to obtain financing with respect to our trade receivables. A material default under the A/R Programs would also constitute an event of default under our Senior Credit Facilities, which could require us to pay off the balance of the Senior Credit Facilities in full and could result in the loss of our Senior Credit Facilities.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

8. DERIVATIVE INSTRUMENTS AND HEDGING ACTIVITIES

We are exposed to market risks, such as changes in interest rates, foreign exchange rates and commodity pricing risks. From time to time, we enter into transactions, including transactions involving derivative instruments, to manage certain of these exposures.

All derivatives, whether designated in hedging relationships or not, are recorded on our balance sheet at fair value. If the derivative is designated as a fair value hedge, the changes in the fair value of the derivative and the hedged items are recognized in earnings. If the derivative is designated as a cash flow hedge, changes in the fair value of the derivative are recorded in accumulated other comprehensive (loss) income, to the extent effective, and will be recognized in the income statement when the hedged item affects earnings. To the extent applicable, we perform effectiveness assessments in order to use hedge accounting at each reporting period. For a derivative that does not qualify as a hedge, changes in fair value are recognized in earnings.

We also hedge our net investment in certain European operations. Changes in the fair value of the hedge in the net investment of certain European operations are recorded in accumulated other comprehensive (loss) income.

Our cash flows and earnings are subject to fluctuations due to exchange rate variation. Our revenues and expenses are denominated in various currencies. From time to time, we may enter into foreign currency derivative instruments to minimize the short-term impact of movements in foreign currency rates. Where practicable, we generally net multicurrency cash balances among our subsidiaries to help reduce exposure to foreign currency exchange rates. Certain other exposures may be managed from time to time through financial market transactions, principally through the purchase of spot or forward foreign exchange contracts (generally with maturities of one year or less). We do not hedge our currency exposures in a manner that would eliminate the effect of changes in exchange rates on our cash flows and earnings. As of March 31, 2010, we had approximately \$117 million notional amount (in U.S. dollar equivalents) outstanding in forward foreign currency contracts.

On December 9, 2009, we entered into a five-year interest rate contract to hedge the variability caused by monthly changes in cash flow due to associated changes in LIBOR under our Senior Credit Facilities. The notional value of the contract is \$50 million and was designated as a cash flow hedge. The effective portion of the changes in the fair value of the swap was recorded in other comprehensive loss. We will pay a fixed 2.6% on the hedge and receive the one-month LIBOR rate. As of March 31, 2010, the fair value of the hedge was less than \$1 million and is recorded in other noncurrent liabilities.

On January 19, 2010, we entered into a five-year interest rate contract to hedge the variability caused by monthly changes in cash flow due to associated changes in LIBOR under our Senior Credit Facilities. The notional value of the contract is \$50 million and was designated as a cash flow hedge. The effective portion of the changes in the fair value of the swap was recorded in other comprehensive loss. We will pay a fixed 2.8% on the hedge and receive the one-month LIBOR rate. As of March 31, 2010, the fair value of the hedge was less than \$1 million and is recorded in other noncurrent liabilities.

In conjunction with our 2020 Senior Subordinated Notes issuance, we entered into cross-currency interest rate contracts with three counterparties. On March 17, 2010, we made payments of \$350 million to these counterparties and received €255 million from these counterparties, and on maturity, March 15, 2015, we are required to pay €255 million to these counterparties and will receive \$350 million from these counterparties. On March 15 and September 15 of each year, we will receive

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

8. DERIVATIVE INSTRUMENTS AND HEDGING ACTIVITIES (Continued)

U.S. dollar interest payments of approximately \$15 million (equivalent to an annual rate of 8.625%) and make interest payments of approximately €11 million (equivalent to an annual rate of approximately 8.41%). This swap is designated as a hedge of net investment for financial reporting purposes. As of March 31, 2010, the fair value of this swap was \$4 million and was recorded as noncurrent assets in our condensed consolidated balance sheet (unaudited). For the three months ended March 31, 2010, the effective portion of the changes in the fair value of \$6 million was recorded in other comprehensive income, with the ineffectiveness of \$2 million recorded in interest expense.

As of and for the three months ended March 31, 2010, the changes in fair value of the realized gains (losses) recorded in the accompanying condensed consolidated statements of operations (unaudited) of our other outstanding foreign currency rate hedging contracts and derivatives were not considered significant.

A significant portion of our intercompany debt is denominated in euros. We also finance certain of our non-U.S. subsidiaries with intercompany loans that are, in many cases, denominated in currencies other than the entities' functional currency. We manage the net foreign currency exposure created by this debt through various means, including cross-currency swaps, the designation of certain intercompany loans as permanent loans because they are not expected to be repaid in the foreseeable future ("permanent loans") and the designation of certain debt and swaps as net investment hedges.

Foreign currency transaction gains and losses on intercompany loans that are not designated as permanent loans are recorded in earnings. Foreign currency transaction gains and losses on intercompany loans that are designated as permanent loans are recorded in other comprehensive loss. From time to time, we review such designation of intercompany loans.

From time to time, we review our non-U.S. dollar denominated debt and swaps to determine the appropriate amounts designated as hedges. As of March 31, 2010, we have designated approximately \in 441 million (\$593 million) of euro-denominated debt and cross-currency interest rate swap as a hedge of our net investments. For the three months ended March 31, 2010, the amount of gain recognized on the hedge of our net investments was \$35 million and was recorded in other comprehensive loss. As of March 31, 2010, we had approximately \in 1,077 million (\$1,447 million) in net euro assets.

9. FAIR VALUE

The fair values of our financial instruments were as follows (dollars in millions):

Huntsman Corporation

	March 3	31, 2010	December 31, 2009			
	Carrying	Estimated	Carrying	Estimated		
	Amount	Fair Value	Amount	Fair Value		
Non-qualified employee benefit plan investments	\$ 11	\$ 11	\$ 10	\$ 10		
Cross-currency interest rate contracts	4	4	_			
Interest rate contracts	(1)	(1)	(1)	(1)		
Long-term debt (including current portion)	(4,175)	(4,248)	(4,212)	(4,390)		

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

9. FAIR VALUE (Continued)

Huntsman International

	March 31, 2010				December 31, 2009			
	Carrying Amount		Estimated Fair Value		Carrying Amount		Estimated Fair Value	
Non-qualified employee benefit plan investments	\$	11	\$	11	\$	10	\$	10
Cross-currency interest rate contracts		4		4		_		_
Interest rate contracts		(1)		(1)		(1)		(1)
Long-term debt (including current portion)	(4	4,175)		(4,248)	(3	3,976)	((3,951)

The carrying amounts reported in the balance sheets for cash and cash equivalents, accounts receivable and accounts payable approximate fair value because of the immediate or short-term maturity of these financial instruments. The fair value of non-qualified employee benefit plan investments is estimated using prevailing market prices. The estimated fair values of our long-term debt other than the Convertible Notes are based on quoted market prices for the identical liability when traded as an asset in an active market. The estimated fair value of our Convertible Notes at December 31, 2009 was based on the present value of estimated future cash flows, calculated using management's best estimates of key assumptions including relevant interest rates, expected share volatility, dividend yields and the probabilities associated with certain features of the Convertible Notes.

The fair value estimates presented herein are based on pertinent information available to management as of March 31, 2010 and December 31, 2009. Although management is not aware of any factors that would significantly affect the estimated fair value amounts, such amounts have not been comprehensively revalued for purposes of these financial statements since March 31, 2010, and current estimates of fair value may differ significantly from the amounts presented herein.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

9. FAIR VALUE (Continued)

The following assets and liabilities are measured at fair value on a recurring basis (dollars in millions):

		Fair Value Amounts Using					
Description	ech 31, 010	Quoted prices in active markets for identical assets (Level 1)		Significant other observable inputs (Level 2)		unol i	nificant oservable nputs evel 3)
Assets:							
Available-for-sale equity securities:							
Equity mutual funds	\$ 11	\$	11	\$	_	\$	_
Derivatives:							
Cross-currency interest rate contracts(1)	4		_		_		4
Total assets	\$ 15	\$	11	\$		\$	4
Liabilities:							
Derivatives:							
Interest rate contracts(2)	\$ 1	\$	_	\$	1	\$	_

⁽¹⁾ The income approach is used to calculate the fair value of these instruments. Fair value represents the present value of estimated future cash flows, calculated using relevant interest rates, exchange rates, and yield curves at stated intervals.

⁽²⁾ The income approach is used to calculate the fair value of these instruments. Fair value represents the present value of estimated future cash flows, calculated using relevant interest rates and yield curves at stated intervals.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

9. FAIR VALUE (Continued)

The following table shows a reconciliation of beginning and ending balances for assets measured at fair value on a recurring basis using significant unobservable inputs (Level 3) (dollars in millions):

	Three months ended March 31, 2010				
Fair Value Measurements Using Significant Unobservable Inputs (Level 3)		ed Interest curitized civables	Cross-Currency Interest Rate Contracts		Total
Beginning balance	\$	262	\$ -	- :	\$ 262
Total gains or losses					
Included in earnings (or changes in net assets)		_	(2	2)	(2)
Included in other comprehensive income (loss)		_		5	6
Purchases, issuances, sales and settlements(1)		(262)	_	-	(262)
Ending balance	\$	_	\$	1 :	\$ 4
The amount of total gains (losses) for the period included in earnings attributable to the change in unrealized gains (losses) relating to assets still held at March 31, 2010	\$	_	\$	5 :	s 6
remains to assets sum nets at that on 51, 2010	4		4		-

⁽¹⁾ Upon adoption of ASU 2009-16, sales of our accounts receivable under our A/R Programs no longer met the criteria for derecognition. Accordingly, beginning January 1, 2010, the amounts outstanding under the A/R Programs were accounted for as secured borrowings and the retained interest in securitized receivables was no longer relevant.

Gains and losses (realized and unrealized) included in earnings (or changes in net assets) for the three months ended March 31, 2010 are reported in interest expense and other comprehensive loss as follows (dollars in millions):

			Other comprehen	sive
	Interes	t expense	loss	
Total net (losses) gains included in earnings (or changes in net assets)	\$	(2)	\$	_
Changes in unrealized gains (losses) relating to assets still held at March 31,				
2010		(2)		6

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

10. EMPLOYEE BENEFIT PLANS

Components of the net periodic benefit costs for the three months ended March 31, 2010 and 2009 were as follows (dollars in millions):

Huntsman Corporation

	Defined B		Oth Postreti Benefit	rement Plans
	Three month	31,	Three r end Marc	led h 31,
	2010	2009	2010	2009
Service cost	\$ 17	\$ 16	\$ 1	\$ 1
Interest cost	36	35	2	2
Expected return on assets	(42)	(35)	_	_
Amortization of prior service cost	(1)	(1)	(1)	(1)
Amortization of actuarial loss	6	9	_	1
Special termination benefits	_	1	_	_
Net periodic benefit cost	\$ 16	\$ 25	\$ 2	\$ 3

Huntsman International

	Defined Pla		Other Postretirement Benefit Plans			
	Three mon Marc		Three end			
	2010	2009	2010	2009		
Service cost	\$ 17	\$ 16	\$ 1	\$ 1		
Interest cost	36	35	2	2		
Expected return on assets	(42)	(35)	_	_		
Amortization of prior service cost	(1)	(1)	(1)	(1)		
Amortization of actuarial loss	8	10	_	1		
Special termination benefits	_	1	_	_		
Net periodic benefit cost	\$ 18	\$ 26	\$ 2	\$ 3		

During the three months ended March 31, 2010 and 2009, we made contributions to our pension and other postretirement benefit plans of \$44 million and \$30 million, respectively. During the remainder of 2010, we expect to contribute an additional amount of \$77 million to these plans.

On March 23, 2010, President Obama signed into law the Patient Protection and Affordable Care Act. On March 30, 2010, President Obama signed into law a reconciliation measure, the Health Care and Education Reconciliation Act of 2010. The passage of these acts has resulted in comprehensive

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

10. EMPLOYEE BENEFIT PLANS (Continued)

health care reform legislation in the United States. We are currently evaluating the impact of this legislation on our condensed consolidated financial statements (unaudited).

11. STOCKHOLDERS' EQUITY

COMMON STOCK DIVIDENDS

On March 31, 2010 and March 31, 2009, we paid cash dividends of \$24 million and \$23 million, respectively, or \$0.10 per share each, to common stockholders of record as of March 15, 2010 and March 16, 2009, respectively.

12. OTHER COMPREHENSIVE LOSS

The components of other comprehensive loss were as follows (dollars in millions):

Huntsman Corporation

	Accumulated other comprehensive loss					Other compro Three mon		
		arch 31, 2010	D	ecember 31, 2009	N	Aarch 31, 2010	M	arch 31, 2009
Foreign currency translation adjustments, net of tax of \$22 and \$15 as of March 31, 2010 and December 31, 2009, respectively	\$	224	\$	274	\$	(50)	\$	(85)
Pension and other postretirement benefit adjustments, net of tax of \$100 and \$102 as of March 31, 2010 and December 31, 2009, respectively		(574)		(580)		6		7
Other comprehensive income (loss) of unconsolidated affiliates Other, net		7		7 6		_		(3)
Total Amounts attributable to noncontrolling interests		(337)		(293)		(44)		(82)
Amounts attributable to Huntsman Corporation	\$	(331)	\$	(287)	\$	(44)	\$	(82)

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

12. OTHER COMPREHENSIVE LOSS (Continued)

Huntsman International

	Accumulated other		_(Other compr				
		comprel			Three mor		onths ended	
		arch 31, 2010	De	cember 31, 2009	N	Iarch 31, 2010	M	Iarch 31, 2009
Foreign currency translation adjustments, net of tax of \$9 and \$2 as of March 31, 2010 and December 31, 2009, respectively	\$	222	\$	273	\$	(51)	\$	(83)
Pension and other postretirement benefit adjustments, net of tax of \$132 and \$134 as of March 31, 2010 and December 31, 2009, respectively		(627)		(635)		8		7
Other comprehensive income (loss) of unconsolidated affiliates		7		7		_		(3)
Other, net		1		1		_		
Total		(397)		(354)		(43)		(79)
Amounts attributable to noncontrolling interests		6		6		_		_
Amounts attributable to Huntsman International LLC	\$	(391)	\$	(348)	\$	(43)	\$	(79)

Items of other comprehensive loss of our Company and our consolidated affiliates have been recorded net of tax, with the exception of the foreign currency translation adjustments related to subsidiaries with earnings permanently reinvested. The tax effect is determined based upon the jurisdiction where the income or loss was recognized and is net of valuation allowances.

13. COMMITMENTS AND CONTINGENCIES

LEGAL MATTERS

Asbestos Litigation

We have been named as a "premises defendant" in a number of asbestos exposure cases, typically claims by non-employees of exposure to asbestos while at a facility. In the past, these cases typically have involved multiple plaintiffs bringing actions against multiple defendants, and the complaints have not indicated which plaintiffs were making claims against which defendants, where or how the alleged injuries occurred or what injuries each plaintiff claimed. These facts, which would be central to any estimate of probable loss, generally have been learned only through discovery.

Where a claimant's alleged exposure occurred prior to our ownership of the relevant "premises," the prior owners generally have contractually agreed to retain liability for, and to indemnify us against, asbestos exposure claims. This indemnification is not subject to any time or dollar amount limitations. Upon service of a complaint in one of these cases, we tender it to the prior owner. None of the complaints in these cases state the amount of damages being sought. The prior owner accepts responsibility for the conduct of the defense of the cases and payment of any amounts due to the claimants. In our sixteen-year experience with tendering these cases, we have not made any payment

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

13. COMMITMENTS AND CONTINGENCIES (Continued)

with respect to any tendered asbestos cases. We believe that the prior owners have the intention and ability to continue to honor their indemnity obligations, although we cannot assure you that they will continue to do so or that we will not be liable for these cases if they do not

The following table presents for the periods indicated certain information about cases for which service has been received that we have tendered to the prior owner, all of which have been accepted.

	Three n end <u>Marcl</u>	ed
	2010	2009
Unresolved at beginning of period	1,138	1,140
Tendered during period	6	6
Resolved during period(1)	5	10
Unresolved at end of period	1,139	1,136

(1) Although the indemnifying party informs us when tendered cases have been resolved, it generally does not inform us of the settlement amounts relating to such cases, if any. The indemnifying party has informed us that it typically manages our defense together with the defense of other entities in such cases and resolves claims involving multiple defendants simultaneously, and that it considers the allocation of settlement amounts, if any, among defendants to be confidential and proprietary. Consequently, we are not able to provide the number of cases resolved with payment by the indemnifying party or the amount of such payments.

We have never made any payments with respect to these cases. As of March 31, 2010, we had an accrued liability of \$16 million relating to these cases and a corresponding receivable of \$16 million relating to our indemnity protection with respect to these cases. We cannot assure you that our liability will not exceed our accruals or that our liability associated with these cases would not be material to our financial condition, results of operations or liquidity; however, we are not able to estimate the amount or range of loss in excess of our accruals. Additional asbestos exposure claims may be made against us in the future, and such claims could be material. However, because we are not able to estimate the amount or range of losses associated with such claims, we have made no accruals with respect to unasserted asbestos exposure claims as of March 31, 2010.

Certain cases in which we are a "premises defendant" are not subject to indemnification by prior owners or operators. The following table presents for the periods indicated certain information about

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

13. COMMITMENTS AND CONTINGENCIES (Continued)

these cases. Cases include all cases for which service has been received by us. Certain prior cases that were filed in error against us have been dismissed.

	Three n end Marc	ed
	2010	2009
Unresolved at beginning of period	39	43
Filed during the period	1	0
Resolved during period	1	2
Unresolved at end of period	39	41

We paid gross settlement costs for asbestos exposure cases that are not subject to indemnification of \$75,000 during the three months ended March 31, 2010. We cannot assure you that our liability will not exceed our accruals or that our liability associated with these cases would not be material to our financial condition, results of operations or liquidity; however, we are not able to estimate the amount or range of loss in excess of our accruals. Additional asbestos exposure claims may be made against us in the future, and such claims could be material. However, because we are not able to estimate the amount or range of losses associated with such claims, we have made no accruals with respect to unasserted asbestos exposure claims as of March 31, 2010.

Antitrust Matters

We have been named as a defendant in civil class action antitrust suits alleging that between 1999 and 2004 we conspired with Bayer, BASF, Dow and Lyondell to fix the prices of MDI, TDI, polyether polyols, and related systems ("polyether polyol products") sold in the U.S. in violation of the federal Sherman Act. These cases are consolidated as the "Polyether Polyols" cases in multidistrict litigation known as In re Urethane Antitrust Litigation, MDL No. 1616, Civil No. 2:04-md-01616-JWL-DJW, pending in the U.S. District Court for the District of Kansas.

In addition, we and the other Polyether Polyol defendants have also been named as defendants in three civil antitrust suits brought by certain direct purchasers of polyether polyol products that opted out of the class certified in MDL No. 1616. While these opt out plaintiffs make similar claims as the class plaintiffs, the court denied defendants' motion to dismiss claims of improper activity outside the class period. Accordingly, the relevant time frame for these cases is 1994-2006. These cases are referred to as the direct action cases and are pending in the U.S. District Court for the District of New Jersey.

Merits discovery was consolidated in the MDL No. 1616 for both the class and direct action cases and is ongoing. The trial is currently scheduled for February 2012.

Two purported class action cases filed May 5 and 17, 2006 pending in the Superior Court of Justice, Ontario Canada and Superior Court, Province of Quebec, District of Quebec, by direct purchasers of MDI, TDI and polyether polyols and by indirect purchasers of these products remain dormant. A purported class action case filed February 15, 2002 by purchasers of products containing rubber and urethanes products and pending in Superior Court of California, County of San Francisco is stayed pending resolution of MDL No. 1616. Finally, we have been named in a proposed third

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

13. COMMITMENTS AND CONTINGENCIES (Continued)

amended complaint by indirect purchasers of MDI, TDI, polyether polyols and polyester polyols pending against Bayer and Chemtura in the U.S. District Court for the District of Massachusetts. The matter is currently stayed pending a settlement of previously asserted claims against Bayer and Chemtura. We filed papers opposing the motion for leave to file the proposed amended complaint adding us as a defendant in that action. The plaintiffs in each of these matters make similar claims against the defendants as the class plaintiffs in MDL No. 1616.

Huntsman International has been named as a defendant in two purported class action civil antitrust suits alleging that it and its codefendants and other co-conspirators conspired to fix prices of titanium dioxide sold in the U.S. between at least March 1, 2002 and the present. The cases were filed on February 9 and 12, 2010 in the U.S. District Court for the District of Maryland and a consolidated complaint was filed on April 12, 2010. The other defendants named in this matter are E.I. Dupont De Nemours and Company, Kronos Worldwide Inc., Millenium Inorganic Chemicals, Inc. and the National Titanium Dioxide Company Limited (d/b/a Cristal).

In all of the antitrust litigation currently pending against us the plaintiffs generally are seeking injunctive relief, treble damages, costs of suit and attorneys fees.

The plaintiffs' pleadings in these various antitrust suits provide few specifics about any alleged illegal conduct on our part, and we are not aware of any illegal conduct by us or any of our employees. For these reasons, we cannot estimate the possibility of loss or range of loss relating to these claims, and therefore we have not accrued a liability for these claims. Nevertheless, we could incur losses due to these claims in the future and those losses could be material.

MTBE Litigation

We are named as a defendant in 18 lawsuits pending in litigation filed between March 23, 2007 and June 24, 2009 in New York federal and state courts alleging liability related to MTBE contamination in groundwater. Numerous other companies, including refiners, manufacturers and sellers of gasoline, as well as manufacturers of MTBE, were named as defendants in these and many other cases that were pending in U.S. courts. The plaintiffs in the 18 cases in which we are named are municipal water districts, a regional water supply authority and municipal corporations that claim that defendants' conduct has caused MTBE contamination of their groundwater. Four cases are pending in the U.S. District Court for the Southern District of New York and 14 are pending in the Supreme Court of the state of New York, nine in Nassau County and five in Suffolk County. The plaintiffs seek injunctive relief, such as monitoring and abatement, compensatory damages, punitive damages and attorney fees. Together with other defendants, we have filed motions to dismiss all of the state court cases. At this time, while we currently have insufficient information to meaningfully assess our potential exposure in these cases, we have joined with a larger group of defendants in an effort to mediate the plaintiffs' claims. Mediation in late 2008 and early 2009 was unsuccessful. A further mediation session was held February 3, 2010 and resulted in a tentative settlement in each of the cases in which we have been named. Our allocated portion of the total settlement is not material to our ongoing operations. We have accrued a liability for these claims equal to our allocated portion of the settlement.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

13. COMMITMENTS AND CONTINGENCIES (Continued)

Shareholder Litigation

From July 5 to July 13, 2007, four putative shareholder class action complaints were filed against our Company and our directors alleging breaches of fiduciary duty in connection with our then-proposed sale to Basell and the receipt of a superior proposal from Hexion. Three actions were filed in Delaware: Cohen v. Archibald, et al., No. 3070, in the Court of Chancery for the State of Delaware (filed July 5, 2007); Augenstein v. Archibald, et al., No. 3076, in the Court of Chancery for the State of Delaware (filed July 9, 2007); and Murphy v. Huntsman, et al., No. 3094, in the Court of Chancery for the State of Delaware (filed July 13, 2007). Another action was filed in Texas: Schwoegler v. Huntsman Corporation, et al., Cause No. 07-07-06993-CV, in the 9th Judicial District Court of Montgomery County, Texas (filed July 6, 2007). As subsequently amended, these lawsuits together allege that we and our directors breached fiduciary duties to the stockholders by, among other things, engaging in an unfair sales process, approving an unfair price per share for the Hexion Merger, and making inadequate disclosures to stockholders, and that Basell, Hexion and MatlinPatterson entities aided and abetted these breaches of fiduciary duty. The lawsuits sought to enjoin the stockholder vote on the Hexion Merger.

On September 20, 2007, the parties entered into a Memorandum of Understanding with plaintiffs' counsel in the Delaware and Texas actions to settle these four lawsuits. As part of the proposed settlement, the defendants deny all allegations of wrongdoing, but we agreed to make certain additional disclosures in the final proxy statement that was mailed to our stockholders on or about September 14, 2007. In connection with the settlement, the parties also reached an agreement with respect to any application that the plaintiffs' counsel would have made for an award of customary attorneys' fees and expenses to be paid following the completion of the now Terminated Merger.

The Memorandum of Understanding is now null and void and of no force and effect because the Hexion Merger was not consummated. The Texas action has been voluntarily dismissed and the Delaware actions have been administratively closed.

Port Arthur Plant Fire Insurance Litigation

On August 31, 2007, an action was brought against our Company and International Risk Insurance Company ("IRIC"), our captive insurer, in the U.S. District Court for the Southern District of Texas, by seventeen reinsurance companies (the "Reinsurers") that reinsure risks under the property insurance policy issued by IRIC to our Company for the period covering the April 29, 2006 fire at our manufacturing facility in Port Arthur, Texas. The action sought to compel our Company and IRIC to arbitrate with the Reinsurers to resolve disputes related to the claim for losses caused by the fire or, in the alternative, to declare judgment in favor of the Reinsurers. On September 26, 2008, the court denied motions to dismiss filed by our Company and IRIC, ordering the parties to engage in a short period of discovery on the issue of arbitrability. In a second and related action filed by our Company against IRIC in state court in Jefferson County, Texas, IRIC filed a third party petition against the Reinsurers, who then removed that action to the U.S. District Court for the Eastern District of Texas. Some of the Reinsurers filed answers and motions to compel arbitration, to stay these proceedings, and to change venue to the U.S. District Court for the Southern District of Texas in order to consolidate the two actions. We filed a motion to remand that action to the state court and opposition to the

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

13. COMMITMENTS AND CONTINGENCIES (Continued)

Reinsurers' motions in that action. On April 23, 2008, the U.S. District Court for the Eastern District of Texas transferred the case to the U.S. District Court for the Southern District of Texas. On September 26, 2008, the court denied our motion to remand that suit to the state court in which it was filed.

Pursuant to a December 29, 2008 agreement among the parties to the actions referenced above: (1) a mediation was scheduled for February 24-25, 2009, (2) if the disputes were not fully resolved in mediation, the parties would submit all coverage and quantum issues to a three-arbitrator panel in the second half of 2009, with a binding award to be entered by September 30, 2009 (see current status in paragraph below), (3) the Reinsurers paid an additional \$40 million on our claim on December 29, 2008 and agreed that all monies paid by the participating Reinsurers on the claim to date are nonrefundable, (4) we waived our noncontractual claims against the Reinsurers, (5) the first action referenced above is stayed pending final resolution and entry of judgment, and (6) the second action referenced above has been dismissed.

Because the non-binding mediation was not successful, we and the Reinsurers are now participating in binding arbitration which began on November 2, 2009. The binding arbitration is ongoing. Reinsurers responsible for a small percentage of our remaining claim were not parties to the two lawsuits and are not parties to the agreement; thus we may need to pursue actions against them separately for their pro rata shares of the unpaid claim. We have paid our deductible on the claim of \$60 million and have been paid \$365 million to date by the Reinsurers. Prior to the commencement of the arbitration proceedings, we had claimed an additional approximately \$242 million plus interest as presently due and owing and unpaid under the applicable policy (the "Policy") for losses caused by the fire. The arbitration panel has made a preliminary ruling disallowing our claim for interest. In addition, the arbitration panel has made certain preliminary rulings on some of the discrete issues that so far effectively reduce the overall amount we have claimed by approximately \$70 million. The arbitration panel has requested additional briefing on the remaining matters and has not indicated when a final ruling will be issued. See "Note 16. Casualty Losses and Insurance Recoveries—Port Arthur, Texas Plant Fire."

Other Proceedings

We are a party to various other proceedings instituted by private plaintiffs, governmental authorities and others arising under provisions of applicable laws, including various environmental, products liability and other laws. Except as otherwise disclosed in this report, we do not believe that the outcome of any of these matters will have a material adverse effect on our financial condition, results of operations or liquidity.

Guarantees

Our unconsolidated joint venture, the Arabian Amines Company, obtained various loan commitments in the aggregate amount of approximately \$195 million (U.S. dollar equivalents) of which \$177 million, including bridge loans, was drawn and outstanding as of March 31, 2010. We have provided certain guarantees of approximately \$14 million for these commitments and our guarantees will terminate upon completion of the project and satisfaction of certain other conditions. We have

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

13. COMMITMENTS AND CONTINGENCIES (Continued)

estimated that the fair value of such guarantees was nil as of the closing date of this transaction and, accordingly, no amounts have been recorded.

14. ENVIRONMENTAL, HEALTH AND SAFETY MATTERS

General

We are subject to extensive federal, state, local and foreign laws, regulations, rules and ordinances relating to safety, pollution, protection of the environment and the generation, storage, handling, transportation, treatment, disposal and remediation of hazardous substances and waste materials. In the ordinary course of business, we are subject to frequent environmental inspections and monitoring and occasional investigations by governmental enforcement authorities. In addition, our production facilities require operating permits that are subject to renewal, modification and, in certain circumstances, revocation. Actual or alleged violations of safety laws, environmental laws or permit requirements could result in restrictions or prohibitions on plant operations, substantial civil or criminal sanctions, as well as, under some environmental laws, the assessment of strict liability and/or joint and several liability. Moreover, changes in environmental regulations could inhibit or interrupt our operations, or require us to modify our facilities or operations. Accordingly, environmental or regulatory matters may cause us to incur significant unanticipated losses, costs or liabilities.

Environmental, Health and Safety Systems

We are committed to achieving and maintaining compliance with all applicable EHS legal requirements, and we have developed policies and management systems that are intended to identify the multitude of EHS legal requirements applicable to our operations, enhance compliance with applicable legal requirements, ensure the safety of our employees, contractors, community neighbors and customers and minimize the production and emission of wastes and other pollutants. Although EHS legal requirements are constantly changing and are frequently difficult to comply with, these EHS management systems are designed to assist us in our compliance goals while also fostering efficiency and improvement and minimizing overall risk to us.

EHS Capital Expenditures

We may incur future costs for capital improvements and general compliance under EHS laws, including costs to acquire, maintain and repair pollution control equipment. For the three months ended March 31, 2010 and 2009, our capital expenditures for EHS matters totaled \$9 million and \$8 million, respectively. Since capital expenditures for these matters are subject to evolving regulatory requirements and depend, in part, on the timing, promulgation and enforcement of specific requirements, we cannot provide assurance that our recent expenditures will be indicative of future amounts required under EHS laws.

Remediation Liabilities

We have incurred, and we may in the future incur, liability to investigate and clean up waste or contamination at our current or former facilities or facilities operated by third parties at which we may

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

14. ENVIRONMENTAL, HEALTH AND SAFETY MATTERS (Continued)

have disposed of waste or other materials. Similarly, we may incur costs for the cleanup of wastes that were disposed of prior to the purchase of our businesses. Under some circumstances, the scope of our liability may extend to damages to natural resources.

Under CERCLA and similar state laws, a current or former owner or operator of real property may be liable for remediation costs regardless of whether the release or disposal of hazardous substances was in compliance with law at the time it occurred, and a current owner or operator may be liable regardless of whether it owned or operated the facility at the time of the release. We have been notified by third parties of claims against us for cleanup liabilities at approximately 10 former facilities or third party sites, including, but not limited to, sites listed under CERCLA. Based on current information and past experiences at other CERCLA sites, we do not expect any of these third party claims to result in material liability to us. Outside the U.S., analogous contaminated property laws, such as those in effect in France and Australia, can hold past owners and/or operators liable for remediation at former facilities.

One of these sites, the North Maybe Canyon Mine CERCLA site, includes an abandoned phosphorous mine near Soda Springs, Idaho that may have been operated by one of our predecessor companies (El Paso Products Company). In 2004, the U.S. Forest Service notified us that we are a CERCLA Potentially Responsible Party (a "PRP") for the mine site involving selenium-contaminated surface water. Under a 2004 administrative order, the current mine lessee, Nu-West Industries, Inc., began undertaking the investigation required for a CERLA removal process. In 2008, the site was transitioned to the CERCLA remedial action process, which requires a Remedial Investigation/Feasibility Study (an "RI/FS"). In 2009, the Forest Service notified the three PRPs (our Company, Nu-West, and Wells Cargo) that it would undertake the RI/FS itself. On February 19, 2010, in conjunction with Wells Cargo, we agreed to jointly comply with a unilateral administrative order (a "UAO") to conduct an RI/FS of the entire West Ridge of the site, although we are alleged to have had only a limited historical presence in the investigation area. The UAO has not yet been executed by the Forest Service. In March 2010, following the initiation of litigation by Nu-West, the Forest Service assumed Nu-West's original investigation obligations. We continue to coordinate with our insurers regarding policy coverage in this matter. At this time, we are unable to estimate the cost of the RI/FS or our ultimate liability in this matter, but we do not believe it will be material to our financial condition.

In addition, under RCRA, and similar state laws, we may be required to remediate contamination originating from our properties as a condition to our hazardous waste permit. Some of our manufacturing sites have an extended history of industrial chemical manufacturing and use, including on-site waste disposal. We are aware of soil, groundwater or surface contamination from past operations at some of our sites, and we may find contamination at other sites in the future. For example, our Port Neches, Texas, and Geismar, Louisiana, facilities are the subject of ongoing remediation requirements under RCRA authority. Similar laws exist in a number of locations in which we currently operate manufacturing facilities, such as Australia, Switzerland and Italy.

In June of 2006, an agreement was reached between the local regulatory authorities and our Advanced Materials site in Pamplona, Spain to relocate our manufacturing operations in order to facilitate new urban development desired by the city. Subsequently, as required by the authorities, soil

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

14. ENVIRONMENTAL, HEALTH AND SAFETY MATTERS (Continued)

and groundwater sampling was performed and followed by a quantitative risk assessment. Although unresolved at this time, some level of remediation of site contamination may be required in the future, but the estimated cost is unknown because the remediation approach and timing has not been determined.

By letter dated March 7, 2006, our Base Chemicals and Polymers facility in West Footscray, Australia, was issued a clean-up notice by the Australian (Victorian) EPA due to concerns about soil and groundwater contamination emanating from the site. The agency revoked the original clean-up notice on September 4, 2007 and issued a revised clean-up notice due to "the complexity of contamination issues" at the site. On March 31, 2009, we submitted the required site remediation action plan to the agency which proposed additional investigation and remediation method trials. We can provide no assurance that the EPA will agree with our proposed plan, will not seek to institute additional requirements for the site or that additional costs will not be associated with the clean up. Additionally, on September 8, 2009, we announced a decision to close this facility in early 2010. In the third quarter of 2009, we recorded a \$30 million liability related to estimated environmental remediation costs at this site.

In many cases, our potential liability arising from historical contamination is based on operations and other events occurring prior to our ownership of a business or specific facility. In these situations, we frequently obtained an indemnity agreement from the prior owner addressing remediation liabilities arising from pre-closing conditions. We have successfully exercised our rights under these contractual covenants for a number of sites and, where applicable, mitigated our ultimate remediation liability. We cannot assure you, however, that all of such matters will be subject to indemnity, that the prior owner will honor its indemnity or that our existing indemnities will be sufficient to cover our liabilities for such matters.

By letter of March 15, 2010, the United States Department of Justice (the "DOJ") notified us that the U.S. EPA has requested that the DOJ bring an action in federal court against us and other PRPs for recovery of costs incurred by the U.S. in connection with releases of hazardous substances from the State Marine Superfund Site in Port Arthur, Texas. As of August 31, 2007, the EPA had incurred and paid approximately \$2.8 million in unreimbursed response costs related to the site. Prior to filing the complaint, the DOJ requested that PRPs sign and return a standard tolling agreement (from March 31, 2010 through September 30, 2010) and participate in settlement discussions. Our Company originally responded to an information request regarding this site on March 7, 2005 and identified historical transactions associated with a predecessor of a company we acquired. The prior owners have contractually agreed to indemnify us in this matter. While the DOJ is aware of the indemnity, we may be required to participate in future settlement discussions, therefore on March 29, 2010, we submitted the signed tolling agreement and offer to negotiate to the DOJ.

Based on available information and the indemnification rights we believe are likely to be available, we believe that the costs to investigate and remediate known contamination will not have a material adverse effect on our financial condition, results of operations or cash flows. However, if such indemnities are unavailable or do not fully cover the costs of investigation and remediation or we are required to contribute to such costs, and if such costs are material, then such expenditures may have a material adverse effect on our financial condition, results of operations or cash flows. At the current

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

14. ENVIRONMENTAL, HEALTH AND SAFETY MATTERS (Continued)

time, we are unable to estimate the full cost, exclusive of indemnification benefits, to remediate any of the known contamination sites.

Environmental Reserves

We have accrued liabilities relating to anticipated environmental cleanup obligations, site reclamation and closure costs and known penalties. Liabilities are recorded when potential liabilities are either known or considered probable and can be reasonably estimated. Our liability estimates are calculated using present value techniques and are based upon requirements placed upon us by regulators, available facts, existing technology and past experience. The environmental liabilities do not include amounts recorded as asset retirement obligations. We had accrued \$41 million each for environmental liabilities as of both March 31, 2010 and December 31, 2009. Of these amounts, \$7 million and \$5 million were classified as accrued liabilities in our consolidated balance sheets as of March 31, 2010 and December 31, 2009, respectively, and \$34 million and \$36 million were classified as other noncurrent liabilities in our consolidated balance sheets as of March 31, 2010 and December 31, 2009, respectively. In certain cases, our remediation liabilities may be payable over periods of up to 30 years. We may incur losses for environmental remediation in excess of the amounts accrued; however, we are not able to estimate the amount or range of such potential excess.

REGULATORY DEVELOPMENTS

In December 2006, the EU parliament and EU council approved a new EU regulatory framework for chemicals called "REACH" (Registration, Evaluation and Authorization of Chemicals). REACH took effect on June 1, 2007, and the program it establishes will be phased in over 11 years. Under the regulation, companies that manufacture in or import into the EEA more than one metric tonne of a chemical substance per year will be required to register such chemical substances and isolated intermediates in a central database. Use authorizations will be granted for a specific chemical if the applicants can show that the risks in using the chemical are adequately controlled; and for chemicals where there are no suitable alternatives substances or technologies available and the applicant can demonstrate that the social and economic benefits of using the chemical outweigh the risks. In addition, specified uses of some hazardous substances may be restricted. Furthermore, all applicants will have to study the availability of alternative chemicals. If an alternative is available, an applicant will have to submit a "substitution" plan to the regulatory agency. The regulatory agency will only authorize persistent bio-accumulative and toxic substances if an alternative chemical is not available. The registration, evaluation and authorization phases of the program will require expenditures and resource commitments in order to, for example, participate in mandatory data-sharing forums; acquire, generate and evaluate data; prepare and submit dossiers for substance registration; obtain legal advice and reformulate products, if necessary. We have established a cross-business European REACH team that is working closely with our businesses to identify and list all substances we purchase or manufacture in, or import into, the EEA. We met pre-registration REACH compliance requirements by the November 30, 2008 regulatory deadline, with the exception of pre-registrations for two substances, for a total of 1,850 preregistrations for substances that we intend to register. We are currently proceeding with the registration of the two substances as provided for under REACH, as well as of the high-volume and high-priority chemicals under the program, which must be registered no later than

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

14. ENVIRONMENTAL, HEALTH AND SAFETY MATTERS (Continued)

November 30, 2010. Although the total long-term cost for REACH compliance is not estimable at this time, we spent approximately \$3 million, \$2 million and \$3 million during the years ended December 31, 2009, 2008 and 2007, respectively, on REACH compliance.

GREENHOUSE GAS REGULATION

In the EU and other jurisdictions committed to compliance with the Kyoto Protocol, there is an increasing likelihood that our manufacturing sites will be affected in some way over the next few years by regulation or taxation of GHG emissions. For example, Australia recently proposed its Carbon Pollution Reduction Scheme, which may impact our Australian operations, and program implementation is currently scheduled for 2011. In addition, although the U.S. is not a signatory to the Kyoto Protocol, several states are implementing their own GHG regulatory programs and a federal program in the U.S. is likely for the future. Draft U.S. federal legislation and the recent U.S. EPA Clean Air Act endangerment findings for carbon dioxide have focused corporate attention on the eventuality of measuring and reporting of GHG emissions for operations in the U.S. The U.S. EPA also recently mandated GHG reporting requirements for U.S. sources in excess of 25,000 tons beginning in 2010. Final details of a comprehensive U.S. GHG management approach are, as yet, uncertain. Nevertheless, we are already managing and reporting GHG emissions, to varying degrees, as required by law for our sites in locations subject to Kyoto Protocol obligations and/or EU emissions trading scheme requirements. Although these sites are subject to existing GHG legislation, few have experienced or anticipate significant cost increases as a result, although it is likely that GHG emission restrictions will increase over time. Potential consequences of such restrictions include capital requirements to modify assets used to meet GHG restriction and/or increases in energy costs above the level of general inflation, as well as direct compliance costs. Currently, however, it is not possible to estimate the likely financial impact of potential future regulation on any of our sites. Finally, it should be noted that some scientists have concluded that increasing concentrations of greenhouse gases in the Earth's atmosphere may produce climate changes that have significant physical effects, such as increased frequency and severity of storms, droughts, and floods and other climatic events. If any of those effects were to occur, they could have an adverse effect on our assets and operations.

CHEMICAL FACILITY ANTI-TERRORISM RULEMAKING

The DHS issued the final rule of their "Chemical Facility Anti-Terrorism Standard" in 2007. The initial phase of the rule required all chemical facilities in the U.S. to evaluate their facilities against the DHS Appendix A list of "Chemicals of Interest." Facilities which have specified chemicals in designated quantities on the Appendix A list were required to submit a "Top Screen" to DHS in 2008. A Top Screen is a questionnaire completed by a facility having Chemicals of Interest in designated threshold quantities. In early 2008, we submitted Top Screens for all of our covered facilities. After reviewing the Top Screens, the DHS determined that four of our sites were "High Risk" facilities. As a result, we were required to perform security vulnerability assessments at the High Risk sites. The security vulnerability assessments were completed and sent to DHS during the fourth quarter of 2008. Based on their assessment of the security vulnerability assessments, we received notice from DHS that one of our sites was elevated to a higher security risk tier. The DHS determined the other three sites to be low security risk tiers. The three lower-tiered sites have submitted Site Security Plans ("SSPs") to

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

14. ENVIRONMENTAL, HEALTH AND SAFETY MATTERS (Continued)

the DHS. The SSPs are based on a list of 18 risk-based performance standards, but security improvements recommended from the SSPs are not anticipated to be material. The tier-elevated site has also submitted an SSP to the DHS, and we believe that security upgrades to the that site will be required; however we do not know what these required updates will be and thus cannot reasonably estimate associated costs at this time, but they could be material. Additionally, on November 26, 2008, the Transportation Safety Administration of the DHS published a final rule regarding "rail security sensitive materials" that are received at, or shipped from, facilities. We have three sites that are subject to this new rule, but at this time do not anticipate that the costs to comply will be material.

MTBE DEVELOPMENTS

We produce MTBE, an oxygenate that is blended with gasoline to reduce vehicle air emissions and to enhance the octane rating of gasoline. Litigation or legislative initiatives restricting the use of MTBE in gasoline may subject us or our products to environmental liability or materially adversely affect our sales and costs. Because MTBE has contaminated some water supplies, its use has become controversial in the U.S. and elsewhere, and its use has been effectively eliminated in the U.S. market. We currently market MTBE, either directly or through third parties, to gasoline additive customers located outside the U.S., although there are additional costs associated with such outside-U.S. sales which may result in decreased profitability compared to historical sales in the U.S. We may also elect to use all or a portion of our precursor tertiary butyl alcohol to produce saleable products other than MTBE. If we opt to produce products other than MTBE, necessary modifications to our facilities will require significant capital expenditures and the sale of such other products may produce a lower level of cash flow than that historically produced from the sale of MTBE.

Numerous companies, including refiners, manufacturers and sellers of gasoline, as well as manufacturers of MTBE, have been named as defendants in more than 150 cases in U.S. courts that allege MTBE contamination in groundwater. Many of these cases were settled after the parties engaged in mediation supervised by a court-appointed special settlement master. Beginning in March 2007 and continuing through June 24, 2009, we have been named as a defendant in 18 of these lawsuits pending in New York state and federal courts. See "Note 13. Commitment and Contingencies—Legal Matters—MTBE Litigation." The plaintiffs in the MTBE groundwater contamination cases generally seek compensatory damages, punitive damages, injunctive relief, such as monitoring and abatement, and attorney fees. While we currently have insufficient information to meaningfully assess our potential exposure in these cases, we have joined with a larger group of defendants in an effort to mediate the plaintiffs' claims. Mediation in late 2008 and early 2009 was unsuccessful. A further mediation session was held February 3, 2010 which resulted in a tentative settlement in each of the cases in which we have been named. Our allocated portion of the total settlement amount is not material and we have accrued a liability for the claims equal to our allocated portion of the settlement. It is possible that we could be named as a defendant in additional existing or future MTBE contamination cases. We cannot provide assurances that adverse results against us in existing or future MTBE contamination cases will not have a material adverse effect on our business, results of operations and financial position.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

14. ENVIRONMENTAL, HEALTH AND SAFETY MATTERS (Continued)

INDIA INVESTIGATION

We have initiated an internal investigation of the operations of Petro Araldite Pvt. Ltd. (PAPL), our majority owned joint venture in India. PAPL manufactures and markets base liquid resins, base solid resins and formulated products in India. The investigation, which is ongoing, initially focused on allegations of illegal disposal of hazardous waste and waste water discharge and related reporting irregularities. Based upon preliminary findings, the investigation was expanded to include a review of certain product sales made by PAPL outside normal distribution channels as well as the accounting for the revenues from such sales and the legality under Indian law and U.S. law, including the U.S. Foreign Corrupt Practices Act, of certain payments (less than \$15,000 specifically identified to date) made by employees of the joint venture to government officials in India.

Even though we have not completed the investigation, we have voluntarily contacted the regional pollution control regulators in India, the U.S. Securities and Exchange Commission and the U.S. Department of Justice to advise them of our ongoing investigation and that we intend to cooperate fully with each of them. In addition, we have taken action to halt all known illegal activity.

The internal investigation is not complete and no conclusion can be drawn at this time as to whether any government agencies will open formal investigations of these matters or what remedies such agencies may seek. Governmental agencies could assess material civil and criminal penalties and fines against PAPL and potentially against us and could issue orders that adversely effect the operations of PAPL. We cannot, however, determine at this time the magnitude of the penalties and fines that will be assessed, the costs to remediate the prior noncompliance, or the effects of implementing any necessary corrective measures on the joint venture's operations.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

15. STOCK-BASED COMPENSATION PLANS

Under the Huntsman Stock Incentive Plan (the "Stock Incentive Plan"), a plan approved by stockholders, we may grant non-qualified stock options, incentive stock options, stock appreciation rights, restricted stock, phantom stock, performance awards and other stock-based awards to our employees, directors and consultants and to employees and consultants of our subsidiaries, provided that incentive stock options may be granted solely to employees. The terms of the grants are fixed at the grant date. We were authorized to grant up to 32.6 million shares under the Stock Incentive Plan. As of March 31, 2010, we had 12.1 million shares remaining under the Stock Incentive Plan available for grant. Option awards have a maximum contractual term of 10 years and generally must have an exercise price at least equal to the market price of our common stock on the date the option award is granted. Stock-based awards generally vest over a three-year period.

The compensation cost under the Stock Incentive Plan for our Company and Huntsman International was as follows (dollars in millions):

			onths irch 31,	,	
	2010		2009	9	
Huntsman Corporation	\$	7	\$	5	
Huntsman International		6		1	

The total income tax benefit recognized in the statements of operations for us and Huntsman International for stock-based compensation arrangements was \$2 million each for each of the three months ended March 31, 2010 and 2009.

STOCK OPTIONS

The fair value of each stock option award is estimated on the date of grant using the Black-Scholes valuation model that uses the assumptions noted in the following table. Expected volatilities are based on the historical volatility of our common stock through the grant date. The expected term of options granted was estimated based on the contractual term of the instruments and employees' expected exercise and post-vesting employment termination behavior. The risk-free rate for periods within the contractual life of the option was based on the U.S. Treasury yield curve in effect at the time of grant. The assumptions noted below represent the weighted average of the assumptions utilized for stock options granted during the period.

	Three months ended March 31,		
	2010	2009	
Dividend yield	3.0%	15.4%	
Expected volatility	69.0%	70.4%	
Risk-free interest rate	3.1%	2.5%	
Expected life of stock options granted during the period	6.6 years	6.6 years	

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

15. STOCK-BASED COMPENSATION PLANS (Continued)

A summary of stock option activity under the Stock Incentive Plan as of March 31, 2010 and changes during the three months then ended is presented below:

Option Awards	Shares (in thousands)	Weighted Average Exercise Price	Weighted Average Remaining Contractual Term (years)	Aggregate Intrinsic Value (in millions)
Outstanding at January 1,				
2010	11,677	\$ 11.30		
Granted	654	13.50		
Exercised	(863)	2.59		
Forfeited	(75)	21.27		
Outstanding at March 31,				
2010	11,393	12.02	7.6	\$ 51
Exercisable at March 31, 2010	6,771	17.29	6.5	14

The weighted-average grant-date fair value of stock options granted during the three months ended March 31, 2010 was \$6.97 per option. As of March 31, 2010, there was \$6 million of total unrecognized compensation cost related to nonvested stock option arrangements granted under the Stock Incentive Plan. That cost is expected to be recognized over a weighted-average period of approximately 1.5 years.

The total intrinsic value of stock options exercised during the three months ended March 31, 2010 was \$9 million. During the three months ended March 31, 2009, no stock options were exercised.

NONVESTED SHARES

Nonvested shares granted under the Stock Incentive Plan consist of restricted stock, which is accounted for as an equity award, and phantom stock, which is accounted for as a liability award

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

15. STOCK-BASED COMPENSATION PLANS (Continued)

because it can be settled in either stock or cash. A summary of the status of our nonvested shares as of March 31, 2010 and changes during the three months then ended is presented below:

	Equity Aw	ards	Liability Awards				
	Shares (in thousands)	Weighted Average Grant-Date Fair Value	Shares (in thousands)	Weighted Average Grant-Date Fair Value			
Nonvested at							
January 1, 2010	3,428	\$ 5.20	1,880	\$ 3.61			
Granted	828	13.50	471	13.50			
)						
Vested	(1,161(1)	6.89	(642)	4.27			
Forfeited	(23)	2.59	_	_			
Nonvested at March 31, 2010	3,072	6.82	1,709	6.09			

⁽¹⁾ As of March 31, 2010, a total of 321,529 restricted stock units were vested, of which 58,130 vested during the three months ended March 31, 2010. These shares have not been reflected as vested shares in this table because, in accordance with the restricted stock unit agreements, shares of common stock are not issued for vested restricted stock units until termination of employment.

As of March 31, 2010, there was \$37 million of total unrecognized compensation cost related to nonvested share compensation arrangements granted under the Stock Incentive Plan. That cost is expected to be recognized over a weighted-average period of approximately 2.2 years. The value of share awards that vested during the three months ended March 31, 2010 and 2009 was \$18 million and \$11 million, respectively.

16. CASUALTY LOSSES AND INSURANCE RECOVERIES

PORT ARTHUR, TEXAS PLANT FIRE

On April 29, 2006, our former Port Arthur, Texas olefins manufacturing plant (which we sold in November 2007) experienced a major fire. With the exception of cyclohexane operations at the site, which were restarted in June 2006, the operations at the site were shutdown until the fourth quarter of 2007. The Port Arthur manufacturing plant was covered by property damage and business interruption insurance. With respect to coverage for this outage, the deductible for property damage is \$10 million and business interruption coverage does not apply for the first 60 days, subject to a combined deductible for property damage and business interruption of \$60 million. See "Note 13. Commitments and Contingencies—Port Arthur Texas Plant Fire Litigation." Future collections on this insured loss, if any, will represent additional income from discontinued operations for us upon final settlement and will be used to repay secured debt in accordance with relevant provisions of agreements governing our Senior Credit Facilities.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

16. CASUALTY LOSSES AND INSURANCE RECOVERIES (Continued)

2005 U.S. GULF COAST STORMS

On September 22, 2005, we sustained property damage at our Port Neches and Port Arthur, Texas facilities as a result of a hurricane. We maintain customary insurance coverage for property damage and business interruption. With respect to coverage of these losses, the deductible for property damage was \$10 million, while business interruption coverage does not apply for the first 60 days.

Through March 31, 2010, we have recorded \$48 million in payments in connection with our insurance claim for property damage and business interruption losses from the 2005 Gulf Coast storms. On July 29, 2009, the reinsurers filed a declaratory judgment action asking the Court to compel arbitration between the parties or to declare that the reinsurers owed nothing further from the storm damage. We filed a counterclaim asking the Court to declare that the reinsurers owed us the remaining amount of our claim. Subsequently, the parties participated in mediation on February 8-9, 2010, and signed a proof of loss and a full and final settlement agreement on March 22, 2010 that resolved the remainder of our insurance claim for a total of \$7 million. The reinsurers paid this amount within 30 days following the execution of the proof of loss and full and final settlement agreement. In connection with this settlement, we recorded a \$7 million pre-tax gain in discontinued operations during the first quarter of 2010, which is included in the \$48 million of insurance recoveries referenced above.

17. INCOME (EXPENSES) ASSOCIATED WITH THE TERMINATED MERGER AND RELATED LITIGATION

On July 12, 2007, we entered into an agreement and plan of merger with Hexion (the "Terminated Merger Agreement" or the "Hexion Merger Agreement"). On June 18, 2008, Hexion, Apollo and certain of their affiliates filed an action in Delaware Chancery Court seeking to terminate the Hexion Merger. We countersued Hexion and Apollo in the Delaware Chancery Court and filed a separate action against Apollo and certain of its affiliates in the District Court of Montgomery County, Texas. On December 13, 2008, we terminated the Hexion Merger Agreement and, on December 14, 2008, we entered into the Apollo Settlement Agreement to settle the Terminated Merger-related litigation and certain other related matters. Pursuant to the Apollo Settlement Agreement, Hexion and certain Apollo affiliates have paid us an aggregate of \$1 billion.

On September 30, 2008, we filed suit in the 9th Judicial District Court in Montgomery County, Texas against the Banks that had entered into a commitment letter to provide funding for the Hexion Merger. On June 22, 2009, we entered into the Texas Bank Litigation Settlement Agreement with the Banks. The Texas Bank Litigation was dismissed with prejudice on June 23, 2009. In accordance with the Texas Bank Litigation Settlement Agreement, the Banks paid us a cash payment of \$632 million, purchased the \$600 million aggregate principal amount $5^{1}/2\%$ senior notes due 2016 (the "2016 Senior Notes") from Huntsman International, and provided Huntsman International with Term Loan C in the principal amount of \$500 million. The 2016 Senior Notes and Term Loan C borrowings were at favorable rates to us and were recorded at a combined fair value of \$864 million. Accordingly, we recognized a gain of \$868 million in connection with the Texas Bank Litigation Settlement Agreement.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

17. INCOME (EXPENSES) ASSOCIATED WITH THE TERMINATED MERGER AND RELATED LITIGATION (Continued)

Expenses associated with the Terminated Merger and related litigation incurred during the three months ended March 31, 2009 of \$7 million related principally to legal fees incurred in connection with the Texas Bank Litigation.

18. INCOME TAXES

For the three months ended March 31, 2010, in accordance with U.S. GAAP, we have computed our provision for income taxes based on the actual effective tax rate for the year-to-date by applying the discrete method. The discrete method was used to calculate the income tax provision as the annual effective tax rate was not considered a reliable estimate of year-to-date income tax expense.

We use the asset and liability method of accounting for income taxes. Deferred income taxes reflect the net tax effects of temporary differences between the carrying amounts of assets and liabilities for financial and tax reporting purposes. We evaluate deferred tax assets to determine whether it is more likely than not that they will be realized. Valuation allowances are reviewed on a tax jurisdictional basis to analyze whether there is sufficient positive or negative evidence to support a change in judgment about the realizability of the related deferred tax assets for each jurisdiction. During the three months ended March 31, 2010, we released a valuation allowance of \$14 million on our Australia net deferred tax assets, primarily as a result of discontinuing the operations of the styrenics business.

During the three months ended March 31, 2010, we recorded a net decrease in unrecognized tax benefits with a corresponding income tax benefit of approximately \$8 million, resulting from the settlement of tax audits, the effective settlement of certain tax positions and the expiration of statutes of limitations. During the three months ended March 31, 2009, we recorded a decrease in unrecognized tax benefits with a corresponding income tax benefit of approximately \$18 million, resulting from of the effective settlement of prior year examinations.

HUNTSMAN CORPORATION

In addition to the tax benefits of the valuation allowance release and unrecognized tax benefit items discussed above, during the three months ended March 31, 2010, we recognized \$12 million of tax benefit on the \$155 million of loss on early extinguishment of debt (the majority of the loss is not tax deductible for tax purposes) and during the three months ended March 31, 2009, we recorded a valuation allowance of \$146 million against the net deferred tax assets in the U.K. Excluding these items, we recorded income tax expense of nil and \$10 million for the three months ended March 31, 2010 and 2009, respectively. Our tax obligations are affected by the mix of income and losses in the tax jurisdictions in which we operate.

HUNTSMAN INTERNATIONAL

In addition to the tax benefits of the valuation allowance release and unrecognized tax benefit items discussed above during the three months ended March 31, 2010, Huntsman International recognized \$3 million of tax benefit on the \$9 million of loss on early extinguishment of debt (the majority of the loss is not tax deductible for tax purposes) and during the three months ended

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

18. INCOME TAXES (Continued)

March 31, 2009, Huntsman International recorded a valuation allowance of \$156 million as of March 31, 2009 against the net deferred tax assets in the U.K. Excluding these items, Huntsman International recorded income tax expense of \$3 million and \$8 million, for the three months ended March 31, 2010 and 2009, respectively. Our tax obligations are affected by the mix of income and losses in the tax jurisdictions in which we operate.

19. DISCONTINUED OPERATIONS

Australian Styrenics Business Shutdown

During the first quarter of 2010, we ceased operation of our former Australian styrenics business. The following results of operations of our former Australian styrenics business have been presented as discontinued operations in the accompanying condensed consolidated statements of operations (unaudited):

		Three mont ended March	
	2	010 2	2009
Revenues	\$	25 \$	13
Costs and expenses		(45)	(20)
Operating loss		(20)	(7)
Income tax benefit		8	_
Loss from discontinued operations, net of tax	\$	(12) \$	(7)

Beginning in the first quarter of 2010, the EBITDA of our former Australian styrenics business was included in discontinued operations for all periods presented.

U.S. Base Chemicals and North American Polymers Dispositions

In 2007, we completed the sale of our former U.S. base chemicals business (the "U.S. Base Chemicals Disposition") and our North American polymers business assets (the "North American Polymers Disposition"). The results of these former businesses are presented as discontinued operations in the accompanying condensed consolidated statements of operations (unaudited).

During the three months ended March 31, 2010, we recorded an after tax loss from discontinued operations of \$1 million, which consisted of \$8 million of loss resulting principally from legal fees related to the ongoing arbitration of the fire insurance claim, offset in part by a \$7 million pretax gain from the settlement of insurance claims related to the 2005 U.S. gulf coast storms. See "Note 16. Casualty Losses and Insurance Recoveries." During the three months ended March 31, 2009, we recorded after tax income from discontinued operations of \$3 million related primarily to the revaluation of outstanding product exchange liabilities related to our former U.S. base chemicals business.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

20. NET LOSS PER SHARE

Basic (loss) income per share excludes dilution and is computed by dividing net loss attributable to Huntsman Corporation common stockholders by the weighted average number of shares outstanding during the period. Diluted loss per share reflects potential dilution and is computed by dividing net loss available to common stockholders by the weighted average number of shares outstanding during the period, increased by the number of additional shares that would have been outstanding if the potential dilutive units had been exercised or converted.

On December 23, 2008, we issued the Convertible Notes in an aggregate principal amount of \$250 million. Prior to their repurchase, the Convertible Notes were convertible into common stock at a conversion price of \$7.857 per share, subject to certain anti-dilution adjustments. On January 11, 2010, we repurchased the entire \$250 million principal amount of the Convertible Notes.

Basic and diluted loss per share is determined using the following information:

	Three Mont March 2010	
Numerator:		
Basic and diluted loss from continuing operations:		
Loss from continuing operations attributable to Huntsman Corporation	\$ (159)	\$ (286)
Convertible notes interest expense, net of tax	_	_
Loss from continuing operations attributable to Huntsman Corporation and assumed conversion	(159)	(286)
Basic and diluted net loss:		
Net loss attributable to Huntsman Corporation	(172)	(290)
Convertible notes interest expense, net of tax	_	_
Net loss attributable to Huntsman Corporation and assumed conversion	\$ (172)	\$ (290)
Shares (denominator):		
Weighted average shares outstanding	234.8	233.7
Dilutive securities:		
Stock-based awards	_	_
Convertible notes conversion	_	_
Total dilutive shares outstanding assuming conversion	234.8	233.7

Additional stock-based awards of 16.9 million and 10.8 million weighted average equivalent shares of stock were outstanding during the three months ended March 31, 2010 and 2009, respectively. In addition, the Convertible Notes would have had a weighted average effect of 3.6 million shares and 31.8 million shares of common stock for the three months ended March 31, 2010 and 2009, respectively, and interest expense, net of tax, of \$1 million and \$5 million would have been included as an adjustment to the numerator of the diluted loss per share calculation for the three months ended March 31, 2010 and 2009, respectively. However, these stock-based awards and the potential effect of assumed conversion of the Convertible Notes were not included in the computation of diluted earnings per share because the effect would be anti-dilutive.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

21. OPERATING SEGMENT INFORMATION

We derive our revenues, earnings and cash flows from the manufacture and sale of a wide variety of differentiated and commodity chemical products. We have reported our operations through five segments: Polyurethanes, Performance Products, Advanced Materials, Textile Effects and Pigments. We have organized our business and derived our operating segments around differences in product lines.

During the first quarter of 2010, we began reporting our LIFO inventory valuation reserves as part of Corporate and other; these reserves were previously reported in our Performance Products segment. Also during the first quarter of 2010, we began reporting the EBITDA of our former Australian styrenics business in discontinued operations, which was previously included in Corporate and other. All segment information for prior periods has been restated to reflect this change.

The major products of each reportable operating segment are as follows:

Segment	Products
Polyurethanes	MDI, PO, polyols, PG, TPU, aniline and MTBE
Performance Products	amines, surfactants, LAB, maleic anhydride, other performance chemicals, EG, olefins and technology licenses
Advanced Materials	epoxy resin compounds and formulations; cross-linking, matting and curing agents; epoxy, acrylic and polyurethane- based adhesives and tooling resin formulations
Textile Effects	textile chemicals and dyes
Pigments	titanium dioxide
	59

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

21. OPERATING SEGMENT INFORMATION (Continued)

Sales between segments are generally recognized at external market prices and are eliminated in consolidation. We use EBITDA to measure the financial performance of our global business units and for reporting the results of our operating segments. This measure includes all operating items relating to the businesses. The EBITDA of operating segments excludes items that principally apply to our Company as a whole. The revenues and EBITDA for each of our reportable operating segments are as follows (dollars in millions):

	 Three mon March	
	2010	 2009
Net Sales:		
Polyurethanes	\$ 767	\$ 600
Performance Products	616	500
Advanced Materials	291	257
Textile Effects	195	152
Pigments	269	196
Eliminations	(44)	(25)
Total	\$ 2,094	\$ 1,680
Huntsman Corporation:		
Segment EBITDA(1):		
Polyurethanes	\$ 52	\$ 26
Performance Products	60	63
Advanced Materials	33	10
Textile Effects	_	(11)
Pigments	28	(29)
Corporate and other(2)	(207)	(26)
Subtotal	(34)	33
Discontinued Operations(3)	(21)	(3)
Total	(55)	30
Interest expense, net	(61)	(55)
Income tax benefit (expense)—continuing operations	34	(138)
Income tax benefit (expense)—discontinued operations	8	(1)
Depreciation and amortization	(98)	(126)
Net loss attributable to		
Huntsman Corporation	\$ (172)	\$ (290)

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

21. OPERATING SEGMENT INFORMATION (Continued)

	Three mon Marc		
	2010	2009	
Huntsman International:			
Segment EBITDA(1):			
Polyurethanes	\$ 52	\$ 26	
Performance Products	60	63	
Advanced Materials	33	10	
Textile Effects	_	(11)	
Pigments	28	(29)	
Corporate and other(2)	(50)	(15)	
Subtotal	123	44	
Discontinued Operations(3)	(21)	(3)	,
Total	102	41	
Interest expense, net	(66)	(55)	
Income tax benefit (expense)—continuing operations	22	(146)	
Income tax benefit (expense)—discontinued operations	8	(1)	
Depreciation and amortization	(92)	(120)	
Net loss attributable to			
Huntsman International LLC	\$ (26)	\$ (281)	

⁽¹⁾ Segment EBITDA is defined as net (loss) income attributable to Huntsman Corporation or Huntsman International LLC, as appropriate, before interest, income tax, depreciation and amortization, and certain Corporate and other items.

⁽²⁾ Corporate and other includes unallocated corporate overhead, loss on early extinguishment of debt, loss on A/R Programs (in 2009), foreign exchange gains or losses, step purchase accounting impacts, LIFO inventory valuation reserve adjustments and other non-operating income (expense).

⁽³⁾ The operating results of our former polymers, base chemicals and Australian styrenics businesses are classified as discontinued operations, and, accordingly, the revenues of these businesses are excluded for all periods presented. The EBITDA of our former polymers, base chemicals and Australian styrenics businesses are included in discontinued operations for all periods presented. For more information, see "Note 19. Discontinued Operations."

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

22. CONDENSED CONSOLIDATING FINANCIAL INFORMATION OF HUNTSMAN INTERNATIONAL LLC (UNAUDITED)

The following condensed consolidating financial information (unaudited) of Huntsman International presents, in separate columns, financial information for the following: Huntsman International (on a parent only basis), with its investment in subsidiaries recorded under the equity method; the Subsidiary Guarantors of Huntsman International's debt on a combined, and where appropriate, consolidated basis; and non-guarantor subsidiaries on a combined, and where appropriate, consolidated basis. Additional columns present eliminating adjustments and consolidated totals as of March 31, 2010 and December 31, 2009 and for the three months ended March 31, 2010 and 2009. There are no contractual restrictions limiting transfers of cash from the Subsidiary Guarantors to Huntsman International. Each Subsidiary Guarantor is 100% owned by Huntsman International and has fully and unconditionally guaranteed Huntsman International's debt on a joint and several basis.

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

22. CONDENSED CONSOLIDATING FINANCIAL INFORMATION OF HUNTSMAN INTERNATIONAL LLC (UNAUDITED) (Continued)

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES CONDENSED CONSOLIDATING BALANCE SHEETS (UNAUDITED) AS OF MARCH 31, 2010 (Dollars in Millions)

	P	'arent							Н	nsolidated untsman ernational
	Co	mpany	Guar	antors	Non-	guarantors	Elin	ninations		LLC
ASSETS							-			
Current assets:										
Cash and cash equivalents	\$	352	\$	11	\$	188	\$	_	\$	551
Restricted cash		_		_		8		_		8
Accounts and notes receivable, net		21		110		1,240		_		1,371
Accounts receivable from affiliates		1,241		2,216		84		(3,465)		76
Inventories		87		205		969		(13)		1,248
Prepaid expenses		9		14		20		(9)		34
Deferred income taxes		10		_		29		(5)		34
Other current assets		131		1		103		(113)		122
Total current assets		1,851		2,557		2,641		(3,605)		3,444
Property, plant and equipment, net		429		900		1,889		2		3,220
Investment in affiliates		4,268		1,144		108		(5,268)		252
Intangible assets, net		77		3		43				123
Goodwill		(18)		84		32		(4)		94
Deferred income taxes		246		_		125		(196)		175
Notes receivable from affiliates		65		928		7		(993)		7
Other noncurrent assets		57		190		187				434
Total assets	\$	6,975	\$	5,806	\$	5,032	\$	(10,064)	\$	7,749
LIABILITIES AND EQUITY										
Current liabilities:										
Accounts payable	\$	27	\$	255	\$	591	\$	_	\$	873
Accounts payable to affiliates		1,820		780		887		(3,465)		22
Accrued liabilities		92		172		419		(121)		562
Deferred income tax		_		7		2		(7)		2
Note payable to affiliate		100		_		_				100
Current portion of debt		164		_		152		_		316
Total current liabilities		2.203		1,214		2,051		(3,593)		1.875
Long-term debt		3,492		12		355		(5,555)		3,859
Notes payable to affiliates		325				998		(993)		330
Deferred income taxes		11		85		68		(82)		82
Other noncurrent liabilities		177		135		504		(1)		815
Total liabilities	_	6,208		1,446		3,976		(4,669)	_	6,961
Equity		0,200		1,110		3,770		(1,00))		0,701
Huntsman International LLC members' equity:										
Members' equity		3,031		4,422		2,032		(6,454)		3,031
Subsidiary preferred stock				.,		1		(1)		
Accumulated deficit		(1,873)		(93)		(643)		736		(1,873)
Accumulated other comprehensive (loss) income		(391)		31		(349)		318		(391)
Total Huntsman International LLC	_									
members' equity		767		4,360		1,041		(5,401)		767
Noncontrolling interests in subsidiaries		_				15		6		21
Total equity		767		4,360		1,056		(5,395)		788
Total liabilities and equity	\$	6,975	\$	5,806	\$	5,032	\$	(10,064)	\$	7,749
Total natinites and equity	Ψ	0,713	Ψ	5,000	Ψ	3,032	Ψ	(10,004)	Ψ	1,179

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

22. CONDENSED CONSOLIDATING FINANCIAL INFORMATION OF HUNTSMAN INTERNATIONAL LLC (UNAUDITED) (Continued)

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES CONDENSED CONSOLIDATING BALANCE SHEETS (UNAUDITED) AS OF DECEMBER 31, 2009 (Dollars in Millions)

		arent mpany	C	ıarantors	No	on-guarantors	FI:.	minations	Н	nsolidated untsman ernational LLC
ASSETS		шрапу	Gi	iai aiitoi s	110	on-guarantors	EIII	mnations	_	LLC
Current assets:										
Cash and cash equivalents	\$	688	\$	24	\$	207	\$	_	\$	919
Restricted cash	Ψ		Ψ		Ψ	5	Ψ	_	Ψ	5
Accounts and notes receivable, net		17		85		916		_		1,018
Accounts receivable from affiliates		1,040		2,178		83		(3,269)		32
Inventories		78		205		908		(7)		1.184
Prepaid expenses		11		27		21		(17)		42
Deferred income taxes		10				28		(5)		33
Other current assets		120		1		100		(112)		109
Total current assets	_	1,964	_	2,520		2,268	_	(3,410)	_	3,342
Property, plant and equipment, net		439		913		2,003		(3,410)		3,357
Investment in affiliates		4.314		1.139		106		(5,309)		250
Intangible assets, net		82		3		44		(3,309)		129
Goodwill		(19)		84		33		(4)		94
Deferred income taxes		236		- 04		118		(196)		158
Notes receivable from affiliates		63		988		8		(1,051)		8
Other noncurrent assets		40		129		186		(1,051)		355
	0		Ф		0		Φ.	(0.0(0)	Φ.	
Total assets	\$	7,119	\$	5,776	\$	4,766	\$	(9,968)	\$	7,693
LIABILITIES AND EQUITY										
Current liabilities:										
Accounts payable	\$	36	\$	177	\$	502	\$	_	\$	715
Accounts payable to affiliates		1,702		744		863		(3,268)		41
Accrued liabilities		86		210		447		(130)		613
Deferred income tax		_		7		2		(7)		2
Note payable to affiliate		25				_		_		25
Current portion of debt		39		_		156		_		195
Total current liabilities		1,888		1,138		1,970		(3,405)		1,591
Long-term debt		3,675		12		94		`		3,781
Notes payable to affiliates		525		_		1,056		(1,051)		530
Deferred income taxes		11		82		68		(82)		79
Other noncurrent liabilities		194		144		527		_		865
Total liabilities		6,293		1,376		3,715		(4,538)		6,846
Equity		0,275		1,070		5,710		(1,220)		0,010
Huntsman International LLC members' equity:										
Members' equity		3,021		4,464		1,986		(6,450)		3,021
Subsidiary preferred stock						1		(1)		
Accumulated deficit		(1,847)		(125)		(636)		761		(1,847)
Accumulated other comprehensive				,		` ′				
(loss) income		(348)		61		(315)		254		(348)
Total Huntsman International LLC	_									
members' equity		826		4,400		1,036		(5,436)		826
Noncontrolling interests in subsidiaries		320		7,700		1,030		(3,430)		21
		024		4.400						
Total equity		826		4,400		1,051		(5,430)		847
Total liabilities and equity	\$	7,119	\$	5,776	\$	4,766	\$	(9,968)	\$	7,693

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

22. CONDENSED CONSOLIDATING FINANCIAL INFORMATION OF HUNTSMAN INTERNATIONAL LLC (UNAUDITED) (Continued)

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

CONDENSED CONSOLIDATING STATEMENTS OF OPERATIONS (UNAUDITED)

THREE MONTHS ENDED MARCH 31, 2010

	Parent Company	Guarantors	Non-guarantors	Eliminations	Consolidated Huntsman International LLC
Revenues:					
Trade sales, services and fees, net	\$ 183	\$ 507	\$ 1,359	\$ —	\$ 2,049
Related party sales	58	108	228	(349)	45
Total revenues	241	615	1,587	(349)	2,094
Cost of goods sold	198	572	1,383	(345)	1,808
Gross profit	43	43	204	(4)	286
Selling, general and administrative	36	20	159		215
Research and development	12	7	17	_	36
Other operating (income) expense	(17)	14	(4)	_	(7)
Restructuring, impairment and plant					
closing costs	2		1		3
Operating income	10	2	31	(4)	39
Interest (expense) income, net	(56)	9	(19)	_	(66)
Equity in (loss) income of investment					
in affiliates and subsidiaries	(2)	(6)	_	9	1
Loss on early extinguishment of debt	(9)				(9)
(Loss) income from continuing					
operations before income taxes	(57)	5	12	5	(35)
Income tax benefit (expense)	24	(2)	_	_	22
(Loss) income from continuing					
operations	(33)	3	12	5	(13)
Income (loss) from discontinued					
operations, net of tax	7	_	(20)	_	(13)
Net (loss) income	(26)	3	(8)	5	(26)
Net loss attributable to noncontrolling interests	_	_	_	_	_
Net (loss) income attributable to Huntsman International LLC	\$ (26)	\$ 3	\$ (8)	\$ 5	\$ (26)

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

22. CONDENSED CONSOLIDATING FINANCIAL INFORMATION OF HUNTSMAN INTERNATIONAL LLC (UNAUDITED) (Continued)

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

CONDENSED CONSOLIDATING STATEMENTS OF OPERATIONS (UNAUDITED)

THREE MONTHS ENDED MARCH 31, 2009

	Parent Company	Guarantors	Non-guarantors	Eliminations	Consolidated Huntsman International LLC
Revenues:					
Trade sales, services and fees, net	\$ 136	\$ 489	\$ 1,048	\$ —	\$ 1,673
Related party sales	33	74	125	(225)	7
Total revenues	169	563	1,173	(225)	1,680
Cost of goods sold	158	457	1,136	(225)	1,526
Gross profit	11	106	37		154
Selling, general and administrative	32	31	126	_	189
Research and development	12	7	17	_	36
Other operating (income) expense	(22) 12	2	_	(8)
Restructuring, impairment and plant closing costs	1	_	13	_	14
Operating (loss) income	(12) 56	(121)		(77)
Interest (expense) income, net	(47	,	(19)		(55)
Loss on accounts receivable	(.,	, 11	(1)		(00)
securitization program	(2	(1)	(1)	_	(4)
Equity in (loss) income of investment		, , ,	,		()
in affiliates and subsidiaries	(244	(248)	3	490	1
Loss from continuing operations		-			
before income taxes	(305	(182)	(138)	490	(135)
Income tax benefit (expense)	24	(23)	. ,	_	(146)
Loss from continuing operations	(281	(205)	(285)	490	(281)
Income (loss) from discontinued		, (: :)	(33)		
operations, net of tax	_	2	(6)	_	(4)
Net loss	(281	(203)	(291)	490	(285)
Net loss attributable to noncontrolling interests	_	, (232) —	4	_	4
Net loss attributable to Huntsman International LLC	\$ (281	\$ (203)	\$ (287)	\$ 490	\$ (281)

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

22. CONDENSED CONSOLIDATING FINANCIAL INFORMATION OF HUNTSMAN INTERNATIONAL LLC (UNAUDITED) (Continued)

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

CONDENSED CONSOLIDATING STATEMENTS OF CASH FLOWS (UNAUDITED)

THREE MONTHS ENDED MARCH 31, 2010

		arent mpany	Guarantors	Non-guarantors	Eliminations	Consolidated Huntsman International LLC
Net cash (used in) provided by operating activities	\$	(342)	\$ 4	\$ (37)	\$ (1)	\$ (376)
Investing activities:	Ψ	(3 12)		(37)	Ψ (1)	ψ (370)
Capital expenditures		(2)	(14)	(21)	_	(37)
Investment in unconsolidated		(-)	(- 1)	(==)		(= 1)
affiliates, net of cash received		_	1	(4)	_	(3)
Investment in affiliate		(27)	5	_	22	
Change in restricted cash		_	_	(3)	_	(3)
Increase in receivable from affiliate		(50)				(50)
Net cash used in investing activities		(79)	(8)	(28)	22	(93)
Financing activities:						
Net borrowings under revolving loan facilities		_	_	3	_	3
Revolving loan facility from A/R						
Programs		254	_	_	_	254
Net repayments on overdraft facilities				(1)		(1)
Repayments of short-term debt		_	_	(44)	_	(44)
Proceeds from short-term debt Repayments of long-term debt		(355)		55 (20)	_	(375)
Proceeds from long-term debt		350		25	_	375
Repayment of note payable to affiliate		(125)				(125)
Intercompany repayments		—	_	(5)	5	—
Repayments of notes payable		(18)	_	(3)	_	(21)
Debt issuance costs paid		(16)	_		_	(16)
Call premiums paid related to early extinguishment of debt						Ý
Contribution from parent, net		(7)	(9)	36	(27)	(7)
Excess tax benefit related to stock-			(2)	30	(27)	
based compensation		4	_	_	_	4
Other, net		(2)	_	_	1	(1)
Net cash provided by (used in)						
financing activities		85	(9)	46	(21)	101
Effect of exchange rate changes on cash						
Decrease in cash and cash equivalents	_	(336)	(13)	(19)		(368)
Cash and cash equivalents at		(550)	(13)	(17)		(300)
beginning of period		688	24	207	_	919
Cash and cash equivalents at end of period	\$	352	\$ 11	\$ 188	\$ —	\$ 551

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) (Continued)

22. CONDENSED CONSOLIDATING FINANCIAL INFORMATION OF HUNTSMAN INTERNATIONAL LLC (UNAUDITED) (Continued)

HUNTSMAN INTERNATIONAL LLC AND SUBSIDIARIES

CONDENSED CONSOLIDATING STATEMENTS OF CASH FLOWS (UNAUDITED)

THREE MONTHS ENDED MARCH 31, 2009

	Par Com		Guarantors	Non-guarantors	Eliminations	Consolidated Huntsman International LLC
Net cash (used in) provided by						
operating activities	\$	(72)	\$ 54	\$ 15	\$ —	\$ (3)
Investing activities:						
Capital expenditures		(2)	(27)	(32)	_	(61)
Proceeds from sale of			, ,	, ,		· í
businesses/assets, net of						
adjustments		_	(1)	_	_	(1)
Investment in affiliate		(32)	(27)	_	59	
Decrease in receivable from affiliate		7		_	_	7
Other, net		_	(2)	2	_	_
Net cash used in investing activities		(27)	(57)	(30)	59	(55)
Financing activities:			<u> </u>			
Net repayments under revolving loan						
facilities		_		(8)		(8)
Net repayments on overdraft facilities		_	_	(1)	_	(1)
Repayments of short-term debt			_	(73)	_	(73)
Proceeds from short-term debt		_	_	47	_	47
Repayments of long-term debt		(16)	_	(1)	_	(17)
Proceeds from note payable to		(-)		()		(')
affiliate		529	_	_	_	529
Intercompany borrowings		_	_	27	(27)	_
Repayments of notes payable		(11)	_	(11)		(22)
Borrowings on notes payable		_	_	11	_	11
Contribution from parent, net		_	3	29	(32)	_
Dividends paid to parent		(23)	_	_		(23)
Other, net			_	1	_	1
Net cash provided by financing						
activities		479	3	21	(59)	444
Effect of exchange rate changes on						
cash		_	_	(3)	_	(3)
Increase in cash and cash equivalents		380		3		383
Cash and cash equivalents at		200		3		203
beginning of period		_	3	84	_	87
- 0 .						
Cash and cash equivalents at end of	\$	380	¢ 2	¢ 07	¢	¢ 470
period	Ф	380	\$ 3	\$ 87	<u> </u>	\$ 470

Table of Contents

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

FORWARD-LOOKING STATEMENTS

Certain information set forth in this report contains "forward-looking statements" within the meaning of the federal securities laws. Forward-looking statements include statements concerning our plans, objectives, goals, strategies, future events, future revenues or performance, capital expenditures, financing needs, plans or intentions relating to acquisitions and other information that is not historical information. In some cases, forward-looking statements can be identified by terminology such as "believes," "expects," "may," "should," "anticipates," or "intends" or the negative of such terms or other comparable terminology, or by discussions of strategy. We may also make additional forward-looking statements from time to time. All such subsequent forward-looking statements, whether written or oral, by us or on our behalf, are also expressly qualified by these cautionary statements.

All forward-looking statements, including without limitation management's examination of historical operating trends, are based upon our current expectations and various assumptions. Our expectations, beliefs and projections are expressed in good faith and we believe there is a reasonable basis for them, but, there can be no assurance that management's expectations, beliefs and projections will result or be achieved. All forward-looking statements apply only as of the date made. We undertake no obligation to publicly update or revise forward-looking statements which may be made to reflect events or circumstances after the date made or to reflect the occurrence of unanticipated events.

There are a number of risks and uncertainties that could cause our actual results to differ materially from the forward-looking statements contained in or contemplated by this report. Any forward-looking statements should be considered in light of the risks referenced in "Part I. Item 1A. Risk Factors" in our Annual Report on Form 10-K for the year ended December 31, 2009.

OVERVIEW

Business

We are a global manufacturer of differentiated organic chemical products and of inorganic chemical products. Our products comprise a broad range of chemicals and formulations, which we market globally to a diversified group of consumer and industrial customers. Our products are used in a wide range of applications, including those in the adhesives, aerospace, automotive, construction products, durable and non-durable consumer products, electronics, medical, packaging, paints and coatings, power generation, refining, synthetic fiber, textile chemicals and dye industries. We are a leading global producer in many of our key product lines, including MDI, amines, surfactants, maleic anhydride, epoxy-based polymer formulations, textile chemicals, dyes and titanium dioxide. We had revenues for the three months ended March 31, 2010 and 2009 of \$2,094 million and \$1,680 million, respectively.

We operate in five segments: Polyurethanes, Performance Products, Advanced Materials, Textile Effects and Pigments. Our Polyurethanes, Performance Products, Advanced Materials and Textile Effects segments produce differentiated organic chemical products and our Pigments segment produces inorganic chemical products. We ceased operation of our Australian styrenics operations during the first quarter of 2010 and report the results of that business as discontinued operations. See "Note 19. Discontinued Operations" to our condensed consolidated financial statements (unaudited).

Table of Contents

RECENT DEVELOPMENTS

Repurchase of Convertible Notes

On January 11, 2010, we repurchased the entire \$250 million principal amount of our outstanding Convertible Notes for approximately \$382 million from Apollo and its affiliates. The Convertible Notes were issued to Apollo in December 2008. The Convertible Notes, which would have matured on December 23, 2018, bore interest at the rate of 7% per year and were convertible into approximately 31.8 million shares of our common stock at any time by the holders. As a result of the repurchase of the Convertible Notes, we recorded a loss on early extinguishment of debt in the first quarter of 2010 of \$146 million.

Senior Subordinated Note Offering

On March 17, 2010 we completed a \$350 million offering of 8.625% senior subordinated notes due 2020. We used the net proceeds to redeem approximately €184 million (approximately \$247 million) senior subordinated notes due 2013 and approximately €59 million (approximately \$79 million) senior subordinated notes due 2015. In connection with the redemption of these notes, we recorded a loss on early extinguishment of debt in the first quarter of 2010 of \$9 million. In connection with these redemptions and the issuance of the 2020 Subordinated Notes, Huntsman International entered into \$350 million notional amount of five-year euro/dollar cross-currency interest rate contracts. In accordance with these swap transactions, the interest payment on the 2020 Subordinated Notes will be denominated in euros at an effective rate of approximately 8.41% for the next five years.

Amendment of Senior Credit Facilities

On March 9, 2010, Huntsman International entered into the Amendment with JPMorgan Chase Bank, N.A. and the other financial institutions party thereto, which amended certain terms of Huntsman International's existing Senior Credit Facilities. Among other things, the Amendment (i) replaces Deutsche Bank AG New York Branch as administrative agent, collateral agent and U.K. security trustee with JPMorgan Chase Bank, N.A. as administrative agent and collateral agent and JPMorgan Chase Bank, N.A. or an affiliate thereof as U.K. security trustee, (ii) extends the stated maturity of the revolving facility to March 9, 2014, and (iii) limits the aggregate amount of the revolving commitments allowable under the revolving facility to an amount up to \$300 million, including \$225 million currently obtained from the lenders. The Amendment increases the applicable LIBOR margin range on the revolving loans by 1.75% to 3.50% per annum and increased the unused commitment fee percentage to a range of 0.50% to 0.75%.

Accounting for our Accounts Receivable Securitization Program

On January 1, 2010, in connection with the adoption of ASU No. 2009-16, we have accounted for sales of accounts receivable under our accounts receivable securitization programs as secured borrowings. See "Note 2. Recently Issued Accounting Pronouncements" and "Note 7. Debt—Accounts Receivable Securitization" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

OUTLOOK

Our positive demand momentum suggests that second quarter 2010 sales volumes should continue to improve. We are aggressively working to increase our product prices to offset the increase we are seeing in raw materials.

RESULTS OF OPERATIONS

For each of our Company and Huntsman International, the following tables set forth the unaudited condensed consolidated results of operations (dollars in millions):

Huntsman Corporation

	Three months ended March 31,			Percent	
		2010		2009	Change
Revenues	\$	2,094	\$	1,680	25%
Cost of goods sold		1,813		1,531	18%
Gross profit		281		149	89%
Operating expenses		256		222	15%
Restructuring, impairment and plant closing costs		3		14	(79)%
Operating income (loss)		22		(87)	NM
Interest expense, net		(61)		(55)	11%
Loss on accounts receivable securitization program		_		(4)	NM
Equity in income of investment in unconsolidated affiliates		1		1	—
Loss on early extinguishment of debt		(155)		_	NM
Expenses associated with the Terminated Merger and related litigation		_		(7)	NM
Loss from continuing operations before income taxes		(193)		(152)	27%
Income tax benefit (expense)		34		(138)	NM
Loss from continuing operations		(159)		(290)	(45)%
Loss from discontinued operations, net of tax		(13)		(4)	225%
Net loss		(172)		(294)	(41)%
Net loss attributable to noncontrolling interests		_		4	NM
Net loss attributable to Huntsman Corporation		(172)		(290)	(41)%
Interest expense, net		61		55	11%
Income tax (benefit) expense from continuing operations		(34)		138	NM
Income tax (benefit) expense from discontinued operations		(8)		1	NM
Depreciation and amortization		98		126	(22)%
EBITDA(1)	\$	(55)	\$	30	NM
Net cash used in operating activities	\$	(408)	\$	(38)	974%
Net cash used in investing activities		(43)		(62)	(31)%
Net cash used in financing activities		(184)		(84)	119%

Huntsman International

	Three months ended March 31,			Percent	
		2010		2009	Change
Revenues	\$	2,094	\$	1,680	25%
Cost of goods sold		1,808		1,526	18%
Gross profit		286		154	86%
Operating expenses		244		217	12%
Restructuring, impairment and plant closing costs		3		14	(79)%
Operating income (loss)		39		(77)	NM
Interest expense, net		(66)		(55)	20%
Loss on accounts receivable securitization program		_		(4)	NM
Equity in income of investment in unconsolidated affiliates		1		1	_
Loss on early extinguishment of debt		(9)		_	NM
Loss from continuing operations before income taxes		(35)		(135)	(74)%
Income tax benefit (expense)		22		(146)	NM
Loss from continuing operations		(13)		(281)	(95)%
Loss from discontinued operations, net of tax		(13)		(4)	225%
Net loss		(26)		(285)	(91)%
Net loss attributable to noncontrolling interests		_		4	NM
Net loss attributable to Huntsman International LLC		(26)		(281)	(91)%
Interest expense, net		66		55	20%
Income tax (benefit) expense from continuing operations		(22)		146	NM
Income tax (benefit) expense from discontinued operations		(8)		1	NM
Depreciation and amortization		92		120	(23)%
EBITDA(1)	\$	102	\$	41	149%
Net cash used in operating activities	\$	(376)	\$	(3)	NM
Net cash used in investing activities		(93)		(55)	69%
Net cash provided by financing activities		101		444	(77)%

NM—Not Meaningful

For each of our Company and Huntsman International, the following tables set forth certain items of income (expense) included in EBITDA (dollars in millions):

Huntsman Corporation

	Three months ended March 31,	
	2010	2009
Foreign exchange gains—unallocated	\$ 1	\$ 2
Loss on early extinguishment of debt	(155)	
Loss on accounts receivable securitization program	_	(4)
Amounts included in discontinued operations	(21)	(3)
Acquisition expenses	_	(1)
Expenses associated with the Terminated Merger and related litigation		(7)
Restructuring, impairment and plant closing credits (costs):		
Polyurethanes		(1)
Performance Products	_	_
Advanced Materials	2	_
Textile Effects	_	1
Pigments	(1)	(13)
Corporate and other	(4)	(1)
Total restructuring, impairment and plant closing costs	(3)	(14)
Total	\$ (178)	\$ (27)

Huntsman International

	Three months ended March 31,		
	2	2010	2009
Foreign exchange gains—unallocated	\$	1	\$ 2
Loss on early extinguishment of debt		(9)	_
Loss on accounts receivable securitization program			(4)
Amounts included in discontinued operations		(21)	(3)
Acquisition expenses		_	(1)
Restructuring, impairment and plant closing credits (costs):			
Polyurethanes			(1)
Performance Products		_	_
Advanced Materials		2	_
Textile Effects			1
Pigments		(1)	(13)
Corporate and other		(4)	(1)
Total restructuring, impairment and plant closing costs		(3)	(14)
Total	\$	(32)	\$ (20)

⁽¹⁾ EBITDA is defined as net (loss) income attributable to Huntsman Corporation or Huntsman International LLC, as appropriate, before interest, income taxes, depreciation and amortization. We believe that EBITDA enhances an investor's understanding of our financial performance and our ability to satisfy principal and interest obligations with respect to our indebtedness. However, EBITDA should not be considered in isolation or viewed as a substitute for net income, cash flow

from operations or other measures of performance as defined by GAAP. Moreover, EBITDA as used herein is not necessarily comparable to other similarly titled measures of other companies due to potential inconsistencies in the method of calculation. Our management uses EBITDA to assess financial performance and debt service capabilities. In assessing financial performance, our management reviews EBITDA as a general indicator of economic performance compared with prior periods. Because EBITDA excludes interest, income taxes, depreciation and amortization, EBITDA provides an indicator of general economic performance that is not affected by debt restructurings, fluctuations in interest rates or effective tax rates, or levels of depreciation and amortization. Accordingly, our management believes this type of measurement is useful for comparing general operating performance from period to period and making certain related management decisions. EBITDA is also used by securities analysts, lenders and others in their evaluation of different companies because it excludes certain items that can vary widely across different industries or among companies within the same industry. For example, interest expense can be highly dependent on a company's capital structure, debt levels and credit ratings. Therefore, the impact of interest expense on earnings can vary significantly among companies. In addition, the tax positions of companies can vary because of their differing abilities to take advantage of tax benefits and because of the tax policies of the various jurisdictions in which they operate. As a result, effective tax rates and tax expense can vary considerably among companies. Finally, companies employ productive assets of different ages and utilize different methods of acquiring and depreciating such assets. This can result in considerable variability in the relative costs of productive assets and the depreciation and amortization expense among companies. Our management also believes that our investors use EBITDA as a measure of our ability to service indebtedness as well as to fund capital expenditures and working capital requirements. Nevertheless, our management recognizes that there are material limitations associated with the use of EBITDA in the evaluation of our Company as compared to net income, which reflects overall financial performance, including the effects of interest, income taxes, depreciation and amortization. EBITDA excludes interest expense. Because we have borrowed money in order to finance our operations, interest expense is a necessary element of our costs and ability to generate revenue. Therefore, any measure that excludes interest expense has material limitations. EBITDA also excludes taxes. Because the payment of taxes is a necessary element of our operations, any measure that excludes tax expense has material limitations. Finally, EBITDA excludes depreciation and amortization expense. Because we use capital assets, depreciation and amortization expense is a necessary element of our costs and ability to generate revenue. Therefore, any measure that excludes depreciation and amortization expense has material limitations. Our management compensates for the limitations of using EBITDA by using it to supplement GAAP results to provide a more complete understanding of the factors and trends affecting the business rather than GAAP results alone. Our management also uses other metrics to evaluate capital structure, tax planning and capital investment decisions. For example, our management uses credit ratings and net debt ratios to evaluate capital structure, effective tax rate by jurisdiction to evaluate tax planning, and payback period and internal rate of return to evaluate capital investments. Our management also uses trade working capital to evaluate its investment in accounts receivable and inventory, net of accounts payable.

We believe that net (loss) income is the performance measure calculated and presented in accordance with GAAP that is most directly comparable to EBITDA and that cash provided by operating activities is the liquidity measure calculated and presented in accordance with GAAP that is most directly comparable to EBITDA. For each of our Company and Huntsman International, the following tables reconcile EBITDA to net (loss) income attributable to

Huntsman Corporation or Huntsman International LLC, as appropriate, and to net cash used in operations (dollars in millions):

Huntsman Corporation

	March 31,		Percent
	2010	2009	Change
EBITDA	\$ (55)	\$ 30	NM
Depreciation and amortization	(98)	(126)	(22)%
Interest expense, net	(61)	(55)	11%
Income tax benefit (expense) from continuing operations	34	(138)	NM
Income tax benefit (expense) from discontinued operations	8	(1)	NM
Net loss attributable to Huntsman Corporation	(172)	(290)	(41)%
Net loss attributable to noncontrolling interests		(4)	NM
Net loss	(172)	(294)	(41)%
Equity in income of investment in unconsolidated affiliates	(1)	(1)	_
Depreciation and amortization	98	126	(22)%
Loss (gain) on disposal of businesses/assets, net	1	(1)	NM
Noncash restructuring, impairment and plant closing costs	_	4	NM
Loss on early extinguishment of debt	155	_	NM
Noncash interest expense	10	2	400%
Deferred income taxes	(20)	131	NM
Net unrealized (gain) loss on foreign currency transactions	(3)	2	NM
Other, net	8	8	_
Changes in operating assets and liabilities	(484)	(15)	NM
Net cash used in operating activities	\$ (408)	\$ (38)	974%

Huntsman International

	March 31,			Percent	
		2010		2009	Change
EBITDA	\$	102	\$	41	149%
Depreciation and amortization		(92)		(120)	(23)%
Interest expense, net		(66)		(55)	20%
Income tax benefit (expense) from continuing operations		22		(146)	NM
Income tax benefit (expense) from discontinued operations		8		(1)	NM
Net loss attributable to Huntsman International LLC		(26)		(281)	(91)%
Net loss attributable to noncontrolling interests		_		(4)	NM
Net loss		(26)	_	(285)	(91)%
Equity in income of investment in unconsolidated affiliates		(1)		(1)	<u>`</u>
Depreciation and amortization		92		120	(23)%
Loss (gain) on disposal of businesses/assets, net		1		(1)	NM
Noncash restructuring, impairment and plant closing costs		_		4	NM
Loss on early extinguishment of debt		9		_	NM
Noncash interest expense		14		5	180%
Deferred income taxes		(18)		143	NM
Net unrealized (gain) loss on foreign currency transactions		(3)		2	NM
Other, net		9		5	80%
Changes in operating assets and liabilities		(453)		5	NM
Net cash used in operating activities	\$	(376)	\$	(3)	NM

NM-Not Meaningful

Three Months Ended March 31, 2010 Compared with Three Months Ended March 31, 2009

For the three months ended March 31, 2010, the net loss attributable to Huntsman Corporation was \$172 million on revenues of \$2,094 million, compared with net loss attributable to Huntsman Corporation of \$290 million on revenues of \$1,680 million for the same period of 2009. For the three months ended March 31, 2010, the net loss attributable to Huntsman International LLC was \$26 million on revenues of \$2,094 million, compared with net loss attributable to Huntsman International LLC of \$281 million on revenues of \$1,680 million for the same period of 2009. The decrease of \$118 million in net loss attributable to Huntsman Corporation and the decrease of \$255 million in net loss attributable to Huntsman International LLC was the result of the following items:

- Revenues for the three months ended March 31, 2010 increased by \$414 million or 25%, as compared with the 2009 period due principally to higher average selling prices in all our segments except Advanced Materials and higher sales volumes in all of our segments except Polyurethanes. See "—Segment Analysis" below.
- Our gross profit and the gross profit of Huntsman International for the three months ended March 31, 2010 increased by \$132 million each, or 89% and 86%, respectively, as compared with the 2009 period resulting from higher gross margins in all of our segments except Performance Products. See "—Segment Analysis" below.
- Our operating expenses and the operating expenses of Huntsman International for the three months ended March 31, 2010 increased by \$34 million and \$27 million, or 15% and 12%, respectively, as compared with the 2009 period. Operating expenses increased by \$11 million in

2010 due to the impact of translating foreign currency amounts to the U.S. dollar, as the U.S. dollar weakened versus other relevant currencies. Also contributing to higher operating expenses was an \$8 million decrease in foreign exchange gains (\$1 million of gains in the 2010 period as compared with \$9 million of gains in the 2009 period) and higher selling, general and administrative expenses.

- Restructuring, impairment and plant closing costs for the three months ended March 31, 2010 decreased to \$3 million from \$14 million in the 2009 period. For more information concerning restructuring activities, see "Note 6. Restructuring, Impairment and Plant Closing Costs" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.
- Our net interest expense and the net interest expense of Huntsman International for the three months ended March 31, 2010 increased by \$6 million and \$11 million, or 11% and 20%, respectively, as compared with the 2009 period. Upon the adoption of new accounting guidance in 2010, sales of our accounts receivable under our A/R Programs no longer met the criteria for derecognition. Accordingly, the amounts outstanding under our A/R Programs are accounted for as secured borrowings beginning January 1, 2010. The increase in interest expense was primarily due to the consolidation of our A/R Programs in the beginning of 2010. For more information, see "Note 7. Debt—Accounts Receivable Securitization" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.
- The loss on early extinguishment of debt in 2010 resulted from the January 11, 2010 repurchase of the Convertible Notes and from the March 17, 2010 partial redemption of the senior subordinated notes due 2013 and the senior subordinated notes due 2015. For more information see "Note 1. General—Recent Developments" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.
- For the three months ended March 31, 2010, our income tax expense decreased by \$172 million and Huntsman International's tax expense decreased by \$168 million, as compared with the same period in 2009. Our and Huntsman International's tax obligations are affected by the mix of income and losses in the tax jurisdictions in which we operate. For the three months ended March 31, 2010, in accordance with U.S. GAAP, we have computed our provision for income taxes based on the actual effective tax rate for the year-to-date by applying the discrete method. The discrete method was used to calculate the income tax provision as the annual effective tax rate was not considered a reliable estimate of year-to-date income tax expense. For further information concerning taxes, see "Note 18. Income Taxes" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.
- Loss from discontinued operations for the three months ended March 31, 2010 increased to \$13 million from \$4 million in the 2009 period. In connection with the closure and abandonment of our Australian styrenics operations in the first quarter of 2010, we have treated this business as discontinued operations beginning in the first quarter of 2010. All relevant information for prior periods has been restated to reflect this change. For more information, see "Note 19. Discontinued Operations" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

Segment Analysis

During the first quarter of 2010, we began reporting our LIFO inventory valuation reserves as part of Corporate and other; these reserves were previously reported in our Performance Products segment. All segment information for prior periods has been restated to reflect this change.

		Three Mont	Percent	
	_	2010	2009	Change
Revenues				
Polyurethanes	\$			00 28%
Performance Products		616		00 23%
Advanced Materials		291		57 13%
Textile Effects		195		52 28%
Pigments		269		96 37%
Eliminations		(44)	(25) 76%
Total	\$	2,094	\$ 1,6	80 25%
Huntsman Corporation				
Segment EBITDA				
Polyurethanes	\$		•	26 100%
Performance Products		60		63 (5)%
Advanced Materials		33		10 230%
Textile Effects				11) NM
Pigments		28	,	29) NM
Corporate and other		(207)		26) 696%
Subtotal		(34)		33 NM
Discontinued Operations		(21)		(3) 600%
Total	\$	(55)	\$	30 NM
Huntsman International	_			
Segment EBITDA				
Polyurethanes	\$	52	•	26 100%
Performance Products		60		63 (5)%
Advanced Materials		33		10 230%
Textile Effects		_	,	11) NM
Pigments		28	,	29) NM
Corporate and other		(50)	(15) 233%
Subtotal		123		44 180%
Discontinued Operations		(21)		(3) 600%
Total	\$	102	\$	41 149%

	Average Selling Price(1)	Sales Volumes(1)
Current Year-Over-Prior Year Increase (Decrease)		
Polyurethanes	37%	(11)%
Performance Products	4%	21%
Advanced Materials	(1)%	16%
Textile Effects	8%	18%
Pigments	4%	33%

	Average	
	Selling	Sales
	Price(1)	Volumes(1)
First Quarter 2010 vs. Fourth Quarter 2009 Increase (Decrease)		
Polyurethanes	13%	(22)%
Performance Products	4%	7%
Advanced Materials	(2)%	12%
Textile Effects	3%	1%
Pigments	(1)%	11%

⁽¹⁾ Excludes revenues and sales volumes from tolling arrangements and byproducts.

NM-Not meaningful

Polyurethanes

The increase in revenues in our Polyurethanes segment for the three months ended March 31, 2010 compared to the same period in 2009 was primarily due to higher MDI sales volumes and higher average selling prices for MTBE. MDI sales volumes were higher with demand recovery across all major markets as a result of the worldwide economic recovery while average selling prices for MTBE increased in response to higher raw material costs and tight supply due in part to a 60-day planned maintenance outage at our Port Neches, Texas PO/MTBE facility. PO/MTBE sales volumes decreased due to the planned outage. Average MDI selling prices decreased primarily due to competitive pressures and lower raw material costs. The increase in EBITDA was primarily due to higher MDI sales volumes partially offset by the approximate \$40 million impact of the planned maintenance outage at our Port Neches, Texas PO/MTBE facility.

Performance Products

The increase in revenues in our Performance Products segment for the three months ended March 31, 2010 compared to the same period in 2009 was due to higher sales volumes and higher average selling prices. Sales volumes increased primarily due to higher demand across nearly all product groups and additional sales of a portion of our ethylene glycol production no longer tolled. Average selling prices increased across most product groups in response to the strength of major European currencies and the Australian dollar against the U.S. dollar and higher raw material costs. The decrease in EBITDA was primarily due to mechanical shut downs at our Port Neches, Texas ethylene and ethylene oxide units which resulted in higher costs of approximately \$11 million.

Advanced Materials

The increase in revenues in our Advanced Materials segment for the three months ended March 31, 2010 compared to the same period in 2009 was due to higher sales volumes partially offset by lower average selling prices. Sales volumes increased primarily due to the worldwide economic recovery. Average selling prices decreased in our specialty components business primarily as a result of

changes in our product mix and competitive pressure in the wind generation and coating systems markets. Average selling prices decreased in our base resins business primarily due to lower raw material costs. There was no change in average selling prices within our formulation systems business. The increase in EBITDA was primarily due to higher sales volumes across our businesses, higher contribution margins in our core formulations systems and specialty components businesses and lower fixed costs.

Textile Effects

The increase in revenues in our Textile Effects segment for the three months ended March 31, 2010 compared to the same period in 2009 was due to higher sales volumes and higher average selling prices. Sales volumes increased across all business lines and in all regions primarily due to the worldwide economic recovery. Average selling prices increased primarily due to the strength of the euro, Indian rupee and Brazilian real against the U.S. dollar as well as favorable changes in product mix. The increase in EBITDA was primarily due to higher sales volumes and higher contribution margins, partially offset by higher fixed costs in part due to our second quarter 2009 acquisition of Baroda.

Pigments

The increase in revenues in our Pigments segment for three months ended March 31, 2010 compared to the same period in 2009 was due to higher sales volumes and higher average selling prices. Sales volumes increased primarily due to demand recovery in all regions as a result of the worldwide economic recovery. Average selling prices increased primarily as a result of the strength of major European currencies against the U.S. dollar and higher local currency selling prices in Asia, Africa, Latin America and Middle East regions and higher selling prices in Europe. The increase in EBITDA in our Pigments segment was primarily due to higher sales volumes, lower raw material and energy costs and lower restructuring, impairment and plant closing costs. During the three months ended March 31, 2010 and 2009, our Pigments segment recorded restructuring, impairment and plant closing charges of \$1 million and \$13 million, respectively. For more information concerning restructuring activities, see "Note 6. Restructuring, Impairment and Plant Closing Costs" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

Corporate and other—Huntsman Corporation

Corporate and other includes unallocated corporate overhead, unallocated foreign exchange gains and losses, LIFO inventory valuation reserve adjustments, loss on early extinguishment of debt, income (expenses) associated with the Terminated Merger and related litigation, net income (loss) attributable to noncontrolling interests, unallocated restructuring impairment and plant closing costs and non-operating income and expense. For the three months ended March 31, 2010, EBITDA from Corporate and other items decreased by \$181 million to a loss of \$207 million from a loss of \$26 million for the same period in 2009. The decrease in EBITDA from Corporate and other for the three months ended March 31, 2010 resulted primarily from a \$155 million loss on early extinguishment of debt and to an increase of LIFO inventory valuation expense of \$30 million. For more information regarding the loss on early extinguishment of debt, see "Note 1. General—Recent Developments" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

Corporate and other—Huntsman International

Corporate and other includes unallocated corporate overhead, unallocated foreign exchange gains and losses, LIFO inventory valuation reserve adjustments, loss on early extinguishment of debt, net income (loss) attributable to noncontrolling interests, unallocated restructuring impairment and plant closing costs, gain and loss on the disposition of assets and non-operating income and expense. For the

three months ended March 31, 2010, EBITDA from Corporate and other items decreased by \$35 million to a loss of \$50 million from a loss of \$15 million for the same period in 2009. The decrease in EBITDA from Corporate and other for the three months ended March 31, 2010 resulted primarily from an increase of LIFO inventory valuation expense of \$30 million and to \$9 million from a loss on early extinguishment of debt for the partial redemption of senior subordinated notes. For more information regarding the loss on early extinguishment of debt associated with the partial redemption of senior subordinates notes, see "Note 1. General—Recent Developments" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

Discontinued Operations

The operating results of our former polymers, base chemicals and Australian styrenics businesses are classified as discontinued operations, and, accordingly, the revenues of these businesses are excluded from revenues for all periods presented. The EBITDA of these former businesses are included in discontinued operations for all periods presented. The income (loss) from discontinued operations represents the operating results, partial fire insurance settlement gains and related litigation costs, and restructuring, impairment and plant closing costs and gain (loss) on disposal with respect to our former businesses. See "Note 19. Discontinued Operations" and "Note 16. Casualty Losses and Insurance Recoveries" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

LIQUIDITY AND CAPITAL RESOURCES

The following is a discussion of our liquidity and capital resources and does not include separate information with respect to Huntsman International in accordance with General Instructions H(1)(a) and (b) of Form 10-Q.

Cash

Net cash used in operating activities for the three months ended March 31, 2010 and 2009 was \$408 million and \$38 million, respectively. The increase in cash used in operations was primarily attributable to a \$469 million unfavorable variance in operating assets and liabilities for the three months ended March 31, 2010 as compared with the same period in 2009, offset in part by an increase in operating income as described in "—Results of Operations" above. Upon the adoption of new accounting guidance on January 1, 2010, sales of accounts receivable under our A/R Programs no longer meet the criteria for derecognition and off-balance sheet treatment. Accordingly, the amounts outstanding under our A/R Programs are accounted for as secured borrowings and are now on balance sheet. As a result of the adoption of this new guidance, accounts receivable increased by \$254 million and a corresponding increase in cash used in operating activities was reflected in the statement of cash flows for the three months ended March 31, 2010.

Net cash used in investing activities for the three months ended March 31, 2010 and 2009 was \$43 million and \$62 million, respectively. During the three months ended March 31, 2010 and 2009, we paid \$37 million and \$61 million, respectively, for capital expenditures. This reduction in capital expenditures was largely attributable to higher 2009 spending on various projects, including the maleic anhydride and MDI expansion projects at our Geismar, Louisiana site.

Net cash used in financing activities for the three months ended March 31, 2010 was \$184 million as compared with \$84 million in the 2009 period. This increase in net cash used in financing activities was primarily due to higher net repayments of debt in the 2010 period as compared to the 2009 period and the resulting call premiums paid in association with that debt, partially offset by the revolving loan facility from the A/R Programs that no longer meet the criteria for derecognition upon adoption of new accounting guidance.

Changes in Financial Condition

The following information summarizes our working capital position (dollars in millions):

	rch 31,	De	cember 31, 2009	`	crease) crease	Percent Change
Cash and cash equivalents	\$ 1,110	\$	1,745	\$	(635)	(36)%
Restricted cash	8		5		3	60%
Accounts receivable, net	1,383		1,019		364	36%
Inventories	1,248		1,184		64	5%
Prepaid expenses	36		42		(6)	(14)%
Deferred income taxes	36		36		_	_
Other current assets	132		109		23	21%
Total current assets	3,953		4,140		(187)	(5)%
Accounts payable	898		755		143	19%
Accrued liabilities	564		623		(59)	(9)%
Deferred income taxes	2		2		_	_
Current portion of debt	316		431		(115)	(27)%
Total current liabilities	1,780		1,811		(31)	(2)%
Working capital	\$ 2,173	\$	2,329	\$	(156)	(7)%

Our working capital decreased by \$156 million as a result of the net impact of the following significant changes:

- The decrease in cash and cash equivalents of \$635 million resulted from the matters identified in the condensed consolidated statements of cash flows (unaudited) included elsewhere in this report.
- Accounts receivable increased by \$364 million mainly due to the inclusion in the March 31, 2010 balance sheet of accounts receivable that were sold into the A/R programs and as a result of higher sales during the first quarter of 2010. Upon the adoption of new accounting guidance in 2010, sales of our accounts receivable under the A/R Programs no longer met the criteria for derecognition and off-balance sheet treatment. Accordingly, the amounts outstanding under the A/R Programs are accounted for as secured borrowings beginning January 1, 2010 and are now on balance sheet. For more information, see "Note 7. Debt—Accounts Receivable Securitization" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.
- Inventories increased by \$64 million mainly due to higher raw material costs.
- The increase in accounts payable of \$143 million was primarily due to higher raw material costs.
- Accrued liabilities decreased by \$59 million principally due to payments of accrued restructuring and plant closing costs, accrued rebates and accrued compensation during the first quarter of 2010.
- Current portion of debt decreased by \$115 million. On April 26, 2010, we prepaid \$124 million of Term Loan B and \$40 million of Term Loan C with proceeds from excess cash flow. Accordingly, this \$164 million prepayment was classified as current as of March 31, 2010. At December 31, 2009, the Convertible Notes were classified as current portion of debt. The Convertible Notes were repaid on January 11, 2010. For more information, see "Note 7. Debt—Accounts Receivable Securitization" and "Note 1. General—Recent Developments—Repurchase

of Convertible Notes" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

Direct and Subsidiary Debt

Huntsman Corporation's direct debt and guarantee obligations consist of the following: our guarantees of certain debt of HPS (our Chinese MDI joint venture); our guarantee of certain debt of the Arabian Amines Company; certain indebtedness incurred from time to time to finance certain insurance premiums; and our guarantee of certain obligations of Huntsman International in its capacity as a contributor and servicer guaranter under the U.S. A/R Program.

Substantially all of our other debt has been incurred by our subsidiaries (primarily Huntsman International); such subsidiary debt is nonrecourse to us and we have no contractual obligation to fund our subsidiaries' respective operations.

Transactions Affecting our Debt

Senior Credit Facilities

As of March 31, 2010, our Senior Credit Facilities consisted of (i) our \$225 million Revolving Facility; (ii) our \$1,509 million Term Loan B; and (iii) our \$495 million (\$441 million carrying value) Term Loan C. As of March 31, 2010, we had no borrowings outstanding under our Revolving Facility, and we had approximately \$40 million (U.S. dollar equivalents) of letters of credit and bank guarantees issued and outstanding under our Revolving Facility. All of our Senior Credit Facilities are obligations of Huntsman International.

On March 9, 2010, Huntsman International entered into the Amendment with JPMorgan Chase Bank, N.A., as successor administrative agent and collateral agent, and the other financial institutions party thereto, which amended certain terms of Huntsman International's existing Senior Credit Facilities. Among other things, the Amendment (i) replaces Deutsche Bank AG New York Branch as administrative agent, collateral agent and U.K. security trustee with JPMorgan Chase Bank, N.A. as administrative agent and collateral agent and JPMorgan Chase Bank, N.A. or an affiliate thereof as U.K. security trustee, (ii) extends the stated maturity of the Revolving Facility to March 9, 2014 and provides for optional extensions of such stated maturity date from time to time with the consent of the lenders and subject to certain specified conditions and exceptions, (iii) limits the aggregate amount of the revolving commitments allowable under the Revolving Facility to an amount up to \$300 million, including \$225 million currently committed by the lenders, (iv) terminated the 2009 Waiver and (v) reduces the maximum letter of credit sublimit to \$75 million (not including existing letters of credit issued by Deutsche Bank AG New York Branch) and the maximum swing line sublimit to \$25 million.

Additionally, the Amendment increases the applicable LIBOR margin range on the revolving loans by 1.75% to 3.50% per annum and increases the unused commitment fee percentage to a range of 0.50% to 0.75%. The Amendment also amends the mandatory prepayment provisions of the Senior Credit Facilities to permit the reinvestment of certain insurance and condemnation proceeds.

Our Senior Credit Facilities are subject to a single financial covenant, the Leverage Covenant, which applies only to the Revolving Facility and is tested at the Huntsman International level. The Leverage Covenant is applicable only if borrowings, letters of credit or guarantees are outstanding under the Revolving Facility (cash collateralized letters of credit or guarantees are not deemed outstanding). Following the termination of the 2009 Waiver, the Leverage Covenant is a net senior secured leverage ratio covenant, with a maximum ratio of senior secured debt to EBITDA (as defined in the applicable agreement) of no more than 3.75 to 1.

At the present time, borrowings under the Revolving Facility, Term Loan B and Term Loan C bear interest at LIBOR plus 3.50%, LIBOR plus 1.75% and LIBOR plus 2.25%, respectively. However, the

applicable interest rate of Term Loan B is subject to a reduction to LIBOR plus 1.5% upon achieving certain secured leverage ratio thresholds. The Revolving Facility matures on March 9, 2014 (subject to optional extensions from time to time with the consent of the lenders and subject to certain specified conditions and exceptions), Term Loan B matures in 2014 and Term Loan C matures in 2016; provided, however, that the maturities of the Revolving Facility and the Dollar Term Loans will accelerate if we do not repay or refinance all but \$100 million of Huntsman International's outstanding debt securities on or before three months prior to the maturity dates of such debt securities.

During the three months ended March 31, 2010, we paid the annual scheduled repayment of \$16 million on our Term Loan B and \$5 million on our Term Loan C. On April 26, 2010, we prepaid \$124 million of Term Loan B and \$40 million of Term Loan C with proceeds from excess cash flow.

Accounts Receivable Securitization

Under our A/R Programs, we grant an undivided interest in certain of our trade receivables to bankruptcy-remote special purpose entities. This undivided interest serves as security for the issuance of debt. The A/R Programs provide for financing through a conduit program (in both U.S. dollars and euros). Receivables transferred under the A/R Programs qualified as sales through December 31, 2009. Upon adoption of new accounting guidance in 2010, sales of accounts receivable under our A/R Programs no longer meet the criteria for derecognition. Accordingly, the amounts outstanding under our new A/R Programs are accounted for as secured borrowings as of January 1, 2010. See "Note 2. Summary of Significant Accounting Policies" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

As of March 31, 2010, the A/R Programs had \$242 million in U.S. dollar equivalents in loans outstanding (consisting of \$55 million and \in 139 million (approximately \$187 million)). As of March 31, 2010, \$538 million of accounts receivable were pledged as collateral under the A/R Programs. As of December 31, 2009, the A/R Programs had \$254 million in U.S. dollar equivalents in loans outstanding (consisting of \$55 million and \in 139 million (approximately \$199 million)).

COMPLIANCE WITH COVENANTS

Our management believes that we are in compliance with the covenants contained in the agreements governing our material debt instruments, including our Senior Credit Facilities, our A/R Programs and the indentures governing our notes.

Our Senior Credit Facilities are subject to a single financial covenant, the Leverage Covenant, which applies only to the Revolving Facility and is tested at the Huntsman International level. The Leverage Covenant is applicable only if borrowings, letters of credit or guarantees are outstanding under the Revolving Facility (cash collateralized letters of credit or guarantees are not deemed outstanding). Following the termination of the 2009 Waiver, the Leverage Covenant is a net senior secured leverage ratio covenant, with a maximum ratio of senior secured debt to EBITDA (as defined in the applicable agreement) of no more than 3.75 to 1.

If in the future Huntsman International is not able to meet the secured leverage ratio under the Leverage Covenant, unless Huntsman International obtains an amendment or waiver (as to which we can provide no assurance), then, for so long as Huntsman International did not meet such secured leverage ratio, we would not have access to the liquidity otherwise available under our Revolving Facility. If Huntsman International fails to meet this secured leverage ratio at a time when we have loans or letters of credit outstanding under the Revolving Facility, Huntsman International would be in default under our Senior Credit Facilities, and, unless we obtain waiver or forbearance with respect to such default (as to which we can provide no assurance), we could be required to pay off the balance of our Senior Credit Facilities in full and would not have further access to such facilities.

Our A/R Programs consist of two facilities: Our \$250 million U.S. A/R Program and our €225 (approximately \$302 million) EU A/R Program. The agreements governing our U.S. A/R Program and the agreements governing our EU A/R Program also contain certain financial covenants. Any material failure to meet the applicable A/R Program's covenants in the future could lead to an event of default under the A/R Programs, which could require us to cease our use of such facilities. Under these circumstances, unless any default was remedied or waived, we would likely lose the ability to obtain financing with respect to our trade receivables. A material default under the A/R Programs would also constitute an event of default under our Senior Credit Facilities, which could require us to pay off the balance of the Senior Credit Facilities in full and could result in the loss of our Senior Credit Facilities.

Short-Term and Long-Term Liquidity

We depend upon our cash, credit facilities, A/R Programs and other debt instruments to provide liquidity for our operations and working capital needs. As of March 31, 2010, we had \$1,468 million of combined cash and unused borrowing capacity, consisting of \$1,118 million in cash and restricted cash, \$185 million in availability under our Revolving Facility, and \$165 million in availability under our A/R Programs.

On March 9, 2010, Huntsman International entered into the Amendment to its existing Senior Credit Facilities. Among other things the Amendment limits the aggregate amount of revolving commitments allowable under the Revolving Facility to an amount up to \$300 million, including \$225 million currently committed by lenders. As of March 31, 2010 we had no borrowings outstanding under our Revolving Facility, and we had approximately \$40 million (U.S. dollar equivalents) of letters of credit and bank guarantees issued and outstanding under our Revolving Facility

Our liquidity can be significantly impacted by various factors. Concerning changes in working capital components for the year ended March 31, 2010, our accounts receivable (excluding the \$254 million effect of the on-balance sheet treatment of the A/R Programs) and inventory, net of accounts payable, increased by approximately \$57 million, as reflected in our condensed consolidated statement of cash flows (unaudited) included elsewhere in this report. We expect volatility in our working capital components to continue.

On September 8, 2009, we announced the closure of our styrenics facility located at West Footscray, Australia. We ceased operation of the Australian styrenics operations during the first quarter of 2010. During 2009, we recorded closure costs of approximately \$63 million (\$25 million primarily in severance, \$8 million of contract termination costs and a \$30 million preliminary estimate of environmental remediation costs) and incurred other closure related costs of approximately \$5 million in the first quarter of 2010. We can provide no assurance that the eventual environmental remediation costs will not be materially different from our current estimate. The closure costs are expected to be funded as they are incurred over the next several years, with severance costs to be paid primarily in 2010. During the three months ended March 31, 2010, we paid approximately \$23 million of related restructuring costs and have remaining accruals of approximately \$45 million for restructuring and environmental remediation costs as of March 31, 2010 to be paid out at a later date. For a discussion of restructuring, impairment and plant closing costs, see "Note 6. Restructuring, Impairment and Plant Closing Costs" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

On April 29, 2006, we experienced fire damage at our Port Arthur, Texas facility. This facility has been subsequently rebuilt and sold. In connection with this fire damage, we have received partial insurance proceeds to date of \$365 million. Binding arbitration to settle our claims began on November 2, 2009. The binding arbitration is ongoing. Prior to the commencement of the arbitration proceedings, we had claimed an additional approximately \$242 million plus interest as presently due and owing and unpaid under the Policy for losses caused by the fire. The arbitration panel has made a

preliminary ruling disallowing our claim for interest. In addition, the arbitration panel has made certain preliminary rulings on some of the discrete issues that so far effectively reduce the overall amount we have claimed by approximately \$70 million. The arbitration panel has requested additional briefing on the remaining matters and has not indicated when a final ruling will be issued. Collections on this insured loss, if any, will represent additional income from discontinued operations for us upon final settlement and will be used to repay secured debt in accordance with relevant provisions of agreements governing our Senior Credit Facilities. See "Note 13. Commitments and Contingencies—Port Arthur Plant Fire Insurance Litigation" and "Note 16 Casualty Losses and Insurance Recoveries—Port Arthur, Texas Plant Fire" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

During the three months ended March 31, 2010, we made contributions to our pension and postretirement benefit plans of \$44 million. During the remainder of 2010, we expect to contribute and additional amount of approximately \$77 million to these plans.

As of March 31, 2010, we currently have \$316 million classified as current portion of debt which consists of certain scheduled term payments and various short term facilities, including but not limited to, a Term loan prepayment made on April 26, 2010 of \$164 million, HPS draft discounting facility in China with \$70 million outstanding, our Australian credit facilities with \$38 million outstanding and certain other short term facilities and scheduled amortization payments totaling \$44 million. Although we cannot provide assurances, we intend to renew, repay or extend the majority of these short-term facilities in the current period. In addition, on January 11, 2010 we redeemed the entire \$250 million principal amount (\$236 million carrying value) of our outstanding Convertible Notes for approximately \$382 million. The carrying value of \$236 million relating to the Convertible Notes was classified as current as of December 31, 2009. See "Note 7. Debt—Convertible Notes" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

Due to our ample liquidity in 2010, we believe that the potentially negative effects of the current credit environment have been limited with regard to our need to access the credit markets. Huntsman International was able to successfully access the credit markets during the three months ended March 31, 2010 to raise \$350 million of subordinated debt to redeem existing subordinated debt and completed an amendment to our Senior Credit Facilities which, among other things, extended the maturity of our Revolving Facility by four years to March 2014. See "Note 7 Debt" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

Capital Expenditures

During 2010, we expect to spend between \$250 million and \$275 million for capital expenditures. We expect to fund capital expenditures through a combination of available cash and cash flows from operations.

Receivables Securitization

Receivables transferred under the A/R Programs qualified as sales through December 31, 2009. Upon adoption of new accounting guidance in 2010, sales of accounts receivable under our A/R Programs no longer meet the criteria for derecognition. Accordingly, the amounts outstanding under our new A/R Programs are accounted for as secured borrowings as of January 1, 2010. For a discussion of our A/R Programs, see "Note 7. Debt—Accounts Receivable Securitization" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

Off-Balance Sheet Arrangements

Guarantees

Our unconsolidated joint venture, the Arabian Amines Company, obtained various loan commitments in the aggregate amount of approximately \$195 million (U.S. dollar equivalents) of which \$177 million, including bridge loans, was drawn and outstanding as of March 31, 2010. We have provided certain guarantees of approximately \$14 million for these commitments and our guarantees will terminate upon completion of the project and satisfaction of certain other conditions. We have estimated that the fair value of such guarantees was nil as of the closing date of this transaction and, accordingly, no amounts have been recorded.

Restructuring, Impairment and Plant Closing Costs

For a discussion of restructuring, impairment and plant closing costs, see "Note 6. Restructuring, Impairment and Plant Closing Costs" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

Legal Proceedings

For a discussion of legal proceedings, see "Note 13. Commitments and Contingencies—Legal Matters" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

Environmental, Health and Safety Matters

For a discussion of environmental, health and safety matters, see "Note 14. Environmental, Health and Safety Matters" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

Recently Issued Accounting Pronouncements

For a discussion of recently issued accounting pronouncements, see "Note 2. Recently Issued Accounting Pronouncements" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

Critical Accounting Policies

Our critical accounting policies are presented in Management's Discussion and Analysis of Financial Condition and Results of Operations included in our Annual Report on Form 10-K for the year ended December 31, 2009.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

On January 19, 2010, we entered into a five-year interest rate contract to hedge the variability caused by monthly changes in cash flow due to associated changes in LIBOR under our Senior Credit Facilities. The notional value of the contract is \$50 million and was designated as a cash flow hedge. The effective portion of the changes in the fair value of the swap was recorded in other comprehensive loss. We will pay a fixed 2.8% on the hedge and receive the one-month LIBOR rate. As of March 31, 2010, the fair value of the hedge was less than \$1 million and is recorded in other noncurrent liabilities.

In conjunction with our 2020 Senior Subordinated Note issuance, we entered into cross-currency interest rate contracts with three counterparties. On March 17, 2010, we made payments of \$350 million to these counterparties and received €255 million from these counterparties, and on maturity, March 15, 2015, we are required to pay €255 million to these counterparties and will receive

\$350 million from these counterparties. On March 15 and September 15 of each year, we will receive U.S. dollar interest payments of approximately \$15 million (equivalent to an annual rate of 8.625%) and make interest payments of approximately €11 million (equivalent to an annual rate of approximately 8.41%). This swap is designated as a hedge of net investment for financial reporting purposes. As of March 31, 2010 the fair value of this swap was \$4 million and was recorded as noncurrent assets in our condensed consolidated balance sheet (unaudited). For the three months ended March 31, 2010, the effective portion of the changes in the fair value of \$6 million was recorded in other comprehensive income, with the ineffectiveness of \$2 million recorded in interest expense.

Otherwise, we do not believe there are any material changes to our market risks from those disclosed in our Annual Report on Form 10-K for the year ended December 31, 2009.

ITEM 4. CONTROLS AND PROCEDURES

Our management, with the participation of our chief executive officer and chief financial officer, has evaluated the effectiveness of our disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934 (the "Exchange Act")) as of March 31, 2010. Based on this evaluation, our chief executive officer and chief financial officer have concluded that, as of March 31, 2010, our disclosure controls and procedures were effective, in that they ensure that information required to be disclosed by us in the reports that we file or submit under the Exchange Act is (1) recorded, processed, summarized and reported within the time periods specified in the Commission's rules and forms, and (2) accumulated and communicated to our management, including our chief executive officer and chief financial officer, as appropriate to allow timely decisions regarding required disclosure.

No changes to our internal control over financial reporting occurred during the three months ended March 31, 2010 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting (as defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act). However, we can only give reasonable assurance that our internal controls over financial reporting will prevent or detect material misstatements on a timely basis. Ineffective internal controls over financial reporting could cause investors to lose confidence in our reported financial information and could result in a lower trading price for our securities.

PART II. OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

On occasion, we receive notices of violation, enforcement or other complaints from regulatory agencies alleging non-compliance with applicable EHS laws. Based on currently available information and our past experience, we do not believe that the resolution of any pending or threatened environmental enforcement proceedings will have a material impact on our financial condition, results of operations or cash flows.

In May 2007, our operation in Wilton, U.K., allegedly caused a discharge of wastewater effluent to be made to Northumbrian Water's Bran Sands treatment facility that contained elevated levels of nitrobenzene. Northumbrian Water alleges that this discharge caused a disruption of its treatment facility which, in turn, exceeded its discharge consent from the U.K. Environmental Agency. The Environmental Agency is investigating a possible prosecution against Northumbrian Water and/or us for the breach. Northumbrian Water has threatened to prosecute our subsidiary in the U.K. To date, however, no charges have been filed.

During the period from 2007 through 2009, our Performance Products business' surfactants manufacturing facility located in New South Wales, Australia experienced four reported releases of ethylene oxide and propylene oxide, both raw materials used in the manufacturing process. As a consequence of these releases, the site has received and responded to information requests and physical inspections from WorkCover NSW and/or the Department of Environment and Climate Change NSW, both regulatory agencies with oversight authority for the facility. Although regulatory enforcement action in this matter is a possibility, no monetary sanctions have been proposed by either agency and this matter remains unresolved at the current time.

OTHER LEGAL PROCEEDINGS

For a discussion of other legal proceedings, see "Note 13. Commitments and Contingencies—Legal Matters" and "Note 14. Environmental, Health and Safety Matters—Remediation Liabilities" to our condensed consolidated financial statements (unaudited) included elsewhere in this report.

ITEM 1A. RISK FACTORS

For information regarding risk factors, see "Part I. Item 1A. Risk Factors," in our Annual Report on Form 10-K for the year ended December 31, 2009.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

ISSUER PURCHASES OF EQUITY SECURITIES

The following table presents shares of restricted stock granted under our stock incentive plan that we withheld upon vesting to satisfy our tax withholding obligations during the three months ended March 31, 2009. We have no publicly announced plans or programs to repurchase our common stock.

<u>Period</u>	Total Number of Shares Purchased	Average Price Paid per Share	Total Number of Shares Purchased as Part of Publicly Announced Plans or Programs	Maximum Number (or Approximate Dollar Value) of Shares that May Yet Be Purchased Under the Plans or Programs
January		\$ —		
February	72,816	13.46	_	_
March	285,565	13.61	_	_
Total	358,381	\$ 13.58		

ITEM 6. EXHIBITS

- 3.1 Third Amended and Restated Bylaws of Huntsman Corporation effective March 24, 2010 (incorporated by reference to Exhibit 3.1(i) to our current report on Form 8-K filed on March 26, 2010)
- 4.1 Indenture, dated as of March 17, 2010, by and among Huntsman International LLC, the subsidiary guarantors therein and Wells Fargo Bank, National Association, as trustee (incorporated by reference to Exhibit 4.1 to our current report on Form 8-K filed on March 19, 2010)
- 4.2 Form of 8⁵/8% Senior Subordinated Note (incorporated by reference to Exhibit 4.2 to our current report on Form 8-K filed on March 19, 2010)
- 4.3 Form of Guarantee (incorporated by reference to Exhibit 4.3 to our current report on Form 8-K filed on March 19, 2010)
- 10.1 Fifth Amendment to Credit Agreement, dated as of March 9, 2010, by and among Huntsman International LLC, JPMorgan Chase Bank, N.A. and the other financial institutions party thereto
- 10.2 Registration Rights Agreement, dated as of March 17, 2010, by and among Huntsman International LLC, the subsidiary guarantors named therein and Goldman, Sachs & Co., J.P. Morgan Securities Inc., Barclays Capital Inc., Banc of America Securities LLC, Citigroup Global Markets Inc. and Credit Suisse Securities (USA) LLC (incorporated by reference to Exhibit 10.1 to our current report on Form 8-K filed on March 19, 2010)
- 31.1 Certification of Chief Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
- 31.2 Certification of Chief Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
- 32.1 Certification of Chief Executive Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
- 32.2 Certification of Chief Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrants have duly caused this report to be signed on their behalf by the undersigned thereunto duly authorized.

Dated: May 7, 2010

HUNTSMAN CORPORATION HUNTSMAN INTERNATIONAL LLC

By: /s/ J. KIMO ESPLIN

J. Kimo Esplin

Executive Vice President and Chief Financial

Officer

(Authorized Signatory and Principal Financial Officer)

By: /s/ L. RUSSELL HEALY

L. Russell Healy
Vice President and Controller
(Authorized Signatory and Principal Accounting
Officer)

EXHIBIT INDEX

- 3.1 Third Amended and Restated Bylaws of Huntsman Corporation effective March 24, 2010 (incorporated by reference to Exhibit 3.1(i) to our current report on Form 8-K filed on March 26, 2010)
- 4.1 Indenture, dated as of March 17, 2010, by and among Huntsman International LLC, the subsidiary guarantors therein and Wells Fargo Bank, National Association, as trustee (incorporated by reference to Exhibit 4.1 to our current report on Form 8-K filed on March 19, 2010)
- 4.2 Form of 8⁵/8% Senior Subordinated Note (incorporated by reference to Exhibit 4.2 to our current report on Form 8-K filed on March 19, 2010)
- 4.3 Form of Guarantee (incorporated by reference to Exhibit 4.3 to our current report on Form 8-K filed on March 19, 2010)
- 10.1 Fifth Amendment to Credit Agreement, dated as of March 9, 2010, by and among Huntsman International LLC, JPMorgan Chase Bank, N.A. and the other financial institutions party thereto
- 10.2 Registration Rights Agreement, dated as of March 17, 2010, by and among Huntsman International LLC, the subsidiary guarantors named therein and Goldman, Sachs & Co., J.P. Morgan Securities Inc., Barclays Capital Inc., Banc of America Securities LLC, Citigroup Global Markets Inc. and Credit Suisse Securities (USA) LLC (incorporated by reference to Exhibit 10.1 to our current report on Form 8-K filed on March 19, 2010)
- 31.1 Certification of Chief Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
- 31.2 Certification of Chief Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
- 32.1 Certification of Chief Executive Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
- 32.2 Certification of Chief Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

FIFTH AMENDMENT TO CREDIT AGREEMENT

This FIFTH AMENDMENT TO CREDIT AGREEMENT (this "Amendment"), dated as of March 9, 2010, is entered into by and among Huntsman International LLC, a Delaware limited liability company (the "Borrower"), and the Lenders party hereto. Terms used herein and not otherwise defined herein shall have the same meanings as specified in the Credit Agreement (as defined below).

RECITALS:

- A. The Borrower, the Lenders, the Agents and Deutsche Bank AG New York Branch ("<u>DB</u>") have heretofore entered into that certain Credit Agreement dated as of August 16, 2005 (as heretofore amended, restated, supplemented or otherwise modified, the "Credit Agreement").
- B. The Borrower has requested that the Credit Agreement be amended to, among other things, extend the Revolver Termination Date to March 9, 2014 and reduce the aggregate Revolving Commitments from \$650,000,000 to no more than \$300,000,000 on the terms and subject to the conditions set forth in this Amendment and the Credit Agreement as amended hereby.
- C. DB has submitted its notice of resignation as Administrative Agent, Collateral Agent and UK Security Trustee (in such capacities, collectively, the "Existing Agent") under the Credit Agreement and the other Loan Documents, and desires to resign as a Swing Line Lender under the Credit Agreement.
- D. The Borrower and the Lenders party hereto desire to appoint JPMorgan Chase Bank, N.A. (or, with respect to successor UK Security Trustee, JPMorgan Chase Bank, N.A.) as successor Administrative Agent, successor Collateral Agent and successor UK Security Trustee (in such capacities, collectively, the "Successor Agent") under the Credit Agreement and the other Loan Documents and the Successor Agent wishes to accept such appointment.
 - E. This Amendment constitutes a Loan Document and these Recitals shall be construed as part of this Amendment.

NOW, THEREFORE, in consideration of the Recitals herein contained and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

- **SECTION 1 Amendments.** Effective as of the Amendment Effective Date (as defined below) and subject to the satisfaction of the terms and conditions set forth herein:
- **1.1.** The Credit Agreement is hereby amended and restated in its entirety to be in the form of <u>Exhibit A</u> attached hereto (as amended, the "Amended Credit Agreement").
- 1.2. <u>Schedule 1.1(a)</u> (Commitments) to the Credit Agreement, solely to the extent relating to Revolving Commitments of Lenders, is hereby amended to reflect the Revolving Commitments set forth in Exhibit B-1 attached hereto.
 - 1.3. Schedule 1.1(d) (Fifth Amendment Existing Letters of Credit) set forth in the

Amended Credit Agreement shall be in the form of Exhibit B-2 attached hereto.

- 1.4. Exhibit 2.1(c) (Form of Swing Line Loan Participation Certificate), Exhibit 2.2(a)(2) (Form of Revolving Note), Exhibit 2.2(a)(3) (Form of Swing Line Note), Exhibit 2.5 (Form of Notice of Borrowing), Exhibit 2.6 (Form of Notice of Conversion or Continuation), Exhibit 2.9(b) (Form of Notice of Issuance) and Exhibit 12.8(c) (Form of Assignment and Assumption Agreement) to the Credit Agreement are each hereby amended and restated in their entirety to be in the form of Exhibit C, Exhibit D, Exhibit E, Exhibit F, Exhibit G, Exhibit H and Exhibit I, respectively, attached hereto.
 - **1.5.** <u>Amendment to Security Documents and Other Loan Documents.</u>

The parties hereto acknowledge and agree that:

- (a) Each instance of the definition of "Overdraft Facilities" in the Security Documents is hereby replaced with the following definition:
 - "Overdraft Facilities" means facilities relating to Indebtedness permitted pursuant to Section 8.2(b)(xii) of the Credit Agreement, including, without limitation, guarantees thereof.
- (b) The definition of "Overdraft Facilities" in the Subsidiary Guaranty is hereby replaced with the following definition:
 - "Overdraft Facilities" means facilities relating to Indebtedness (as such term is defined in the Credit Agreement) permitted pursuant to Section 8.2(b)(xii) of the Credit Agreement, including, without limitation, guarantees thereof.

(c) Each instance of the words "Deutsche Bank AG New York Branch" and "DB" in the Security Documents and the Subsidiary Guaranty is hereby replaced with "JPMorgan Chase Bank, N.A." and "JPMCB", respectively.

SECTION 2 Extension of Revolving Commitments; Termination.

all or a portion of their Revolving Commitments, and such Revolving Lenders that have executed and delivered to J.P. Morgan Securities Inc. ("<u>IPM</u>") an Extending Revolving Lender Consent (as defined below) on or prior to 5:00 p.m. (New York time), on March 9, 2010, or on such later date or time as JPM and the Borrower may agree (the "<u>Consent Due Date</u>"), (each, an "<u>Extending Revolving Lender</u>"), shall be a Revolving Lender under the Amended Credit Agreement, and its Revolving Commitment shall be a Revolving Commitment thereunder with respect to the aggregate amount of its Revolving Commitment that is specified on its Extending Revolving Lender Consent; and at the request of the Borrower, certain financial institutions party hereto have agreed to become Revolving Lenders (the "New Revolving Lenders") having Revolving Commitments set forth in Exhibit B-1 attached hereto (the "New Revolving Commitments"); provided, that the aggregate amount of Revolving Commitments after giving effect to this Amendment shall be no more than \$300,000,000; provided further, that to the extent JPM receives Extending Revolving Lender Consents that, together with the New Revolving Commitments, represent Revolving Commitments after giving effect to the Amendment Effective Date in excess of \$300,000,000 in the aggregate, the Borrower shall reduce the Revolving Commitments specified by one or more Extending Revolving Lenders in their Extending Revolving Lender Consents (whether on a pro rata basis or otherwise) such that the aggregate Revolving Commitments after giving effect to the Amendment Effective Date in excess of \$300,000,000.

2

- 2.2. Each Revolving Lender that is not an Extending Revolving Lender (other than each New Revolving Lender) (each a "Non-Extending Revolving Lender") shall, effective upon the Amendment Effective Date, no longer be a Revolving Lender under the Amended Credit Agreement and (i) its Revolving Commitment shall be voluntarily terminated by the Borrower upon the Amendment Effective Date pursuant to Section 4.1(a) of the Credit Agreement and (ii) any outstanding Revolving Loans shall be paid in full together with all accrued and unpaid interest thereon, Commitment Fees and LC Commissions pursuant to Section 4.2(b) hereof.
- **2.3.** On the Amendment Effective Date, the Risk Participation (as defined below) of each Non-Extending Revolving Lender shall be transferred to and assumed by the Extending Revolving Lenders and the New Revolving Lenders on a pro rata basis in accordance with their respective Revolving Commitments under the Amended Credit Agreement. "Risk Participation" means, at any date, the aggregate amount of funded and unfunded obligations of any Revolving Lender to purchase participations from or make payments for the account of (i) any Swing Line Lender pursuant to Sections 2.1(c)(iii) and (iv) of the Credit Agreement and (ii) any Facing Agent pursuant to Section 2.9(d) of the Credit Agreement.
- **2.4.** Each Extending Revolving Lender agrees to promptly return to the Borrower any promissory notes evidencing its Revolving Loans previously issued to such Extending Revolving Lender.

SECTION 3 Appointment of Successor Agent; Waiver of Notice Requirement.

- 3.1. Appointment of Successor Agent. (i) The Borrower and the Lenders party hereto hereby waive, solely for the purpose of this Amendment, any notice, timing or other requirement provided for under the Loan Documents in respect of such resignation of the Existing Agent and the appointment of the Successor Agent and (ii) the Borrower and the Lenders party hereto hereby consent to the appointment of the Successor Agent. The Lenders party hereto hereby authorize the Successor Agent, on behalf of the Lenders, to enter into such supplements, modifications and amendments to the Security Documents with the Credit Parties as it may deem necessary or advisable to evidence the appointment of the Successor Agent and to effectuate the terms hereof and to carry out the purposes and intent of this Amendment.
- 3.2. <u>Waiver of Notice Requirement</u>. The Lenders party hereto hereby waive pursuant to <u>Section 12.1(a)</u> of the Credit Agreement, solely for the purpose of this Amendment, any notice, timing or other requirement of the Credit Agreement (including, without limitation, pursuant to Section 4.1(a) of the Credit Agreement) with respect to a voluntary reduction in Revolving Commitments.
- SECTION 4 Conditions Precedent to the Effectiveness of this Amendment. The provisions of Section 1, Section 2, Section 3 and Section 5 of this Amendment shall become effective upon the date of the satisfaction of all of the conditions set forth in this Section 4 (the "Amendment Effective Date"), with any documents delivered to JPM dated the Amendment Effective Date unless otherwise noted:
- 4.1. Proper Execution and Delivery of Amendment. JPM shall have received (i) this Amendment, duly executed and delivered by the Borrower, the Lenders constituting the Required Lenders, DB (in its capacities as Administrative Agent, Swing Line Lender, Facing Agent, Collateral Agent and UK Security Trustee immediately prior to the Amendment Effective Date) and JPMCB (in its capacities as Administrative Agent, Swing Line Lender, Facing Agent, Collateral Agent and UK Security Trustee immediately upon the Amendment Effective Date), (ii) Extending Revolving Lender Consents, in the form attached hereto as Annex A (the "Extending Revolving Lender Consents") and (iii) the Consent and Reaffirmation, in the form attached hereto as Annex B executed by the Borrower and each of the

4.2. <u>Fees, Interest and Expenses</u>.

- (a) As consideration for the execution of this Amendment (and provided that all of the other conditions set forth in this Section 4 are satisfied), the Borrower agrees to pay to (x) each Extending Revolving Lender and New Revolving Lender the upfront fees as have been posted to Intralinks to such Revolving Lender in connection with the making, extension or increase of its Revolving Commitment and (y) JPM for the account of each Lender (other than New Revolving Lenders) for which JPM shall have received (by facsimile or otherwise) an executed Amendment (or a release from escrow of an executed Amendment previously delivered in escrow for this Amendment) by the Consent Due Date, an amendment fee equal to 0.10% of (i) with respect to each Term Lender, such Term Lender's outstanding Term Loans as of February 28, 2010 and (ii) with respect to each Revolving Lender, such Revolving Lender's Revolving Commitment in effect as of February 28, 2010; provided, that each Lender executing this Amendment acknowledges and consents to the payment of such amendment fee to JPM for the benefit of such Lender.
- (i) repayment of the outstanding Revolving Loans, if any, (ii) all accrued and unpaid interest on the Revolving Loans and all accrued and unpaid Commitment Fees and (ii) all accrued and unpaid LC Commissions with respect to each outstanding Letter of Credit pursuant to Section 2.9(e)(ii) of the Credit Agreement, in each case, to but excluding the Amendment Effective Date (whether or not otherwise due and payable).
- (c) The Borrower shall have paid, to the extent invoiced, the reasonable costs and expenses of JPM and the Existing Agent in connection with the preparation, reproduction, execution and delivery of this Amendment and all other Loan Documents entered into in connection herewith (including, without limitation, the reasonable fees and out-of-pocket expenses of counsel for JPM and the Existing Agent with respect thereto).

4.3. Delivery of Credit Party Documents.

- (a) <u>Notes</u>. The Borrower shall have duly executed and delivered to JPM notes in the form of <u>Exhibit D</u> attached hereto (the "<u>Revolving Notes</u>") payable to each applicable Extending Revolving Lender and New Revolving Lender which has requested a note in the amount of its respective Revolving Commitments after giving effect to this Amendment, all of which shall be in full force and effect
- (b) <u>Opinion of Counsel.</u> JPM shall have received from Vinson & Elkins L.L.P., special counsel to the Borrower, an opinion in the form attached hereto as <u>Exhibit J</u>, addressed to the Existing Agent and each of the Lenders and dated the Amendment Effective Date.
- **4.4.** Successor Agency Agreement. The Successor Agent, the Existing Agent and the Borrower shall have entered into that certain Successor Agency Agreement, dated as of the date hereof, in form and substance satisfactory to the Successor Agent.
- 4.5. <u>Compliance with Flood Insurance Regulations</u>. The Successor Agent shall have received a completed "Life-of-Loan" Federal Emergency Management Agency Standard Flood Hazard Determination with respect to each Mortgaged Property and, for any Mortgaged Property on which improvements are located in a special flood hazard area, a notice duly executed by the Borrower acknowledging the special flood hazard area status together with evidence of flood insurance in form and

4

substance reasonably satisfactory to the Successor Agent.

SECTION 5 Post-Effective Date Requirements.

- **5.1.** Within 120 days after the Amendment Effective Date (or such later date acceptable to the Successor Agent in its sole discretion in writing), the Borrower shall deliver to the Successor Agent:
- (a) Mortgage amendments reflecting the appointment of the Successor Agent as Administrative Agent, Collateral Agent and UK Security Trustee, as applicable (the "Mortgage Amendments"), each in form and substance reasonably satisfactory to the Successor Agent, with respect to the Mortgaged Property, each duly executed and delivered by an authorized officer of each party thereto and in form suitable for filing and recording in all filing or recording offices that the Successor Agent may deem necessary or desirable.
- (b) Date-down endorsements to the title insurance policies with respect to the Mortgaged Real Property reflecting the appointment of the Successor Agent as Administrative Agent, Collateral Agent and UK Security Trustee, as applicable, each in form and substance reasonably satisfactory to the Successor Agent.
- (c) An opinion with respect to each Mortgage Amendment from local counsel to the Borrower addressed to the Successor Agent and each of the Lenders and dated the date of the Mortgage Amendments, in form and substance reasonably satisfactory to the Successor Agent.
- (d) Evidence that the reasonable fees, costs and expenses have been paid, to the extent invoiced, in connection with the preparation, execution, filing and recordation of the Mortgage Amendments, including, without limitation, reasonable attorneys' fees, title insurance premiums, filing and recording fees, title insurance company coordination fees, documentary stamp, mortgage and intangible taxes and title search charges and other charges incurred in connection with the recordation of the Mortgage Amendments and the other matters described in this Section 5.1.

- **5.2.** Within 90 days after the Amendment Effective Date (or such later date acceptable to the Successor Agent in its sole discretion in writing), the Borrower shall deliver to the Successor Agent a Perfection Certificate in form and substance reasonably satisfactory to the Successor Agent and dated as of the date of delivery thereof.
- **5.3.** Within 30 days after the Amendment Effective Date (or such later date acceptable to the Successor Agent in its sole discretion in writing), the Borrower shall deliver to the Successor Agent:
- (a) An opinion of Dickinson Dees LLP, special United Kingdom counsel to the Borrower, or another firm reasonably acceptable to the Successor Agent, in form and substance reasonably satisfactory to the Successor Agent, addressed to the Successor Agent and each of the Lenders and dated as of the date of delivery thereof.
- (b) An opinion of Alvord and Alvord, special railcar counsel to the Borrower, or another firm reasonably acceptable to the Successor Agent, in form and substance reasonably satisfactory to the Successor Agent, addressed to the Successor Agent and each of the Lenders and dated as of the date of delivery thereof.

SECTION 6 Representations and Warranties. On and as of the Amendment Effective Date,

5

after giving effect to this Amendment, the Borrower hereby represents and warrants to each Lender as follows:

- (a) this Amendment has been duly authorized, executed and delivered by the Borrower and each Subsidiary Guarantor, as applicable, and constitutes a legal, valid and binding obligation of the Borrower and each Subsidiary Guarantor, as applicable, enforceable against the Borrower and each Subsidiary Guarantor, as applicable, in accordance with its terms, and the Credit Agreement after giving effect to this Amendment, constitutes the legal, valid and binding obligation of the Borrower enforceable against the Borrower in accordance with its terms (in each case, except to the extent that the enforceability hereof or thereof may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws generally affecting creditors' rights and by equitable principles (regardless of whether enforcement is sought in equity or at law));
- (b) each of the representations and warranties contained in <u>Article VI</u> of the Credit Agreement and in the other Loan Documents are true and correct as of the Amendment Effective Date, except to the extent such representations and warranties are expressly made as of a specific date, in which event such representations and warranties are true and correct as of such specified date; <u>provided, however</u>, that references therein to the "Credit Agreement" shall be deemed to refer to the Credit Agreement after giving effect to this Amendment;
 - (c) no Event of Default or Unmatured Event of Default has occurred and is continuing;
- (d) other than with respect to Letters of Credit issued by DB and outstanding as of the Amendment Effective Date, the Borrower does not currently intend to Cash Collateralize LC Obligations as of March 31, 2010; and
- (e) the Senior Secured Leverage Ratio as of the Amendment Effective Date, after giving effect to the Amended Credit Agreement, on a Pro Forma Basis, does not exceed 3.75 to 1.00.

SECTION 7 References to and Effect on the Credit Agreement.

- **7.1.** On and after the Amendment Effective Date each reference in the Credit Agreement to "this Agreement," "hereunder," "herein," or words of like import, and each reference to the Credit Agreement, as the case may be, in the Loan Documents and all other documents (the "Ancillary Documents") delivered in connection with the Credit Agreement shall mean and be a reference to the Credit Agreement as amended hereby.
- **7.2.** Except as specifically amended above, the Credit Agreement, and the other Loan Documents and all other Ancillary Documents shall remain in full force and effect and are hereby ratified and confirmed.
- **7.3.** The execution, delivery and effectiveness of this Amendment shall not, except as expressly provided herein, operate as a waiver of any right, power or remedy of the Lenders or the Existing Agent under the Credit Agreement, the Loan Documents or the Ancillary Documents.

SECTION 8 Miscellaneous.

8.1. Execution in Counterparts. This Amendment may be executed in one or more counterparts, each of which, when executed and delivered, shall be deemed to be an original and all of which counterparts, taken together, shall constitute but one and the same document with the same force

6

electronic (pdf) transmission shall be deemed to constitute delivery of an originally executed signature page hereto.

- **8.2.** Governing Law. THIS AMENDMENT SHALL BE DEEMED TO BE A CONTRACT MADE UNDER THE LAWS OF THE STATE OF NEW YORK, AND FOR ALL PURPOSES SHALL BE CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS AND DECISIONS OF SAID STATE.
- **8.3.** Headings used in this Amendment are for convenience of reference only and shall not affect the construction of this Amendment.
- **8.4.** Integration. This Amendment, the other agreements and documents executed and delivered pursuant to this Amendment and the Credit Agreement constitute the entire agreement among the parties hereto with respect to the subject matter hereof.
- **8.5. Binding Effect**. This Amendment shall be binding upon and inure to the benefit of and be enforceable by the Borrower and the Administrative Agent and the Lenders and their respective successors and assigns. Except as expressly set forth to the contrary herein, this Amendment shall not be construed so as to confer any right or benefit upon any Person other than the Borrower, the Administrative Agent and the Lenders and their respective successors and permitted assigns.

8.6. <u>Consent to Jurisdiction; Waiver of Jury Trial</u>.

- (a) Each party hereby irrevocably submits to the non-exclusive jurisdiction of any United States federal or New York State court sitting in the City of New York in any action or proceeding arising out of or relating to this Amendment, and hereby irrevocably agrees that all claims in respect to such action or proceeding may be heard and determined in any such United States federal or New York State court and the Borrower and each Lender irrevocably waive any objection, including, without limitation, any objection to the laying of venue or based on the grounds of forum non conveniens which any of them may now or hereafter have to the bringing of any such action or proceeding in such respective jurisdictions.
- (b) Each of the parties hereto irrevocably waives trial by jury in any action or proceeding with respect to this Amendment or any other Loan Document.
- 8.7. <u>Termination of Waiver</u>. Effective upon receipt by JPM of this Amendment, duly executed and delivered by the Borrower and the Lenders constituting the Majority Revolving Facility Lenders, and irrespective of whether the conditions set forth in Section 4 of this Amendment are satisfied, the Waiver to Credit Agreement, dated as of April 16, 2009, entered into by and among the Borrower and the Revolving Lenders party thereto (the "<u>Waiver</u>"), shall terminate and all of the terms and provisions of the Waiver shall no longer be in full force and effect (other than those expressly stated to survive termination); provided, however, that the terms of Section 4(b) (Fees, Interest and Expenses, Etc.) of the Waiver shall remain in full force and effect unless and until the Amendment Effective Date occurs, whereupon the terms of Section 4(b) of the Waiver shall terminate.

[Signature Page Follows]

7

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed and delivered by their duly authorized officers as of the day and year first above written.

HUNTSMAN INTERNATIONAL LLC

By: /s/ John R. Heskett

Name: John R. Heskett

Title: Vice President, Planning and Treasurer

SIGNATURE PAGE TO HUNTSMAN INTERNATIONAL LLC FIFTH AMENDMENT TO CREDIT AGREEMENT

DEUTSCHE BANK AG NEW YORK BRANCH, as

Administrative Agent, Collateral Agent, Facing Agent, Swing Line Lender and UK Security Trustee, in each case immediately prior to the Amendment Effective Date

By: /s/ Omayra Laucella
Name: Omayra Laucella

Title: Vice President

By: /s/ Paul O'Leary

Name: Paul O'Leary
Title: Director

SIGNATURE PAGE TO HUNTSMAN INTERNATIONAL LLC FIFTH AMENDMENT TO CREDIT AGREEMENT

JPMORGAN CHASE BANK, N.A., as Administrative Agent, Collateral Agent, UK Security Trustee, Facing Agent and Swing Line Lender (in each case immediately upon the Amendment Effective Date) and as Lender

By: /s/ Jennifer Heard
Name: Jennifer Heard
Title: Vice President

SIGNATURE PAGE TO HUNTSMAN INTERNATIONAL LLC FIFTH AMENDMENT TO CREDIT AGREEMENT

EXHIBIT A TO FIFTH AMENDMENT TO THE HUNTSMAN INTERNATIONAL LLC CREDIT AGREEMENT

CREDIT AGREEMENT

among

HUNTSMAN INTERNATIONAL LLC,

as the Borrower,

JPMORGAN CHASE BANK, N.A.,

as Administrative Agent

J.P. MORGAN SECURITIES INC.,

as Sole Lead Arranger and Book Runner

and

VARIOUS LENDING INSTITUTIONS,

as Lenders

Dated as of August 16, 2005

as amended by:

Consent and First Amendment to Credit Agreement dated as of December 12, 2005,
Consent and Second Amendment to Credit Agreement and
Amendment to Security Documents dated as of June 30, 2006, and
Third Amendment to Credit Agreement dated as of April 19, 2007
Fourth Amendment to Credit Agreement dated as of June 22, 2009
Fifth Amendment to Credit Agreement dated as of March 9, 2010

TABLE OF CONTENTS

	Refinitions Terms; Financial Statements	5½
ARTICLE I	I AMOUNT AND TERMS OF CREDIT	53
2.1	The Commitments	53
2.2	Notes	57
2.3	Minimum Amount of Each Borrowing; Maximum Number of Borrowings	57
2.4	Interest Rate Options	58
2.5	Notice of Borrowing	58
2.6	Conversion or Continuation	59
2.7	Disbursement of Funds	60
2.8	Pro Rata Borrowings	61
2.9	Amount and Terms of Letters of Credit	61
2.10	Revolving Commitment Increase	69
2.11	Extension of Stated Termination Date	71
ARTICLE I	II INTEREST AND FEES	72
3.1	Interest	72
3.2	Fees	73
3.3	Computation of Interest and Fees	74
3.4	Interest Periods	74
3.5	Compensation for Funding Losses	75
3.6	Increased Costs, Illegality, Etc.	76
3.7	Replacement of Affected Lenders	78
3.8	Impaired Lenders	79
ARTICLET	V REDUCTION OF COMMITMENTS; PAYMENTS AND PREPAYMENTS	81
4.1	Voluntary Reduction of Commitments	81
4.2	Mandatory Reductions of Commitments	82
4.3	Voluntary Prepayments	83
4.4	Mandatory Prepayments	84
	i	
4.5	Application of Prepayments	0.5
	Application of Frepayments	
		87
4.6 4.7	Method and Place of Payment	88
4.6 4.7		
4.7	Method and Place of Payment	88
4.7	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT	88 89
4.7 ARTICLE V	Method and Place of Payment Net Payments	88 89 92
4.7 ARTICLE V 5.1	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date	88 89 92 92
4.7 ARTICLE V 5.1 5.2	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events	88 89 92 92 98
4.7 ARTICLE V 5.1 5.2 5.3 5.4	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments	88 89 92 92 98 99
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments // REPRESENTATIONS AND WARRANTIES	88 89 92 92 98 99 99
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / I REPRESENTATIONS AND WARRANTIES Corporate Status	88 89 92 92 98 99 99
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / I REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority	88 89 92 92 98 99 99
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / I REPRESENTATIONS AND WARRANTIES Corporate Status	88 89 92 92 98 99 99
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / I REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority	88 89 92 92 98 99 99
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / I REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc.	88 89 92 92 98 99 99 100 100 100 101 101
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / I REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation	88 89 92 92 98 99 99 100 100 100 101
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6 6.7	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / I REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation Disclosure	88 89 92 92 98 99 99 100 100 100 101 101 101 103 103
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / I REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation Disclosure Use of Proceeds; Margin Regulations	88 89 92 92 98 99 99 100 100 100 101 101 101 103 103
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / I REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation Disclosure Use of Proceeds; Margin Regulations Tax Returns and Payments	88 89 92 92 98 99 99 100 100 101 101 101 103 103 103
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / I REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation Disclosure Use of Proceeds; Margin Regulations Tax Returns and Payments Compliance With ERISA	88 89 92 92 98 99 99 100 100 101 101 101 103 103 103 104
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation Disclosure Use of Proceeds; Margin Regulations Tax Returns and Payments Compliance With ERISA Ownership of Property	88 89 92 98 99 99 100 100 101 101 101 103 103 103
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11 6.12	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments // REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation Disclosure Use of Proceeds; Margin Regulations Tax Returns and Payments Compliance With ERISA Ownership of Property Capitalization of the Borrower	88 89 92 98 99 99 100 100 101 101 101 103 103 103
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11 6.12 6.13	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments // REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation Disclosure Use of Proceeds; Margin Regulations Tax Returns and Payments Compliance With ERISA Ownership of Property Capitalization of the Borrower Subsidiaries	88 89 92 98 99 99 100 100 101 101 101 103 103 103
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11 6.12 6.13 6.14	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation Disclosure Use of Proceeds; Margin Regulations Tax Returns and Payments Compliance With ERISA Ownership of Property Capitalization of the Borrower Subsidiaries Compliance With Law, Etc.	88 89 92 98 99 99 100 100 101 101 101 103 103 103
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11 6.12 6.13 6.14 6.15	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation Disclosure Use of Proceeds; Margin Regulations Tax Returns and Payments Compliance With ERISA Ownership of Property Capitalization of the Borrower Subsidiaries Compliance With Law, Etc. Investment Company Act	88 89 92 98 99 99 100 100 101 101 101 103 103 103
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11 6.12 6.13 6.14 6.15 6.16	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation Disclosure Use of Proceeds; Margin Regulations Tax Returns and Payments Compliance With ERISA Ownership of Property Capitalization of the Borrower Subsidiaries Compliance With Law, Etc. Investment Company Act Subordination Provisions	88 89 92 98 99 99 100 100 101 101 101 103 103 103
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11 6.12 6.13 6.14 6.15 6.16 6.17	Method and Place of Payment Net Payments // CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments // REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation Disclosure Use of Proceeds; Margin Regulations Tax Returns and Payments Compliance With ERISA Ownership of Property Capitalization of the Borrower Subsidiaries Compliance With Law, Etc. Investment Company Act Subordination Provisions Environmental Matters	88 89 92 92 98 99 99 100 100 101 101 101 103 103 103 104 104 105 106 106 107 107
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11 6.12 6.13 6.14 6.15 6.16 6.17 6.18	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments // REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation Disclosure Use of Proceeds; Margin Regulations Tax Returns and Payments Compliance With ERISA Ownership of Property Capitalization of the Borrower Subsidiaries Compliance With Law, Etc. Investment Company Act Subordination Provisions Environmental Matters Labor Relations	88 89 92 98 99 99 100 100 100 101 101 101
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11 6.12 6.13 6.14 6.15 6.16 6.17 6.18 6.19	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation Disclosure Use of Proceeds; Margin Regulations Tax Returns and Payments Compliance With ERISA Ownership of Property Capitalization of the Borrower Subsidiaries Compliance With Law, Etc. Investment Company Act Subordination Provisions Environmental Matters Labor Relations Intellectual Property, Licenses, Franchises and Formulas	88 89 92 98 99 99 100 100 100 101 101 101
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11 6.12 6.13 6.14 6.15 6.16 6.17 6.18 6.19 6.20	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments // REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation Disclosure Use of Proceeds; Margin Regulations Tax Returns and Payments Compliance With ERISA Ownership of Property Capitalization of the Borrower Subsidiaries Compliance With Law, Etc. Investment Company Act Subordination Provisions Environmental Matters Labor Relations Intellectual Property, Licenses, Franchises and Formulas Certain Fees	88 89 92 92 98 99 99 100 100 100 101 101 101 103 103 103 104 104 105 106 106 107 107 107 107 107 107 108 108
4.7 ARTICLE V 5.1 5.2 5.3 5.4 ARTICLE V 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11 6.12 6.13 6.14 6.15 6.16 6.17 6.18 6.19	Method and Place of Payment Net Payments / CONDITIONS OF CREDIT Conditions Precedent to the Closing Date Conditions Precedent to All Credit Events Additional Conditions to All Credit Events Additional Conditions to Usage of Revolving Commitments / REPRESENTATIONS AND WARRANTIES Corporate Status Corporate Power and Authority No Violation Governmental and Other Approvals Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc. Litigation Disclosure Use of Proceeds; Margin Regulations Tax Returns and Payments Compliance With ERISA Ownership of Property Capitalization of the Borrower Subsidiaries Compliance With Law, Etc. Investment Company Act Subordination Provisions Environmental Matters Labor Relations Intellectual Property, Licenses, Franchises and Formulas	88 89 92 98 99 99 100 100 100 101 101 101

ARTICLE	VII AFFIRMATIVE COVENANTS	111
7.1	Financial Statements	111
7.2	Certificates; Other Information	112
7.3	Notices	113
7.4	Conduct of Business and Maintenance of Existence	115
7.5	Payment of Obligations	115
7.6	Inspection of Property, Books and Records	115
7.7	ERISA	116
7.8	Maintenance of Property, Insurance	117
7.9	Environmental Laws	118
7.10	Use of Proceeds	119
7.11	Additional Security; Further Assurances	119
7.12	End of Fiscal Years; Fiscal Quarters	122
7.12	Maintenance of Ratings	122
7.13	Certain Fees Indemnity	122
7.14	Successor Agency Transfer	122
7.13	Successor Agency Transier	122
ARTICLE	VIII NEGATIVE COVENANTS	123
8.1	Liens	123
8.2	Indebtedness	125
8.3		123
	Consolidation, Merger, Purchase or Sale of Assets, etc.	
8.4	Dividends or Other Distributions	130
8.5	Certain Restrictions on Subsidiaries	131
8.6	Issuance of Stock	132
8.7	Loans and Investments	132
8.8	Transactions with Affiliates	135
8.9	Lines of Business	136
8.10	Fiscal Year	136
8.11	Limitation on Voluntary Payments and Modifications of Indebtedness; Modifications of Certificate of	
	Incorporation, By-Laws and Certain Other Agreements; Etc.	136
8.12	Accounting Changes	137
8.13	Permitted Accounts Receivable Securitization and Foreign Factoring Transactions	138
	iii	
ARTICLE 9.1	IX FINANCIAL COVENANTS Senior Secured Leverage Ratio	138 138
ADTICLE	V EVENTO OF DEFAULT	120
	X EVENTS OF DEFAULT	138
	Events of Default	138
10.2	Rights Not Exclusive	143
ADTICLE	VI THE ADMINISTRATIVE ACENT	143
11.1	XI THE ADMINISTRATIVE AGENT	143
	Appointment	
11.2	Nature of Duties	144
11.3	Exculpation, Rights Etc.	144
11.4	Reliance	145
11.5	Indemnification	145
11.6	The Administrative Agent In Its Individual Capacity	146
11.7		146
11.8	Notice of Default	
11.9	Holders of Obligations	146
11.10	Holders of Obligations Resignation by the Administrative Agent	146 146
11.11	Holders of Obligations Resignation by the Administrative Agent Administrative Agent or the Collateral Agent as UK Security Trustee	
	Holders of Obligations Resignation by the Administrative Agent	146
	Holders of Obligations Resignation by the Administrative Agent Administrative Agent or the Collateral Agent as UK Security Trustee	146
	Holders of Obligations Resignation by the Administrative Agent Administrative Agent or the Collateral Agent as UK Security Trustee The Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior	146 148
ARTICLE	Holders of Obligations Resignation by the Administrative Agent Administrative Agent or the Collateral Agent as UK Security Trustee The Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents XII MISCELLANEOUS	146 148 149
ARTICLE 12.1	Holders of Obligations Resignation by the Administrative Agent Administrative Agent or the Collateral Agent as UK Security Trustee The Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents	146 148 149
12.1 12.2	Holders of Obligations Resignation by the Administrative Agent Administrative Agent or the Collateral Agent as UK Security Trustee The Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents XII MISCELLANEOUS	146 148 149
12.1	Holders of Obligations Resignation by the Administrative Agent Administrative Agent or the Collateral Agent as UK Security Trustee The Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents XII MISCELLANEOUS No Waiver; Modifications in Writing	146 148 149 150 150
12.1 12.2	Holders of Obligations Resignation by the Administrative Agent Administrative Agent or the Collateral Agent as UK Security Trustee The Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents XII MISCELLANEOUS No Waiver; Modifications in Writing Further Assurances	146 148 149 150 150 152
12.1 12.2 12.3	Holders of Obligations Resignation by the Administrative Agent Administrative Agent or the Collateral Agent as UK Security Trustee The Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents XII MISCELLANEOUS No Waiver; Modifications in Writing Further Assurances Notices, Etc.	146 148 149 150 150 152 152
12.1 12.2 12.3 12.4	Holders of Obligations Resignation by the Administrative Agent Administrative Agent or the Collateral Agent as UK Security Trustee The Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents XII MISCELLANEOUS No Waiver; Modifications in Writing Further Assurances Notices, Etc. Costs, Expenses and Taxes Confirmations	146 148 149 150 150 152 152 154
12.1 12.2 12.3 12.4 12.5	Holders of Obligations Resignation by the Administrative Agent Administrative Agent or the Collateral Agent as UK Security Trustee The Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents XII MISCELLANEOUS No Waiver; Modifications in Writing Further Assurances Notices, Etc. Costs, Expenses and Taxes Confirmations Adjustment; Setoff	146 148 149 150 150 152 152 154 156
12.1 12.2 12.3 12.4 12.5 12.6 12.7	Holders of Obligations Resignation by the Administrative Agent Administrative Agent or the Collateral Agent as UK Security Trustee The Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents XII MISCELLANEOUS No Waiver; Modifications in Writing Further Assurances Notices, Etc. Costs, Expenses and Taxes Confirmations Adjustment; Setoff Execution in Counterparts	146 148 149 150 150 152 152 154 156 157 158
12.1 12.2 12.3 12.4 12.5 12.6 12.7 12.8	Holders of Obligations Resignation by the Administrative Agent Administrative Agent or the Collateral Agent as UK Security Trustee The Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents XII MISCELLANEOUS No Waiver; Modifications in Writing Further Assurances Notices, Etc. Costs, Expenses and Taxes Confirmations Adjustment; Setoff Execution in Counterparts Binding Effect; Assignment; Addition and Substitution of Lenders	146 148 149 150 152 152 154 156 157 158 158
12.1 12.2 12.3 12.4 12.5 12.6 12.7 12.8 12.9	Holders of Obligations Resignation by the Administrative Agent Administrative Agent or the Collateral Agent as UK Security Trustee The Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents XII MISCELLANEOUS No Waiver; Modifications in Writing Further Assurances Notices, Etc. Costs, Expenses and Taxes Confirmations Adjustment; Setoff Execution in Counterparts Binding Effect; Assignment; Addition and Substitution of Lenders CONSENT TO JURISDICTION; MUTUAL WAIVER OF JURY TRIAL	146 148 149 150 150 152 152 154 156 157 158 158
12.1 12.2 12.3 12.4 12.5 12.6 12.7 12.8	Holders of Obligations Resignation by the Administrative Agent Administrative Agent or the Collateral Agent as UK Security Trustee The Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents XII MISCELLANEOUS No Waiver; Modifications in Writing Further Assurances Notices, Etc. Costs, Expenses and Taxes Confirmations Adjustment; Setoff Execution in Counterparts Binding Effect; Assignment; Addition and Substitution of Lenders CONSENT TO JURISDICTION; MUTUAL WAIVER OF JURY TRIAL	146 148 149 150 152 152 154 156 157 158 158

12.14	Confidentiality	163
12.15	Concerning the Collateral and the Loan Documents	164
12.16	Intentionally Omitted	166
12.17	Registry	166
12.18	Accounts Receivable Securitization	167
12.19	Certain Guarantee Obligations	167
12.20	Redesignation of Unrestricted Subsidiaries	168
12.21	Administrative Agent and Collateral Agent as Joint Creditors	169
12.22	Amendment With Respect to Revolver Events of Default	169
12.23	Term C Dollar Lenders	169

v

INDEX OF EXHIBITS AND SCHEDULES

$\underline{Exhibits}$

Exhibit 2.1(c)	Form of Swing Line Loan Participation Certificate
Exhibit 2.2(a)(1)	Form of Term B Dollar Note
Exhibit 2.2(a)(2)	Form of Revolving Note
Exhibit 2.2(a)(3)	Form of Swing Line Note
Exhibit 2.5	Form of Notice of Borrowing
Exhibit 2.6	Form of Notice of Conversion or Continuation
Exhibit 2.9(b)	Form of Notice of Issuance
Exhibit 4.7(d)	Form of Section 4.7(d)(i) Certificate
Exhibit 5.1(c)	Form of Pledge Agreement
Exhibit 5.1(d)(i)	Form of Subsidiary Guaranty Agreement
Exhibit 5.1(d)(ii)	Form of Headquarters Subsidiary Guaranty Agreement
Exhibit 5.1(f)	Form of Perfection Certificate
Exhibit 5.1(s)(i)	Form of Vinson & Elkins L.L.P. Legal Opinion
Exhibit 5.1(s)(ii)	Form of Stoel Rives LLP Legal Opinion
Exhibit 5.1(s)(iii)	Form of Alvord and Alvord Legal Opinion
Exhibit 5.1(v)	Form of Tax Sharing Agreement
Exhibit 7.2(b)	Form of Compliance Certificate
Exhibit 8.7(g)	Form of Subordination Provisions
Exhibit 12.8(c)	Form of Assignment and Assumption Agreement
	Tri

Schedules

Schedule 1.1(b)	Calculation of the Mandatory Cost
Schedule 1.1(c)	Unrestricted Subsidiaries
Schedule 1.1(d)	Fifth Amendment Existing Letters of Credit
Schedule 2.9(j)	Outstanding Letters of Credit
Schedule 5.1(i)(iii)	List of Foreign Intercompany Loan Security Document Deliveries
Schedule 6.5(c)	Existing Liabilities
Schedule 6.5(e)	Projections
Schedule 6.12(a)	Capitalization of the Borrower
Schedule 6.13	List of Subsidiaries
Schedule 6.21(c)	Owned and Leased Properties
Schedule 7.8	Insurance Levels
Schedule 8.1(h)	Existing Liens
Schedule 8.2(b)(ii)	Existing Indebtedness
Schedule 8.5(a)	Existing Restrictions on Subsidiaries
Schedule 8.7(b)	Existing Investments
Schedule 8.7(h)	Investments With Respect to Forgiveness of Certain Intercompany Debt
Schedule 8.9	IRIC Account Procedures
Schedule 12.3	Notice Information
	vii

CREDIT AGREEMENT

THIS CREDIT AGREEMENT is dated as of August 16, 2005 and is made by and among Huntsman International LLC, a Delaware limited liability company (the "Borrower"), the financial institutions party hereto, in their capacities as lenders hereunder (collectively, the "Lenders," and each individually, a "Lender"), JPMorgan Chase Bank, N.A., as Administrative Agent (acting in such capacity, the "Administrative Agent") for the Lenders, and J.P. Morgan Securities Inc., as Sole Lead Arranger and Book Runner.

WITNESSETH:

WHEREAS, on the Closing Date, Huntsman LLC, a Delaware limited liability company ("HLLC") will merge with and into the Borrower, with the Borrower as the surviving entity (the "Merger") pursuant to the terms of the Merger Agreement (as defined herein);

WHEREAS, this Agreement will repay in full, extinguish and replace (i) that certain Revolving Credit Agreement dated as of October 14, 2004 by and among HLLC, Deutsche Bank Trust Company Americas ("DBTCA"), as administrative agent and the lenders party thereto (the "Prior HLLC Revolving Credit Agreement"), (ii) that certain Credit Agreement dated as of October 14, 2004 by and among HLLC, DBTCA, as administrative agent and collateral agent and the lenders party thereto (the "Prior HLLC Term Credit Agreement") and (iii) that certain Amended and Restated Credit Agreement dated as of July 13, 2004 by and among the Borrower, Huntsman International Holdings LLC, the financial institutions party thereto, DBTCA, as administrative agent, and the co-lead arrangers, co-syndication agents and co-documentation agents identified therein (the "Prior HI Credit Agreement" and together with the Prior HLLC Revolving Credit Agreement and the Prior Term HLLC Credit Agreement the "Prior Credit Agreements");

NOW THEREFORE, in consideration of the premises and of the mutual covenants herein contained the parties hereto agree as follows:

SECTION 9

DEFINITIONS AND ACCOUNTING TERMS

9.1. Definitions

As used herein, and unless the context requires a different meaning, the following terms have the meanings indicated:

"Accounts Receivable" means presently existing and hereafter arising or acquired accounts receivable, notes, drafts, acceptances, general intangibles, choses in action and other forms of obligations and receivables relating in any way to Inventory or arising from the sale of Inventory or the rendering of services by the Borrower or its Subsidiaries or howsoever otherwise arising, including the right to payment of any interest or finance charges with respect thereto and all proceeds of insurance with respect thereto, together with all of the Borrower's or

1

its Subsidiaries' rights as an unpaid vendor, all pledged assets, guaranty claims, liens and security interests held by or granted to the Borrower or its Subsidiaries to secure payment of any Accounts Receivable and all books, customer lists, ledgers, records and files (whether written or stored electronically) relating to any of the foregoing.

"Acquired Debt" has the meaning assigned to such term in Section 8.2(b)(xi).

"Acquisition" has the meaning assigned to that term in Section 8.7(m).

"Additional Security Documents" means all mortgages, pledge agreements, security agreements, reaffirmations and other security documents entered into pursuant to Section 7.11 with respect to additional Collateral, in each case, as amended, supplemented or otherwise modified from time to time.

"Additional Term Loans" means term loans and commitments to make term loans whose contractual priority of payment is pari passu in all respects with the Term B Dollar Loans made pursuant to Section 2.1(a)(i) that (i) have a Weighted Average Life to Maturity of not less than the Term Loan with the then longest Weighted Average Life to Maturity and a final maturity no earlier than latest Term Maturity Date; (ii) are on terms and conditions substantially similar to those applicable to the existing Term Facilities and (iii) have applicable margins (which, for such purposes only, shall be deemed to include all upfront or similar compensation or original issue discount (amortized over an assumed three year life) payable to all Lenders providing such loans, but exclusive of any arrangement, structuring or other similar fees payable in connection therewith that are not shared with all Lenders providing such loans) determined as of the initial funding date for such loans not greater than 0.50% above the applicable margins then in effect for Term B Dollar Loans (which, for such purposes only, shall be deemed to include all upfront or similar compensation or original issue discount (amortized over an assumed three year life) paid to all Term B Dollar Lenders as of the initial funding date for such Term B Dollar Loans, but exclusive of any arrangement, structuring or other similar fees payable in connection therewith that are not shared with all Term B Dollar Lenders).

"Administrative Agent" has the meaning assigned to that term in the introduction to this Agreement, and includes any successor Administrative Agent in such capacity.

"Affiliate" means, with respect to any Person, any Person or group acting in concert in respect of the Person in question that, directly or indirectly, controls (including but not limited to all directors and officers of such Person) or is controlled by or is under common control with such Person; provided that no Agent nor any Affiliate of an Agent shall be deemed to be an Affiliate of the Borrower. For the purposes of this definition, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to any Person or group of Persons, shall mean the possession, directly or indirectly, of the power to direct or cause the direction of management and policies of such Person, whether through the ownership of Voting Securities or by contract or otherwise. A Person shall be deemed to control a Person if such first Person possesses, directly or indirectly, the power to vote 15% or more of the securities having ordinary voting power for the election of directors, managers or similar governing body of such Person.

2

"Agent" has the meaning assigned to that term in the introduction to this agreement and includes any successor agent in any such capacity.

"Agreement" means this Credit Agreement, as the same may at any time be amended, supplemented or otherwise modified in accordance with the terms hereof and in effect.

"<u>Airstar Aircraft Financing Documents</u>" means operating leases and related documents entered into by Airstar Corporation relating to aircraft owned or acquired by it and any agreements or documents entered into by Airstar Corporation.

"Alternative Currency" means, with respect to (i) Revolving Loans, Euros and Sterling, and (ii) any Letter of Credit or Swing Line Loans, Euros, Sterling and any currency which is freely transferable and convertible into Dollars.

"Applicable Base Rate Margin" means at any date, (i) with respect to Revolving Loans denominated in Dollars, (A) during the period commencing on the Fifth Amendment Effective Date and ending on the date that financial statements for the Fiscal Quarter ending March 31, 2010 are delivered to the Lenders pursuant to Section 7.1, 2.50% per annum, and (B) thereafter, the applicable percentage set forth in the following table under the column Applicable Base Rate Margin for Revolving Loans opposite the Most Recent Senior Secured Leverage Ratio as of such date, (ii) with respect to Term C Dollar Loans, 1.25% and (iii) with respect to Term B Dollar Loans, the applicable percentage set forth under the column Applicable Base Rate Margin for Term B Dollar Loans opposite the Most Recent Senior Secured Leverage Ratio as of such date:

Most Recent Senior Secured Leverage Ratio	Margin for Revolving Loans
Less than or equal to 2.00 to 1	2.00 %
Greater than 2.00 to 1 but less than or equal to 2.50 to 1	2.25 %
Greater than 2.50 to 1	2.50 %
Most Recent Senior Secured Leverage Ratio	Applicable Base Rate Margin for Term B Dollar Loans
Less than or equal to 2.25 to 1	0.50 %
	0.75 %

"Applicable Commitment Fee Percentage" means at any date, (i) during the period commencing on the Fifth Amendment Effective Date and ending on the date that financial statements for the Fiscal Quarter ending March 31, 2010 are delivered to the Lenders pursuant to Section 7.1, 0.75% per annum and (ii) thereafter, the applicable percentage set forth in the following table under the column Applicable Commitment Fee Percentage opposite the Most Recent Senior Secured Leverage Ratio as of such date:

3

Most Recent	Applicable Commitment	
Senior Secured Leverage Ratio	Fee Percentage	
Less than or equal to 2.00 to 1	0.50 %	
Greater than 2.00 to 1 but less than or equal to 2.50 to 1	0.625 %	
Greater than 2.50 to 1	0.75 %	

"Applicable Currency" means as to any particular payment or Loan, Dollars or the Alternative Currency in which such payment or Loan is denominated or is payable.

"Applicable Eurocurrency Margin" means at any date, (i) with respect to Term B Dollar Loans, the applicable percentage set forth in the following table under the column Applicable Eurocurrency Margin for Term B Dollar Loans opposite the Most Recent Senior Secured Leverage Ratio on such date, (ii) with respect to Term C Dollar Loans, 2.25% and (iii) with respect to Revolving Loans, (A) during the period commencing on the Fifth Amendment Effective Date and ending on the date that financial statements for the Fiscal Quarter ending March 31, 2010 are delivered to the Lenders pursuant to Section 7.1, 3.50% per annum and (B) thereafter, the applicable percentage set forth in the following table under the column Applicable Eurocurrency Margin for Revolving Loans opposite the Most Recent Senior Secured Leverage Ratio as of such date:

Most Recent	Applicable	
Senior Secured Leverage	Eurocurrency Margin for	
Ratio	Revolving Loans	
Less than or equal to 2.00 to 1	3.00 %	
Greater than 2.00 to 1 but less than or equal to 2.50 to 1	3.25 %	
Greater than 2.50 to 1	3.50%	

4

	Applicable Eurocurrency	
Most Recent	Margin for Term B	
Senior Secured Leverage Ratio	Dollar Loans	
Less than or equal to 2.25 to 1	1.50 %	
Greater than 2.25 to 1	1.75%	

"Asset Disposition" means any sale, lease, transfer or other disposition (or series of related sales, leases, transfers or dispositions) by the Borrower or any of its Subsidiaries (other than Dividends or Tax Distributions to the extent permitted under Section 8.4 or dispositions constituting a Recovery Event) of all or any part of an interest in shares of Capital Stock of a Subsidiary of the Borrower (other than directors' qualifying shares and similar arrangements required by Requirements of Law), property or other assets (each referred to for the purposes of this definition as a "disposition"); provided that a disposition permitted by Section 8.3(a) through 8.3(g) or Section 8.3(j) through 8.3(n) shall not constitute an Asset Disposition for purposes of this definition.

"Assigned Dollar Value" shall mean (i) in respect of any Borrowing denominated in Dollars, the amount thereof, (ii) in respect of the undrawn amount of any Letter of Credit denominated in an Alternative Currency, the Dollar Equivalent thereof based upon the applicable Exchange Rate as of (a) the date of issuance of such Letter of Credit, and (b) thereafter as of the first Business Day of each month, (iii) in respect of any Letter of Credit reimbursement obligations denominated in an Alternative Currency, the Dollar Equivalent thereof determined based upon the applicable Exchange Rate as of the date such reimbursement obligation was incurred and (iv) in respect of a Borrowing denominated in an Alternative Currency, the Dollar Equivalent thereof based upon the applicable Exchange Rate as of the last Exchange Rate Determination Date; provided, however, in the case of Borrowings in an Alternative Currency, if, as of the end of any Interest Period in respect of such Borrowing, the Dollar Equivalent thereof determined based upon the applicable Exchange Rate as of the date that is three Business Days before the end of such Interest Period would be at least 5% more, or 5% less, than the "Assigned Dollar Value" thereof that would otherwise be applicable, then on and after the end of such Interest Period the "Assigned Dollar Value" of such Borrowing shall be adjusted to be the Dollar Equivalent thereof determined based upon the Exchange Rate that

gave rise to such adjustment (subject to further adjustment in accordance with this proviso thereafter), and the Administrative Agent shall give the Borrower notice of such adjustment; <u>provided</u>, <u>however</u>, that failure to give such notice shall not affect the Borrower's Obligations hereunder or result in any liability to the Administrative Agent. The Assigned Dollar Value of a Loan included in any Borrowing shall equal the pro rata portion of the Assigned Dollar Value of such Borrowing represented by such Loan.

"Assignee" has the meaning assigned to that term in Section 12.8(c).

"Assignment and Assumption Agreement' means an Assignment and Assumption Agreement substantially in the form of Exhibit 12.8(c), executed by any applicable Lender, as assignor, and such Lender's assignee in accordance with Section 12.8.

5

"Attorney Costs" means all reasonable fees and disbursements of any law firm or other external counsel and the reasonable allocated cost of internal legal services, including all reasonable disbursements of internal counsel.

"Attributable Debt" means as of the date of determination thereof with respect to an Operating Financing Lease, the net present value (discounted according to GAAP at the cost of debt implied in the lease) of the obligations of the lessee for rental payments during the then remaining term of such Operating Financing Lease.

"Available Equity Proceeds" means, as of any date of determination, all Net Equity Proceeds received by the Borrower after the Effective Date less the sum of the amounts attributed to such proceeds that are utilized for (i) Restricted Payments pursuant to Section 8.4(b), (ii) Unrestricted Investments pursuant to Section 8.7(o) and (iii) payments in respect of Public Notes pursuant to Section 8.11(i).

"Available Liquidity" means, as of any date of determination, Cash, Cash Equivalents and Foreign Cash Equivalents of the Borrower and its Subsidiaries plus the Total Available Revolving Commitment as of such date.

"Available Revolving Commitment" means, as to any Lender at any time an amount equal to the amount, if any, by which (i) such Lender's Revolving Commitment exceeds (ii) the sum of (w) the Assigned Dollar Value of the aggregate principal amount of Revolving Loans made by such Lender then outstanding, (x) such Lender's Pro Rata Share of the Assigned Dollar Value of LC Obligations, (y) such Lender's Pro Rata Share of the Assigned Dollar Value of the aggregate principal amount of Swing Line Loans then outstanding and (z) such Lender's Pro Rata Share of the Overdraft Reserve, if any, at such time.

"Available Unrestricted Subsidiary Investment Basket" means, as of any date of determination, an amount equal to the Unrestricted Subsidiary Investment Basket, as of such date, less the sum of the aggregate outstanding amount of Investments made in Permitted Unconsolidated Ventures or Unrestricted Subsidiaries pursuant to Section 8.7(j)(i).

"Bank Guarantee" means a direct guarantee issued for the account of the Borrower, and, if requested, a Subsidiary of the Borrower, pursuant to this Agreement by a Facing Agent, in form acceptable to the Facing Agent, ensuring that a liability acceptable to the Facing Agent of the Borrower or a Subsidiary of a Borrower to a third Person will be met.

"<u>Bankruptcy Code</u>" means Title I of the Bankruptcy Reform Act of 1978, as amended, as set forth in Title 11 of the United States Code, as hereafter amended.

"Base Rate" means the greater of (i) the rate most recently announced by JPMCB at its principal office as its "prime rate", which is not necessarily the lowest rate made available by JPMCB or (ii) the Federal Funds Rate plus 1/2 of 1% per annum. The "prime rate" announced by JPMCB is evidenced by the recording thereof after its announcement in such internal publication or publications as JPMCB may designate. Any change in the Base Rate resulting from a change in such "prime rate" announced by JPMCB shall become effective without prior notice to the Borrower as of 12:01 a.m. (New York City time) on the Business Day

6

on which each change in such "prime rate" is announced by JPMCB. JPMCB may make commercial or other loans to others at rates of interest at, above or below its "prime rate".

"Base Rate Loan" means any Loan which bears interest at a rate determined with reference to the Base Rate.

"Benefited Lender" has the meaning assigned to that term in Section 12.6(a).

"Board" means the Board of Governors of the Federal Reserve System.

"Borrower" has the meaning assigned to that term in the introduction to this Agreement.

"Borrowing" means a group of Loans of a single Type made by the Lenders or the Swing Line Lender, as appropriate, on a single date (or resulting from one or more conversions or continuations, or a combination thereof, on a single date) and in the case of Eurocurrency Loans, as to which a single Interest Period is, or on such date of conversion or continuation, will be in effect; provided that Base Rate Loans or Eurocurrency Loans acquired by a Replacement Lender pursuant to Section 3.7 shall be considered part of any related Borrowing of Eurocurrency Loans made, converted or continued by the Replaced Lender.

"Business Day" means (i) as it relates to any payment, determination, funding or notice to be made or given in connection with any Dollar-denominated Loan, or otherwise to be made or given to or from the Administrative Agent, a day other than a Saturday, Sunday or other day on which commercial banks in New York City are authorized or required by law to close; provided, however, that when used in connection with a Eurocurrency Loan, the term "Business Day" shall also exclude any day on which banks are not open for dealings in dollar deposits in the London interbank market; provided, further, that when used in connection with any Letter of Credit, the term "Business Day" shall also exclude any day on which commercial banks in the city in which the Facing Agent for such Letter of Credit is domiciled are required by law to close; and (ii) as it relates to any payment, determination, funding or notice to be made or given in connection with any Loan or Letter of Credit denominated in an Alternative Currency, any day (x) on which dealings in deposits in the relevant Alternative Currency are carried out in the London interbank market, (y) on which commercial banks and foreign exchange markets are open for business in London, New York City and the principal financial center for such Alternative Currency or (z) any day on which the Trans-European Real-time Gross Settlement Operating System (or any successor operating system) is not operating (as determined in good faith by the Administrative Agent). For purposes of this Agreement (other than for purposes of determining the end of any applicable Interest Period and other than for purposes of any Loan, Letter of Credit or action required to be taken outside of the United States), "Business Day" shall not include Pioneer Day as recognized in the State of Utah in any year.

"<u>C4 Business Sale</u>" means the sale of Huntsman Petrochemical Corporation and Huntsman Fuels, L.P. to Texas Petrochemicals L.P. pursuant to that certain Asset Purchase Agreement dated as of April 5, 2006.

"Capital Stock" means, with respect to any Person, any and all shares, interests, participations, rights in or other equivalent ownership interests (however designated) in such

7

Person's capital stock, partnership interests, membership interests or other equivalent interests and any rights (other than debt securities convertible into or exchangeable for capital stock), warrants or options exchangeable for or convertible into any such ownership interests.

"<u>Capitalized Lease</u>" means, at the time any determination thereof is to be made, any lease of property, real or personal, in respect of which the present value of the minimum rental commitment is capitalized on the balance sheet of the lessee in accordance with GAAP.

"Capitalized Lease Obligation" means, at the time any determination thereof is to be made, the amount of the liability in respect of a Capitalized Lease which would at such time be required to be capitalized on the balance sheet of the lessee in accordance with GAAP.

"Cash" means money, currency or the available credit balance in a Deposit Account.

"Cash Collateralize" means, in respect of an obligation, provide and pledge (as a first priority perfected security interest) Cash in Dollars (or cash in an Alternative Currency if the Administrative Agent, relevant Facing Agent and/or the Swing Line Lender so approves), at a location and pursuant to documentation in form and substance reasonably satisfactory to the Administrative Agent, relevant Facing Agent and/or the Swing Line Lender, as applicable (and "Cash Collateralization" has a corresponding meaning).

"Cash Equivalents" means any Investment in (i) a marketable obligation, maturing within two years after issuance thereof, issued or guaranteed by the United States of America or any instrumentality or agency thereof, (ii) a certificate of deposit or banker's acceptance, maturing within one year after issuance thereof, issued by any Lender, or a national or state bank or trust company organized and existing under the laws of the United States or any state thereof or a European, Canadian or Japanese bank, in each case having capital, surplus and undivided profits of at least \$100,000,000 and whose long-term unsecured debt has a rating of "A" or better by S&P or "A2" or better by Moody's or the equivalent rating by any other nationally recognized rating agency (provided that the aggregate face amount of all Investments in certificates of deposit or bankers' acceptances issued by the principal offices of or branches of European or Japanese banks located outside the United States shall not at any time exceed 33-1/3% of all Investments described in this definition), (iii) open market commercial paper, maturing within 270 days after issuance thereof, which has a rating of "A1" or better by S&P or "P1" or better by Moody's, or the equivalent rating by any other nationally recognized rating agency, (iv) repurchase agreements and reverse repurchase agreements with a term not in excess of one year with any financial institution which has been elected a primary government securities dealer by the Board or whose securities are rated "AA-" or better by S&P or "Aa3" or better by Moody's or the equivalent rating by any other nationally recognized rating agency relating to marketable direct obligations issued or unconditionally guaranteed by the United States of America or any agency or instrumentality thereof and backed by the full faith and credit of the United States of America, (v) "Money Market" preferred stock maturing within six months after issuance thereof or municipal bonds issued by a corporation organized under the laws of any state of the United States, which has a rating of "A" or better by S&P or Moody's or the equivalent rating by any other nationally recognized rating agency, (vi) tax exempt floating rate option tender bonds backed by letters of credit issued by a national or state bank whose

8

long-term unsecured debt has a rating of "AA" or better by S&P or "Aa2" or better by Moody's or the equivalent rating by any other nationally recognized rating agency, and (vii) shares of any money market mutual fund rated at least AAA or the equivalent thereof by S&P or at least Aaa or the equivalent thereof by Moody's or any other mutual fund holding assets consisting (except for *de minimis* amounts) of securitized products, such as asset backed securities and of the type specified in clauses of (i) through (vi) above.

"Change of Control" means the occurrence of one or more of the following events: (x) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act), other than Mr. Jon M. Huntsman, his spouse, direct descendants, an entity controlled by any of the foregoing and/or by a trust of the type described hereafter, and/or a trust for the benefit of any of the foregoing (the "Huntsman Group") or MatlinPatterson Global Opportunities Partners L.P., or any Affiliate thereof (the "MP Group"), is or becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act, except that a person shall be deemed to have "beneficial ownership" of all securities that such Person has the right to acquire, whether such right is exercisable immediately or only after the passage of time), directly or indirectly, of 40% or more of the then outstanding Voting Securities of Huntsman Corporation; (y) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act), other than the Huntsman Group, the MP Group or Huntsman Corporation (or any successor entity) is or becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act, except that a person shall be deemed to have "beneficial ownership" of all securities that such Person has the right to acquire, whether such right is exercisable immediately or only after the passage of time), directly or indirectly, of 40% or more of the then outstanding Voting Securities of any Parent Company or the Borrower; or (z) the replacement of a majority of the Board of Directors of Huntsman Corporation, the Board of Directors or other group with similar powers of any Parent Company or the Board of Managers of the Borrower over a two-year period from the managers or directors who constituted such group, at the beginning of such period, and such replacement shall not (A) have been approved by a vote of at least a majority of the Board of Directors of Huntsman Corporation, the Board of Directors or other group with similar powers of any Parent Company or the Board of Managers of the Borrower, as the case may be, then still in office who either were members of such group at the beginning of such period or whose election as a member of such group was previously so approved or (B) have been elected or nominated for election by one or more members of the Huntsman Group. Change of Control shall also mean any "Change of Control" as defined in any Public Note Documents that occurs at a time when the market price (as determined by the Administrative Agent) of the Public Notes to which the "Change of Control" applies is less than 103% of the principal amount thereof, or if as a result thereof the holders of such Public Notes collectively put or tender in excess of \$50,000,000 in the aggregate of such Public Notes.

"Closing Date" means August 16, 2005.

"Code" means the Internal Revenue Code of 1986, as from time to time amended, including the regulations proposed or promulgated thereunder, or any successor statute and the regulations proposed or promulgated thereunder.

Ç

"Collateral" means all "Collateral" as defined in each of the Security Documents and all other assets of each Credit Party pledged pursuant to any Security Document and any other collateral pledged by any Credit Party to secure the Obligations.

"Collateral Agent" means JPMCB in its capacity as Collateral Agent under the Collateral Security Agreement, the Pledge Agreement or any other applicable Security Document, or any successor Collateral Agent.

"Collateral Security Agreement" has the meaning assigned to that term in Section 5.1(b).

"Commercial Letter of Credit" means any letter of credit or similar instrument issued pursuant to this Agreement for the purpose of supporting trade obligations of the Borrower or any of its Subsidiaries in the ordinary course of business.

"Commitment" means, with respect to each Lender, the aggregate of the Revolving Commitment and Term Commitment, of such Lender and "Commitments" means such commitments of all of the Lenders collectively.

"Commitment Fee" has the meaning assigned to that term in Section 3.2(a).

"Commitment Period" means, the period from and including the Fifth Amendment Effective Date to but not including the Revolver Termination Date or, in the case of the Swing Line Commitment, five (5) Business Days prior to the Revolver Termination Date.

"Compliance Certificate" has the meaning assigned to that term in Section 7.2(b).

"Consenting Revolving Lender" has the meaning assigned to that term in Section 2.11(b).

"Consolidated Cash Interest Expense" means, for any period, (i) Consolidated Interest Expense, but excluding, however, interest expense not payable in cash, amortization of discount and deferred financing costs, plus or minus, as the case may be (ii) net amounts paid or received under Interest Rate Agreements (with cap payments amortized over the life of the cap) and minus interest income received in Cash or Cash Equivalents in respect of Investments permitted hereunder. Subject to adjustment as described in Section 1.2(c) for events occurring after the Effective Date, for purposes of computing Consolidated Cash Interest Expense, the Consolidated Cash Interest Expense for the second, third and fourth Fiscal Quarters of 2006 shall be \$74.8 million, \$74.9 million and \$74.8 million, respectively.

"Consolidated Debt" means, at any time, without duplication, the sum of (i) all Indebtedness of the Borrower and its Subsidiaries determined on a consolidated basis that would be required to be shown as debt on a balance sheet prepared in accordance with GAAP and other Indebtedness under Operating Financing Leases incurred pursuant to Section 8.2(b)(iv), less Cash, Cash Equivalents and Foreign Cash Equivalents of the Borrower and its Subsidiaries not subject to any Lien (other than a Lien in favor of the Administrative Agent and/or the Collateral Agent) or transfer restriction and (ii) Indebtedness of Borrower and its Subsidiaries of the type referred to in clause (x) of the definition of such term.

"Consolidated EBITDA" means, with respect to any Person, for any applicable period, the sum (without duplication) of (i) Consolidated Net Income minus, to the extent Consolidated Net Income has been increased thereby, any Port Arthur Fire Insurance Income, and (ii) to the extent Consolidated Net Income has been reduced thereby, (A) all income taxes of such Person and its Subsidiaries paid or accrued in accordance with GAAP for such period (other than income taxes attributable to extraordinary, unusual or nonrecurring gains or losses or taxes attributable to sales or dispositions outside the ordinary course of business) and Tax Distributions paid during such period, (B) Consolidated Interest Expense, (C) Permitted Non-Cash Impairment and Restructuring Charges less any non-cash items increasing Consolidated Net Income for such period, (D) the amount of net loss resulting from the payment of any premiums or similar amounts that are required to be paid under the express terms of the instrument(s) governing any Indebtedness of the Borrower upon the repayment or other extinguishment of such Indebtedness by the Borrower in accordance with the express terms of such Indebtedness and (E) the Port Arthur Fire Add Back, all as determined on a consolidated basis for such Person and its Subsidiaries in accordance with GAAP. For purposes of computing Consolidated EBITDA, (i) all components of Consolidated EBITDA for any such applicable period shall be computed without giving effect to any extraordinary gains or losses (in accordance with GAAP) for such period and (ii) subject to adjustment as described in Section 1.2(c) for events occurring after the Effective Date, the Consolidated EBITDA for the second, third and fourth Fiscal Quarters of 2006 shall be \$395.9 million, \$282.5 million and \$231.7 million, respectively.

"Consolidated Interest Expense" means, for any period, the total interest expense (including that attributable to Capitalized Leases in accordance with GAAP) of the Borrower and its Subsidiaries on a consolidated basis with respect to all outstanding Indebtedness of the Borrower and its Subsidiaries, including, without limitation, all commissions, discounts and other fees and charges owed with respect to letters of credit and bankers' acceptance financing, all as determined on a consolidated basis for the Borrower and its Subsidiaries in accordance with GAAP, as modified by the last sentence of this definition. Notwithstanding anything in this definition to the contrary, as used in this definition, the term "interest" shall include, without limitation, any discount in respect of sales of Receivables Facility Assets pursuant to a Permitted Accounts Receivable Securitization (regardless of whether such discount would constitute interest expense as determined in accordance with GAAP) and any net payments made or received by the Borrower and its Subsidiaries with respect to Other Hedging Agreements entered into by the Borrower or any of its Subsidiaries to protect against fluctuations in currency values in connection with the Permitted Accounts Receivable Securitization, and the term "discount" shall include any amounts which would be interest under GAAP if the Permitted Accounts Receivable Securitization were a debt financing.

"Consolidated Net Income" and "Consolidated Net Loss" mean, respectively, with respect to any Person, for any period, the sum of: (x) the aggregate net income (or loss) of such Person and its Subsidiaries for such period on a consolidated basis, determined in accordance with GAAP plus (y) cash dividends or distributions paid to such Person or a Subsidiary of such Person by any other Person (the "Payor") other than a Subsidiary of the referent Person, to the extent not otherwise included in Consolidated Net Income, which have been derived from operating cash flow of the Payor; provided that there shall be excluded therefrom (a) after-tax gains and losses from Asset Dispositions or abandonments or reserves relating thereto, (b) after-

11

tax items classified as extraordinary or nonrecurring gains, (c) the net income of any Person acquired in a "pooling of interests" transaction accrued prior to the date it becomes a Subsidiary of the Person or is or is merged or consolidated with the Person or any Subsidiary of the Person, (d) the net income (but not loss) of any Subsidiary of the Person to the extent that the declaration of Dividends or similar distributions by that Subsidiary of that income is restricted; provided however, that the net income of Foreign Subsidiaries shall only be excluded in any calculation of Consolidated Net Income of the Borrower as a result of application of this clause (d) if the restriction on Dividends or similar distributions results from consensual restrictions, (e) the net income or loss of any Person, other than a Subsidiary or the Person, except to the extent of cash Dividends or distributions paid to the Person or to a Subsidiary that is a Wholly-Owned Subsidiary of the Person by such Person, (f) income or loss attributable to discontinued operations (including, without limitation, operations disposed of during such period whether or not such operations were classified as discontinued), (g) in the case of a successor to the referent Person by consolidation or merger or as a transferee of the referent Person's assets, any earnings of the successor corporation prior to such consolidation, merger or transfer of assets, (h) non-cash charges relating to asset impairments, which charges do not require an accrual of or a reserve for cash charges for any future period, (i) all gains or losses from the cumulative effect of any change in accounting principals and (j) the net amount of all Tax Distributions made during such period.

"Consolidated Net Tangible Assets" means, for any Person, the total assets of such Person and its Subsidiaries, as determined from a consolidated balance sheet of such Person and its consolidated Subsidiaries prepared in accordance with GAAP, but excluding therefrom all items that are treated as goodwill and other intangible assets under GAAP.

"Contaminant" means any material with respect to which any Environmental Law imposes a duty, obligation or standard of conduct, including without limitation any pollutant contaminant (as those terms are defined in 42 U.S.C. §9601(33)), toxic pollutant (as that term is defined in 33 U.S.C. §1362(13)), hazardous substance (as that term is defined in 42 U.S.C. §9601(14)), hazardous chemical (as that term is defined by 29 CFR §1910.1200(c)), hazardous waste (as that term is defined in 42 U.S.C. §6903(5)), or any state or local equivalent of such laws and regulations, including, without limitation, radioactive material, special waste, polychlorinated biphenyls, asbestos, petroleum, including crude oil or any petroleum-derived substance, (or any fraction thereof), solid waste (as that term is defined in 42 U.S.C. § 6903(27)), or breakdown or decomposition product thereof, or any constituent of any such substance or waste, including but not limited to polychlorinated biphenyls and asbestos.

"Contractual Obligation" means, as to any Person, any provision of any Securities issued by such Person or of any indenture or credit agreement or any agreement, instrument or other undertaking to which such Person is a party or by which it or any of its property is bound or to which such property may be subject.

"Credit Event" means the making of any Loan or the issuance of any Letter of Credit.

"Credit Exposure" has the meaning assigned to that term in Section 12.8(b).

12

"Credit Party" means the Borrower, each Subsidiary Guarantor and any guarantor which may hereafter enter into a Guaranty with respect to the Obligations.

"Customary Permitted Liens" means:

- (i) Liens for taxes, assessments, governmental charges or levies not yet due and payable or which are being contested in good faith by appropriate proceedings diligently pursued, provided that (x) any proceedings commenced for the enforcement of such Liens shall have been stayed or suspended within 30 days of the commencement thereof and (y) provision for the payment of all such taxes, assessments, governmental charges or levies known to such Person has been made on the books of such Person to the extent required by GAAP;
- (ii) mechanics', processor's, materialmen's, carriers', warehouse-men's, landlord's and similar Liens arising by operation of law and arising in the ordinary course of business and securing obligations of such Person that are not overdue for a period of more than 30 days or are being contested in good faith by appropriate proceedings diligently pursued, provided that (x) any proceedings commenced for the enforcement of such Liens shall have been stayed or suspended within 30 days of the commencement thereof and (y) provision for the payment of such Liens has been made on the books of such Person to the extent required by GAAP;
- (iii) Liens arising in connection with worker's compensation, unemployment insurance, old age pensions and social security benefits which are not overdue or are being contested in good faith by appropriate proceedings diligently pursued, provided that (A) any proceedings commenced for the enforcement of such Liens shall have been stayed or suspended within 30 days of the commencement thereof and (B) provision for the payment of such Liens has been made on the books of such Person to the extent required by GAAP;
- (iv) (x) Liens incurred or deposits made in the ordinary course of business to secure the performance of bids, tenders, statutory obligations, fee and expense arrangements with trustees and fiscal agents (exclusive of obligations incurred in connection with the borrowing of money or the payment of the deferred purchase price of property) and customary deposits granted in the ordinary course of business under Operating Financing Leases and (y) Liens securing surety, indemnity, performance, appeal and release bonds, provided that full provision for the payment of all such obligations has been made on the books of such Person to the extent required by GAAP;
 - (v) Permitted Real Property Encumbrances;
- (vi) attachment, judgment or other similar Liens arising in connection with court or arbitration proceedings involving individually and in the aggregate liability of \$50,000,000 or less at any one time, provided the same are discharged, or that execution or enforcement thereof is stayed pending appeal, within 60 days or, in the case of any stay of execution or enforcement pending appeal, within such lesser time during which such appeal may be taken;

13

- (vii) leases or subleases granted to others not interfering in any material respect with the business of the Borrower or any of its Subsidiaries and any interest or title of a lessor under any lease permitted by this Agreement or the Security Documents;
- (viii) customary rights of set off, revocation, refund or chargeback under deposit agreements or under the UCC of banks or other financial institutions where the Borrower or any of its Subsidiaries maintains deposits in the ordinary course of business permitted by this Agreement; and
- Environmental Liens, to the extent that (w) any proceedings commenced for the enforcement of such Liens shall have been suspended or are being contested in good faith, (x) provision for all liability and damages that are the subject of said Environmental Liens has been made on the books of such Person to the extent required by GAAP and (y) such Liens do not relate to obligations exceeding \$10,000,000 in the aggregate at any one time; and (z) Liens in favor of customs and revenue authorities arising as a matter of law to secure payment of custom duties in connection with the importation of goods so long as such Liens attach only to the imported goods and provided that such duties are not yet due or are being contested in good faith and by appropriate proceedings diligently conducted, if adequate reserves with respect thereto are maintained on the books of the applicable Person in accordance with GAAP.

"DB" means Deutsche Bank AG New York Branch and its successors.

"DBTCA" has the meaning assigned to that term in the Recitals to this Agreement.

"Declining Revolving Lender" has the meaning assigned to that term in Section 2.11(b).

"Default Rate" means a variable rate per annum which shall be two percent (2%) per annum plus either (i) the then

applicable interest rate hereunder in respect of the amount on which the Default Rate is being assessed or (ii) if there is no such applicable interest rate, the Base Rate plus the Applicable Base Rate Margin, but in no event in excess of that permitted by applicable law.

"Defaulting Lender" means any Lender with respect to which a Lender Default is in effect.

"<u>Deposit Account</u>" means a demand, time, savings, passbook or like account with a bank, savings and loan association, credit union or like organization, other than an account evidenced by a negotiable certificate of deposit.

"Determination Date" has the meaning assigned to that term in Section 12.23.

"Dividends" has the meaning assigned to that term in Section 8.4(a).

"Documents" means the Loan Documents and the Transaction Documents.

14

"Dollar" and "S" means the lawful currency of the United States of America.

"<u>Dollar Equivalent</u>" means, at any time, (i) as to any amount denominated in Dollars, the amount thereof at such time, and (ii) as to any amount denominated in any Alternative Currency, the equivalent amount in Dollars as determined by the Administrative Agent at such time on the basis of the Spot Rate.

"<u>Domestic Subsidiary</u>" means any Subsidiary other than a Foreign Subsidiary not a party to the Subsidiary Guaranty or a guaranty delivered pursuant to <u>Section 7.11(h)</u>.

"Effective Date" means the "Third Amendment Effective Date" as defined in the Third Amendment.

"Eligible Assignee" means a commercial bank, investment company, financial institution, financial company, Fund (whether a corporation, partnership, trust or other entity) or insurance company in each case, together with its Affiliates or Related Funds, which makes, purchases, holds or otherwise invests in commercial loans or other similar extensions of credit in the ordinary course of its business or any other Person approved by the Administrative Agent and the Borrower, such approval not to be unreasonably withheld or delayed.

"Environmental Claim" means any notice of violation, claim (including common law claims), suit, written demand, abatement order, or other order or directive (conditional or otherwise), by any Governmental Authority or any Person for any damage, personal injury (including sickness, disease or death), tangible or intangible property damage, contribution, cost recovery, indemnity, indirect or consequential damages, damage to the environment, or for nuisance, pollution, contamination or other adverse effects on the environment, human health, or natural resources, or for fines, penalties, restrictions or injunctive relief, resulting from or based upon (i) the occurrence or existence of a Release or substantial threat of a material Release (whether sudden or non-sudden or accidental or non-accidental) of, or exposure to, any Contaminant in, into or onto the environment at, in, by, from or related to any real estate owned, leased or operated at any time by the Borrower or any of its Subsidiaries (the "Premises"), (ii) the use, handling, generation, transportation, storage, treatment or disposal of Contaminants in connection with the operation of any Premises, or (iii) the violation, or alleged violation, of any Environmental Laws relating to environmental matters connected with the Borrower's operations or any Premises.

"Environmental Laws" means any and all applicable foreign, federal, state or local laws, statutes, ordinances, codes, rules or regulations or orders, decrees, judgments or directives issued by a Governmental Authority, or Environmental Permits or Remedial Action standards, levels or objectives imposing liability or standards of conduct for or relating to the protection of health, safety or the environment, including, but not limited to, the following United States statutes, as now written and hereafter amended: the Water Pollution Control Act, as codified in 33 U.S.C. §1251 et seq., the Clean Air Act, as codified in 42 U.S.C. §7401 et seq., the Toxic Substances Control Act, as codified in 15 U.S.C. §2601 et seq., the Solid Waste Disposal Act, as codified in 42 U.S.C. §6901 et seq., the Comprehensive Environmental Response, Compensation and Liability Act, as codified in 42 U.S.C. §9601 et seq., the Emergency Planning and Community Right-to-Know Act of 1986, as codified in 42 U.S.C. §11001 et seq., and the Safe

15

Drinking Water Act, as codified in 42 U.S.C. §300fet seq., and any related regulations, as well as all state and local equivalents.

"Environmental Lien" means a Lien in favor of any Governmental Authority for (i) any liability under Environmental Laws or Environmental Permits, or (ii) damages relating to, or costs incurred by such Governmental Authority in response to, a Release or threatened Release of a Contaminant into the environment.

"Environmental Permits" means any and all permits, licenses, certificates, authorizations or approvals of any Governmental Authority required by Environmental Laws or necessary or reasonably required for the current and anticipated future operation of the business of the Borrower or any Subsidiary of the Borrower.

"ERISA" means the Employee Retirement Income Security Act of 1974, as from time to time amended.

"ERISA Affiliate" means, with respect to any Person, any trade or business (whether or not incorporated) which,

together with such Person, is under common control as described in Section 414(c) of the Code, is a member of a "controlled group", as defined in Section 414(b) of the Code, or is a member of an "affiliated service group", as defined in Section 414(m) of the Code which includes such Person. Unless otherwise qualified, all references to an "ERISA Affiliate" in this Agreement shall refer to an ERISA Affiliate of the Borrower or any Subsidiary.

"<u>Euro</u>" means the lawful currency adopted by or which is adopted by participating member states of the European Community relating to Economic and Monetary Union.

"Eurocurrency Loan" means any Loan bearing interest at a rate determined by reference to the Eurocurrency Rate.

"Eurocurrency Rate" means

- (a) in the case of Dollar denominated loans, (A) the rate per annum equal to the rate determined by Administrative Agent to be the offered rate that appears on the Reuters Screen LIBOR01 Page for deposits in Dollars (for delivery on the first day of such interest period) with a term equivalent to such interest period, determined as of approximately 11:00 a.m. (London time) on the applicable Interest Rate Determination Date and, in the event such rate is not available, (B) the arithmetic average (rounded up to the nearest 1/100th of 1%) of the offered quotation in the interbank eurodollar market by the Reference Lenders to first class banks for Dollar deposits of amounts in immediately available funds with a term comparable to the interest period for which a Eurocurrency Rate is determined, as of 11:00 a.m. (London time) on the applicable Interest Rate Determination Date; or
- (b) in the case of Euro denominated loans, (A) the rate per annum equal to the rate determined by Administrative Agent to be the offered rate that appears on the appropriate page of the Telerate screen that displays EURIBOR

16

(for delivery on the first day of such interest period) with a term equivalent to such interest period, determined as of approximately 11:00 a.m. (London time) on the applicable Interest Rate Determination Date and, in the event such rate is not available, (B) the arithmetic average (rounded up to the nearest 1/100th of 1%) of the offered quotation in the European interbank market by the Reference Lenders for Euro deposits of amounts in immediately available funds with a term comparable to the interest period for which a Eurocurrency Rate is determined, as of 11:00 a.m. (London time) on the applicable Interest Rate Determination Date; or

(c) in the case of Sterling denominated loans, (A) the rate per annum equal to the rate determined by Administrative Agent to be the offered rate that appears on the appropriate page of the Telerate screen that displays LIBOR (for delivery on the first day of such interest period) with a term equivalent to such interest period, determined as of approximately 11:00 a.m. (London time) on the applicable Interest Rate Determination Date and, in the event such rate is not available, (B) the arithmetic average (rounded up to the nearest 1/100th of 1%) of the offered quotation in the London interbank market by the Reference Lenders for Sterling deposits of amounts in immediately available funds with a term comparable to the interest period for which a Eurocurrency Rate is determined, as of 11:00 a.m. (London time) on the applicable Interest Rate Determination Date.

In the case of Swing Line Loans maintained at the Quoted Rate and Eurocurrency Loans, the cost of the Lenders of complying with any Eurocurrency Reserve Requirements will be added to the interest rate computed in the manner set forth in <u>Schedule 1.1(b)</u>.

"Eurocurrency Reserve Requirements" means, for any day as applied to a Eurocurrency Loan, the aggregate (without duplication) of the maximum rates (expressed as a decimal fraction) of reserve liquid asset or similar requirements in effect on such day (including, without limitation, basic, supplemental, marginal and emergency reserves under any regulations of the Board or other Governmental Authority having jurisdiction with respect thereto), including without limitation, under regulations issued from time to time by (a) the Board, (b) any Governmental Authority of the jurisdiction of the relevant currency or (c) any Governmental Authority of any jurisdiction in which advances in such currency are made to which banks in any jurisdiction are subject for any category of deposits or liabilities customarily used to fund loans in such currency or by reference to which interest rates applicable to loans in such currency are determined, including Mandatory Costs.

"Event of Default" has the meaning assigned to that term in Section 10.1.

"Excess Cash Flow" means, an amount not less than zero calculated as of the close of business on March 31 of each year (commencing March 31, 2008), equal to (i) the sum of (a) the average daily aggregate Total Available Revolving Commitment during the period of February 1 through and including March 31 of such year <u>plus</u> (b) the amount, if any, by which the average available capacity under Receivables Documents in existence during such period and subject to any limitations on availability contained therein during the period of February 1

Equivalents, held during the period February 1 through and including March 31 of such year, <u>less</u> (ii) the sum of (w) the aggregate amount of Net Sale Proceeds from Asset Dispositions during the preceding twelve months, to the extent not reinvested prior to March 31 of such year, <u>plus</u> (x) the aggregate amount of proceeds from the issuance of Indebtedness issued to refinance other Indebtedness and being held pending the payment of such other Indebtedness <u>plus</u> (y) the aggregate amount during the preceding twelve months of cash proceeds from Recovery Events received by the Borrower or any of its Subsidiaries during the preceding twelve months, to the extent not reinvested prior to March 31 of such year, <u>plus</u> (z) \$850,000,000.

"Exchange Act" means the Securities Exchange Act of 1934, as amended and as codified in 15 U.S.C. §78aet seq. and as hereafter amended.

"Exchange Rate" shall mean, on any day, (i) with respect to any Alternative Currency, the Spot Rate at which Dollars are offered on such day by the Administrative Agent in London or New York (as selected by the Administrative Agent) for such Alternative Currency at approximately 11:00 A.M. (London time or New York time, as applicable), and (ii) with respect to Dollars in relation to any specified Alternative Currency, the Spot Rate at which such specified Alternative Currency is offered on such day by the Administrative Agent in London or New York for Dollars at approximately 11:00 A.M. (London time or New York time, as applicable). The Administrative Agent shall provide the Borrower with the then current Exchange Rate from time to time upon the Borrower's request therefor.

"Exchange Rate Determination Date" means (i) for purposes of the determination of the Exchange Rate of any stated amount on any Business Day in relation to any Borrowing of Revolving Loans or Swing Line Loans in an Alternative Currency, (A) the date which is two Business Days prior to such Borrowing in the case of a Borrowing denominated in Euros or (B) the date of such Borrowing in the case of a Borrowing denominated in Sterling, (ii) for purposes of the determination of the Exchange Rate of any Stated Amount in relation to any issuance of any Letter of Credit, on the date of such issuance and (iii) for the purpose of determining the Exchange Rate to make determinations pursuant to Section 4.4(a), the last Business Day of each calendar month.

"Existing Stated Termination Date" has the meaning assigned to that term in Section 2.11(a).

"Extension Effective Date" has the meaning assigned to that term in Section 2.11(b).

"<u>Facility</u>" means any of the credit facilities established under this Agreement, <u>i.e.</u>, any of the Term Facilities or the Revolving Facility.

"Facing Agent" means each of (i) DB as to each Fifth Amendment Existing Letter of Credit, (ii) JPMCB and (iii) any other Revolving Lender agreed to by such Revolving Lender,

18

the Borrower and the Administrative Agent (in each case, acting through any branch or Affiliate).

"Facing Agent Sublimit" means \$75,000,000.

"Federal Funds Rate" means on any one day, the rate per annum equal to the weighted average (rounded upwards, if necessary, to the nearest 1/100th of 1%) of the rate on overnight federal funds transactions with members of the Federal Reserve System only arranged by federal funds brokers, as published as of such day by the Federal Reserve Bank of New York, or, if such rate is not so published, the average of the quotations for such day on such transactions received by JPMCB from three federal funds brokers of recognized standing selected by JPMCB.

"<u>Fee Letter</u>" means the letter agreement with respect to fees related to this Agreement between the Borrower, DB and Deutsche Bank Securities Inc. dated on or before the Effective Date.

"Fifth Amendment" means the Fifth Amendment to Credit Agreement dated as of March 9, 2010.

"Fifth Amendment Effective Date" means the "Amendment Effective Date" as defined in the Fifth Amendment.

"Fifth Amendment Existing Letter of Credit" means any letter of credit issued by DB pursuant to this Agreement prior to the Fifth Amendment Effective Date and set forth on Schedule 1.1(d), to the extent such letter of credit is not increased (other than in accordance with the terms thereof as in effect on the Fifth Amendment Effective Date), extended, auto-extended or renewed, on or after the Fifth Amendment Effective Date.

"<u>Fifth Amendment Fee Letter</u>" means the letter agreement with respect to fees related to this Agreement among the Borrower, J.P. Morgan Securities Inc. and JPMCB dated on or before the Fifth Amendment Effective Date.

"Fiscal Quarter" has the meaning assigned to that term in Section 7.12.

"Fiscal Year" has the meaning assigned to that term in Section 7.12.

"Flood Insurance Laws" means, collectively, (i) the National Flood Insurance Act of 1968 as now or hereafter in effect or any successor statute thereto, (ii) the Flood Disaster Protection Act of 1973 as now or hereafter in effect or any successor statute thereto, (iii) the National Flood Insurance Reform Act of 1994 as now or hereafter in effect or any successor statute thereto and (iv) the Flood Insurance Reform Act of 2004 as now or hereafter in effect or any successor statute thereto.

long-term unsecured debt rating is provided by either)(each such jurisdiction, an "approved jurisdiction"), or any agency or instrumentality of an approved jurisdiction, provided that the full faith and credit of the approved jurisdiction is pledged in support of such debt securities or such debt securities constitute a general obligation of the approved jurisdiction and (ii) debt securities in an aggregate principal amount not to exceed the Dollar Equivalent of \$40,000,000 with a maturity of 365 days or less issued by any nation in which the Borrower or its Subsidiaries has cash which is the subject of restrictions on export, any agency or instrumentality of such nation or any bank or other organization organized in such nation.

"<u>Foreign Factoring Transactions</u>" means transactions (other than pursuant to (a) any Permitted Accounts Receivable Securitization or (b) a transaction described in <u>Section 8.3(e)</u>) for the sale or discounting of (i) the Accounts Receivable of a Foreign Subsidiary and/or (ii) letters of credit the beneficiary of which is a Foreign Subsidiary.

"Foreign Pension Plan" means any plan, fund (including, without limitation, any super-annuation fund) or other similar program established or maintained outside of the United States of America by the Borrower or one or more of its Subsidiaries or its Affiliates primarily for the benefit of employees of the Borrower or such Subsidiaries or its Affiliates residing outside the United States of America, which plan, fund, or similar program provides or results in, retirement income, a deferral of income in contemplation of retirement or payments to be made upon termination of employment, and which is not subject to ERISA or the Code.

"<u>Foreign Subsidiary</u>" means any Subsidiary that is organized under the laws of a jurisdiction other than the United States of America or any state thereof or the District of Columbia and that is not a Subsidiary Guarantor.

"Fourth Amendment" means the Fourth Amendment to this Agreement dated as of June 22, 2009.

"Fourth Amendment Effective Date" has the meaning assigned to that term in the Fourth Amendment.

"<u>Fund</u>" means a Person that is a fund that makes, purchases, holds or otherwise invests in commercial loans or similar extensions of credit in the ordinary course of its business.

"GAAP" means generally accepted accounting principles in the United States of America as in effect from time to time.

"Governmental Authority" means any nation or government, any state or other political subdivision thereof and any entity exercising executive, legislative, judicial, taxing, regulatory or administrative functions of government.

"Guarantee Obligations" means, as to any Person, without duplication, any direct or indirect obligation of such Person guaranteeing or intended to guarantee any Indebtedness, Capitalized Lease, Operating Financing Lease ("primary obligations") of any other Person (the "primary obligor") in any manner, whether directly or indirectly, including, without limitation, any obligation of such Person, whether or not contingent: (i) to purchase any such primary obligation or any property constituting direct or indirect security therefor; (ii) to advance or

20

supply funds (x) for the purchase or payment of any such primary obligation, or (y) to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor; (iii) to purchase property, securities or services primarily for the purpose of assuring the owner of any such primary obligation of the ability of the primary obligor to make payment of such primary obligation; or (iv) otherwise to assure or hold harmless the owner of such primary obligation against loss in respect thereof; provided, however, that the term Guarantee Obligations shall not include any endorsements of instruments for deposit or collection in the ordinary course of business. The amount of any Guarantee Obligation at any time shall be deemed to be an amount equal to the lesser of (x) the stated or determinable amount at such time of the primary obligation in respect of which such Guarantee Obligation is made or (y) the maximum amount for which such Person may be liable pursuant to the terms of the instrument embodying such Guarantee Obligation; or, if not stated or determinable, the maximum liability (assuming full performance) in respect thereof reasonably anticipated at such time.

"Guaranteed Obligations" means (i) the full and prompt payment when due (whether at the stated maturity, by acceleration or otherwise) of the principal and interest (whether such interest is allowed as a claim in a bankruptcy proceeding with respect to the Borrower or otherwise) on each Note issued by the Borrower to each Lender, and Loans made under this Agreement and all reimbursement obligations and Unpaid Drawings with respect to Letters of Credit, together with all other obligations (including obligations which, but for the automatic stay under Section 362(a) of the Bankruptcy Code, would become due) and liabilities (including, without limitation, indemnities, fees and interest thereon) of the Borrower to such Lender now existing or hereafter incurred under, arising out of or in connection with this Agreement or any other Loan Documents and the due performance and compliance with all terms, conditions and agreements contained in the Loan Documents by the Borrower and (ii) the full and prompt payment when due (whether by acceleration or otherwise) of all obligations (including obligations which, but for the automatic stay under Section 362(a) of the Bankruptcy Code, would become due) of the Borrower (or, if permitted by Section 8.2, its Subsidiaries) owing under any Interest Rate

Agreement or Other Hedging Agreement or any Overdraft Facility entered into by the Borrower or any of its Subsidiaries with any Lender or any Affiliate thereof (even if such Lender subsequently ceases to be a Lender under this Agreement for any reason) so long as such Lender or Affiliate participates in such Interest Rate Agreement or Other Hedging Agreement or Overdraft Facility, as the case may be, and their subsequent assigns, if any, whether or not in existence or hereafter arising, and the due performance and compliance with all terms, conditions and agreements contained therein.

"Guaranty" means, collectively, (i) the Subsidiary Guaranty and (ii) each guaranty delivered by a Foreign Subsidiary pursuant to Section 7.11(h), in each case as the same may be amended, supplemented or otherwise modified from time to time.

"<u>Headquarters Subsidiary Guaranty Agreement</u>" has the meaning assigned to that term in <u>Section 5.1(d)(ii)</u> of this Agreement.

"<u>Huntsman Affiliate</u>" means Huntsman Corporation or any of its Affiliates (other than the Borrower and its Subsidiaries).

21

"Huntsman Corporation" means Huntsman Corporation, a Delaware corporation.

"<u>Huntsman Finco</u>" means Huntsman International Financial LLC, a direct Wholly-Owned Subsidiary of the Borrower that is a limited liability company formed under the laws of Delaware.

"Huntsman Group" has the meaning assigned to that term in the definition of "Change of Control" in this Section 1.1.

"<u>Huntsman Parent Company</u>" means Huntsman Corporation or any entity of which the Borrower is a direct or indirect Wholly-Owned Subsidiary.

"Immaterial Subsidiary" means any Subsidiary of the Borrower, the Consolidated Net Tangible Assets of which are less than 1% of the Borrower's Consolidated Net Tangible Assets (as of the end of the most recently completed Fiscal Quarter of the Borrower for which financial statements are available) and which did not account for more than 1% of the consolidated revenues of the Borrower and its Subsidiaries for such period.

"Impaired Lender" means, at any time, a Revolving Lender as determined by the Administrative Agent, that (i) is a Defaulting Lender, (ii)(x) has notified the Administrative Agent, any Facing Agent, the Swing Line Lender or the Borrower, or has stated publicly, that it will not comply with its obligations under this Agreement to make a Loan, make a payment to any Facing Agent in respect of a Letter of Credit Payment and/or make a payment to the Swing Line Lender in respect of a Swing Line Loan (each a "funding obligation") or (y) has defaulted on its funding obligations under any other loan agreement or credit agreement or other similar agreement, unless such default in respect of funding obligations is the subject of a good faith dispute, (iii) has, for three or more Business Days, failed to confirm in writing to the Administrative Agent, in response to a written request of the Administrative Agent, that it will comply with its funding obligations hereunder, or (iv) as to which a Lender Insolvency Event has occurred and is continuing (provided that the reallocation of funding obligations provided for in Section 3.8(a) as a result of a Revolving Lender being an Impaired Lender will not by itself cause the relevant Impaired Lender to become a Non-Impaired Lender). Any determination that a Revolving Lender is an Impaired Lender under clauses (i) through (iv) above will be made by the Administrative Agent in its reasonable discretion acting in good faith. Notwithstanding anything to the contrary above, a Revolving Lender will not be an Impaired Lender solely by virtue of the ownership or acquisition of any Capital Stock in such Revolving Lender or Lender Parent Company by a Governmental Authority.

"Indebtedness" means, as applied to any Person (without duplication):

- (i) all obligations of such Person for borrowed money;
- (ii) the deferred and unpaid balance of the purchase price of assets or services (other than trade payables and other accrued liabilities incurred in the ordinary course of business that are not overdue by more than 90 days unless being contested in good faith) which purchase price is (x) due more than six months from the date of incurrence of the obligation in respect thereof (or in the case of long-term supply

22

agreements, from the date of delivery of such assets or services) or (y) evidenced by a note or a similar written instrument;

- (iii) all Capitalized Lease Obligations;
- (iv) all indebtedness secured by any Lien (other than Customary Permitted Liens) on any property owned by such Person, whether or not such indebtedness has been assumed by such Person or is nonrecourse to such Person;
- (v) notes payable and drafts accepted representing extensions of credit whether or not representing obligations for borrowed money (other than such notes or drafts for the deferred purchase price of assets or services which does not constitute Indebtedness pursuant to clause (ii) above);
 - (vi) indebtedness or obligations of such Person, in each case, evidenced by bonds, notes or similar written

instruments;

- (vii) the face amount of all letters of credit and bankers' acceptances issued for the account of such Person, and without duplication, all drafts drawn thereunder other than, in each case, commercial or standby letters of credit or the functional equivalent thereof issued in connection with performance, bid or advance payment obligations incurred in the ordinary course of business, including, without limitation, performance requirements under workers compensation or similar laws;
 - (viii) all obligations of such Person under Interest Rate Agreements or Other Hedging Agreements;
 - (ix) Guarantee Obligations of such Person;
- (x) the aggregate outstanding amount of Receivables Facility Attributed Indebtedness or the gross proceeds from any similar transaction, regardless of whether such transaction is effected without recourse to such Person or in a manner that would not otherwise be reflected as a liability on a balance sheet of such Person in accordance with GAAP; and
- (xi) the Attributable Debt of any synthetic lease, tax retention operating lease, off-balance sheet loan or similar off-balance sheet financing product to which such Person is a party, where such transaction is considered borrowed money indebtedness for tax purposes but is classified as an operating lease in accordance with GAAP;

<u>provided</u>, <u>however</u>, notwithstanding the foregoing, "Indebtedness" shall not include deferred taxes or indebtedness of Borrower and/or its Subsidiaries incurred to finance insurance premiums, if (a) such indebtedness is unsecured (except as permitted by <u>Section 8.1(k)</u>), and (b) is in a principal amount not in excess of the casualty and other insurance premiums to be paid by Borrower and/or its Subsidiaries for a three year period beginning on the date of any incurrence of such indebtedness.

"Indemnified Party" has the meaning assigned to that term in Section 12.4(a).

23

"Initial Borrowing" means the first Borrowing by the Borrower under this Agreement.

"Initial Loan" means the first Loan made by the Lenders under this Agreement.

"Intellectual Property" has the meaning assigned to that term in Section 6.19.

"Intercompany Loan" has the meaning assigned to that term in Section 8.7(g).

"Intercreditor Agreement" means that certain Intercreditor Agreement, dated the date hereof, by and among the Collateral Agent, Administrative Agent, DB as beneficiary of the Mortgages, HSBC Bank USA, National Association (as successor to HSBC Bank USA), as trustee for the Senior Secured Notes and the Borrower, in the form of Exhibit B to the Collateral Security Agreement, as amended, modified or supplemented in accordance with the terms thereof.

"Interest Coverage Ratio" means, for any period, the ratio of Consolidated EBITDA to Consolidated Cash Interest Expense for such period.

"Interest Payment Date" means (i) as to any Base Rate Loan, each Quarterly Payment Date to occur while such Loan is outstanding, (ii) as to any Eurocurrency Loan having an Interest Period of three months or less, the last day of the Interest Period applicable thereto and (iii) as to any Eurocurrency Loan having an Interest Period longer than three months, each three (3) month anniversary of the first day of the Interest Period applicable thereto and the last day of the Interest Period applicable thereto; provided, however, that, in addition to the foregoing, each of (A) the Revolver Termination Date, (B) the Term B Loan Maturity Date and (C) the Term C Loan Maturity Date shall be deemed to be an "Interest Payment Date" with respect to any interest which is then accrued hereunder for such Loan.

"Interest Period" has the meaning assigned to that term in Section 3.4.

"Interest Rate Agreement" means any interest rate swap agreement, cross-currency interest rate swap agreement, interest rate cap agreement, interest rate futures contract, interest rate option contract or other similar agreement or arrangement to which the Borrower or any Subsidiary is a party.

"Interest Rate Determination Date" means the date for calculating the Eurocurrency Rate for an Interest Period, which date shall be (i) in the case of any Eurocurrency Loan in Dollars, the second Business Day prior to first day of the related Interest Period for such Loan or (ii) in the case of any Eurocurrency Loan in an Alternative Currency, the date on which quotations would ordinarily be given by prime banks in the London interbank market for deposits in the Applicable Currency for value on the first day of the related Interest Period for such Eurocurrency Loan; provided, however, that if for any such Interest Period with respect to an Alternative Currency Loan, quotations would ordinarily be given on more than one date, the Interest Rate Determination Date shall be the last of those dates.

"Inventory" means, inclusively, all inventory as defined in the Uniform Commercial Code in effect in the State of New York from time to time and all goods,

merchandise and other personal property wherever located, now owned or hereafter acquired by the Borrower or any of its Subsidiaries of every kind or description which are held for sale or lease or which are furnished or to be furnished under a contract of service or are raw materials, work-in-process or materials used or consumed or to be used or consumed in the Borrower's or any of its Subsidiaries' businesses.

"Investment" means, as applied to any Person, (i) any direct or indirect purchase or other acquisition by that Person of, or a beneficial interest in, Securities of any other Person, or a capital contribution by that Person to any other Person, (ii) any direct or indirect loan or advance by that person to any other Person (other than prepaid expenses or Accounts Receivable created or acquired in the ordinary course of business), including all Indebtedness of such Person arising from a sale of property by such first Person other than in the ordinary course of its business or (iii) any purchase by that Person of all or a significant part of the assets of a business conducted by another Person. The amount of any Investment by any Person on any date of determination shall be the sum of the acquisition price of the gross assets acquired by such Person (including the amount of any liability assumed in connection with the acquisition by such Person to the extent such liability would be reflected as a liability on a balance sheet prepared in accordance with GAAP) <u>plus</u> all additional capital contributions or purchase price and earnout adjustments (positive or negative) paid (or credited) in respect thereof, without any adjustments for increases or decreases in value, or write-ups, write-downs or write-offs with respect to such Investment <u>minus</u> the amount of all cash returns of principal or capital thereon, cash dividends thereon and other cash returns on investment thereon or liabilities expressly assumed by another Person (other than the Borrower or another Subsidiary of the Borrower) in connection with the sale of such Investment. Whenever the term "outstanding" is used in this Agreement with reference to an Investment, it shall take into account the matters referred to in the preceding sentence.

"IRIC" means International Risk Insurance Company, a Utah corporation.

"IRS" means the United States Internal Revenue Service, or any successor or analogous organization.

"Issuer" means the entity acting as issuer or similar funding vehicle under the relevant Receivables Documents.

"Joint Venture" means any corporation, partnership, limited liability company, joint venture or other similar legal arrangement (whether created by contract or conducted through a separate legal entity) now or hereafter formed by the Borrower or any of its Subsidiaries with another Person in order to conduct a common venture or enterprise with such Person.

"JPMCB" means JPMorgan Chase Bank, N.A. and its successors.

"LC Commission" has the meaning assigned to that term in Section 2.9(e)(ii).

"<u>LC Obligations</u>" means, at any time, an amount equal to the sum of (i) the Assigned Dollar Value of the aggregate Stated Amount of the then outstanding Letters of Credit and (ii) the Assigned Dollar Value of the aggregate amount of drawings under Letters of Credit

25

which have not then been reimbursed pursuant to <u>Section 2.9(c)</u>. The LC Obligation of any Revolving Lender at any time shall mean the Dollar Equivalent of its Pro Rata Share of the Assigned Dollar Value of the aggregate LC Obligations outstanding at such time.

"Lender" and "Lenders" have the respective meanings assigned to those terms in the introduction to this Agreement and shall include any Person that becomes a "Lender" in connection with the issuance of Additional Term Loans pursuant to Section 2.1(a)(ii).

"Lender Default" means (i) the refusal (which has not been retracted) of a Lender (x) to make available its portion of any Borrowing when the conditions precedent thereto, in the determination of the Administrative Agent, have been met, or (y) to fund its portion of any unreimbursed payment under Section 2.9(d) or (ii) a Lender having notified in writing the Borrower and/or the Administrative Agent that it does not intend to comply with its obligations under Section 2.1 or Section 2.9(d), as a result of any takeover of such Lender by any regulatory authority or agency.

"Lender Insolvency Event" means that (i) a Revolving Lender or its Lender Parent Company is insolvent, or is generally unable to pay its debts as they become due, or admits in writing its inability to pay its debts as they become due, or makes a general assignment for the benefit of its creditors, or (ii) such Revolving Lender or its Lender Parent Company is the subject of a bankruptcy, insolvency, reorganization, liquidation or similar proceeding, or a receiver, trustee, conservator, intervenor or sequestrator or the like has been appointed for such Revolving Lender or its Lender Parent Company, or such Revolving Lender or its Lender Parent Company has taken any action in furtherance of or indicating its consent to or acquiescence in any such proceeding or appointment.

"<u>Lender Parent Company</u>" means, with respect to a Revolving Lender, the bank holding company (as defined in Federal Reserve Board Regulation Y), if any, of such Revolving Lender, and/or any Person owning, beneficially or of record, directly or indirectly, a majority of the shares of such Revolving Lender.

"Lending Office" means, with respect to each Lender, the office specified under such Lender' name on the administrative questionnaire delivered to the Administrative Agent on or prior to the Effective Date, or on the signature page to any Assignment and Assumption Agreement, with respect to each Type of Loan or such other office as such Lender may designate in writing from time to time to Borrower and Administrative Agent with respect thereto.

"<u>Letter of Credit Payment</u>" means, as applicable (i) all payments made by a Facing Agent pursuant to either a draft or demand for payment under a Letter of Credit or (ii) all payments by Revolving Lenders to a Facing Agent in respect thereof (whether or not in accordance with their Pro Rata Share).

"<u>Letters of Credit</u>" means, collectively, all Fifth Amendment Existing Letters of Credit, Commercial Letters of Credit, Standby Letters of Credit and Bank Guarantees, in each case as issued pursuant to this Agreement, and "<u>Letter of Credit</u>" means any one of such Letters of Credit.

26

"Leverage Ratio" means, for any Test Period, the ratio of Consolidated Debt as of the last day of such Test Period to Consolidated EBITDA for such Test Period.

"Lien" means (i) any judgment lien or execution, attachment, levy, distraint or similar legal process and (ii) any mortgages, pledge, hypothecation, collateral assignment, security interest, encumbrance, lien, charge or deposit arrangement (other than a deposit to a Deposit Account in the ordinary course of business and not intended as security) of any kind (including, without limitation, any conditional sale or other title retention agreement or lease in the nature thereof, any agreement to give any of the foregoing, or any sale of receivables with recourse against the seller or any Affiliate of the seller).

"Loan" means any Term B Dollar Loan, Term C Dollar Loan, Swing Line Loan or Revolving Loan, and "Loans" means all such Loans, collectively.

"Loan Documents" means, collectively, this Agreement, the Notes, each Letter of Credit, each Security Document, each Guaranty and all other agreements, instruments and documents executed in connection therewith, in each case as the same may at any time be amended, supplemented, restated or otherwise modified and in effect.

"LOU Claim Proceeds" means all insurance proceeds received by the Borrower or any of its Subsidiaries after the Fifth Amendment Effective Date in respect of the Port Arthur Plant Fire.

"LPC" mean Louisiana Pigment Company, and its successors and assigns.

"Majority Lenders" of any Facility means those Non-Defaulting Lenders which would constitute the Required Lenders under, and as defined in, this Agreement if all outstanding Obligations of other Facilities under this Agreement were repaid in full and all Commitments with respect thereto were terminated.

"Mandatory Cost" means the cost imputed to the Lender(s) of compliance with the mandatory liquid assets requirements of the Bank of England and/or the banking supervision or other costs of the Financial Services Authority or European Central Bank or any successor body exercising their functions in this respect as determined in accordance with Schedule 1.1(b).

"Material Adverse Effect" means a material adverse effect on (i) the business, condition (financial or otherwise), assets, liabilities or operations of the Borrower and its Subsidiaries taken as a whole, (ii) the ability of the Borrower or any of its Subsidiaries to perform its respective obligations under any Loan Document to which it is a party, or (iii) the validity or enforceability of this Agreement or any of the Security Documents or the material rights or remedies of the Administrative Agent, Collateral Agent or and the Lenders hereunder or thereunder.

"Material Agreement" means (i) any Contractual Obligation of the Borrower or any of its Subsidiaries, the breach of which or the failure to maintain would be reasonably likely to result in a Material Adverse Effect, (ii) the Public Note Documents and (iii) any material Contractual Obligation entered into in connection with an Acquisition.

27

"Material Subsidiary" means any Subsidiary of the Borrower, the Consolidated Net Tangible Assets of which were more than 2% of the Borrower's Consolidated Net Tangible Assets as of the end of the most recently completed Fiscal Year of the Borrower for which audited financial statements are available; provided that, in the event the aggregate of the Consolidated Total Assets of all Subsidiaries that do not constitute Material Subsidiaries exceeds 5% of the Borrower's Consolidated Total Assets as of such date, the Borrower (or the Administrative Agent, in the event the Borrower has failed to do so within 10 days of request therefor by the Administrative Agent) shall, to the extent necessary, designate sufficient Subsidiaries to be deemed to be "Material Subsidiaries" to eliminate such excess, and such designated Subsidiaries shall thereafter constitute Material Subsidiaries. Assets of Foreign Subsidiaries shall be converted into Dollars at the rates used for purposes of preparing the consolidated balance sheet of the Borrower included in such audited financial statements.

"Maximum Commitment" means, when used with reference to any Lender, the aggregate of such Lender's Term Commitments and Revolving Commitment in the amounts not to exceed those set forth opposite the name of such Lender on Schedule 1.1(a) hereto, subject to reduction from time to time in accordance with the terms of this Agreement.

"Minimum Borrowing Amount" means, with respect to (i) Base Rate Loans, \$3,000,000, (ii) with respect to Eurocurrency Loans, \$5,000,000, in the case of a Borrowing in Dollars, £2,000,000, in the case of a Borrowing in Sterling, and €5,000,000, in the case of a Borrowing in Euros and (iii) with respect to Swing Line Loans, \$500,000 or the Dollar Equivalent thereof in

an Alternative Currency (or such other amount as the Swing Line Lender may agree.)

"Minimum Floor Amount" has the meaning assigned to that term in Section 12.23.

"Moody's" means Moody's Investors Service, Inc. or any successor to the rating agency business thereof.

"Mortgage" has the meaning assigned to that term in <u>Section 5.1(e)(i)</u> and shall also include any mortgage or similar documents executed pursuant to <u>Section 7.11</u>.

"Mortgage Policies" has the meaning assigned to that term in Section 5.1(e)(ii)

"Mortgaged Property" means the owned or leased real property subject to a Mortgage as indicated on <u>Schedule 6.21(c)</u> and shall also include any owned or leased real property subject to a Mortgage pursuant to <u>Section 7.11</u>.

"Most Recent Leverage Ratio" means, at any date, the Leverage Ratio for the Test Period ending as of the most recently ended Fiscal Quarter for which financial statements have been delivered to the Lenders pursuant to Section 7.1; provided, however, that if the Borrower fails to deliver such financial statements as required by Section 7.1 and further fails to remedy such default within five days of notice thereof from the Administrative Agent, then, without prejudice to any other rights of any Lender hereunder, the Most Recent Leverage Ratio shall be deemed to be the highest level as of the date such financial statements were required to be delivered under Section 7.1. The Most Recent Leverage Ratio for the period from the Closing

28

Date to the first date on which financial statements are required to be delivered to the Lenders pursuant to <u>Section 7.1</u> shall be deemed to be 3.0 to 1.0.

"Most Recent Senior Secured Leverage Ratio" means, at any date, the Senior Secured Leverage Ratio for the Test Period ending as of the most recently ended Fiscal Quarter for which financial statements have been delivered to the Lenders pursuant to Section 7.1; provided, however, that if the Borrower fails to deliver such financial statements as required by Section 7.1 and further fails to remedy such default within five days of notice thereof from the Administrative Agent, then, without prejudice to any other rights of any Lender hereunder, the Most Recent Senior Secured Leverage Ratio shall be deemed to be the highest level as of the date such financial statements were required to be delivered under Section 7.1. The Most Recent Senior Secured Leverage Ratio shall be deemed to be greater than 2.25 to 1.0 for the period from the Effective Date to the earlier of (A) the date of delivery of the financial statements required by Section 7.1 and a Compliance Certificate with respect to the Fiscal Quarter of the Borrower ended September 30, 2007 or (B) the date three (3) Business Days following delivery by the Borrower of a certificate of a Responsible Financial Officer stating that (i) the US Commodity Business Sale has been consummated, (ii) no Event of Default or Unmatured Event of Default has occurred and (iii) certifying as to the Senior Secured Leverage Ratio calculated on a Pro Forma Basis after giving effect to the US Commodity Business Sale and the repayment of any Indebtedness with the Net Sale Proceeds therefrom that has occurred by the date of such certificate.

"MP Group" has the meaning assigned to that term in the definition of "Change of Control" in this Section 1.1.

"<u>Multiemployer Plan</u>" means any plan described in Section 4001(a)(3) of ERISA to which contributions are or have, within the preceding six years, been made, or are or were, within the preceding six years, required to be made, by the Borrower or any of its ERISA Affiliates or any Subsidiary of the Borrower or ERISA Affiliates of such Subsidiary.

"Net Equity Proceeds" means the cash proceeds received from (i) any capital contribution from any member of the Borrower or (ii) the issuance of Capital Stock of the Borrower (other than to a Subsidiary or an employee stock ownership plan), net of the actual liabilities for reasonably anticipated cash taxes in connection with such incurrence, if any, any underwriting, brokerage and other customary selling commissions incurred in connection with such incurrence, and reasonable legal, advisory and other fees and expenses, incurred in connection with such incurrence.

"Net Recovery Proceeds" means, with respect to any Recovery Event, an amount equal to the sum of the aggregate cash payments received by the Borrower or any Subsidiary of the Borrower from such Recovery Event minus the direct costs and expenses incurred in connection therewith (including, without limitation, all legal fees and other professional fees) and minus any provision for taxes in respect thereof made in accordance with GAAP. Any proceeds, costs expenses or taxes denominated in a currency other than Dollars shall, for purposes of the calculation of the amount of Net Recovery Proceeds, be in an amount equal to the Dollar Equivalent thereof as of the date of receipt of such Net Recovery Proceeds by the Borrower or any Subsidiary of the Borrower.

20

"Net Sale Proceeds" means, with respect to any Asset Disposition, an amount equal to the sum of the aggregate cash payments received by the Borrower or any Subsidiary of the Borrower from such Asset Disposition (including, without limitation, cash received by way of deferred payment pursuant to a note receivable, conversion of non-cash consideration, cash payments in respect of purchase price adjustments or otherwise, but only as and when such cash is received) minus the direct costs and expenses incurred in connection therewith (including in the case of any Asset Disposition, the payment of the outstanding principal amount of, premium, if any, and interest on any Indebtedness (other than hereunder) required to be repaid as a result of such Asset Disposition) and minus any provision for taxes in respect thereof made in accordance with GAAP. Any proceeds received in a currency other than Dollars shall, for purposes of the calculation of the amount of Net Sale Proceeds, be in an amount equal to the Dollar Equivalent thereof as of the date of

receipt thereof by the Borrower or any Subsidiary of the Borrower.

"New Revolving Lender" has the meaning assigned to that term in Section 2.10(a).

"Non-Defaulting Lender" means each Lender which is not a Defaulting Lender.

"Non-Impaired Lender" means, at any time, any Revolving Lender which is not an Impaired Lender.

"Non-U.S. Participant" means any Lender that is not a United States person within the meaning of Code section 7701(a)

(30).

"Note" means any of the Swing Line Note, the Revolving Notes or the Term Notes and "Notes" means all of such Notes collectively.

"Notice of Borrowing" has the meaning assigned to that term in Section 2.5.

"Notice of Conversion or Continuation" has the meaning assigned to that term in Section 2.6.

"Notice of Issuance" has the meaning assigned to that term in Section 2.9(b).

"Notice of Revolving Commitment Increase" has the meaning assigned to that term in Section 2.10(b).

"Notice Office" means the office of the Administrative Agent located at 1111 Fannin, 10th Floor, Houston, Texas 77002, Attention: Monica M. Espitia (Facsimile No. 713-427-6307), or such other office as the Administrative Agent may designate to the Borrower and the Lenders from time to time.

"Obligations" means all liabilities and obligations of the Borrower and its Subsidiaries now or hereafter arising under this Agreement and all of the other Loan Documents, whether for principal, interest, fees, expenses, indemnities or otherwise, and whether primary, secondary, direct, indirect, contingent, fixed or otherwise (including obligations of performance).

30

"Operating Financing Lease" means a lease of the type described in clause (xi) of the definition of "Indebtedness".

"Organizational Documents" means, with respect to any Person, such Person's memorandum, articles or certificate of incorporation, certificates of formation, bylaws, partnership agreement, limited liability company agreement, joint venture agreement or other similar governing documents and any document setting forth the designation, amount and/or relative rights, limitations and preferences of any class or series of such Person's Capital Stock.

"Other Hedging Agreement" means any foreign exchange contract, currency swap agreement, futures contract, commodity agreements, option contract, synthetic cap or other similar agreement other than an Interest Rate Agreement to which the Borrower or any Subsidiary is a party.

"Overdraft Facility" has the meaning assigned to that term in Section 8.2(b)(xii).

"Overdraft Reserve" shall mean an amount, if any, equal to the amount by which Indebtedness incurred by the Borrower or any of its Subsidiaries pursuant to Section 8.2(b)(xii) exceeds \$60,000,000 (or the Dollar Equivalent thereof).

"<u>Parent Company</u>" means each Person which owns, directly or indirectly, at least a majority of the Voting Securities of the Borrower.

"Participants" has the meaning assigned to that term in Section 12.8(b).

"<u>Participating Subsidiary</u>" means any Subsidiary of the Borrower or other entity formed as necessary or customary under the laws of the relevant jurisdiction that is a participant in a Permitted Accounts Receivable Securitization.

"Patriot Act" has the meaning assigned to that term in Section 6.22.

"Payment Office" means (i) with respect to the Administrative Agent or the Swing Line Lender, for payments with respect to Dollar-denominated Loans, the address located at 1111 Fannin, 10th Floor, Houston, Texas 77002, Attention: Monica M. Espitia (Facsimile No. 713-427-6307), or such other address as the Administrative Agent or the Swing Line Lender, as the case may be, may from time to time specify in accordance with Section 12.3 or (ii) with respect to the Administrative Agent or the Swing Line Lender, for payments in an Alternative Currency or with respect to a Letter of Credit denominated in an Alternative Currency, such account at such bank or office in London (or such other location) as the Administrative Agent or the Swing Line Lender, as the case may be, shall designate by notice to the Person required to make the relevant payment.

"PBGC" means the Pension Benefit Guaranty Corporation created by Section 4002(a) of ERISA.

"Perfection Certificates" has the meaning assigned to that term in Section 5.1(f).

"Permitted Accounts Receivable Securitization" means any receivables financing program providing for the sale, conveyance or contribution to capital of Receivables Facility Assets or interests therein by the Borrower and its Participating Subsidiaries to a Receivables Subsidiary in transactions purporting to be sales, which Receivables Subsidiary shall finance the purchase of such Receivables Facility Assets by the direct (or, to the extent approved by the Administrative Agent as evidenced by its written approval thereof, indirect) sale, transfer, conveyance, lien, grant of participation or other interest or pledge of such Receivables Facility Assets or interests therein to one or more limited purpose financing companies, special purpose entities, trusts and/or financial institutions, in each case, on a limited recourse basis as to the Borrower and the Participating Subsidiaries (except to the extent a limitation on recourse is not customary for similar transactions or is prohibited in the relevant jurisdiction); provided that any such transaction shall be consummated pursuant to documentation necessary or customary for such transactions in the relevant jurisdiction (or otherwise satisfactory to the Administrative Agent as evidenced by its written approval thereof) and shall provide for purchase price percentages reasonably satisfactory to the Administrative Agent. Each of the Receivables Securitization Programs shall be considered a Permitted Accounts Receivables Securitization hereunder.

"Permitted Entrustment Loan Arrangement" means a coordinated credit-linked deposit and loan facility arranged in compliance with the laws of the People's Republic of China pursuant to which a Foreign Subsidiary of the Borrower organized under the laws of the People's Republic of China makes loans or advances to another Foreign Subsidiary of the Borrower through the use of an intermediary financial institution in the People's Republic of China.

"Permitted Non-Cash Impairment and Restructuring Charges" means, with respect to any Person, for any period, (i) the aggregate depreciation, amortization and other non-cash charges of such Person and its Subsidiaries reducing Consolidated Net Income of such Person and its Subsidiaries for such period, determined on a consolidated basis in accordance with GAAP (excluding any such charges constituting an extraordinary item or loss or any such charge which requires an accrual of or a reserve for cash charges for any future period) and (ii) cash charges taken after the Effective Date in an aggregate amount not to exceed \$100,000,000.

"Permitted Liens" has the meaning assigned to that term in Section 8.1.

"Permitted Real Property Encumbrances" means (i) those liens, encumbrances and other matters affecting title to any Mortgaged Property listed in the applicable title policy in respect thereof (or any update thereto) and found, on the date of delivery of such title policy to the Administrative Agent in accordance with the terms hereof, reasonably acceptable by the Administrative Agent, (ii) as to any particular real property at any time, such easements, encroachments, covenants, restrictions, rights of way, minor defects, irregularities or encumbrances on title which do not, in the reasonable opinion of the Administrative Agent, materially impair such real property for the purpose for which it is held by the mortgagor or owner, as the case may be, thereof, or the Lien held by the Administrative Agent, (iii) municipal and zoning laws, regulations, codes and ordinances, which are not violated in any material respect by the existing improvements and the present use made by the mortgagor or owner, as the case may be, of such real property, (iv) general real estate taxes and assessments not yet delinquent, and (v) such other items as the Administrative Agent may consent to.

32

"Permitted Refinancing Indebtedness" means, with respect to any Indebtedness, any Indebtedness refinancing, extending, renewing or refunding such Indebtedness; provided, however, that any such refinancing Indebtedness shall (i) be issued by the same obligor as the Indebtedness being so refinanced (or by Huntsman Corporation or a Parent Company) and be on terms, taken as a whole, not more restrictive than the terms of the documents governing the Indebtedness being so refinanced; (ii) if the Indebtedness being so refinanced is subordinated to the Obligations, be subordinated to the Obligations on substantially the same terms (or on terms at least as favorable to the Lenders) as Indebtedness being so refinanced; (iii) be in a principal amount (as determined as of the date of the incurrence of such refinancing Indebtedness in accordance with GAAP) not exceeding the principal amount of the Indebtedness being refinanced on such date plus any call premiums, prepayment fees, costs and expenses paid in connection with such refinancing; (iv) not have a Weighted Average Life to Maturity less than the Indebtedness being refinanced; (v) if the Indebtedness being refinanced is Public Notes, be unsecured Indebtedness maturing no earlier than the then latest Term Maturity Date; and (vi) be upon terms and subject to documentation which is in form and substance reasonably satisfactory in all material respects to the Administrative Agent.

"Permitted Technology Licenses" has the meaning assigned to that term in Section 8.3(f) of this Agreement.

"<u>Permitted Unconsolidated Ventures</u>" means an Investment in a Person not constituting a Subsidiary of the Borrower which Person is not engaged in any business other than that permitted under <u>Section 8.9</u> for the Borrower and its Subsidiaries.

"<u>Person</u>" means an individual or a corporation, partnership, limited liability company, trust, incorporated or unincorporated association, joint venture, joint stock company, Governmental Authority or other entity of any kind.

"Plan" means any plan described in Section 4021(a) of ERISA and not excluded pursuant to Section 4021(b) thereof, which is or has, within the preceding six years, been established or maintained, or to which contributions are or have, within the preceding six years, been made, by the Borrower or any of its ERISA Affiliates or any Subsidiary of the Borrower or any ERISA Affiliates of such Subsidiary, but not including any Multiemployer Plan.

"Plan Administrator" has the meaning assigned to the term "administrator" in Section 3(16)(A) of ERISA.

"Plan Sponsor" has the meaning assigned to the term "plan sponsor" in Section 3(16)(B) of ERISA.

"Pledge Agreement" has the meaning assigned to that term in Section 5.1(c) of this Agreement.

"Pledged Receivables Subsidiary Notes" means the subordinated notes of the Receivables Subsidiary, if any, issued to the Borrower or any Participating Subsidiary in connection with a Permitted Accounts Receivable Securitization, which subordinated notes are pledged pursuant to the Receivables Subsidiary Pledge Agreement.

33

"<u>Pledged Receivables Subsidiary Stock</u>" means all the issued and outstanding shares of capital stock of the Receivables Subsidiary, which shares are pledged pursuant to the Receivables Subsidiary Pledge Agreement.

"<u>Pledged Securities</u>" means, collectively, "Pledged Securities" as defined in the Collateral Security Agreement or any other pledged securities under any Security Document.

"Port Arthur Fire Add Back" means, for any period that includes any Fiscal Quarter ending after December 31, 2006 and on or before December 31, 2007, the sum of (i) an amount not to exceed \$50,000,000 per Fiscal Quarter (provided, that any excess above \$50,000,000 may be carried forward to the next Fiscal Quarter) ending after December 31, 2006 and on or before December 31, 2007 included in such period (and \$0 for each Fiscal Quarter that ends thereafter included in such period), equal to the net business interruption losses (calculated in accordance with GAAP) for the applicable period, including the retained portion of any business interruption claims that relate to the Port Arthur Plant Fire and that are, or are expected to be, the subject of insurance claims by the Borrower or its Subsidiaries but only to the extent such claims have not been denied and have been, or in the reasonable judgment of the Borrower, are likely to be, paid by the Borrower's or its Subsidiaries' insurance carriers or represent the retained portion of any business interruption claims pertaining to the deductible <u>plus</u> (ii) to the extent (A) deducted in determining Consolidated Net Income for such period and (B) such charges are, or are expected to be, the subject of insurance claims by the Borrower or its Subsidiaries but only to the extent such claims have not been denied and have been, or in the reasonable judgment of the Borrower, are likely to be, paid by the Borrower's or its Subsidiaries' insurance carriers or represent the retained portion of any physical property insurance claims pertaining to the deductible, any charges payable in cash for the maintenance or repair of property damaged in the Port Arthur Plant Fire to the extent of such damage. The Port Arthur Fire Add Back shall be set forth on the Compliance Certificate delivered pursuant to Section 7.2(b) for each Fiscal Quarter ending after December 31, 2006 and on or before September 30, 2008, and the Borrower shall, upon the request of the Administrative Agent, deliver such detailed computations as are necessary to support such amount.

"Port Arthur Fire Insurance Income" means, for any period, the amount of income of the Borrower or any of its Subsidiaries for such period (as determined in accordance with GAAP) related to insurance proceeds received or expected to be received as a result of the Port Arthur Plant Fire, including, without limitation, any related (i) payments in respect of claims on policies related to business interruption insurance during such period, (ii) physical property insurance proceeds received by the Borrower or any of its Subsidiaries during such period and (iii) accruals for expected insurance income recoveries during such period.

"Port Arthur Plant Fire" means the fire that occurred at the Borrower's Port Arthur, Texas olefins manufacturing plant on or about April 29, 2006.

"Pro Forma Basis" means, (a) with respect to the preparation of a pro forma financial statement for any purpose relating to an Acquisition or for calculation of Consolidated EBITDA for any period, a pro forma financial statement or calculation prepared on the basis that (i) any Indebtedness incurred or assumed in connection with such Acquisition was incurred or assumed on the first day of the applicable period, (ii) if such Indebtedness bears a floating

34

interest rate, such interest shall be paid over such period at the rate in effect on the date of such Acquisition, and (iii) all income and expense associated with the assets or entity acquired in connection with such Acquisition for the most recently ended four fiscal quarter period for which such income and expense amounts are available shall be treated as being earned or incurred by Borrower over the applicable period on a pro forma basis without giving effect to any cost savings other than, to the extent desired to be included by the Borrower, Pro Forma Cost Savings, (b) with respect to the preparation of a pro forma financial statement for any purpose relating to an Asset Disposition or for calculation of Consolidated EBITDA, a pro forma financial statement or calculation prepared on the basis that (i) any Indebtedness prepaid out of the proceeds of such Asset Disposition shall be deemed to have been prepaid as of the first day of the applicable period, and (ii) all income and expense (other than such expenses as the Borrower, in good faith, estimates will not be reduced or eliminated as a consequence of such Asset Disposition) associated with the assets or entity disposed of in connection with such Asset Disposition shall be deemed to have been eliminated as of the first day of the applicable period and (c) with respect to the preparation of a pro forma financial statement for any purpose relating to an incurrence of Indebtedness or the making of any payment (or any repayment), or both, a pro forma financial statement or calculation prepared on the basis that (i) such Indebtedness incurred was incurred or such payment (or repayment) was made on the first day of the applicable period and (ii) if such Indebtedness bears a floating rate of interest, such interest was paid over such period at the rate in effect on the date of incurrence of such Indebtedness. For the avoidance of doubt, for the purpose of determining any financial ratio hereunder on a "Pro Forma Basis" for any period (the "Subject Period"), any Acquisition, Asset Disposition, incurrence of Indebtedness or payment (or repayment) made at any time during the period commencing after the last day of the Subject Period through and including the date that such determination is required to be made, shall be taken into account for purposes of this definition.

"<u>Pro Forma Cost Savings</u>" means, with respect to the determination of Consolidated Net Income on a Pro Forma Basis, such cost savings as would be permitted pursuant to Rule 11.02 of Regulation S-X (assuming Regulation S-X applied to the acquisition in question), if prior to the consummation of any Acquisition permitted by <u>Section 8.7(m)</u>, the Borrower's certified public accountants shall

have issued a comfort letter (in a manner consistent with example d of SAS 72) or shall have performed procedures agreed upon by the Borrower and Administrative Agent, in each case related to the determination of such Consolidated Net Income on a Pro Forma Basis in accordance with the applicable accounting requirements of Rule 11.02 of Regulation S-X.

"Pro Rata Share" means, when used with reference to any Lender and any described aggregate or total amount of any Facility or Facilities, an amount equal to the result obtained by multiplying such described aggregate or total amount by a fraction the numerator of which shall be such Lender's Commitment with respect to such Facility or Facilities and the denominator of which shall be the aggregate Commitments outstanding for all Lenders or, if no Commitments are then outstanding, with respect to such Facilities such Lender's aggregate Loans with respect to such Facilities to the total Loans outstanding hereunder with respect to such Facility or Facilities, and when used with reference to any Lender's percent interest, such fraction, expressed as a percentage.

"Projections" has the meaning assigned to that term in Section 6.5(e).

35

"<u>Public Note Documents</u>" means the Senior Secured (2010) Note Documents, the Senior Note (2012) Documents, the Senior Subordinated Note (2013) Documents, the Senior Subordinated Note (2014) Documents, the Senior Subordinated Note (2015) Documents, the Senior Note (2016) Documents and all documents evidencing, guaranteeing or otherwise governing any Permitted Refinancing Indebtedness of any Public Notes.

"<u>Public Notes</u>" means the Senior Secured Notes (2010), the Senior Notes (2012), Senior Subordinated Notes (2013), the Senior Subordinated Notes (2014), the Senior Subordinated Notes (2015) and the Senior Notes (2016), in each case in the amounts outstanding on the Fifth Amendment Effective Date, and any notes evidencing any Permitted Refinancing Indebtedness of any of the foregoing.

"Quarterly Payment Date" means each March 31, June 30, September 30 and December 31 of each year.

"Quoted Rate" means the rate of interest per annum with respect to a Swing Line Loan denominated in an Alternative Currency as determined by the Swing Line Lender at the time such Swing Line Loan is made to the Borrower.

"Receivables Documents" shall mean all documentation relating to any receivables financing program providing for the sale of Receivables Facility Assets by the Borrower and its Subsidiaries (whether or not to a Receivables Subsidiary) in transactions purporting to be sales and shall include the Receivables Securitization Program Documents.

"Receivables Facility Assets" shall mean any Accounts Receivable (whether now existing or arising in the future) of the Borrower or any of its Subsidiaries, and any assets related thereto, including without limitation (i) all collateral given by the respective account debtor or on its behalf (but not by the Borrower or any of its Subsidiaries) securing such Accounts Receivable, (ii) all contracts and all guarantees (but not by the Borrower or any of its Subsidiaries) or other obligations directly related to such Accounts Receivable, (iii) other related assets and (iv) proceeds of all of the foregoing. Receivables Securitization Program Facility Assets shall be deemed to constitute Receivables Facility Assets.

"Receivables Facility Attributed Indebtedness" at any time shall mean the aggregate Dollar Equivalent net outstanding amount theretofore paid, directly or indirectly, by a funding source in respect of the Receivables Facility Assets or interests therein sold, conveyed, contributed or transferred or pledged pursuant to the relevant Receivables Documents (including in connection with a Permitted Accounts Receivable Securitization) (it being the intent of the parties that the amount of Receivables Facility Attributed Indebtedness at any time outstanding approximate as closely as possible the principal amount of Indebtedness which would be outstanding at such time under the Receivables Documents if the same were structured as a secured lending agreement rather than an agreement providing for the sale, conveyance, contribution to capital, transfer or pledge of such Receivables Facility Assets or interests therein).

36

"Receivables Securitization Program Documents" means all documents and deliveries in connection with the Receivables Securitization Programs, as such documents may be amended or modified from time to time to the extent permitted under this Agreement.

"Receivables Securitization Program Facility Assets" means all "Receivables" and other "Receivable Assets" (as defined in the Receivables Securitization Program Loan Agreements) of the Borrower and the Receivables Program Participating Subsidiaries.

"Receivables Securitization Program Loan Agreements" means, collectively, (i) that certain European Receivables Loan Agreement, dated as of October 16, 2009, among Huntsman Receivables Finance LLC, Huntsman (Europe) B.V.B.A, the several entities party thereto as lenders, the several financial institutions party thereto as funding agents, Barclays Bank plc, as administrative agent and collateral agent, and (ii) that certain U.S. Receivables Loan Agreement, dated as of October 16, 2009, among Huntsman Receivables Finance II LLC, Huntsman (Europe) B.V.B.A., the several entities party thereto as lenders, the several financial institutions party thereto as funding agents, the several commercial paper conduits party thereto as conduit lenders, the several financial institutions party thereto as committed lenders, Wachovia Bank National Association, as administrative agent and collateral agent, each as amended, supplemented or otherwise modified from time to time, or either such receivables loan agreement, as the context requires.

"Receivables Securitization Program Participating Subsidiaries" means each Subsidiary of the Borrower from time to time party to any of the Receivables Securitization Programs.

"Receivables Securitization Program Subsidiaries" means, collectively, (i) Huntsman Receivables Finance LLC and (ii) Huntsman Receivables Finance II LLC, each a limited liability company organized under the laws of the State of Delaware, or either such entity, as the context requires.

"Receivables Securitization Programs" means each of the receivables financing programs providing for the sale, contribution or other transfer of Receivables Securitization Program Facility Assets pursuant to the relevant Receivables Securitization Program Documents by the Borrower and/or the Receivables Securitization Program Participating Subsidiaries to the relevant Receivables Securitization Program Subsidiary(directly or through the Borrower, another Receivables Securitization Program Participating Subsidiary or a financial institution) in which the Borrower and/or the Receivables Securitization Program Subsidiaries shall finance the purchase of such Receivables Securitization Program Facility Assets by the sale, transfer, conveyance, lien, grant of a participation or other interest or pledge of such Receivables Securitization Program Facility Assets directly or indirectly to one or more limited purpose financing companies, special purpose entities and/or financial institutions, in each case, on a limited recourse basis as to the Borrower and the Receivables Securitization Program Participating Subsidiaries.

"Receivables Subsidiary" means a special purpose, bankruptcy remote Wholly-Owned Subsidiary of the Borrower which may be formed for the sole and exclusive purpose of engaging in activities in connection with the purchase, sale and financing of Accounts

37

Receivable in connection with and pursuant to one or more Permitted Accounts Receivable Securitizations; provided, however, that if the law of a jurisdiction in which the Borrower proposes to create a Receivables Subsidiary does not provide for the creation of a bankruptcy remote entity that is acceptable to the Borrower or requires the formation of one or more additional entities (whether or not Subsidiaries of the Borrower), the Administrative Agent may in its discretion permit the Borrower to form such other type of entity in such jurisdiction to serve as a Receivables Subsidiary as is necessary or customary for similar transactions in such jurisdiction. Each of the Receivables Securitization Program Subsidiaries shall be considered a Receivables Subsidiary hereunder.

"Receivables Subsidiary Pledge Agreement" means the Pledge Agreement, Collateral Security Agreement and/or such other pledge or security agreement in form reasonably satisfactory to the Administrative Agent pursuant to which the Borrower or a Participating Subsidiary pledges the Pledged Receivables Subsidiary Stock and the Pledged Receivables Subsidiary Notes to the Collateral Agent for the benefit of the Lenders to secure the "Secured Obligations" described in the Pledge Agreement or the Collateral Security Agreement, as applicable.

"Recovery Event" means the receipt by the Borrower or any of its Subsidiaries of any insurance or condemnation proceeds payable (i) by reason of any theft, physical destruction or damage or any other similar event with respect to any properties or assets of the Borrower or any of its Subsidiaries, (ii) by reason of any condemnation, taking, seizing or similar event with respect to any properties or assets of the Borrower or any of its Subsidiaries and (iii) under any policy of insurance required to be maintained under Section 7.8 provided, however, that in no event shall payments made under business interruption insurance constitute a Recovery Event.

"Reference Lenders" means the principal office of JPMCB in the relevant jurisdiction or such other banks as may be appointed by the Administrative Agent in consultation with the Borrower.

"Refinanced Facility Debt" has the meaning assigned to that term in Section 12.23(a).

"Refunded Swing Line Loans" has the meaning assigned to that term in Section 2.1(c)(ii).

"Regulation D" means Regulation D of the Board as from time to time in effect and any successor to all or a portion thereof establishing reserve requirements.

"Related Fund" means, with respect to any Lender which is a Fund, any other Fund that is administered or managed by the same investment advisor of such Lender or by an Affiliate of such investment advisor.

"Release" means any release, spill, emission, leaking, pumping, pouring, emptying, dumping, injection, deposit, disposal, discharge, dispersal, escape, leaching or migration into the indoor or outdoor environment, including the movement of Contaminants through or in the air, soil, surface water or groundwater.

38

"Remedial Action" means actions required to (i) clean up, remove, treat or in any other way address Contaminants in the indoor or outdoor environment; (ii) prevent, minimize or otherwise address the Release or substantial threat of a material Release of Contaminants so they do not migrate or endanger or threaten to endanger public health or welfare or the indoor or outdoor environment; or (iii) perform pre-response or post-response studies and investigations and post-response monitoring and care or any other studies, reports or investigations relating to Contaminants.

"Replaced Lender" has the meaning assigned to that term in Section 3.7.

"Replacement Lender" has the meaning assigned to that term in Section 3.7.

"Reportable Event" means a "reportable event" described in Section 4043(c) of ERISA or in the regulations thereunder with respect to a Plan other than a reportable event for which the 30 day notice requirement to the PBGC has been waived, any event requiring disclosure under Section 4063(a) or 4062(e) of ERISA, receipt of a notice of withdrawal liability with respect to a Multiemployer Plan pursuant to Section 4202 of ERISA or receipt of a notice of reorganization or insolvency with respect to a Multiemployer Plan pursuant to Section 4242 or 4245 of ERISA.

"Required Lenders" means Non-Defaulting Lenders the sum of whose outstanding Term Loans and Revolving Commitments (or, after the Total Revolving Commitment has been terminated, the Assigned Dollar Value of outstanding Revolving Loans and the Assigned Dollar Value of LC Obligations) constitute greater than 50% of the sum of (i) the total outstanding Dollar Equivalent amount of Term Loans of Non-Defaulting Lenders and (ii) the Total Revolving Commitment less the aggregate Revolving Commitments of Defaulting Lenders (or, after the Total Revolving Commitment has been terminated, the total Assigned Dollar Value of outstanding Revolving Loans of Non-Defaulting Lenders and the aggregate Pro Rata Share of all Non-Defaulting Lenders of the total Assigned Dollar Value of outstanding LC Obligations at such time).

"Required Note Offer Amount Proceeds" shall mean (i) with respect to any Asset Disposition, the amount of any Net Sale Proceeds which would be required by the terms of the Senior Secured Notes (2010) Indenture to be applied to offer to purchase Senior Secured Notes (2010) and (ii) with respect to any Recovery Event, any Net Recovery Proceeds which would be required by the terms of the Senior Secured Notes (2010) Indenture to offer to purchase Senior Secured Notes (2010); in each case (x) including amounts which are required under the terms of the Senior Secured Notes (2010) Indenture to be accumulated to make such an offer, (y) assuming no reinvestment of such proceeds or expenditure of such proceeds to purchase replacement properties or assets and (z) after giving effect to any prepayment of Loans to the maximum extent permitted by the Senior Secured Notes (2010) Indenture without requiring an offer to repurchase Senior Secured Notes (2010).

"Requirement of Law" means, as to any Person, any law (including common law), treaty, rule or regulation or judgment, decree, determination or award of an arbitrator or a court or other Governmental Authority, including without limitation, any Environmental Law, in each

39

case applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject.

"Resigning Agent" has the meaning assigned to that term in Section 11.9(h).

"Responsible Financial Officer" means, as to any Person, the chief financial officer, principal accounting officer, a financial vice president, controller, manager (in the case of a limited liability company) having responsibility for financial matters, treasurer or assistant treasurer of such Person.

"Responsible Officer" means, as to any Person, any of the chairman or vice chairman of the board of directors, the president, any executive vice president, the vice president-controller, any vice president, manager (in the case of a limited liability company) or any Responsible Financial Officer of such Person.

"Restricted Payment" has the meaning assigned to that term in Section 8.4(a).

"Revolver Event of Default" means any Event of Default under Section 10.1(c) as a result of the Borrower failing to be in compliance with Section 9.1.

"Revolver Termination Date" means the Stated Termination Date or such earlier date as the Revolving Commitments shall have been terminated or otherwise reduced to \$0 pursuant to this Agreement; provided, that, if there shall occur a date (an "earlier date") on which in excess of \$100,000,000 of Public Notes and/or Term Loans which have not been repaid or refinanced with Permitted Refinancing Indebtedness in accordance with the terms of this Agreement is scheduled to come due within three months, "Revolver Termination Date" means such earlier date.

"Revolving Commitment" means, with respect to any Revolving Lender, the obligation of such Revolving Lender to make Revolving Loans and participate in Letters of Credit and Swing Line Loans, as such commitment may be adjusted from time to time pursuant to this Agreement, which commitment as of the Fifth Amendment Effective Date is the amount set forth opposite such Lender's name on Schedule 1.1(a) hereto under the caption "Amount of Revolving Commitment" as the same may be adjusted from time to time pursuant to the terms hereof and "Revolving Commitments" means such commitments collectively, which commitments will not exceed \$300,000,000 in the aggregate.

"Revolving Commitment Increase" has the meaning assigned to that term in Section 2.10(a).

"Revolving Commitment Increase Date" has the meaning assigned to that term in Section 2.10(a).

"Revolving Facility" means the credit facility under this Agreement evidenced by the Revolving Commitments and the Revolving Loans.

40

"Revolving Loan" and "Revolving Loans" have the meanings given in Section 2.1(b)(i).

"Revolving Note" has the meaning assigned to that term in Section 2.2(a).

"Rubicon" means Rubicon Inc., and its successors and assigns.

"Sale and Leaseback Transaction" means any arrangement, directly or indirectly, whereby a seller or transferor shall sell or otherwise transfer any real or personal property and then or thereafter lease, or repurchase under an extended purchase contract, conditional sales or other title retention agreement, the same or similar property.

"Scheduled Term B Dollar Repayments" means, with respect to the principal payments on the Term B Dollar Loans for each date set forth below, that percentage of the aggregate outstanding principal amount of Term B Dollar Loans on the Effective Date set forth opposite thereto:

Scheduled Term B Dollar Repayments

Date	Principal Payment
March 31, 2008	1% of the aggregate principal amount as of the Effective Date
March 21, 2000	10/ of the aggregate minerical amount as of the Effective Date
March 31, 2009	1% of the aggregate principal amount as of the Effective Date
March 31, 2010	1% of the aggregate principal amount as of the Effective Date
,	
March 31, 2011	1% of the aggregate principal amount as of the Effective Date
March 31, 2012	1% of the aggregate principal amount as of the Effective Date
March 31, 2013	1% of the aggregate principal amount as of the Effective Date
17141011 31, 2013	1/0 of the aggregate principal amount as of the Effective Date
Term B Loan Maturity	100% of the aggregate principal amount of Term B Dollar Loans
Date	outstanding on the Term B Loan Maturity Date.

"Scheduled Term C Dollar Repayments" means, with respect to the principal payments on the Term C Dollar Loans for each date set forth below, that percentage of the aggregate outstanding principal amount of Term C Dollar Loans on the Fourth Amendment Effective Date set forth opposite thereto:

41

Scheduled Term C Dollar Repayments

Date	Principal Payment
March 31, 2010	1% of the aggregate principal amount as of the Fourth Amendment Effective Date
March 31, 2011	1% of the aggregate principal amount as of the Fourth Amendment Effective Date
March 31, 2012	1% of the aggregate principal amount as of the Fourth Amendment Effective Date
March 31, 2013	1% of the aggregate principal amount as of the Fourth Amendment Effective Date
March 31, 2014	1% of the aggregate principal amount as of the Fourth Amendment Effective Date
March 31, 2015	1% of the aggregate principal amount as of the Fourth Amendment Effective Date
March 31, 2016	1% of the aggregate principal amount as of the Fourth Amendment Effective Date

Term C Loan Maturity Date 100% of the aggregate principal amount of Term C Dollar Loans outstanding on the Term C Loan Maturity Date.

"Scheduled Term Repayments" mean, for any Term Facility, the scheduled principal payments set forth in the "Scheduled Term Repayments" definition applicable to such Term Facility.

"SEC" means the Securities and Exchange Commission or any successor thereto.

"Secured Parties" has the meaning provided in the respective Security Documents to the extent defined therein and shall include any Person who is granted a security interest pursuant to any Loan Document.

"Securities" means any stock, shares, voting trust certificates, bonds, debentures, options, warrants, notes, or other evidences of indebtedness, secured or unsecured, convertible, subordinated or otherwise, or in general any instruments commonly known as "securities" or any certificates of interest, shares or participations in temporary or interim certificates for the purchase or acquisition of, or any right to subscribe to, purchase or acquire, any of the foregoing.

42.

"Security Documents" means, collectively the Collateral Security Agreement, the Pledge Agreement, the Mortgages, the UK Security Documents and all other agreements, assignments, security agreements, instruments and documents executed in connection therewith, in each case as the same may at any time be amended, supplemented, restated or otherwise modified and in effect. For purposes of this Agreement, "Security Documents" shall also include all guaranties, mortgagees, pledge agreements, collateral assignments, subordination agreements and other collateral documents and any reaffirmation of the foregoing in the nature thereof entered into by the Borrower or any Subsidiary of the Borrower on and after the date of this Agreement in favor of the Collateral Agent for the benefit of the Secured Parties in satisfaction of the requirements of this Agreement.

"Senior Note (2012) Documents" means the Senior Notes (2012), the Senior Notes (2012) Indenture and all other documents evidencing, guaranteeing or otherwise governing the terms of the Senior Notes (2012).

"Senior Note (2016) Documents" means the Senior Notes (2016), the Senior Notes (2016) Indenture and all other documents evidencing, guaranteeing or otherwise governing the terms of the Senior Notes (2016).

"Senior Notes (2012)" means those certain 11-1/2% senior notes due July 15, 2012 issued pursuant to the terms of the Senior Notes (2012) Indenture.

"Senior Notes (2016)" means those certain 5-1/2% senior notes due 2016 issued pursuant to the terms of the Senior Notes (2016) Indenture.

"Senior Notes (2012) Indenture" means that certain Indenture dated as of June 22, 2004 among Borrower (as successor to Huntsman LLC), the guarantors named therein and HSBC Bank USA, National Association, as trustee (as the same may be amended in compliance with this Agreement, including pursuant to the Supplemental Indenture dated as of July 11, 2005) and any supplemental indenture or additional indenture to be entered into with respect to the Senior Notes (2012) to the extent permitted under Section 8.11.

"Senior Notes (2016) Indenture" means that certain Indenture dated as of July 6, 2009 among Borrower, the guarantors named therein and Wilmington Trust FSB, as trustee (as the same may be amended in compliance with this Agreement) and any supplemental indenture or additional indenture to be entered into with respect to the Senior Notes (2016) to the extent permitted under Section 8.11.

"Senior Secured (2010) Note Documents" means the Senior Secured Notes (2010), the Senior Secured Notes (2010) Indenture and all other documents evidencing, guaranteeing or otherwise governing the terms of the Senior Secured Notes (2010).

"Senior Secured Leverage Ratio" means, as of any date of determination, the ratio of (i) Consolidated Debt that is either (x) secured by any Lien or (y) is Indebtedness of the Borrower or any Subsidiary of the type referred to in clause (x) of the definition of such term (other than any such Indebtedness of Foreign Subsidiaries that are not secured by Liens) to (ii) Consolidated EBITDA for the most recently ended four Fiscal Quarters for which financial statements have been delivered pursuant to Section 7.1.

43

"Senior Secured Notes (2010)" means those certain 11-5/8% senior secured notes due October 15, 2010 issued pursuant to the terms of the Senior Secured Notes (2010) Indenture, and secured by the Collateral on a pari passu basis with the Obligations.

"Senior Secured Notes (2010) Indenture" means that certain Indenture dated as of September 30, 2003 among Borrower (as successor to Huntsman LLC), the guarantors named therein and HSBC Bank USA, National Association (as successor to HSBC Bank USA), as trustee (as the same may be amended in compliance with this Agreement, including pursuant to the Supplemental Indenture dated as of July 13, 2005) and any supplemental indenture or additional indenture to be entered into with respect to the Senior Secured Notes (2010) to the extent permitted under Section 8.11.

(2010) Indenture, as evidenced by the Senior Secured Notes (2010).

"Senior Subordinated Note (2013) Documents" means the Senior Subordinated Notes (2013), the Senior Subordinated Notes (2013/2014) Indenture and all other documents evidencing, guaranteeing or otherwise governing the terms of the Senior Subordinated Notes (2013).

"Senior Subordinated Notes (2013)" means those certain 6-7/8% senior subordinated notes due November 2013 issued by the Borrower pursuant to the terms of the Senior Subordinated Notes (2013/2014) Indenture.

"Senior Subordinated Notes (2013/2014) Indenture" means that certain Indenture dated as of November 13, 2006 among the Borrower, the guarantors named therein and Wells Fargo, National Association, as trustee (as the same may be amended in compliance with this Agreement, including pursuant to the Supplemental Indenture dated February 2007) and any supplemental indenture or additional indenture to be entered into with respect to the Senior Subordinated Notes (2013) or the Senior Subordinated Notes (2014) to the extent permitted under Section 8.11.

"Senior Subordinated Note (2014) Documents" means the Senior Subordinated Notes (2014), the Senior Subordinated Notes (2013/2014) Indenture and all other documents evidencing, guaranteeing or otherwise governing the terms of the Senior Subordinated Notes (2014).

"Senior Subordinated Notes (2014)" means those certain 7-7/8% senior subordinated notes due November 2014 issued by the Borrower pursuant to the terms of the Senior Subordinated Notes (2013/2014) Indenture.

"Senior Subordinated Note (2015) Documents" means the Senior Subordinated Notes (2015), the Senior Subordinated Notes (2015) Indenture and all other documents evidencing, guaranteeing or otherwise governing the terms of the Senior Subordinated Notes (2015).

44

"Senior Subordinated Notes (2015)" shall mean (i) those certain 7-3/8% senior subordinated notes due January 1, 2015 denominated in Dollars issued by the Borrower pursuant to the terms of the Senior Subordinated Notes (2015) Indenture and (ii) those certain 7-1/2% senior subordinated notes due January 1, 2015 denominated in Euros issued by the Borrower pursuant to the terms of the Senior Subordinated Notes (2015) Indenture.

"Senior Subordinated Notes (2015) Indenture" shall mean that certain Indenture dated as of December 17, 2004 among Borrower, the guarantors named therein and Wells Fargo Bank Minnesota, National Association, as trustee (as the same may be amended in compliance with this Agreement) and any supplemental indenture or additional indenture to be entered into with respect to the Senior Subordinated Notes (2015) to the extent permitted under Section 8.11.

"Solvent" means, when used with respect to (i) any Person (other than subject to clause (ii)), that (x) the fair saleable value of its assets is in excess of the total amount of its liabilities (including for purposes of this definition all liabilities, whether or not reflected on a balance sheet prepared in accordance with GAAP, and whether direct or indirect, fixed or contingent, disputed or undisputed), (y) it is able to pay its debts or obligations in the ordinary course as they mature and (z) it has capital sufficient to carry on its business and all business in which it is about to engage and (ii) for any Person other than a Domestic Subsidiary, such Person has the ability to pay its debts as and when they fall due and could not be deemed to be insolvent for the purposes of the law of such Person's jurisdiction of formation. For purposes of Section 6.5(b) "debt" means any liability on a claim, and "claim" means (A) any right to payment, whether or not such a right is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, legal, equitable, secured, or unsecured (including all obligations, if any, under any Plan or the equivalent for unfunded past service liability, and any other unfunded medical and death benefits) or (B) any right to an equitable remedy for breach of performance if such breach gives rise to a payment, whether or not such right to an equitable remedy is reduced to judgment, fixed, contingent, matured, unmatured, disputed, undisputed, secured or unsecured. In computing the amount of contingent or unliquidated liabilities at any time, such liabilities will be computed at the amount which, in light of all the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability.

"Spot Rate" means, for any currency at any date, the rate quoted in Bloomberg World Currency Value as the spot rate for the purchase of such currency with another currency on the applicable determination date.

"S&P" means Standard & Poor's Ratings Service, a division of the McGraw-Hill Companies, Inc., or any successor to the rating agency business thereof.

"Standby Letters of Credit" means any of the irrevocable standby letters of credit issued pursuant to this Agreement, in form acceptable to the Facing Agent issuing such standby letters of credit, together with any increases or decreases in the Stated Amount thereof and any renewals, amendments and/or extensions thereof.

"Stated Amount" or "Stated Amounts" means, (i) with respect to any Letter of Credit issued in Dollars, the stated or face amount of such Letter of Credit to the extent available

at the time for drawing (subject to presentment of all requisite documents), and (ii) with respect to any Letter of Credit issued in any currency other than Dollars, the Assigned Dollar Value of the stated or face amount of such Letter of Credit to the extent available at the time for drawing (subject to presentment of all requisite documents), in either case, as the same may be increased or decreased from time to time in accordance with the terms of such Letter of Credit. For purposes of calculating the Stated Amount of any Letter of Credit at any time:

- (A) any increase in the Stated Amount of any Letter of Credit by reason of any amendment to any Letter of Credit shall be deemed effective under this Agreement as of the date the relevant Facing Agent actually issues an amendment purporting to increase the Stated Amount of such Letter of Credit, whether or not such Facing Agent receives the consent of the Letter of Credit beneficiary or beneficiaries to the amendment, except that if the Borrower has required that the increase in Stated Amount be given effect as of an earlier date and such Facing Agent issues an amendment to that effect, then such increase in Stated Amount shall be deemed effective under this Agreement as of such earlier date requested by the Borrower; and
- (B) any reduction in the Stated Amount of any Letter of Credit by reason of any amendment to any Letter of Credit shall be deemed effective under this Agreement as of the later of (x) the date the applicable Facing Agent actually issues an amendment purporting to reduce the Stated Amount of such Letter of Credit, whether or not the amendment provides that the reduction be given effect as of an earlier date, or (y) the date the applicable Facing Agent receives the written consent (including by authenticated telex, cable, SWIFT messages or facsimile transmission (with, in the case of a facsimile transmission, a follow-up original hard copy)) of the Letter of Credit beneficiary or beneficiaries to such reduction, whether written consent must be dated on or after the date of the amendment issued by such Facing Agent purporting to effect such reduction.

"Stated Termination Date" means March 9, 2014, as the same may be extended from time to time pursuant to

Section 2.11.

"Sterling" means the lawful currency of the United Kingdom.

"Subsidiary" of any Person means any corporation, partnership (limited or general), limited liability company, trust or other entity of which a majority of the Capital Stock (or equivalent beneficial ownership or voting interest) having ordinary voting power to elect a majority of the board of directors (if a corporation) or to select the trustee or equivalent controlling interest, shall, at the time such reference becomes operative, be directly or indirectly owned or controlled by such Person or one or more of the other subsidiaries of such Person or any combination thereof. Unless otherwise qualified, all references to a "Subsidiary" or to "Subsidiaries" in this Agreement shall refer to a Subsidiary or Subsidiaries of the Borrower.

46

Unless otherwise expressly provided, an Unrestricted Subsidiary shall not be considered a "Subsidiary" of the Borrower for purposes of this Agreement.

"Subsidiary Guarantor" means each Subsidiary of the Borrower that is or becomes a party to the Subsidiary Guaranty or delivers a guaranty pursuant to Section 7.11.

"Subsidiary Guaranty" means the guaranty executed by the Subsidiary Guarantors, in form and substance satisfactory to the Administrative Agent, and delivered as of the Closing Date, as the same may be amended, supplemented or otherwise modified from time to time.

"Swing Line Commitment" means, with respect to the Swing Line Lender at any date, the obligation of the Swing Line Lender to make Swing Line Loans pursuant to Section 2.1(c) in the amount referred to therein.

"Swing Line Lender" means JPMCB in such capacity (acting through any branch or Affiliate).

"Swing Line Lender Sublimit" means \$25,000,000.

"Swing Line Loan Participation Certificate" means a certificate, substantially in the form of Exhibit 2.1(c).

"Swing Line Loans" has the meaning assigned to that term in Section 2.1(c)(i).

"Swing Line Note" has the meaning assigned to that term in Section 2.2(a).

"Tax Distributions" has the meaning provided in Section 8.4.

"<u>Tax Sharing Agreement</u>" means that certain tax sharing agreement dated as of the date hereof by and among Borrower, Huntsman Corporation and other Subsidiaries of Huntsman Corporation referred to therein, initially substantially in the form of <u>Exhibit 5.1(v)</u>, as amended or otherwise modified from time to time in accordance with <u>Section 8.11</u>.

"Taxes" has the meaning assigned to that term in Section 4.7(a).

"<u>Technology</u>" has the meaning assigned to that term in <u>Section 8.3(f)</u>.

"Term B Dollar Commitment" means, with respect to any Term B Lender signatory to the Third Amendment, the principal amount set forth opposite such Lender's name on Schedule 1.1(a) hereto or in any Assignment and Assumption Agreement under the caption "Amount of Term B Dollar Commitment", as such commitment may be adjusted from time to time pursuant to this Agreement or increased pursuant to Section 2.1(a)(ii), and "Term B Dollar Commitments" means such commitments collectively, which commitments equal \$1,640,000,000 in the aggregate on the Effective Date.

"<u>Term B Dollar Facility</u>" means the credit facility under this Agreement evidenced by the Term B Dollar Commitments and the Term B Dollar Loans.

47

"<u>Term B Dollar Lender</u>" means any Lender which has a Term B Dollar Commitment or has made (or a portion thereof) a Term B Dollar Loan.

"Term B Dollar Loan" and "Term B Dollar Loans" have the meanings assigned to those terms in Section 2.1(a)(i).

"Term B Dollar Note" and "Term B Dollar Notes" have the meanings assigned to those terms in Section 2.2(a).

"Term B Loan Facility" means the Term B Dollar Facility.

"Term B Loan Maturity Date" means April 19, 2014; provided, that, if there shall occur a date (an "earlier date") on which in excess of \$100,000,000 of Public Notes which have not been repaid or refinanced with Permitted Refinancing Indebtedness in accordance with the terms of this Agreement is scheduled to come due within three months, "Term B Loan Maturity Date" means such earlier date.

"Term C Dollar Commitment" means, with respect to any Term C Dollar Lender signatory to the Fourth Amendment, the principal amount set forth opposite such Lender's name on Schedule 1.1(a) hereto under the caption "Amount of Term C Dollar Commitment", as such commitment may be adjusted from time to time pursuant to this Agreement or increased pursuant to Section 2.1(a) (ii), and "Term C Dollar Commitments" means such commitments collectively, which commitments equal \$500,000,000 in the aggregate on the Fourth Amendment Effective Date.

"<u>Term C Dollar Facility</u>" means the credit facility under this Agreement evidenced by the Term C Dollar Commitments and the Term C Dollar Loans.

"<u>Term C Dollar Lender</u>" means any Lender which has a Term C Dollar Commitment or has made (or a portion thereof) a Term C Dollar Loan.

"Term C Dollar Loan" and "Term C Dollar Loans" have the meanings assigned to those terms in Section 2.1(d).

"Term C Loan Maturity Date" means June 30, 2016.

"Term Commitment" means, with respect to each Lender and any Term Facility, the principal amount set forth opposite such Lender's name on Schedule 1.1(a) hereto or in any Assignment and Assumption Agreement under the caption for the amount of commitment to such Term Facility, as such commitments may be adjusted from time to time pursuant to this Agreement, and "Term Commitments" means such commitments collectively.

"Term Facilities" means the Facilities under the Agreement other than the Revolving Facility.

"Term Loan Ratable Share" shall mean, as of any date of determination, a fraction, the numerator of which is the total outstanding principal amount of Term Loans as of such date and the denominator of which is an amount equal to the sum of (i) the total principal

48

amount of Term Loans outstanding as of such date and (ii) the total principal amount of Senior Secured Notes (2010) outstanding as of such date.

"Term Loans" means the Loans under the Term Facilities, collectively.

"Term Maturity Date" means, with respect to any Term Facility, the scheduled maturity date for such Term Facility under this Agreement.

"Term Note" and "Term Notes" means the Term C Dollar Notes that evidence the Term C Dollar Loans and the notes provided for in Section 2.2 that evidence indebtedness under the Term Facilities, collectively.

"<u>Term Percentage</u>" means, at any time with respect to any Term Facility, a fraction (expressed as a percentage) the numerator of which is equal to the aggregate Assigned Dollar Value of all Loans under such Term Facility outstanding at such time and the denominator of which is equal to the aggregate Assigned Dollar Value of all Term Loans outstanding at such time.

"Test Period" means, at any time the four Fiscal Quarters of the Borrower then last ended.

"<u>TG</u>" means Tioxide Group Unlimited, a direct Subsidiary of the Borrower that is a private unlimited company incorporated under the laws of England and Wales with company number 00249759.

"<u>Thai Holding Companies</u>" means the Domestic Subsidiaries of the Borrower whose sole asset, if any, is an ownership interest in Huntsman (Thailand) Ltd., a corporation organized under the laws of Thailand, and identified as such on <u>Schedule 6.13</u>.

"Third Amendment" means the Third Amendment to this Agreement dated as of April 19, 2007.

"<u>Total Assets</u>" means, at any date of determination, the total assets of the Borrower and its Subsidiaries, as determined in accordance with GAAP at the end of the most recent Fiscal Quarter for which financial statements were delivered pursuant to Section 7.1.

"<u>Total Available Revolving Commitment</u>" means, at the time of any determination thereof is made, the sum of the respective Available Revolving Commitments of the Lenders at such time.

"<u>Total Commitment</u>" means, at the time any determination thereof is made, the sum of the Term Commitments and the Revolving Commitments at such time.

"<u>Total Revolving Commitment</u>" means, at any time, the sum of the Revolving Commitments of each of the Lenders at such time.

"Transferee" has the meaning assigned to that term in Section 12.8(d).

49

"Type" means any type of Loan, namely, a Base Rate Loan or a Eurocurrency Loan.

"UCC" means the Uniform Commercial Code as in effect from time to time in the relevant jurisdiction.

"<u>UK Business Sale</u>" means the sale all of the outstanding equity interests of Huntsman Petrochemicals (UK) Limited to SABIC and the assumption by SABIC of unfunded pension liabilities.

"<u>UK Debenture</u>" has the meaning assigned to that term <u>Section 5.1(b)</u>.

"<u>UK Holdco 1</u>" means Huntsman (Holdings) UK, a direct Wholly-Owned Subsidiary of TG that is a private unlimited company incorporated under the laws of England and Wales with company number 03768308.

"<u>UK Pledge Agreements</u>" means the English law governed charges over shares in respect of the shares held by the Borrower or a Subsidiary Guarantor in TG or UK Holdco 1 granted in favor of the UK Security Trustee (as modified, supplemented or amended from time to time).

"<u>UK Security Trustee</u>" means JPMCB, J.P. Morgan Europe Limited or any other designated affiliate, in its capacity as UK Security Trustee under the UK Security Documents or any successor UK Security Trustee.

"UK Security Documents" means the UK Pledge Agreements and the UK Debenture.

"Unmatured Event of Default" means an event, act or occurrence which with the giving of notice or the lapse of time (or both) would become an Event of Default.

"Unpaid Drawing" has the meaning assigned to that term in Section 2.9(c).

"Unrestricted Investment" has the meaning assigned to that term in Section 8.4.

"Unrestricted Subsidiary" means (i) each of the Persons identified on Schedule 1.1(c) hereto, (ii) any Subsidiary of the Borrower designated as an Unrestricted Subsidiary in the manner provided below, (iii) any Subsidiary of an Unrestricted Subsidiary created at or after the designation of its parent company as an Unrestricted Subsidiary pursuant to clause (ii) above and (iv) any Permitted Unconsolidated Venture that, as a result of an Investment permitted under Section 8.7(j), (o) or (p), would otherwise become a Subsidiary of the Borrower; provided, however, that no Receivables Subsidiary may be an Unrestricted Subsidiary. The Borrower may designate any Subsidiary (including any newly acquired or newly formed Subsidiary but excluding any Receivables Subsidiary) to be an Unrestricted Subsidiary unless such Subsidiary owns any Capital Stock of, or owns or holds any Lien on any property of, the Borrower or any Subsidiary of the Borrower that is not a Subsidiary of the Subsidiary to be so designated; provided, that (x) each Subsidiary to be so designated an Unrestricted Subsidiary and each of its Subsidiaries has not, at the time of designation, and does not thereafter, create, incur, assume,

any of the assets of the Borrower or any of its Subsidiaries (other than Unrestricted Subsidiaries), (y) immediately before and immediately after the effectiveness of such designation, no Unmatured Event of Default or Event of Default exists or will exist and (z) the Investment made in such Unrestricted Subsidiary (valued at the fair market value of the Subsidiary at the time that such Subsidiary is designated an Unrestricted Subsidiary and valued thereafter in accordance with the definition of "Investments") is permitted pursuant to Section 8.7. Any such designation of an Unrestricted Subsidiary shall be evidenced by delivery to the Administrative Agent of (A) a copy of the resolutions of the Borrower giving effect to such designation and (B) an officer's certificate signed by two Responsible Financial Officers of the Borrower certifying that such designation complies with the foregoing provisions. If, at any time, any Unrestricted Subsidiary would fail to meet the requirements set forth in clause (x) above for an Unrestricted Subsidiary, it shall thereafter cease to be an Unrestricted Subsidiary for purposes of this Agreement and the other Loan Documents and Borrower shall take such applicable actions as required by Section 12.20 to redesignate such Unrestricted Subsidiary as a Subsidiary.

"Unrestricted Subsidiary Investment Basket" means, as of any date of determination, an amount equal to the sum of (i) the greater of (A) \$300,000,000 or (B) 3% of Total Assets <u>plus</u> (ii) the after-tax amount of any cash returns of principal or capital on Investments listed on <u>Schedule 1.1(c)</u> hereto or made pursuant to <u>Section 8.7(j)</u>, cash dividends thereon and other cash returns on investment thereon, as the case may be, in each case, after the Effective Date.

"<u>US Commodity Business Sale</u>" means the proposed sale of the Borrower's U.S. base chemicals and polymers business to Flint Hills Resources pursuant to the terms of the US Commodity Business Sale Agreement.

"US Commodity Business Sale Agreement" means that certain Asset Purchase Agreement, dated February 15, 2007 among Flint Hills Resources, LLC, the Borrower, Huntsman Petrochemical Corporation, Huntsman International Chemicals Corporation, Huntsman Polymers Holdings Corporation, Huntsman Expandable Polymers Company, LLC, Huntsman Polymers Corp. and Huntsman Chemical Company of Canada, Inc., together with any amendments or modifications thereto to the extent not materially adverse to the Lenders.

"Voting Securities" means any class of Capital Stock of a Person pursuant to which the holders thereof have, at the time of determination, the general voting power under ordinary circumstances to vote for the election of directors, managers, trustees or general partners of such Person (irrespective of whether or not at the time any other class or classes will have or might have voting power by reason of the happening of any contingency).

"Weighted Average Life to Maturity" means, when applied to any Indebtedness at any date, the number of years obtained by dividing (i) the then outstanding principal amount of such Indebtedness into (ii) the sum of each of the products obtained by multiplying (x) the amount of each then remaining installment, sinking fund, serial maturity or other required payments of principal, including payment at final maturity, in respect thereof by (y) the number

51

of years (calculated to the nearest one-twelfth) that will elapse between such date and the scheduled due date of each such remaining payment.

"Wholly-Owned Subsidiary" means, with respect to any Person, any Subsidiary of such Person, all of the outstanding shares of capital stock of which (other than qualifying shares required to be owned by directors of Foreign Subsidiaries, or similar *de minimis* issuances of capital stock to comply with Requirements of Law) are at the time owned directly or indirectly by such Person and/or one or more Wholly-Owned Subsidiaries of such Person; <u>provided</u>, that each of Huntsman Corporation Hungary Rt. and its Wholly-Owned Subsidiaries shall be deemed to be a Wholly-Owned Subsidiary. For purposes of this definition, 'capital stock' shall include equivalent ownership or controlling interests having ordinary voting power in entities other than corporations.

"written" or "in writing" means any form of written communication or a communication by means of telecopier device or authenticated telex, telegraph or cable.

The foregoing definitions shall be equally applicable to both the singular and plural forms of the defined terms. In the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including" and the words "to" and "until" each mean "to but excluding." The words "herein," "hereof" and words of similar import as used in this Agreement shall refer to this Agreement as a whole and not to any particular provision in this Agreement. References to "Articles", "Sections", "paragraphs", "Exhibits" and "Schedules" in this Agreement shall refer to Articles, Sections, paragraphs, Exhibits and Schedules of this Agreement unless otherwise expressly provided; references to Persons include their respective permitted successors and assigns or, in the case of governmental Persons, Persons succeeding to the relevant functions of such persons; and all references to statutes and related regulations shall include any amendments of same and any successor statutes and regulations. Unless otherwise expressly provided herein, references to constitutive and Organizational Documents and to agreements (including the Loan Documents) and other contractual instruments shall be deemed to include subsequent amendments, restatements, extensions, supplements and other modifications thereto, but only to the extent that such amendments, restatements, extensions, supplements and other modifications are not prohibited by any Loan Document.

9.2. Accounting Terms; Financial Statements

A. Except as otherwise expressly provided herein (including without limitation, any modification to the terms hereof pursuant to <u>Section 8.12</u>), all accounting terms used herein but not expressly defined in this Agreement and all computations and determinations for purposes of determining compliance with the financial requirements of this Agreement shall be made in accordance with GAAP in effect on the date hereof and on a basis consistent with the presentation of the financial statements and projections delivered pursuant to, or otherwise referred to in, <u>Sections 6.5(a)</u> and <u>6.5(e)</u>. Notwithstanding the foregoing sentence, the financial

attributable to any Person that is not a Subsidiary of such Person. For purposes of the financial terms set forth herein, whenever a reference is made to a determination which is required to be made on a consolidated basis (whether in accordance with GAAP or otherwise) for the Borrower and its Subsidiaries, such determination shall be made as if each Unrestricted Subsidiary were wholly-owned by a Person not an Affiliate of the Borrower.

- B. Solely for purposes of delivery of the financial statements required by Sections 6.5, 7.1(a) and (b), Unrestricted Subsidiaries shall be deemed to be Subsidiaries; provided, however, concurrently with the delivery of the officer's certificate required by Section 7.2(b), for purposes of calculating compliance with the financial covenants hereof, Borrower shall also deliver to Administrative Agent statements excluding the Unrestricted Subsidiaries, and, upon request of Administrative Agent, reflecting the relevant calculations pertaining to the Unrestricted Subsidiaries existing on the Effective Date, in each case in form and substance satisfactory to the Administrative Agent.
- C. Without limiting the definition of "Pro Forma Basis", for purposes of computing the financial ratios as of the end of any period, all components of such ratios for the applicable period shall include or exclude, as the case may be, without duplication, such components of such ratios attributable to any business or assets that have been acquired or disposed of by the Borrower or any Subsidiary of the Borrower (including through mergers or consolidations) after the first day of such period and prior to the end of such period on a pro forma basis, as if such acquisition or disposition had occurred on the first day of such period, as determined in good faith by the Borrower and certified to by a Responsible Officer of the Borrower to the Administrative Agent.

SECTION 10

AMOUNT AND TERMS OF CREDIT

10.1. The Commitments

A. Term B Loan.

- (a) Subject to the terms and conditions hereof and in the Third Amendment, each Term B Dollar Lender agrees to make a loan in Dollars (the "Term B Dollar Loans") to the Borrower on the Third Amendment Effective Date in the aggregate principal amount of such Lender's Term B Dollar Commitment. No amount of a Term B Loan which is repaid or prepaid by the Borrower may be reborrowed hereunder. The Term B Dollar Loans shall be denominated in Dollars, shall be maintained as and/or converted into Base Rate Loans or Eurocurrency Loans or a combination thereof, provided, that all Term B Dollar Loans made by the Term B Dollar Lenders pursuant to the same Borrowing shall, unless otherwise specifically provided herein, consist entirely of Term B Dollar Loans of the same Type.
- (b) (A) The Borrower shall have the right at any time following the date on which the Administrative Agent notifies the Borrower that the initial

53

syndication of the Loans and Commitments with respect to this Agreement has occurred to the Administrative Agent's satisfaction (so long as (x) no Unmatured Event of Default or Event of Default then exists and is continuing, (y) the Borrower shall have delivered to Administrative Agent a Compliance Certificate for the period of four full Fiscal Quarters immediately preceding the incurrence described below (prepared in good faith and in a manner and using such methodology which is consistent with the most recent financial statements delivered pursuant to Section 7.1) giving pro forma effect to such incurrence and evidencing compliance with the covenant set forth in Article IX) and (z) such incurrence is not prohibited by the terms of any Public Note Document), to incur from one or more existing Lenders and/or other Persons that are Eligible Assignees and which, in each case, agree to make such term loans to the Borrower, Additional Term Loans in Dollars or Euros (in maximum amounts described below), which loans may be incurred as one or more tranches of additional term loans as determined by Administrative Agent in an aggregate principal amount not to exceed \$1,000,000,000 so long as the proceeds of such Additional Term Loans are used solely for (i) capital expenditures, (ii) repayment of secured Indebtedness of the Borrower, (iii) repayment of Senior Notes (2012) and (iv) to finance Acquisitions permitted by Section 8.3(b) hereof.

(B) In the event that the Borrower desires to incur Additional Term Loans, the Borrower will enter into an amendment with the lenders (who shall by execution thereof become Lenders hereunder if not theretofore Lenders) to provide for such Additional Term Loans, which amendment shall set forth any terms and conditions of the Additional Term Loans not covered by this Agreement as agreed by the Borrower and such Lenders, and shall provide for the issuance of promissory notes to evidence the Additional Term Loans if requested by the lenders advancing Additional Term Loans (which notes shall constitute Term Notes for purposes of this Agreement) and for such reaffirmations of or amendments to Security Documents as Administrative Agent may reasonably require, with such amendment to be in form and substance reasonably acceptable to Administrative Agent and consistent with the terms of this Section 2.1(a)(ii) and

of the other provisions of this Agreement. No consent of any Lender (other than any Lender making Additional Term Loans) is required to permit the Loans contemplated by this <u>Section 2.1(a)(ii)</u> or the aforesaid amendment to effectuate the Additional Term Loans. This section shall supersede any provisions contained in this Agreement, including, without limitation, <u>Section 12.1</u>, to the contrary.

B. Revolving Loans. Each Revolving Lender, severally and for itself alone, hereby agrees, on the terms and subject to the conditions hereinafter set forth and in reliance upon the representations and warranties set forth herein and in the other Loan Documents, to make loans to the Borrower denominated in Dollars or an Alternative Currency on a revolving basis from time to time during the Commitment Period, in an amount that will not (i) exceed its Pro Rata Share of the Total Available Revolving Commitment (each such loan by any Lender, a "Revolving Loan" and collectively, the "Revolving Loans") or (ii) cause the Assigned Dollar Value of Revolving Loans, Swing Line Loans and LC Obligations denominated in an Alternative

54

Currency to exceed \$200,000,000. All Revolving Loans comprising the same Borrowing hereunder shall be made by the Revolving Lenders simultaneously and in proportion to their respective Revolving Commitments. Prior to the Revolver Termination Date, Revolving Loans may be repaid and reborrowed by the Borrower in accordance with the provisions hereof and, except as otherwise specifically provided in Section 3.6, all Revolving Loans comprising the same Borrowing shall at all times be of the same Type. On the Revolver Termination Date, 100% of the aggregate principal amount of Revolving Loans then outstanding shall be due and payable.

C. Swing Line Loans

- (a) Swing Line Commitment. Subject to the terms and conditions hereof, the Swing Line Lender in its individual capacity agrees to make swing line loans in Dollars, Euros or Sterling (or, at the option of the Swing Line Lender, any other Alternative Currency) ("Swing Line Loans") to the Borrower on any Business Day from time to time during the Commitment Period in an aggregate principal amount at any one time outstanding not to exceed the Dollar Equivalent of the Swing Line Lender Sublimit; provided, however, that in no event may the amount of any Borrowing of Swing Line Loans (A) exceed the Total Available Revolving Commitment immediately prior to such Borrowing (after giving effect to the use of proceeds thereof), (B) cause the outstanding Revolving Loans of any Lender, when added to such Lender's Pro Rata Share of the then outstanding Swing Line Loans and Pro Rata Share of the aggregate LC Obligations (exclusive of Unpaid Drawings relating to LC Obligations which are repaid with the proceeds of, and simultaneously with the incurrence of, Revolving Loans or Swing Line Loans) to exceed such Lender's Revolving Commitment or (C) cause the Assigned Dollar Value of Revolving Loans, Swing Line Loans and LC Obligations denominated in an Alternative Currency to exceed \$150,000,000. Amounts borrowed by the Borrower under this Section 2.1(c)(i) may be repaid and reborrowed until the Revolver Termination Date.
- (b) Refunding of Swing Line Loans. The Swing Line Lender, at any time in its sole and absolute discretion, may on behalf of the Borrower (which hereby irrevocably directs the Swing Line Lender to so act on its behalf) notify each Revolving Lender (including the Swing Line Lender) to make a Revolving Loan in an amount equal to such Lender's Pro Rata Share of the Dollar Equivalent of the principal amount of such Swing Line Loans (the "Refunded Swing Line Loans") outstanding on the date such notice is given, provided, however, that such notice shall be deemed to have automatically been given upon the occurrence of an Event of Default under Sections 10.1(e) or 10.1(f) or upon the occurrence of a Change of Control. Unless any of the events described in Sections 10.1(e) or 10.1(f) shall have occurred (in which event the procedures of Section 2.1(c)(iii) shall apply) and regardless of whether the conditions precedent set forth in this Agreement to the making of a Revolving Loan are then satisfied, each Lender shall make the proceeds of its Revolving Loan available to the Swing Line Lender at the Payment Office prior to 11:00 a.m., New York City time, in funds immediately available on the third Business Day next succeeding the date such

55

notice is given. The proceeds of such Revolving Loans shall be immediately applied to repay the Refunded Swing Line Loans.

- (c) Participation in Swing Line Loans. If, prior to refunding a Swing Line Loan with a Revolving Loan pursuant to Section 2.1(c)(ii), one of the events described in Sections 10.1(e) or 10.1(f) shall have occurred, or if for any other reason a Revolving Loan cannot be made pursuant to Section 2.1(c)(ii), then, subject to the provisions of Section 2.1(c)(iv) below, each Revolving Lender will, on the date such Revolving Loan was to have been made, purchase (without recourse or warranty) from the Swing Line Lender an undivided participation interest in the Swing Line Loan in an amount equal to its Pro Rata Share of the Dollar Equivalent of such Swing Line Loan. Upon request, each Revolving Lender will immediately transfer to the Swing Line Lender, in immediately available funds, the amount of its participation and upon receipt thereof the Swing Line Lender will deliver to such Revolving Lender a Swing Line Loan Participation Certificate dated the date of receipt of such funds and in such amount.
- (d) <u>Lenders' Obligations Unconditional</u>. Each Revolving Lender's obligation to make Revolving Loans in accordance with <u>Section 2.1(c)(ii)</u> and to purchase participating interests in accordance with <u>Section 2.1(c)(iii)</u> above shall, except as set forth in the last sentence of this clause (iv), be absolute and unconditional and shall not, except as set forth in the last sentence of this clause (iv), be affected by any circumstance, including, without limitation, (A) any set-off, counterclaim, recoupment, defense or other right which such Revolving Lender may have against the Swing Line

Lender, the Borrower or any other Person for any reason whatsoever; (B) the occurrence or continuance of any Event of Default or Unmatured Event of Default; (C) any adverse change in the condition (financial or otherwise) of the Borrower or any other Person; (D) any breach of this Agreement by the Borrower or any other Person; (E) any inability of the Borrower to satisfy the conditions precedent to borrowing set forth in this Agreement on the date upon which such participating interest is to be purchased or (F) any other circumstance, happening or event whatsoever, whether or not similar to any of the foregoing. If any Revolving Lender does not make available to the Swing Line Lender the amount required pursuant to Section 2.1(c)(ii) or (iii) above, as the case may be, the Swing Line Lender shall be entitled to recover such amount on demand from such Revolving Lender, together with interest thereon for each day from the date of non-payment until such amount is paid in full at the Federal Funds Rate for the first two Business Days and at the Base Rate thereafter. Notwithstanding the foregoing provisions of this Section 2.1(c)(iv), no Revolving Lender shall be required to make a Revolving Loan to the Borrower for the purpose of refunding a Swing Line Loan pursuant to Section 2.1(c)(iii) above or to purchase a participating interest in a Swing Line Loan pursuant to Section 2.1(c)(iii) if an Event of Default or Unmatured Event of Default has occurred and is continuing and, prior to the making by the Swing Line Lender of such Swing Line Loan, the Swing Line Lender has received written notice from such Revolving Lender specifying that such Event of Default or Unmatured Event

56

of Default has occurred and is continuing, describing the nature thereof and stating that, as a result thereof, such Revolving Lender shall cease to make such Refunded Swing Line Loans and purchase such participating interests, as the case may be; <u>provided</u>, <u>however</u>, that the obligation of such Revolving Lender to make such Refunded Swing Line Loans and to purchase such participating interests shall be reinstated upon the earlier to occur of (y) the date upon which such Revolving Lender notifies the Swing Line Lender that its prior notice has been withdrawn and (z) the date upon which the Event of Default or Unmatured Event of Default specified in such notice no longer is continuing.

D. Term C Loans. Subject to the terms and conditions hereof and in the Fourth Amendment, each Term C Dollar Lender agrees to make a loan in Dollars (the "Term C Dollar Loans") to the Borrower on the Fourth Amendment Effective Date in the aggregate principal amount of such Lender's Term C Dollar Commitment. No amount of a Term C Dollar Loan which is repaid or prepaid by the Borrower may be reborrowed hereunder. The Term C Dollar Loans shall be denominated in Dollars, shall be maintained as and/or converted into Base Rate Loans or Eurocurrency Loans or a combination thereof, provided, that all Term C Dollar Loans made by the Term C Dollar Lenders pursuant to the same Borrowing shall, unless otherwise specifically provided herein, consist entirely of Term C Dollar Loans of the same Type.

10.2. Notes

A. Evidence of Indebtedness. The Borrower's obligation to pay the principal of and interest on all the Loans made to it by each Lender shall, if requested by a Lender, be evidenced, (1) if Term B Dollar Loans, by a promissory note (each, a "Term B Dollar Note" and, collectively, the "Term B Dollar Notes") duly executed and delivered by the Borrower substantially in the form of Exhibit 2.2(a)(1) hereto, with blanks appropriately completed in conformity herewith, (2) if Revolving Loans, by a promissory note (each, a "Revolving Note" and, collectively, the "Revolving Notes") duly executed and delivered by the Borrower substantially in the form of Exhibit 2.2(a)(2) hereto, with blanks appropriately completed in conformity herewith and (3) if Swing Line Loans, by a promissory note (each, a "Swing Line Note" and, collectively, the "Swing Line Notes") duly executed and delivered by the Borrower substantially in the form of Exhibit 2.2(a)(3) hereto, with blanks appropriately completed in conformity herewith.

B. <u>Notation of Payments</u>. Each Lender will note on its internal records the amount of each Loan made by it, and each payment in respect thereof and will, prior to any transfer of any of its Notes, endorse on the reverse side thereof the outstanding principal amount of Loans evidenced thereby. Failure to make any such notation, or any error in any such notation, shall not affect the Borrower's or any guarantor's obligations hereunder or under the other applicable Loan Documents in respect of such Loans.

10.3. Minimum Amount of Each Borrowing; Maximum Number of Borrowings

The aggregate principal amount of each Borrowing by the Borrower hereunder shall be not less than the Minimum Borrowing Amount and, if greater (other than with respect to

57

Swing Line Loans), shall be in integral multiples of (i) in the case of a Borrowing in Dollars, \$1,000,000, (ii) in the case of a Borrowing in Sterling, £750,000, or (iii) in the case of a Borrowing in Euros, epsilon1,000,000, above such minimum (or, if less, the then Total Available Revolving Commitment). More than one Borrowing may be incurred on any date; <u>provided</u> that at no time shall there be outstanding more than six Borrowings of Eurocurrency Loans for any Facility.

10.4. Interest Rate Options

The Term Loans and the Revolving Loans shall, at the option of the Borrower except as otherwise provided in this Agreement, be (i) Base Rate Loans, (ii) Eurocurrency Loans, or (iii) part Base Rate Loans and part Eurocurrency Loans; provided, that non-Dollar denominated Revolving Loans may only be made as Eurocurrency Loans. Dollar denominated Swing Line Loans shall be made only as Base Rate Loans. Non-Dollar denominated Swing Line Loans shall bear interest at the applicable Quoted Rate and shall have such interest periods and interest payment dates as are determined by the Swing Line Lender and the Borrower at the time of

Borrowing thereof. As to any Eurocurrency Loan, any Lender may, if it so elects, fulfill its commitment by causing a foreign branch or affiliate to make or continue such Loan, provided that in such event that Lender's Loan shall, for the purposes of this Agreement, be considered to have been made by that Lender and the obligation of the Borrower to repay that Lender's Loan shall nevertheless be to that Lender and shall be deemed held by that Lender, for the account of such branch or affiliate.

10.5. Notice of Borrowing

Whenever the Borrower desires to make a Borrowing of any Loan hereunder, it shall give the Administrative Agent at its office located at the Notice Office (or such other address as the Administrative Agent may hereafter designate in writing to the parties hereto) at least one Business Day's prior written notice (or telephonic notice promptly confirmed in writing), given not later than 1:00 p.m. (New York City time) of each Base Rate Loan, and at least three Business Days' prior written notice (or telephonic notice promptly confirmed in writing), given not later than 1:00 p.m. (New York City time), of each Dollar denominated Eurocurrency Loan to be made hereunder and at least four Business Days prior written notice (or telephone notice promptly confirmed in writing) given not later than 1:00 p.m. (New York time), of each Loan denominated in an Alternative Currency; provided, however, that a Notice of Borrowing with respect to the Initial Borrowing may, at the discretion of the Administrative Agent, be delivered later than the time specified above. Whenever the Borrower desires that the Swing Line Lender make a Swing Line Loan under Section 2.1(c), it shall deliver to the Swing Line Lender prior to 1:00 p.m. (New York City time) on the date of Borrowing (at least two Business Days prior to the date of Borrowing for Swing Line Loans denominated in an Alternative Currency) written notice (or telephonic notice promptly confirmed in writing). Each such notice (each a "Notice of Borrowing"), which shall be in the form of Exhibit 2.5 hereto, shall be irrevocable, shall be deemed a representation by the Borrower that all conditions precedent to such Borrowing have been satisfied and shall specify (i) the aggregate principal amount of the Loans to be made pursuant to such Borrowing, (ii) the date of Borrowing (which shall be a Business Day), (iii) whether the Loans being made pursuant to such Borrowing are to be Base Rate Loans or Eurocurrency Loans or both, (iv) with respect to Eurocurrency Loans, the

58

Interest Period to be applicable thereto, (v) with respect to Revolving Loans, the amount of the Overdraft Reserve at such time and (vi) the account details of the Borrower. The Administrative Agent shall as promptly as practicable give each Lender written or telephonic notice (promptly confirmed in writing) of each proposed Borrowing, of such Lender's Pro Rata Share thereof and of the other matters covered by the Notice of Borrowing. Without in any way limiting the Borrower's obligation to confirm in writing any telephonic notice, the Administrative Agent or the Swing Line Lender (in the case of Swing Line Loans) may act without liability upon the basis of telephonic notice believed by the Administrative Agent in good faith to be from a Responsible Officer of the Borrower prior to receipt of written confirmation. The Administrative Agent's records shall, absent manifest error, be final, conclusive and binding on the Borrower with respect to evidence of the terms of such telephonic Notice of Borrowing.

10.6. Conversion or Continuation

With respect to Dollar denominated Loans (other than Swing Line Loans), the Borrower may elect (i) on any Business Day at any time after the earlier of (x) the third Business Day following the Closing Date and (y) the date the Administrative Agent notifies the Borrower that Base Rate Loans or any portion thereof may be converted to Eurocurrency Loans and (ii) at the end of any Interest Period with respect thereto, to convert Eurocurrency Loans or any portion thereof into Base Rate Loans or to continue such Eurocurrency Loans or any portion thereof for an additional Interest Period; provided, however, that the aggregate principal amount of the Eurocurrency Loans that will, upon such conversion, constitute a single Borrowing that must satisfy Section 2.3. With respect to Euro or Sterling denominated Loans, the Borrower may elect, in the same manner as described for conversions, to continue such Eurocurrency Loans or any portion thereof for an additional Interest Period. Each conversion or continuation of Loans of each applicable Facility shall be allocated among the Loans of the Lenders for such Facility in accordance with their respective Pro Rata Shares. Each such election shall be in substantially the form of Exhibit 2.6 hereto (a "Notice of Conversion or Continuation") and shall be made by giving the Administrative Agent at least three Business Days' (or one Business Day in the case of a conversion into Base Rate Loans or four Business Days' in the case of continuation of a non-Dollar denominated Revolving Loan) prior written notice thereof to the Notice Office given not later than 1:00 p.m. (New York City time) specifying (i) the amount and type of conversion or continuation, (ii) in the case of a conversion to or a continuation of Eurocurrency Loans, the Interest Period therefor, and (iii) in the case of a conversion, the date of conversion (which date shall be a Business Day and, if a conversion from Eurocurrency Loans, shall also be the last day of the Interest Period therefor). Notwithstanding the foregoing, no conversion in whole or in part of Base Rate Loans to Eurocurrency Loans, and no continuation in whole or in part of Dollar denominated Eurocurrency Loans upon the expiration of any Interest Period therefor, shall be permitted at any time at which an Unmatured Event of Default or an Event of Default shall have occurred and be continuing. The Borrower shall not be entitled to specify an Interest Period in excess of one month for any non-Dollar denominated Revolving Loan if an Unmatured Event of Default or an Event of Default has occurred and is continuing. If, within the time period required under the terms of this Section 2.6, the Administrative Agent does not receive a Notice of Conversion or Continuation from the Borrower containing a permitted election to continue any Eurocurrency Loans for an additional Interest Period or to convert any such Loans, then, upon the expiration of the Interest Period therefor, such Loans will be automatically converted to Base Rate Loans or, in the case of non-Dollar denominated Revolving Loans, Eurocurrency

59

Loans with an Interest Period of one month. Each Notice of Conversion or Continuation shall be irrevocable.

No later than 1:00 p.m. (local time at the place of funding) on the date specified in each Notice of Borrowing, each Lender will make available its Pro Rata Share of Loans, to fund the Borrowing requested to be made on such date in Dollars, Euro or Sterling, as the case may be, and in immediately available funds, at the Payment Office (for the account of such non-U.S. office of the Administrative Agent as the Administrative Agent may direct in the case of Eurocurrency Loans) and the Administrative Agent will make available to the Borrower at its Payment Office the aggregate of the amounts so made available by the Lenders not later than 2:00 p.m. (local time in the place of payment). Unless the Administrative Agent shall have been notified by any Lender at least one Business Day prior to the date of Borrowing that such Lender does not intend to make available to the Administrative Agent such Lender's portion of the Borrowing to be made on such date, the Administrative Agent may assume that such Lender has made such amount available to the Administrative Agent on such date of Borrowing and the Administrative Agent may, but shall not be required to, in reliance upon such assumption, make available to the Borrower a corresponding amount. If such corresponding amount is not in fact made available to the Administrative Agent by such Lender on the date of Borrowing, the Administrative Agent shall be entitled to recover such corresponding amount on demand from such Lender. If such Lender does not pay such corresponding amount forthwith upon the Administrative Agent's demand therefor, the Administrative Agent shall promptly notify the Borrower and, if so notified, the Borrower shall immediately pay such corresponding amount to the Administrative Agent. The Administrative Agent shall also be entitled to recover from the Borrower interest on such corresponding amount in respect of each day from the date such corresponding amount was made available by the Administrative Agent to the Borrower to the date such corresponding amount is recovered by the Administrative Agent, at a rate per annum equal to the rate for Base Rate Loans or Eurocurrency Loans, applicable to the Type of Loan to which such corresponding amount related, for the period in question; provided, however, that any interest paid to the Administrative Agent in respect of such corresponding amount shall be credited against interest payable by the Borrower to such Lender under Section 3.1 in respect of such corresponding amount. Any amount due hereunder to the Administrative Agent from any Lender which is not paid when due shall bear interest payable by such Lender, from the date due until the date paid, at the Federal Funds Rate for amounts in Dollars (and at the Administrative Agent's cost of funds for amounts in Euros or Sterling or any other Alternative Currency) for the first three days after the date such amount is due and thereafter at the Federal Funds Rate (or such cost of funds rate) plus 1%, together with the Administrative Agent's standard interbank processing fee. Further, such Lender shall be deemed to have assigned any and all payments made of principal and interest on its Loans, amounts due with respect to its Letters of Credit (or its participations therein) and any other amounts due to it hereunder first to the Administrative Agent to fund any outstanding Loans made available on behalf of such Lender by the Administrative Agent pursuant to this Section 2.7 until such Loans have been funded (as a result of such assignment or otherwise) and then to fund Loans of all Lenders other than such Lender until each Lender has outstanding Loans equal to its Pro Rata Share of all Loans (as a result of such assignment or otherwise). Such Lender shall not have recourse against the Borrower with respect to any amounts paid to the Administrative Agent or any Lender with respect to the

60

preceding sentence; <u>provided</u>, that such Lender shall have full recourse against the Borrower to the extent of the amount of such Loans such Lender has been deemed to have made pursuant to the preceding sentence. Nothing herein shall be deemed to relieve any Lender from its obligation to fulfill its Commitment hereunder or to prejudice any rights which the Borrower may have against the Lender as a result of any default by such Lender hereunder.

10.8. Pro Rata Borrowings

All Borrowings in respect of Loans (other than Swing Line Loans) under this Agreement shall be advanced by the Lenders pro rata on the basis of their Commitments to the applicable Facilities. No Lender shall be responsible for the failure of any Defaulting Lender or Impaired Lender to make a Loan required under this Agreement and each Lender shall be obligated to make the Loans provided to be made by it hereunder, regardless of the occurrence of any Lender becoming a Defaulting Lender or an Impaired Lender.

10.9. Amount and Terms of Letters of Credit

A. <u>Letter of Credit Commitments, Terms of Letters of Credit.</u>

Subject to and upon the terms and conditions herein set forth, at any time and from time to time on or after the Closing Date until a date which is thirty (30) days prior to the Revolver Termination Date, each Facing Agent (other than DB) agrees, severally not jointly, to issue each in its own name, but for the ratable account of all Revolving Lenders (including the applicable Facing Agent), one or more Letters of Credit, denominated in Dollars or an Alternative Currency, for the account of the Borrower in a Stated Amount which together with the aggregate Stated Amount of all other Letters of Credit (other than Fifth Amendment Existing Letters of Credit) then outstanding does not exceed the Facing Agent Sublimit; provided, however, that a Facing Agent shall not issue or extend the expiration of any Letter of Credit if, immediately after giving effect to such issuance or extension, (A) the sum of the Assigned Dollar Value of the Revolving Loans, the Swing Line Loans and the LC Obligations would exceed the Total Revolving Commitment minus the Overdraft Reserve and (B) the sum of the Assigned Dollar Value of the LC Obligations would exceed the Total Revolving Commitment on the date of such issuance or extension. Each Revolving Lender severally, but not jointly, agrees to participate in each such Letter of Credit issued by the applicable Facing Agent in an amount equal to its Pro Rata Share and to make available to the applicable Facing Agent such Lender's Pro Rata Share of any payment made to the beneficiary of such Letter of Credit to the extent not reimbursed by the Borrower; provided, however, that no Revolving Lender shall be required to participate in any Letter of Credit to the extent that such participation therein would exceed such Revolving Lender's then applicable Available Revolving Commitment. No Lender's obligation to participate in any Letter of Credit or to make available to the applicable Facing Agent such Revolving Lender's Pro Rata Share of any Letter of Credit Payment made by the applicable Facing Agent shall be affected by any other Revolving Lender's failure to participate in the same or any other Letter of Credit or by any

other Revolving Lender's failure to make available to the applicable Facing Agent such other Lender's Pro Rata Share of any Letter of Credit Payment.

- (b) Each Letter of Credit issued or to be issued hereunder shall be issued on a sight basis, and (other than Bank Guarantees) shall have an expiration date of one (1) year or less from the issuance date thereof; provided, however, that each Standby Letter of Credit may provide by its terms that it will be automatically extended for additional successive periods of up to one (1) year unless the applicable Facing Agent shall have given notice to the applicable beneficiary (with a copy to the Borrower) of the election by the applicable Facing Agent (such election to be in the sole and absolute discretion of the applicable Facing Agent) not to extend such Letter of Credit, such notice to be given prior to the then current expiration date of such Letter of Credit; provided, further, that no Standby Letter of Credit or extension thereof shall be stated to expire later than the date five (5) days prior to the Revolver Termination Date and no Commercial Letter of Credit or extension thereof shall be stated to expire later than the day thirty (30) days prior to the Revolver Termination Date.
- Procedure for Issuance and Amendment of Letters of Credit. Whenever the Borrower desires the issuance of a Letter of Credit hereunder, it shall give the Administrative Agent and the applicable Facing Agent at least three (3) Business Days' prior written notice (or such shorter period as may be agreed to by the Borrower, the Administrative Agent and the applicable Facing Agent) specifying the day of issuance thereof (which day shall be a Business Day), such notice to be given prior to 12:00 p.m. (New York time) on the date specified for the giving of such notice. Each such notice (each, a "Notice of Issuance") shall be in the form of Exhibit 2.9(b) hereto and shall specify (A) the proposed issuance date and expiration date, (B) whether such Letter of Credit is to be issued as a letter of credit or bank guarantee. (C) the Borrower as the account party and, if desired by the Borrower, one or more Subsidiaries as additional account parties, (D) the name and address of the beneficiary, (E) the Stated Amount of such proposed Letter of Credit, (F) the currency in which such proposed Letter of Credit is to be issued and (G) such other information as the applicable Facing Agent may reasonably request. In addition, each Notice of Issuance shall contain a description of the terms and conditions to be included in such proposed Letter of Credit (all of which terms and conditions shall be acceptable in form to the applicable Facing Agent). In the event of any inconsistency between the terms and conditions of this Agreement and the terms and conditions of any form of letter of credit application or other agreement submitted by the Borrower to, or entered into by the Borrower with, a Facing Agent (other than DB) relating to any Letter of Credit issued by it, the terms and conditions of this Agreement shall control. Promptly after issuance or extension of any Letter of Credit, the Borrower and the applicable Facing Agent (to the extent such Facing Agent is not the same entity as, or an Affiliate of, the Administrative Agent) shall notify the Administrative Agent of such issuance or extension and such notice of a Standby Letter of Credit shall be accompanied by a copy of the Standby Letter of Credit. Unless otherwise specified, all Standby Letters of Credit and Commercial Letters of Credit will be governed by the "Uniform Customs and Practice for Documentary Credits" and all Bank Guarantees will be governed by the "Uniform Rules for Demand Guarantees", in each case as issued by the International Chamber of Commerce and as in effect on the date of issuance of such Letter of Credit. On the Business Day specified by the Borrower and upon confirmation from the Administrative Agent that the

62

applicable conditions set forth in Article V have been fulfilled or waived, the applicable Facing Agent will issue the requested Letter of Credit to the applicable beneficiary. From time to time while a Letter of Credit is outstanding and prior to the Revolver Termination Date, the applicable Facing Agent will, upon the written request of the Borrower received by the Facing Agent (with a copy sent by the Borrower to the Administrative Agent) at least three Business Days (or such shorter time as the Facing Agent and the Administrative Agent may agree in a particular instance in their sole discretion) prior to the proposed date of amendment, amend any Letter of Credit issued by it. Each such request for amendment of a Letter of Credit shall be made by facsimile, confirmed immediately in an original writing and shall specify in form and detail satisfactory to the Facing Agent: (i) the Letter of Credit to be amended; (ii) the proposed date of amendment of the Letter of Credit (which shall be a Business Day); (iii) the nature of the proposed amendment; and (iv) such other matters as the Facing Agent may require. The Facing Agent shall be under no obligation to amend any Letter of Credit if: (A) the Facing Agent would have no obligation at such time to issue such Letter of Credit in its amended form under the terms of this Agreement, or (B) the beneficiary of any such Letter of Credit does not accept the proposed amendment to the Letter of Credit. The Facing Agent will provide a copy of any amendment to the Administrative Agent and the Administrative Agent will promptly notify the Lenders of the receipt by it of any amendment to a Letter of Credit. In the event that the Facing Agent is other than the Administrative Agent, such Facing Agent will send by facsimile transmission to the Administrative Agent, promptly on the first Business Day of each week its daily maximum Dollar Equivalent amount available to be drawn under the Commercial Letters of Credit issued by such Facing Agent for the previous week. The Administrative Agent shall deliver to each Lender upon such calendar month end, and upon each commercial letter of credit fee payment, a report setting forth the daily maximum Dollar Equivalent amount available to be drawn for all Facing Agents during such period.

Credit by the beneficiary thereof, the applicable Facing Agent shall give telephonic notice to the Borrower and the Administrative Agent (x) confirming such drawing and (y) of the date on or before which such Facing Agent intends to honor such drawing, and the Borrower shall reimburse the applicable Facing Agent on the day on which such drawing is honored in an amount in same day funds equal to the amount of such drawing (each such amount so paid until reimbursed, an "Unpaid Drawing"); provided, however, that, anything contained in this Agreement to the contrary notwithstanding, (i) unless the Borrower shall have notified the Administrative Agent and the applicable Facing Agent prior to 10:00 a.m. (New York City time) on the Business Day on which the applicable Facing Agent intends to honor such drawing that the Borrower intends to reimburse the applicable Facing Agent for the amount of such drawing with funds other than the proceeds of Revolving Loans, the Borrower shall be deemed to have timely given a Notice of Borrowing to the Administrative Agent requesting each Revolving Lender to make Dollar denominated Base Rate Loans on the date on which such drawing is honored in an amount equal to the Dollar Equivalent of the amount of such drawing and the Administrative Agent shall, if such Notice of Borrowing is

63

applicable Facing Agent for the amount of such drawing; and <u>provided</u>, <u>further</u>, that, if for any reason, proceeds of Revolving Loans are not received by the applicable Facing Agent on such date in an amount equal to the amount of the Dollar Equivalent of such drawing, the Borrower shall reimburse the applicable Facing Agent, on the Business Day immediately following the date such drawing is honored, in an amount in same day funds equal to the excess of the amount of the Dollar Equivalent of such drawing over the Dollar Equivalent of the amount of such Revolving Loans, if any, which are so received, plus accrued interest on such amount at the rate set forth in <u>Section 3.1(a)</u>.

Lenders' Participation in Letters of Credit. In the event that (1) the Borrower shall fail to reimburse the applicable Facing Agent as provided in Section 2.9(c) in an amount equal to the amount of any drawing honored by the applicable Facing Agent under a Letter of Credit issued by it in accordance with the terms hereof, or (2) any Bank Guarantee shall remain outstanding on the Revolver Termination Date, the applicable Facing Agent shall promptly notify the Administrative Agent and the Administrative Agent shall promptly notify each Revolving Lender, of the unreimbursed amount of such drawing or the amount of such Bank Guarantee, as applicable, and of such Lender's respective participation therein. Each such Revolving Lender shall purchase a participation interest in such LC Obligation and shall make available to the applicable Facing Agent the Dollar Equivalent of an amount equal to its Pro Rata Share of such drawing in same day funds, at the office of the applicable Facing Agent specified in such notice, in each case not later than 1:00 p.m. (New York City time) on the Business Day after the date such Lender is notified by the Administrative Agent. In the event that any such Lender fails to make available to the applicable Facing Agent the amount of such Lender's participation in such Letter of Credit as provided in this Section 2.9(d), the applicable Facing Agent shall be entitled to recover such amount on demand from such Lender together with interest at the Federal Funds Rate for two Business Days and thereafter at the Base Rate. Nothing in this Section 2.9(d) shall be deemed to prejudice the right of any Lender to recover from the applicable Facing Agent any amounts made available by such Lender to the applicable Facing Agent pursuant to this Section 2.9(d) in the event that it is determined by a court of competent jurisdiction that the payment with respect to a Letter of Credit by the applicable Facing Agent in respect of which payment was made by such Lender constituted gross negligence or willful misconduct on the part of the applicable Facing Agent. The applicable Facing Agent shall distribute to each other Lender which has paid all amounts payable by it under this Section 2.9(d) with respect to any Letter of Credit issued by the applicable Facing Agent such other Revolving Lender's Pro Rata Share of all payments received by the applicable Facing Agent from the Borrower in reimbursement of drawings honored by the applicable Facing Agent under such Letter of Credit when such payments are received. Each Lender's obligation to make Revolving Loans pursuant to Section 2.9(c) or to purchase participations pursuant to this Section 2.9(d) as a result of a drawing under a Letter of Credit shall be absolute and unconditional and without recourse to the applicable Facing Agent and shall not be affected by any circumstance, including (i) any set-off, counterclaim, recoupment, defense or other right which such Lender may have against the Facing Agent, the Borrower or any other Person for any reason whatsoever; (ii) the occurrence or continuance of an Event of Default or a Material Adverse Effect; or (iii) any other circumstance, happening or event whatsoever, whether or not similar to any of the foregoing; provided, however, that each Lender's obligation to make Revolving Loans under Section 2.9(c) is subject to the conditions set forth in Section 5.2.

64

E. Fees for Letters of Credit.

- (a) <u>Facing Agent Fees</u>. The Borrower agrees to pay the following amount to the applicable Facing Agent with respect to the Letters of Credit issued by it for the account of the Borrower:
- (i) with respect to drawings made under any Letter of Credit, interest, payable on demand, on the amount paid by such Facing Agent in respect of each such drawing from the date a drawing is honored up to (but not including) the date such amount is reimbursed by the Borrower (including any such reimbursement out of the proceeds of Revolving Loans, as the case may be, pursuant to Section 2.9(c)) at a rate which is at all times equal to 2% per annum in excess of the Base Rate;
- (ii) with respect to the issuance or amendment of each Letter of Credit and each payment made thereunder, documentary and processing charges in accordance with the applicable Facing Agent's standard schedule for such charges in effect at the time of such issuance, amendment, transfer or payment, as the case may be; and
- (iii) a facing fee as agreed from time to time by the Borrower and the applicable Facing Agent for the applicable Letter of Credit or, with respect to DB as Facing Agent with respect to that portion of the outstanding LC Obligations attributable to the Fifth Amendment Existing Letters of Credit, a facing fee as may be separately agreed in writing between the Borrower and DB, which fee, unless otherwise agreed, shall be payable with respect to the Assigned Dollar Value of the maximum Stated Amount under such outstanding Letters of Credit payable in arrears on each Quarterly Payment Date, on the Revolver Termination Date and thereafter, on demand together with customary issuance and payment charges payable pursuant to clause (B) above; provided, however, if calculation of the facing fee in the manner set forth above would result in a facing fee of less than \$500 per year per Letter of Credit issued by any Facing Agent, the Borrower shall be obligated to pay such additional amount to such Facing Agent so as to provide for a minimum facing fee of \$500 per year per Letter of Credit.

(b) Participating Lender Fees. The Borrower agrees to pay to the Administrative Agent for distribution to each participating Revolving Lender in respect of all Letters of Credit outstanding such Revolving Lender's Revolving Pro Rata Share of a commission equal to the then Applicable Eurocurrency Margin for Revolving Loans per annum with respect to the maximum Stated Amount under such outstanding Letters of Credit (the "LC Commission"), payable in Dollars in arrears on each Quarterly Payment Date, on the Revolver Termination Date and thereafter, on demand. The LC Commission shall be computed from the first day of issuance of each Letter of Credit and on the basis of the actual number of days elapsed over a year of 360 days.

65

Promptly upon receipt by a Facing Agent or the Administrative Agent of any amount described in clause (i)(A) or (ii) of this Section 2.9(e), the applicable Facing Agent or the Administrative Agent shall distribute to each Revolving Lender its Pro Rata Share, as the case may be, of such amount as long as, in the case of amounts described in clause (i)(A), such Lender has reimbursed the applicable Facing Agent in accordance with Section 2.9(c). Amounts payable under clause (i)(B) and (C) of this Section 2.9(e) shall be paid directly to the applicable Facing Agent.

- F. <u>LC Obligations Unconditional</u>. The obligation of the Borrower to reimburse a Facing Agent (or any Lender that has purchased a participation from or made a Loan to enable the Borrower to reimburse the applicable Facing Agent) for drawings made under any Letter of Credit issued by it and the obligations of each Lender to purchase participations pursuant to <u>Section 2.9(d)</u> as a result of such a drawing shall be unconditional and irrevocable and shall be paid strictly in accordance with the terms of this Agreement under all circumstances, including, without limitation, any of the following circumstances:
 - (a) any lack of validity or enforceability of such Letter of Credit;
 - (b) the existence of any claim, setoff, defense or other right which the Borrower or any of its Affiliates may have at any time against a beneficiary or any transferee of such Letter of Credit (or any persons or entities for which any such beneficiary or transferee may be acting), the applicable Facing Agent, any Lender or any other Person, whether in connection with this Agreement, the transactions contemplated herein or any unrelated transaction (including any underlying transaction between the Borrower or one of its Subsidiaries and the beneficiary of such Letter of Credit);
 - (c) any draft, demand, certificate or any other document presented under such Letter of Credit proving to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect;
 - (d) payment by the applicable Facing Agent under such Letter of Credit against presentation of a demand, draft or certificate or other document which does not comply with the terms of such Letter of Credit;
 - (e) any other circumstance or happening whatsoever, whether or not similar to any of the foregoing; or
 - (f) the fact that an Event of Default or an Unmatured Event of Default shall have occurred and be continuing.

Notwithstanding the foregoing, neither the Borrower nor the Lenders (other than the applicable Facing Agent in its capacity as such) shall be liable for any obligation resulting from the gross negligence or willful misconduct of the applicable Facing Agent, as determined by a court of competent jurisdiction, with respect to any Letter of Credit.

66

<u>Indemnification</u>. In addition to amounts payable as elsewhere provided in this Agreement, the Borrower hereby agrees to protect, indemnify, pay and save the applicable Facing Agent and the Lenders harmless from and against any and all claims, demands, liabilities, damages, losses, costs, charges and expenses (including Attorney Costs) (other than for Taxes, which shall be covered by Section 4.7) which the applicable Facing Agent and the Lenders may incur or be subject to as a consequence, direct or indirect, of (i) the issuance of the Letters of Credit, other than as a result of the gross negligence or willful misconduct of the applicable Facing Agent, as determined by a court of competent jurisdiction, or (ii) the failure of the applicable Facing Agent to honor a drawing under any Letter of Credit as a result of any act or omissions, whether rightful or wrongful, of any present or future de jure or de facto Governmental Authority (all such acts or omissions herein called "Government Acts") other than arising out of the gross negligence or willful misconduct, as determined by a court of competent jurisdiction, of the applicable Facing Agent. As between the Borrower on the one hand, and the applicable Facing Agent and the Lenders, on the other hand, the Borrower assumes all risks of the acts and omissions of, or misuse of the Letters of Credit issued by the applicable Facing Agent or by the respective beneficiaries of such Letters of Credit. In furtherance and not in limitation of the foregoing, neither the applicable Facing Agent nor any of the Lenders shall be responsible: (i) for the form, validity, sufficiency, accuracy, genuineness or legal effect of any document submitted by any party in connection with the application for and issuance of or any drawing under such Letters of Credit, even if it should in fact prove to be in any or all respects invalid, insufficient, inaccurate, fraudulent or forged; (ii) for the validity or sufficiency of any instrument transferring or assigning or purporting to transfer or assign any such Letter of Credit or the rights or benefits thereunder or proceeds thereof, in whole or in part, which may prove to be invalid or ineffective for any reason; (iii) for failure of the beneficiary of any such Letter of Credit to comply fully with conditions required in order to draw upon such Letter of Credit; (iv) for errors, omissions, interruptions or delays in transmission or delivery of any messages, by mail, cable, telegraph, telex, or otherwise, whether or not they be in cipher; (v) for errors in interpretation of technical terms; (vi) for any loss or delay in the transmission or otherwise of any document required in order to make a drawing under any such Letter of Credit or of the proceeds thereof; (vii) for the misapplication by the beneficiary of any such Letter of Credit of the proceeds of any drawing under such Letter of Credit; and (viii) for any consequences arising from causes beyond the control of the applicable Facing Agent, including, without limitation, any Government Acts. None of the above shall effect, impair, or prevent the vesting of any of the applicable Facing Agent's or any Lender's rights or powers hereunder.

In furtherance and extension and not in limitation of the specific provisions hereinabove set forth, any action taken or omitted by the applicable Facing Agent under or in connection with the Letters of Credit issued by it or the related certificates, if taken or omitted in good faith, shall not put the applicable Facing Agent under any resulting liability to the Borrower. Notwithstanding anything to the contrary contained in this Agreement, the Borrower shall have no obligation to indemnify or hold harmless the applicable Facing Agent in respect of any claims, demands, liabilities, damages, losses, costs, charges or expenses (including Attorney Costs) incurred by the applicable Facing Agent to the extent arising out of the gross negligence or willful misconduct of the applicable Facing Agent, as determined by a court of competent jurisdiction. The right of indemnification in the first paragraph of this Section 2.9(g) shall not prejudice any rights that the Borrower may otherwise have against the applicable Facing Agent with respect to a Letter of Credit issued hereunder.

67

- H. <u>Stated Amount</u>. The Stated Amount of each Letter of Credit shall not be less than the Dollar Equivalent of One Hundred Thousand Dollars (\$100,000) or such lesser amount as the applicable Facing Agent has agreed to.
- <u>Increased Costs.</u> Subject to <u>Section 4.7</u>, if any time after the date hereof any Facing Agent or any Lender determines that the introduction of or any change after the Closing Date in any applicable law, rule, regulation, order, guideline or request or in the interpretation or administration thereof by any Governmental Authority charged with the interpretation or administration thereof, or compliance by the applicable Facing Agent or such Lender with any request or directive issued after the Closing Date by any such authority (whether or not having the force of law), shall either (i) impose, modify or make applicable any reserve, deposit, capital adequacy or similar requirement against letters of credit issued by the applicable Facing Agent or participated in by any Revolving Lender, or (ii) impose on the applicable Facing Agent or any Revolving Lender any other conditions relating, directly or indirectly, to the provisions of this Agreement relating to Letters of Credit or any Letter of Credit; and the result of any of the foregoing is to increase the cost to the applicable Facing Agent or any Lender of issuing, maintaining or participating in any Letter of Credit, or reduce the amount of any sum received or receivable by the applicable Facing Agent or any Lender hereunder or reduce the rate of return on its capital with respect to Letters of Credit, then, upon demand to the Borrower by the applicable Facing Agent or any Lender (a copy of which demand shall be sent by the applicable Facing Agent or such Lender to the Administrative Agent), the Borrower shall pay to the applicable Facing Agent or any Lender, as the case may be, such additional amount or amounts as will compensate the applicable Facing Agent or such Lender for such increased cost or reduction in the amount receivable or reduction on the rate of return on its capital. In determining such additional amounts pursuant to the preceding sentence, the applicable Facing Agent or such Lender will act reasonably and in good faith and will, to the extent the increased costs or reductions in amounts receivable or reductions in rates of return relate to the Facing Agent's or such Lender's letters of credit in general and are not specifically attributable to the Letters of Credit hereunder, use averaging and attribution methods which are reasonable and which cover all letters of credit similar to the Letters of Credit issued by or participated in by the applicable Facing Agent or such Lender whether or not the documentation for such other Letters of Credit permit the applicable Facing Agent or such Lender to receive amounts of the type described in this Section 2.9(i). The applicable Facing Agent or any Lender, upon determining that any additional amounts will be payable pursuant to this Section 2.9(i), will give prompt written notice thereof to the Borrower, which notice shall include a certificate submitted to the Borrower by the applicable Facing Agent or such Lender (a copy of which certificate shall be sent by the applicable Facing Agent or such Lender to the Administrative Agent), setting forth in reasonable detail the basis for the calculation of such additional amount or amounts necessary to compensate the applicable Facing Agent or such Lender, although failure to give any such notice shall not release or diminish the Borrower's obligations to pay additional amounts pursuant to this Section 2.9(i); provided, however, if the applicable Facing Agent or such Lender, as applicable, has intentionally withheld or delayed such notice, the applicable Facing Agent or such Lender, as the case may be, shall not be entitled to receive additional amounts pursuant to this Section 2.9(i) for periods occurring prior to the 180th day before the giving of such notice. The certificate required to be delivered pursuant to this Section 2.9(i) shall, absent manifest error, be final, conclusive and binding on the Borrower.

68

- J. Outstanding Letters of Credit. The letters of credit set forth under the caption "Letters of Credit outstanding on the Closing Date" as set forth on Schedule 2.9(j) attached to the First Amendment, were issued prior to the Closing Date pursuant to one of the Prior Credit Agreements and will remain outstanding as of the Closing Date and the letters of credit and bank guarantees set forth under the caption "Letters of Credit and Bank Guarantees outstanding on the First Amendment Effective Date" were issued pursuant to the credit facility of Huntsman Advanced Materials Holdings LLC or otherwise issued by Facing Agent prior to the First Amendment Effective Date and will remain outstanding as of the First Amendment Effective Date (collectively, the "Outstanding Letters of Credit"). The Borrower, each Facing Agent and each of the Lenders hereby agree with respect to the Outstanding Letters of Credit that each such Outstanding Letters of Credit, for all purposes under this Agreement, shall, from and after the Closing Date, be deemed to be Letters of Credit governed by the terms and conditions of this Agreement. Each Revolving Lender further agrees to participate in each such Outstanding Letter of Credit in an amount equal to its Pro Rata Share of the Stated Amount of such Outstanding Letters of Credit.
- K. <u>Fifth Amendment Existing Letters of Credit.</u> The Borrower agrees that if DB receives any proceeds in respect of any Letter of Credit providing credit support for a Fifth Amendment Existing Letter of Credit, the Borrower will cause a reduction to occur in the amount of then outstanding Fifth Amendment Existing Letters of Credit in an amount equal to such proceeds.

- A. Subject to the terms and conditions set forth herein, the Borrower shall have the right from time to time to cause an increase in the Revolving Commitments of the Revolving Lenders (a "Revolving Commitment Increase") by adding to this Agreement one or more additional Eligible Assignees that are not already Revolving Lenders hereunder and that are reasonably satisfactory to the Administrative Agent, each Facing Agent and the Swing Line Lender (each, a "New Revolving Lender") or by allowing one or more existing Revolving Lenders to increase their respective Revolving Commitments; provided that (i) both before and immediately after giving effect to such Revolving Commitment Increase, no Event of Default or Unmatured Event of Default shall have occurred and be continuing as of the effective date of such Revolving Commitment Increase (such date, the "Revolving Commitment Increase Date"), (ii) no such Revolving Commitment Increase shall be in an amount less than \$10,000,000 (unless the aggregate amount of the Revolving Commitments then in effect is greater than \$290,000,000), (iii) after giving effect to such Revolving Commitment Increase, the aggregate amount of the Revolving Commitments shall not exceed \$300,000,000 and (iv) no Revolving Lender's Revolving Commitment shall be increased without such Revolving Lender's prior written consent (which consent may be given or withheld in such Revolving Lender's sole and absolute discretion).
- B. The Borrower shall provide the Administrative Agent with written notice (a "Notice of Revolving Commitment Increase") of its intention to increase the Revolving Commitments pursuant to this Section 2.10. Each such Notice of Revolving Commitment Increase shall specify (i) the proposed Revolving Commitment Increase Date, which date shall be no earlier than five (5) Business Days after receipt by the Administrative Agent of such

69

Notice of Revolving Commitment Increase, (ii) the amount of the requested Revolving Commitment Increase, (iii) as applicable, the identity of each New Revolving Lender and Revolving Lender that has agreed in writing to increase its Revolving Commitment hereunder, and (iv) the amount of the respective Revolving Commitments of the then existing Revolving Lenders and the New Revolving Lenders from and after the Revolving Commitment Increase Date.

- C. On any Revolving Commitment Increase Date, the Revolving Lenders shall purchase and assume (without recourse or warranty) from the Revolving Lenders (i) Revolving Loans, to the extent that there are any Revolving Loans then outstanding, and (ii) undivided participation interests in any outstanding LC Obligations and any outstanding Swing Line Loans, in each case, to the extent necessary to ensure that after giving effect to the Revolving Commitment Increase, each Revolving Lender has outstanding Revolving Loans and participation interests in outstanding LC Obligations and Swing Line Loans equal to its Pro Rata Share of the Revolving Commitments. Each Revolving Lender shall make any payment required to be made by it pursuant to the preceding sentence via wire transfer to the Administrative Agent on the Revolving Commitment Increase Date. Each existing Revolving Lender (i) shall be automatically deemed to have assigned any outstanding Revolving Loans on the Revolving Commitment Increase Date and (ii) agrees to take any further steps reasonably requested by the Administrative Agent, in each case to the extent deemed necessary by the Administrative Agent to effectuate the provisions of the preceding sentences. If, on such Revolving Commitment Increase Date or Extension Effective Date, any Revolving Loans have been funded, then the Borrower shall be obligated to pay any breakage fees or costs that are payable pursuant to Section 3.5 in connection with the reallocation of such outstanding Revolving Loans to effectuate the provisions of this Section 2.10(c).
- D. Each Revolving Commitment Increase shall become effective on its Revolving Commitment Increase Date and upon such effectiveness: (i) the Administrative Agent shall record in the register each then New Revolving Lender's information as provided in the applicable Notice of Commitment Increase and pursuant to an administrative questionnaire that shall be executed and delivered by each New Revolving Lender to the Administrative Agent on or before such Revolving Commitment Increase Date, (ii) Schedule 1.1(a) hereof shall be amended and restated to set forth all Revolving Lenders (including any New Revolving Lenders) that will be Revolving Lenders hereunder after giving effect to such Revolving Commitment Increase (which amended and restated Schedule 1.1(a) shall be set forth in Annex I to the applicable Notice of Revolving Commitment Increase) and the Administrative Agent shall distribute to each Revolving Lender (including each New Revolving Lender) a copy of such amended and restated Schedule 1.1(a), and (iii) each New Revolving Lender identified on the Notice of Revolving Commitment Increase for such Revolving Commitment Increase shall be a "Lender" and a "Revolving Lender" for all purposes under this Agreement.
- E. As a condition precedent to any Revolving Commitment Increase, the Borrower shall deliver to the Administrative Agent a certificate of the Borrower dated as of the Revolving Commitment Increase Date signed by a Responsible Officer of the Borrower certifying and attaching the resolutions adopted by the Borrower approving or consenting to such Revolving Commitment Increase and certifying that, before and after giving effect to such Revolving Commitment Increase, (i) the representations and warranties contained in Article-VI

70

made by it are true and correct on and as of the Revolving Commitment Increase Date, except to the extent that such representations and warranties specifically refer to an earlier date and (ii) no Event of Default or Unmatured Event of Default exists or will exist as of the Revolving Commitment Increase Date.

10.11. Extension of Stated Termination Date

A. At any time and from time to time, the Borrower may, upon notice to the Administrative Agent (which shall promptly notify the Revolving Lenders), request an extension of the Stated Termination Date then in effect (such existing Stated Termination Date being the "Existing Stated Termination Date being the "Existing Stated Termination Date") to a date specified in such notice. Within 10 Business Days of delivery of such notice (or such other period as the Borrower and the Administrative Agent shall mutually agree upon), each Revolving Lender

shall notify the Administrative Agent whether or not it consents to such extension (which consent may be given or withheld in such Revolving Lender's sole and absolute discretion). Any Revolving Lender not responding within the above time period shall be deemed not to have consented to such extension. The Administrative Agent shall promptly notify the Borrower and the Revolving Lenders of the Revolving Lenders' responses.

The Stated Termination Date shall be extended only if the Majority Lenders of the Revolving Facility (calculated excluding any Impaired Lender and after giving effect to any replacements of Revolving Lenders pursuant to Section 4.1(c)) (the Revolving Lenders that so consent being the "Consenting Revolving Lenders" and the Revolving Lenders that declined being the "Declining Revolving Lenders") have consented thereto. If so extended, the Stated Termination Date, as to the Consenting Revolving Lenders, shall be extended to the date specified in the notice referred to in Section 2.11(a) above, which shall become the new Stated Termination Date as to such Consenting Revolving Lenders. The Administrative Agent and the Borrower shall promptly confirm to the Revolving Lenders such extension, specifying the effective date of such extension (the "Extension Effective Date"), the Existing Stated Termination Date, and the new Stated Termination Date (after giving effect to such extension). As a condition precedent to such extension, the Borrower shall deliver to the Administrative Agent (i) a certificate of the Borrower dated as of the Extension Effective Date signed by a Responsible Officer of the Borrower certifying and attaching the resolutions adopted by the Borrower approving or consenting to such extension and certifying that, before and after giving effect to such extension, the representations and warranties contained in Article VI made by it are true and correct on and as of the Extension Effective Date, except to the extent that such representations and warranties specifically refer to an earlier date, and no Event of Default or Unmatured Event of Default exists or will exist as of the Extension Effective Date and (ii) if requested by the Administrative Agent, a legal opinion reasonably satisfactory to it. The Borrower shall pay to the Administrative Agent for the account of each Declining Revolving Lender the then unpaid principal amount of such Declining Revolving Lender's Revolving Loans outstanding on the Existing Stated Termination Date (and pay any additional amounts required pursuant to Section 3.5). In addition, (i) the Borrower shall prepay any Revolving Loans outstanding on the Existing Stated Termination Date (and pay any additional amounts required pursuant to Section 3.5) to the extent necessary to maintain outstanding Revolving Loans ratable with any revised and new Pro Rata Shares of all the Revolving Lenders effective as of the Existing Stated Termination Date and (ii) on the Existing Stated Termination Date, the Risk

71

Participation (as defined below) of each Declining Revolving Lender shall be transferred to and assumed by the Consenting Revolving Lenders on a pro rata basis in accordance with their respective Revolving Commitments. "Risk Participation" means, at any date, the aggregate amount of funded and unfunded obligations of any Revolving Lender to purchase participations from or make payments for the account of (i) the Swing Line Lender pursuant to <u>Sections 2.1(c)(iii)</u> and <u>(iv)</u> and (ii) any Facing Agent pursuant to <u>Section 2.9(d)</u>.

SECTION 11

INTEREST AND FEES

11.1. Interest

- A. <u>Base Rate Loans</u>. The Borrower agrees to pay interest in respect of the unpaid principal amount of each Base Rate Loan, at a rate per annum equal to the Base Rate plus the Applicable Base Rate Margin, from the date the proceeds thereof are made available to the Borrower (or, if such Base Rate Loan was converted from a Eurocurrency Loan, the date of such conversion) until the earlier of (i) the maturity (whether by acceleration or otherwise) of such Base Rate Loan and (ii) the conversion of such Base Rate Loan to a Eurocurrency Loan pursuant to <u>Section 2.6</u>.
- B. <u>Eurocurrency Loans</u>. The Borrower agrees to pay interest in respect of the unpaid principal amount of each Eurocurrency Loan at a rate per annum equal to the relevant Eurocurrency Rate plus the Applicable Eurocurrency Margin, from the date the proceeds thereof are made available to the Borrower (or, if such Eurocurrency Loan was converted from a Base Rate Loan, the date of such conversion) until the earlier of (i) the maturity (whether by acceleration or otherwise) of such Eurocurrency Loan and (ii) the conversion of such Eurocurrency Loan to a Base Rate Loan pursuant to Section 2.6.
- C. <u>Payment of Interest</u>. Interest on each Loan shall be payable in arrears on each Interest Payment Date; <u>provided, however</u>, that interest accruing pursuant to <u>Section 3.1(e)</u> and as otherwise set forth in the penultimate sentence of this <u>Section 3.1(c)</u> shall be payable from time to time on demand. Interest shall also be payable on all then outstanding Revolving Loans on the Revolver Termination Date and on all Loans on the date of repayment (including prepayment) thereof (except that accrued interest need not be paid in connection with voluntary prepayments pursuant to <u>Section 4.3</u> of Swing Line Loans and Revolving Loans that are Base Rate Loans made on any day other than a Quarterly Payment Date or the Revolver Termination Date; provided such accrued interest is paid on the next Quarterly Payment Date) or on the date of maturity (by acceleration or otherwise), as applicable, of such Loans. During the existence of any Event of Default, interest on any Loan shall be payable on demand. Interest to be paid with respect to Loans denominated in (x) Dollars shall be paid in Dollars and (y) in an Alternative Currency shall be in such Alternative Currency.
- D. <u>Notification of Rate</u>. The Administrative Agent, upon determining the interest rate for any Borrowing of Eurocurrency Loans for any Interest Period, shall promptly notify the Borrower and the Lenders thereof. Such determination shall, absent manifest error and subject to <u>Section 3.6</u>, be final, conclusive and binding upon all parties hereto.

72.

of any Event of Default and for so long thereafter as any such Event of Default shall be continuing, and effective immediately upon any failure to pay any Obligations or any other amounts due under any of the Loan Documents when due, whether by acceleration or otherwise, the principal balance of each Loan then outstanding and, to the extent permitted by applicable law, any interest payment on each Loan not paid when due or other amounts then due and payable shall bear interest payable on demand, after as well as before judgment, at a rate per annum equal to the Default Rate.

F. <u>Maximum Interest</u>. If any interest payment or other charge or fee payable hereunder exceeds the maximum amount then permitted by applicable law, the Borrower shall be obligated to pay the maximum amount then permitted by applicable law and the Borrower shall continue to pay the maximum amount from time to time permitted by applicable law until all such interest payments and other charges and fees otherwise due hereunder (in the absence of such restraint imposed by applicable law) have been paid in full.

11.2. Fees

- A. <u>Commitment Fees.</u> The Borrower shall pay to the Administrative Agent for pro rata distribution to each Non-Impaired Lender having a Revolving Commitment (based on its Pro Rata Share) a commitment fee (the "<u>Commitment Fee</u>") for the period commencing on the Closing Date to and including the Revolver Termination Date, computed at a rate equal to the Applicable Commitment Fee Percentage per annum on the average daily Total Available Revolving Commitment, which shall be calculated based upon the time period as to which such Commitment Fee is being paid. Unless otherwise specified, accrued Commitment Fees shall be due and payable (i) on each Quarterly Payment Date occurring after the Closing Date, (ii) on the Revolver Termination Date and (iii) upon any reduction or termination in whole or in part of the Revolving Commitments, as the case may be (but only, in the case of a reduction, on the portion of the Revolving Commitments then being reduced).
- B. Impaired Lender Fees. Notwithstanding anything herein to the contrary, so long as a Revolving Lender is an Impaired Lender, such Impaired Lender shall not be entitled to any fees accruing during such period pursuant to Section 3.2(a) and Section 2.9(e) (without prejudice to the rights of the Revolving Lenders other than Impaired Lenders in respect of such fees); provided, that (i) to the extent that a Pro Rata Share of the LC Obligations or Swing Line Loans of such Impaired Lender is reallocated to the Non-Impaired Lenders pursuant to Section 3.8(a), such fees that would have accrued for the benefit of such Impaired Lender will instead accrue for the benefit of and be payable to such Non-Impaired Lenders, pro rata in accordance with their respective Revolving Commitments, and (ii) to the extent any portion of such LC Obligations or Swing Line Loans cannot be so reallocated and the Borrower does not Cash Collateralize such portion in an amount not less than 105% of the amount of such portion (or prepay such Swing Line Loans) or make other arrangements pursuant to Section 3.8(a)(ii), then such fees will instead accrue for the benefit of and be payable to the applicable Facing Agent and the Swing Line Lender as their interests appear.

73

C. <u>Arranger and Agency Fees.</u> The Borrower shall pay to the Administrative Agent for its own account (or the account of its Affiliates), agency and other Loan fees in the amount and at the times set forth in the Fifth Amendment Fee Letter.

11.3. Computation of Interest and Fees

Interest on all Loans and fees payable hereunder shall be computed on the basis of the actual number of days elapsed over a year of 360 days; provided that interest on all Base Rate Loans and Sterling denominated Loans shall be computed on the basis of the actual number of days elapsed over a year of 365 or 366 days, as the case may be. The Administrative Agent shall, at any time and from time to time upon request of the Borrower, deliver to the Borrower a statement showing the quotations used by the Administrative Agent in determining any interest rate applicable to Revolving Loans pursuant to this Agreement. Each change in the Applicable Base Rate Margin or Applicable Eurodollar Margin or the Applicable Commitment Fee Percentage or any change in the LC Commission as a result of a change in the Borrower's Most Recent Leverage Ratio shall become effective on the date upon which such change in such ratio occurs.

11.4. Interest Periods

At the time it gives any Notice of Borrowing or a Notice of Conversion or Continuation with respect to any Borrowing bearing interest with reference to the Eurocurrency Rate, the Borrower shall elect, by giving the Administrative Agent written notice, the interest period (each an "Interest Period") applicable thereto, which Interest Period shall, at the option of the Borrower, be one, two, three or six months or, if available or otherwise satisfactory to each of the applicable Lenders (as determined by each such applicable Lender in its sole discretion) a nine or twelve month period, provided that:

- (a) all Eurocurrency Loans comprising a single Borrowing shall at all times have the same Interest Period;
- (b) the initial Interest Period for any Eurocurrency Loan shall commence on the date of such Borrowing of such Eurocurrency Loan (including the date of any conversion thereto from a Loan of a different Type) and each Interest Period occurring thereafter in respect of such Eurocurrency Loan shall commence on the last day of the immediately preceding Interest Period;
- (c) if any Interest Period relating to a Eurocurrency Loan begins on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period, such Interest Period shall end on the last Business Day of such calendar month;
 - (d) if any Interest Period would otherwise expire on a day which is not a Business Day, such Interest

- (e) at any time when an Event of Default is then in existence, no Interest Period (a) of more than one month may be selected with respect to any Loan denominated in an Alternative Currency and (b) may be selected with respect to any Loan denominated in Dollars;
- (f) no Interest Period shall extend beyond the applicable Term Maturity Date for any Term Loan or the Revolver Termination Date for any Revolving Loan; and
- (g) no Interest Period in respect of any Borrowing of Term Loans for any Term Facility shall be selected which extends beyond any date upon which a Scheduled Term Repayment will be required to be made under Section 4.4(b) for such Term Facility if the aggregate principal amount of all Term Loans which have Interest Periods which will expire after such date will be in excess of the aggregate principal amount of Term B Loans then outstanding less the aggregate amount of such required prepayment.

Notwithstanding anything to the contrary herein (i) with respect to Term B Dollar Loans, the Borrower may only have Base Rate Loans and Eurocurrency Loans with a one month Interest Period for the first 30 days after the Effective Date or, if earlier, the date on which the Administrative Agent informs the Borrower of the completion of the syndication of the Term B Dollar Loans and (ii) if requested by the Borrower, the Administrative Agent, in its discretion, may offer Interest Periods shorter than one month.

11.5. Compensation for Funding Losses

The Borrower shall compensate each Lender, upon its written request (which request shall set forth the basis for requesting such amounts, showing the calculation thereof in reasonable detail), for all losses, expenses and liabilities (including, without limitation, any interest paid by such Lender to lenders of funds borrowed by it to make or carry its Eurocurrency Loans to the extent not recovered by the Lender in connection with the liquidation or re-employment of such funds and including the compensation payable by such Lender to a Participant) and any loss sustained by such Lender in connection with the liquidation or re-employment of such funds (including, without limitation, a return on such liquidation or re-employment that would result in such Lender receiving less than it would have received had such Eurocurrency Loan remained outstanding until the last day of the Interest Period applicable to such Eurocurrency Loans) which such Lender may sustain as a result of: (i) the continuation or Borrowing of, the conversion from or into, a Eurocurrency Loan not occurring on the date specified therefor in a Notice of Borrowing or Notice of Conversion or Continuation (whether or not withdrawn for any reason (other than a default by such Lender or the Administrative Agent)); (ii) any payment, prepayment or conversion or continuation of any of its Eurocurrency Loans occurring for any reason whatsoever on a date which is not the last day of an Interest Period applicable thereto; (iii) any repayment of any of its Eurocurrency Loans not being made on the date specified in a notice of payment given by the Borrower; or (iv) (A) any other failure by the Borrower to repay its Eurocurrency Loans when required by the terms of this Agreement or (B) an election made by the Borrower pursuant to Section 3.7; provided that such written request, including the calculation as to additional amounts owed such Lender under this Section 3.5, is

75

delivered to the Borrower and the Administrative Agent by such Lender shall be delivered within 30 days of such event. Absent manifest error, such request shall be final, conclusive and binding for all purposes. Calculation of all amounts payable to a Lender under this Section 3.5 shall be made as though that Lender had actually funded its relevant Eurocurrency Loan through the purchase of a Eurocurrency deposit bearing interest at the Eurocurrency Rate in an amount equal to the amount of that Loan, having a maturity comparable to the relevant Interest Period and through the transfer of such Eurocurrency deposit from an offshore office of that Lender to a domestic office of that Lender in the United States of America; provided, however, that each Lender may fund each of its Eurocurrency Loans in any manner it sees fit and the foregoing assumption shall be utilized only for the calculation of amounts payable under this Section 3.5.

11.6. Increased Costs, Illegality, Etc.

- A. <u>Generally</u>. Except as provided in <u>Section 4.7</u>, in the event that any Lender shall have determined, in its reasonable, good faith judgment, (which determination shall, absent manifest error, be final and conclusive and binding upon all parties hereto but, with respect to clause (i) below, may be made only by the Administrative Agent):
 - (a) on any Interest Rate Determination Date that, by reason of any change arising after the date of this Agreement affecting the interbank Eurocurrency market, adequate and fair means do not exist for ascertaining the applicable interest rate on the basis provided for in the definition of Eurocurrency Rate; or
 - (b) at any time that such Lender shall incur increased costs or reduction in the amounts received or receivable hereunder with respect to any Eurocurrency Loan because of (x) any change arising after the date of this Agreement in any applicable law or governmental rule, regulation, order, guideline or request (whether or not having the force of law) or in the interpretation or administration thereof and including the introduction of any new law or governmental rule, regulation, order, guideline or request, such as, for example, but not limited to: (A) a change in the

basis of taxation of payments to such Lender of the principal of or interest on the Notes or any other amounts payable hereunder (except for taxes described in Sections 4.7(a)(i) through (vi) or (B) a change in official reserve requirements (but, in all events, excluding reserves required under Regulation D to the extent included in the computation of the Eurocurrency Rate) and/or (y) other circumstances arising after the date of this Agreement affecting such Lender or the interbank Eurocurrency market or the position of such Lender in such market (excluding, however, differences in such Lender's cost of funds from those of the Administrative Agent which are solely the result of credit differences between such Lender and the Administrative Agent); or

(c) at any time that (x) the making or continuance of any Eurocurrency Loan has been made (1) unlawful by any law or governmental rule, regulation or order, or (2) impracticable as a result of a contingency occurring after the date of this Agreement which materially and adversely affects the interbank

76

Eurocurrency market or (y) compliance by such Lender, acting in good faith, with any governmental request (whether or not having force of law) is impossible;

then, and in any such event, such Lender (or the Administrative Agent, in the case of clause (i) above) shall promptly give notice (by telephone confirmed in writing) to the Borrower and, except in the case of clause (i) above, to the Administrative Agent of such determination (which notice the Administrative Agent shall promptly transmit to each of the other Lenders). Thereafter (x) in the case of clause (i) above, Eurocurrency Loans shall no longer be available until such time as the Administrative Agent notifies the Borrower and the Lenders that the circumstances giving rise to such notice by the Administrative Agent no longer exist, and any Notice of Borrowing or Notice of Conversion or Continuation given by the Borrower with respect to Eurocurrency Loans (other than with respect to conversions to Base Rate Loans) which have not yet been advanced (including by way of conversion) shall be deemed rescinded by the Borrower; (y) in the case of clause (ii) above, the Borrower shall pay to such Lender, upon written demand therefor, such additional amounts (in the form of an increased rate of, or a different method of calculating, interest, as such Lender in its reasonable good faith judgment shall determine) as shall be required to compensate such Lender for such increased costs or reductions in amounts received or receivable hereunder, (such written demand as to the additional amounts owed to such Lender, which shall show the basis for the calculation thereof, submitted to the Borrower by such Lender shall, absent manifest error, be final and conclusive and binding on all the parties hereto; provided, that such Lender shall not be entitled to receive additional amounts pursuant to this Section 3.6(a)(y) for any change or circumstance referred to in Section 3.6(a) occurring prior to the 180th day before the making of such demand (unless the change is due to a law enacted with retroactive effect, in which case the 180 day period shall be extended to include the period of retroactive effect); and (z) in the case of clause (iii) above, the Borrower shall take one of the actions specified in Section 3.6(b) as promptly as possible and, in any event, within the time period required by law. In determining such additional amounts pursuant to clause (y) of the immediately preceding sentence, each affected Lender shall act reasonably and in good faith and will, to the extent the increased costs or reductions in amounts receivable relate to such Lender's loans in general and are not specifically attributable to a Loan hereunder, use averaging and attribution methods which are reasonable and which cover all loans similar to the Loans made by such Lender whether or not the loan documentation for such other loans permits the Lender to receive increased costs of the type described in this Section 3.6(a).

B. <u>Eurocurrency Loans</u>. At any time that any Eurocurrency Loan is affected by the circumstances described in <u>Section 3.6(a)(ii)</u> or (iii), the Borrower may (and, in the case of a Eurocurrency Loan affected by the circumstances described in <u>Section 3.6(a)(iii)</u>, shall) either (i) if the affected Eurocurrency Loan has not yet been made or the conversion to such Eurocurrency Loan has not yet been accomplished, by giving the Administrative Agent telephonic notice (confirmed in writing) on the same date that the Borrower was notified by the affected Lender or the Administrative Agent pursuant to <u>Section 3.6(a)(ii)</u> or (iii), cancel the respective Borrowing, or (ii) if the affected Eurocurrency Loan is then outstanding, upon at least three Business Days' written notice to Administrative Agent, require the affected Lender to convert such Eurocurrency Loan into a Base Rate Loan, <u>provided</u>, that if more than one Lender is affected at any time, then all affected Lenders must be treated the same pursuant to this <u>Section 3.6(b)</u>.

77

Capital Requirements. If any Lender determines that the introduction of or any change in any applicable law or governmental rule, regulation, order, guideline or request (whether or not having the force of law) concerning capital adequacy, or any change in interpretation or administration thereof by any Governmental Authority, central bank or comparable agency (in each case after the date of this Agreement), will have the effect of increasing the amount of capital required or expected to be maintained by such Lender or any corporation controlling such Lender based on the existence of such Lender's Commitments hereunder or its obligations hereunder, then the Borrower shall pay to such Lender, upon its written notice as hereafter described, such additional amounts as shall be required to compensate such Lender or such other corporation for the increased cost to such Lender or such other corporation or the reduction in the rate of return to such Lender or such other corporation as a result of such increase of capital. In determining such additional amounts, each Lender will act reasonably and in good faith and will use averaging and attribution methods which are reasonable and which will, to the extent the increased costs or reduction in the rate of return relates to such Lender's commitments or obligations in general and are not specifically attributable to the Commitments and obligations hereunder, cover all commitments and obligations similar to the Commitments and obligations of such Lender hereunder whether or not the loan documentation for such other commitments or obligations permits such Lender to make the determination specified in this Section 3.6(c), and such Lender's determination of compensation owing under this Section 3.6(c) shall, absent manifest error, be final and conclusive and binding on all the parties hereto. Each Lender, upon determining that any additional amounts will be payable pursuant to this Section 3.6(c), will give prompt written notice thereof to the Borrower, which notice shall show in reasonable detail the basis for calculation of such additional amounts; provided that no Lender shall be entitled to receive additional amounts pursuant to this Section 3.6(c) for increased costs incurred prior to the 180th day before the giving of such notice, unless such increased costs are due to a change in law that provides for retroactive effect, in which

case the 180 day period shall be extended to include the period of retroactive effect).

D. <u>Change of Lending Office</u>. Each Lender which is or will be owed compensation pursuant to <u>Section 3.6(a)</u> or (c) or 4.7 will, if requested by the Borrower, use reasonable efforts (subject to overall policy considerations of such Lender) to cause a different Lending Office to make or continue a Loan or Letter of Credit if such designation will avoid the need for, or materially reduce the amount of, such compensation to such Lender and will not, in the judgment of such Lender, be otherwise disadvantageous to such Lender or Lending Office. Nothing in this <u>Section 3.6(d)</u> shall affect or postpone any of the obligations of the Borrower or the right of any Lender provided for herein.

11.7. Replacement of Affected Lenders

(x) So long as no Event of Default or Unmatured Event of Default then exists, if any Revolving Lender becomes an Impaired Lender, (y) if any Lender (or in the case of Section 2.9(i), Facing Agent) is owed increased costs under Section 2.9(i), Section 3.6(a)(ii) or (iii), or Section 3.6(c), or the Borrower is required to make any payments under Section 4.7(a) or (c) to any Lender, or (z) as provided in Section 12.1(b) in the case of certain refusals by a Lender to consent to certain proposed amendments, changes, supplements, waivers, discharges or terminations with respect to this Agreement which have been approved by the Required Lenders,

78

the Borrower shall have the right to replace such Lender (the "Replaced Lender") with one or more other Eligible Assignees acceptable to the Administrative Agent, provided that no such Eligible Assignee is an Impaired Lender at the time of such replacement (collectively, the "Replacement Lender"), provided further that (i) at the time of any replacement pursuant to this Section 3.7, the Replaced Lender and Replacement Lender shall enter into one or more assignment agreements, in form and substance satisfactory to such parties and the Administrative Agent, pursuant to which the Replacement Lender shall acquire, at par, all of the Commitments and outstanding Loans of, and participation in Letters of Credit and Swing Line Loans by, the Replaced Lender (with the assignment fee paid by either the Replacement Lender or the Borrower) and (ii) all obligations of the Borrower owing to the Replaced Lender (including, without limitation, such increased costs and including those specifically described in clause (y) above but excluding principal and interest in respect of which the assignment purchase price has been, or is concurrently being paid at par) shall be paid in full to such Replaced Lender concurrently with such replacement. Upon the execution of the respective assignment documentation, the payment of amounts referred to in clause (ii) above and the par purchase price referred to in (i) above, and, if so requested by the Replacement Lender, delivery to the Replacement Lender of the appropriate Note or Notes executed by the Borrower, the Replacement Lender shall become a Lender hereunder and, unless the Replaced Lender continues to have outstanding Loans hereunder, the Replaced Lender shall cease to constitute a Lender hereunder, except with respect to indemnification provisions under this Agreement, which shall survive as to such Replaced Lender. Notwithstanding anything to the contrary contained above, no Lender that acts as a Facing Agent may be replaced hereunder at any time during which such Facing Agent has Letters of Credit outstanding hereunder, unless arrangements satisfactory to such Facing Agent (including (1) the furnishing of a standby letter of credit in form and substance, and issued by an issuer, satisfactory to such Facing Agent or (2) the depositing of cash collateral into a collateral account in amounts and pursuant to arrangements satisfactory to such Facing Agent) have been made with respect to such outstanding Letters of Credit. The Replaced Lender shall be required to deliver for cancellation its applicable Notes to be canceled on the date of replacement, or if any such Note is lost or unavailable, such other assurances or indemnification therefor as the Borrower may reasonably request.

11.8. Impaired Lenders

- A. <u>Reallocation of Impaired Lender Commitment</u>. If a Revolving Lender becomes an Impaired Lender, the following provisions shall apply with respect to any outstanding LC Obligations and any outstanding Swing Line Loans for so long as such Revolving Lender is an Impaired Lender:
 - (a) the Pro Rata Share of such Impaired Lender with respect to any LC Obligations and any outstanding Swing Line Loans shall be reallocated (effective on the date such Lender becomes an Impaired Lender, but without releasing such Impaired Lender from any of its obligations or commitments) among the Revolving Lenders that are Non-Impaired Lenders pro rata in accordance with their respective Revolving Commitments but only to the extent the sum of the aggregate outstanding principal amount of each of the Non-Impaired Lenders' Revolving Loans plus each of the Non-Impaired Lenders' Pro Rata Share of the

79

LC Obligations and Swing Line Loans does not exceed the total of all Non-Impaired Lenders' Revolving Commitments; provided, that no such reallocation will constitute a waiver or release of any claim the Borrower, the Administrative Agent, any Facing Agent, the Swing Line Lender or any other Revolving Lender may have against such Impaired Lender (including any claim for contribution made by a Non-Impaired Lender in respect of amounts funded by such Non-Impaired Lender pursuant to such reallocation of commitments) or cause such Impaired Lender to be a Non-Impaired Lender;

(b) to the extent that any portion (the "unreallocated portion") of the Pro Rata Share of such Impaired Lender with respect to any LC Obligations and any outstanding Swing Line Loans cannot be so reallocated, whether by reason of clause (i) above or otherwise, the Borrower will, not later than 10 Business Days after demand by the Administrative Agent (at the direction of any Facing Agent and/or the Swing Line Lender, as the case may be), (A) Cash Collateralize the obligations of the Borrower to any Facing Agent in respect of such LC Obligations in an amount at least equal to 105% of the aggregate amount of the unreallocated portion of such LC Obligations, or (B) in the case of such

outstanding Swing Line Loans, prepay the unreallocated portion thereof, or (C) make other arrangements satisfactory to the Borrower and the Administrative Agent and to the applicable Facing Agent and the Swing Line Lender, as the case may be, in their sole discretion to protect them against the risk of non-payment by such Impaired Lender; and

any amount paid by the Borrower or otherwise received for the account of an Impaired Lender under this Agreement (whether on account of principal, interest, fees, indemnity payments or other amounts) will not be distributed to such Impaired Lender within the meaning of clause (i), (ii)(x) or (iii) of the definition of Impaired Lender (but, for the avoidance of doubt, the Obligations for which such amount is paid for the account of such Impaired Lender will be satisfied and discharged in full when paid by the Borrower to the Administrative Agent), and such amount will instead be retained by the Administrative Agent in a segregated, non-interest bearing account until (subject to Section 2.10) the termination of the Revolving Commitments and payment in full of all Obligations of the Borrower hereunder and will be applied by the Administrative Agent, to the fullest extent permitted by law, to the making of payments from time to time in the following order of priority: first to the payment of any amounts owing by such Impaired Lender to the Administrative Agent under this Agreement, second to the payment of any amounts owing by such Impaired Lender to any Facing Agent or the Swing Line Lender (pro rata as to the respective amounts owing to each of them) under this Agreement, third to the payment of post-default interest and then current interest due and payable to the Revolving Lenders hereunder other than Impaired Lenders, ratably among them in accordance with the amounts of such interest then due and payable to them, fourth to the payment of fees then due and payable to the Non-Impaired Lenders hereunder, ratably among them in accordance with the amounts of such fees then due and payable to them, fifth to pay principal of the Revolving Loans and any

80

reimbursement obligations with respect to any Letter of Credit then due and payable to the Non-Impaired Lenders hereunder ratably in accordance with the amounts thereof then due and payable to them, <u>sixth</u> to the ratable payment of other amounts then due and payable under this Agreement to the Non-Impaired Lenders, and <u>seventh</u> after the termination of the Revolving Commitments and payment in full of all Obligations of the Borrower hereunder, to pay amounts owing under this Agreement to such Impaired Lender or as a court of competent jurisdiction may otherwise direct.

- B. <u>Termination of Impaired Lender Commitments</u>. The Borrower may terminate the unused amount of the Revolving Commitment of an Impaired Lender upon not less than 10 Business Days' prior notice to the Administrative Agent (which will promptly notify the Revolving Lenders thereof); provided, that such termination will not be deemed to be a waiver or release of any claim the Borrower, the Administrative Agent, any Facing Agent, the Swing Line Lender or any Revolving Lender may have against such Impaired Lender.
- C. Cure. If the Borrower and the Administrative Agent, each Facing Agent and the Swing Line Lender agree in writing in their discretion that a Revolving Lender that is an Impaired Lender should no longer be deemed to be an Impaired Lender, the Administrative Agent will so notify the Borrower and the Revolving Lenders, whereupon as of the effective date specified in such notice, such Revolving Lender will, to the extent applicable, purchase such portion of outstanding Revolving Loans of the other Revolving Lenders (or the other Revolving Lenders will purchase from the formerly Impaired Lender) and/or make such other adjustments as the Administrative Agent may reasonably determine to be necessary to cause such Revolving Lender's Pro Rata Share to be on a pro rata basis in accordance with its Revolving Commitment, whereupon such Revolving Lender will cease to be an Impaired Lender and will be a Non-Impaired Lender; provided, that no adjustments will be made retroactively with respect to fees accrued or payments made by or on behalf of the Borrower while such Revolving Lender was an Impaired Lender; and provided, further, that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Impaired Lender to Non-Impaired Lender will constitute a waiver or release of any claim of any party hereunder arising from such Revolving Lender having been an Impaired Lender.

SECTION 12

REDUCTION OF COMMITMENTS; PAYMENTS AND PREPAYMENTS

12.1. Voluntary Reduction of Commitments

A. Upon at least three Business Days' prior written notice (or telephonic notice confirmed in writing) to the Administrative Agent at the Notice Office (which notice the Administrative Agent shall promptly transmit to each Lender), the Borrower shall have the right, without premium or penalty, to terminate the unutilized portion of the Revolving Commitments and/or, the Swing Line Commitment, as the case may be, in part or in whole; <u>provided</u> that (i) any such voluntary termination of the Revolving Commitments shall apply proportionately to, and shall permanently reduce, the Revolving Commitments of each Revolving Lender; (ii) any partial voluntary reduction of the Revolving Commitments pursuant to this <u>Section 4.1</u> shall be

8

in the amount of at least \$10,000,000 and integral multiples of \$5,000,000 in excess of that amount and (iii) any such voluntary termination of the Revolving Commitment shall occur simultaneously with a voluntary prepayment, pursuant to Section 4.3 such that the total of the Revolving Commitments shall not be reduced below the sum of the Assigned Dollar Value of the aggregate principal amount of outstanding Revolving Loans, Swing Line Loans and LC Obligations plus any Overdraft Reserve.

terminations with respect to this Agreement which have been approved by the Required Lenders as provided in Section 12.1(b), the Borrower shall have the right, upon five (5) Business Days' prior written notice to the Administrative Agent (which notice the Administrative Agent shall promptly transmit to each of the Lenders), to terminate the entire Revolving Commitment of such Lender, so long as (i) the Borrower repays all Loans, together with accrued and unpaid interest, fees and all other amounts, due and owing to such Lender pursuant to Section 4.3(b) concurrently with the effectiveness of such termination at which time Schedule 1.1(a) shall be deemed modified to reflect such changed amounts and (ii) the Borrower cash collateralizes such Lender's Pro Rata Share of the LC Obligations (in the manner set forth in Section 4.4(a)) then outstanding. At such time, such Lender shall no longer constitute a "Lender" for purposes of this Agreement, except with respect to indemnifications in favor of such Lender under this Agreement which shall survive as to such repaid Lender.

C. In the event that any Revolving Lender does not consent to an increase in its Revolving Commitment pursuant to a Revolving Commitment Increase, or if any Revolving Lender does not consent (or is deemed not to have consented) to an extension of the Stated Termination Date pursuant to Section 2.11, the Borrower shall have the right, upon five (5) Business Days' prior written notice to the Administrative Agent (which notice the Administrative Agent shall promptly transmit to each of the Revolving Lenders), to terminate the entire Revolving Commitment of such Revolving Lender, so long as the Borrower repays all Revolving Loans, together with any accrued and unpaid interest and fees thereon, concurrently with the effectiveness of such termination at which time Schedule 1.1(a) shall be deemed modified to reflect such changed amounts and such Revolving Lender's Pro Rata Share of the LC Obligations and Swing Line Loans shall be reallocated pursuant to Section 2.10(c) or Section 2.11(b), as the case may be. At such time, such Revolving Lender shall no longer constitute a "Revolving Lender" for purposes of this Agreement, except with respect to indemnifications in favor of such Revolving Lender under this Agreement which shall survive as to such repaid Revolving Lender.

12.2. Mandatory Reductions of Commitments

- A. <u>Reduction of Revolving Commitments</u>. The Revolving Commitments shall be reduced at the time and in the amounts required to be reduced pursuant to Section 4.4(c).
- B. <u>Reduction of Term B Dollar Commitments</u>. The Term B Dollar Commitments shall terminate on the Effective Date after giving effect to the Term B Dollar Loans on such date.

82

- C. <u>Proportionate Reductions</u>. Each reduction or adjustment to the Commitments pursuant to this <u>Section 4.2</u> shall apply proportionately to the Commitments of each Lender under the applicable Facility.
- D. <u>Reduction of Term C Dollar Commitments</u>. The Term C Dollar Commitments shall terminate on the Fourth Amendment Effective Date after giving effect to the Term C Dollar Loans on such date.

12.3. Voluntary Prepayments

The Borrower shall have the right to prepay the Revolving Loans, any of the Term Loans or the Swing Line Loans in any combination, in whole or in part, from time to time, without premium or penalty except as set forth in Section 4.5(c), on the following terms and conditions: (i) the Borrower shall give the Administrative Agent irrevocable written notice at its Notice Office (or telephonic notice promptly confirmed in writing) of its intent to prepay the Loans, whether such Loans are Term Loans, Revolving Loans or Swing Line Loans, the amount of such prepayment and the specific Borrowings to which such prepayment is to be applied, which notice shall be given by the Borrower to the Administrative Agent by 12:00 p.m. (New York City time) at least three Business Days prior in the case of Eurocurrency Loans and at least one Business Day prior in the case of Base Rate Loans to the date of such prepayment and which notice shall (except in the case of Swing Line Loans) promptly be transmitted by the Administrative Agent to each of the applicable Lenders; (ii) each partial prepayment of any Borrowing (other than a Borrowing of Swing Line Loans) shall be in an aggregate Dollar Equivalent principal amount of at least \$5,000,000 and each partial prepayment of a Swing Line Loan shall be in an aggregate principal amount of at least \$500,000; provided, that any partial prepayment of Eurocurrency Loans made pursuant to a single Borrowing that reduces the aggregate principal amount of the outstanding Loans made pursuant to such Borrowing to an amount less than the Minimum Borrowing Amount applicable thereto shall be subject to the ante-penultimate sentence of Section 4.5(a); (iii) Eurocurrency Loans may be prepaid pursuant to this Section 4.3 on the last day of an Interest Period applicable thereto, or subject to Section 3.5 on any other day; (iv) each prepayment in respect of any Borrowing shall be applied pro rata among the Loans comprising such Borrowing; provided, that such prepayment shall not be applied to any Revolving Loans of an Impaired Lender at any time when the aggregate amount of Revolving Loans of any Non-Impaired Lender exceeds such Non-Impaired Lender's Pro Rata Share of all Revolving Loans then outstanding; (v) each voluntary prepayment of Term Loans shall be applied first to the Scheduled Term Repayments of the Term Facility being repaid due within the 12 month period following the date of such prepayment in direct order of maturity and, thereafter, shall be applied to reduce the remaining Scheduled Term Repayments on a pro rata basis (based upon the then remaining principal amount of such Scheduled Term Repayments). Unless otherwise specified by the Borrower, such prepayment shall be applied first to the payment of Base Rate Loans and second to the payment of such Eurocurrency Loans as the Borrower shall request (and in the absence of such request, as the Administrative Agent shall determine). The notice provisions, the provisions with respect to the minimum amount of any prepayment, and the provisions requiring prepayments in integral multiples above such minimum amount of this Section 4.3 are for the benefit of the Administrative Agent and may be waived unilaterally by the Administrative Agent.

terminations with respect to this Agreement which have been approved by the Required Lenders as provided in Section 12.1(b), the Borrower shall have the right, upon five (5) Business Days' prior written notice to the Administrative Agent (which notice the Administrative Agent shall promptly transmit to each of the Lenders), to repay all Loans, together with accrued and unpaid interest, fees and all other amounts due and owing to such Lender in accordance with said Section 12.1(b), so long as (A) in the case of the repayment of Revolving Loans of any Revolving Lender pursuant to this clause (b), the Revolving Commitment of such Revolving Lender is terminated concurrently with such repayment pursuant to Section 4.1(b) and (B) in the case of the repayment of Loans of any Lender, the consents required by Section 12.1(b) in connection with the repayment pursuant to this clause (b) shall have been obtained.

12.4. Mandatory Prepayments

- A. Prepayment Upon Overadvance. The Borrower shall prepay the outstanding principal amount of Revolving Loans and/or Swing Line Loans on any date on which the Assigned Dollar Value of all outstanding Revolving Loans, Swing Line Loans and LC Obligations (after giving effect to any other repayments or prepayments on such day) plus any Overdraft Reserve exceeds the Total Revolving Commitment then in effect (including, without limitation, solely as a result of fluctuation in Exchange Rates), in the amount of such excess and in the applicable currency; provided, however, that if such excess is solely as a result of fluctuation in Exchange Rates, (i) the Borrower shall not be obligated to pay such amount until four Business Days after notice from the Administrative Agent and (ii) the Borrower shall not be obligated to pay such amount unless such excess is greater than the Dollar Equivalent of an amount equal to 5% of the Total Revolving Commitment. If, after giving effect to the prepayment of all outstanding Revolving Loans and Swing Line Loans pursuant to this Section 4.4(a), the aggregate Assigned Dollar Value of LC Obligations plus any Overdraft Reserve exceeds the Total Revolving Commitment then in effect, the Borrower shall cash collateralize LC Obligations by depositing, pursuant to a cash collateral agreement to be entered into in form and substance reasonably satisfactory to the Administrative Agent, cash with the Administrative Agent in an amount equal to the difference between the Assigned Dollar Value of such LC Obligations plus any Overdraft Reserve and the Total Revolving Commitment then in effect. The Administrative Agent shall establish in its name for the benefit of the Revolving Lenders a collateral account into which it shall deposit such cash to hold as collateral security for the LC Obligations.
- B. <u>Scheduled Term Repayments</u>. The Borrower shall cause to be paid Scheduled Term Repayments for each Term Facility on the Term Loans until the Term Loans are paid in full in the amounts and currencies and at the times specified in each of the Scheduled Term Repayment definitions to the extent that prepayments have not previously been applied to such Scheduled Term Repayments (and such Scheduled Term Repayments have not otherwise been reduced) pursuant to the terms hereof. Payments to be made pursuant to this <u>Section 4.4(b)</u> with respect to Term B Dollar Loans shall be paid in Dollars. Payments to be made pursuant to this <u>Section 4.4(b)</u> with respect to Term C Dollar Loans shall be paid in Dollars.
 - C. <u>Mandatory Prepayment Upon Asset Disposition or Recovery Event.</u>

84

On the first Business Day after the date of receipt thereof by the Borrower and/or any of its Subsidiaries of Net Sale Proceeds from any Asset Disposition (other than the US Commodity Business Sale, the UK Business Sale or the C4 Business Sale) or Net Recovery Proceeds of a Recovery Event (or in the case of a receipt thereof by a Foreign Subsidiary of the Borrower, such later date as is practicable (but in any event not later than the 360th day or such earlier day as the Borrower is obligated to make an offer to purchase any Public Notes due to such Asset Disposition or Recovery Event) in the event that such mandatory repayment would result in the provisions of Sections 151 et seq. of the Companies Act 1985 of England being breached or in any Foreign Subsidiary breaching any similar applicable law in its country of incorporation), the Borrower shall cause to be paid a mandatory repayment of principal of Loans pursuant to the terms of Section 4.5(a) in an amount equal to the greater of (A) the Term Loan Ratable Share of such Net Sale Proceeds or such Net Recovery Proceeds and (B) 100% of such Net Sale Proceeds or such Net Recovery Proceeds less the applicable Required Note Offer Amount Proceeds; provided, that so long as no Event of Default or Unmatured Event of Default then exists, if either (i) the Net Recovery Proceeds of any single or series of related Recovery Events or (ii) the Net Sale Proceeds of any single or series of related Asset Dispositions are less than \$10,000,000 in the aggregate, then no prepayment shall be required pursuant to this Section 4.4(c)(i), with respect to such Recovery Event(s) or Asset Disposition(s) (but if greater than \$10,000,000, the entire amount of the Net Recovery Proceeds or the Net Sale Proceeds, as applicable, shall be required to be prepaid and not only the portion of the Net Recovery Proceeds or the Net Sale Proceeds, as applicable, in excess of \$10,000,000); provided, further, that the Net Recovery Proceeds of a Recovery Event and the Net Sales Proceeds of an Asset Disposition (if such Asset Disposition is permitted by Section 8.3(h) or (i)), shall not be required to be so applied on such date to the extent that (x) no Event of Default or Unmatured Event of Default then exists and (y) the Borrower delivers a certificate to the Administrative Agent on or prior to such date stating that an amount equal to such Net Sale Proceeds or Net Recovery Proceeds, as applicable, (1) has been used, or (2) is expected to be used within 360 days following the date of receipt of such Net Sale Proceeds or Net Recovery Proceeds, as applicable, to purchase assets used or to be used in the businesses referred to in Section 8.9 (which certificate shall set forth the estimates of the proceeds to be so expended, if applicable) or to redeem, repurchase or otherwise acquire up to \$300,000,000 in aggregate principal amount of obligations under the Senior Secured Notes (2010) or the Senior Notes (2012) in accordance with the terms of Section 8.11(i); provided, however, that, in the case of an Asset Disposition, if the Net Sale Proceeds from such Asset Disposition is for an amount in excess of 7.5% of the Consolidated Net Tangible Assets of the Borrower and its Subsidiaries as of the end of the immediately preceding Fiscal Quarter for which the Borrower has delivered financial statements required by Section 7.1, the Net Sale Proceeds in excess of the amount equal to 7.5% of Consolidated Net Tangible Assets as of such date shall be required to be so applied as a mandatory repayment of principal of Loans pursuant to the terms of Section 4.5(a) on the date thirty (30) days

following receipt thereof unless the Borrower has received from S&P or Moody's by such date a reaffirmation of its debt ratings giving effect to such Asset Disposition and the proposed use of the proceeds therefrom; provided, further, that (1) if all or any portion of such Net Sale Proceeds or Net Recovery Proceeds, as applicable, not so applied to the repayment of Loans have not been used or are not so used (or contractually committed to be used) within such 360 day period as provided above, or not applied in accordance with the terms of Section 8.11(i), such remaining portion shall be applied on the last day of the period or such earlier date as the Borrower is obligated to make an offer to purchase Senior Secured Notes (2010) due to such Asset Disposition or Recovery Event, as applicable, as a mandatory repayment of principal of outstanding Loans as provided above in this Section 4.4(c) and (2) if all or any portion of such Net Sale Proceeds are a result of an Asset Disposition involving the sale of Collateral owned by the Borrower or a Domestic Subsidiary (other than the Capital Stock of a Foreign Subsidiary) or if all or any portion of such Net Recovery Proceeds are a result of a Recovery Event involving Collateral owned by the Borrower or a Domestic Subsidiary, respectively, then such Net Sale Proceeds and Net Recovery Proceeds shall be required to be reinvested in assets located in the United States constituting Collateral (to the extent not used to repay Loans pursuant to this Section 4.4(c)). Notwithstanding the foregoing, on the fifth Business Day after the date of receipt of any LOU Claim Proceeds, the Borrower shall cause to be paid a mandatory repayment of principal of Loans pursuant to the terms of Section 4.5(a) in an amount equal to such LOU Claim Proceeds.

- (b) The Lenders, by execution hereof, hereby irrevocably instruct the Administrative Agent, upon the request of the Borrower, to waive (A) up to \$200,000,000 of Net Sale Proceeds otherwise required to prepay the Loans hereunder in any Fiscal Year, (B) up to \$50,000,000 in the aggregate of Net Sale Proceeds received in connection with Sale and Leaseback Transactions of transportation assets and (C) to the extent not previously waived, Net Sales Proceeds from the C4 Business Sale and the UK Business Sale.
- (c) Notwithstanding anything to the contrary in this Section 4.4(c), if the Borrower or any of its Subsidiaries are required by the terms of any Public Note Document to make an offer to purchase Public Notes due to an Asset Disposition (other than the US Commodity Business Sale and the Asset Dispositions described in clause (ii) (C) above) but would not otherwise be required to make a mandatory prepayment of the Loans applied pursuant to the terms of Section 4.5(a), the Borrower shall cause to be made a mandatory prepayment of the Loans pursuant to Section 4.5(a) in an amount equal to the greater of (A) the Term Loan Ratable Share of the Net Sale Proceeds from such Asset Disposition and (B) 100% of such Net Sale Proceeds from such Asset Disposition less the applicable Required Note Offer Amount Proceeds.
- D. <u>Mandatory Prepayment With Excess Cash Flow.</u> As soon as practicable and in any event by April 30th of each year (commencing April 30, 2006), the Borrower shall cause to be paid a mandatory repayment of principal of Loans applied pursuant to the terms of

86

Section 4.5(a) in an amount equal to 50% of Excess Cash Flow of the Borrower and its Subsidiaries for the Fiscal Year then most recently ended; provided, that so long as no Event of Default or Unmatured Event of Default then exists, (i) if the Most Recent Leverage Ratio is less than 3.5 to 1.0, then, instead of 50%, an amount equal to 25% of Excess Cash Flow of the Borrower and its Subsidiaries for such Fiscal Year shall be applied as a mandatory repayment of Loans as provided in Section 4.5 and (ii) if the Most Recent Leverage Ratio is less than 3.0 to 1.0 then no mandatory prepayment shall be required to be applied from Excess Cash Flow.

12.5. Application of Prepayments.

Prepayments. Except as expressly provided in this Agreement, all prepayments of principal made by the Borrower pursuant to Sections 4.4(c) and (d) shall be applied (i) first, to the payment of the unpaid principal amount of the Term Loans (with, except as provided in the next succeeding sentence, the Term Percentage for each Term Facility of such repayment to be applied as a repayment of Term Loans of such Term Facility), second, to the prepayment of the then outstanding balance of Swing Line Loans, third, to the payment, pro rata, of the then outstanding balance of the Revolving Loans (and the Revolving Commitments shall be permanently reduced by the amount of the required prepayment not applied to the Term Loans), and fourth, to the cash collateralization of LC Obligations; (ii) within each of the foregoing Loans, first to the payment of Base Rate Loans and second to the payment of Eurocurrency Loans, and (iii) with respect to Eurocurrency Loans, in such order as the Borrower shall request (and in the absence of such request, as the Administrative Agent shall determine). Each prepayment of Term Loans made pursuant to Section 4.4(c) and (d) shall be allocated first to the Term Loans based on the aggregate principal amount of the Scheduled Term Repayments due within the twelve month period following the date of such prepayment in direct order of maturity, and, thereafter, shall be allocated second to the Term Loans in proportional amounts equal to the Term Percentage for each Term Facility (in each case, after giving effect to the prepayments made to the Scheduled Term Repayments due within such twelve month period as specified above), as the case may be, of such remaining prepayment, if any, and, within each Term Loan, shall be applied to reduce the remaining Scheduled Term Repayments on a pro rata basis (based upon the then remaining principal amount of such Scheduled Term Repayments). If any prepayment of Eurocurrency Loans made pursuant to a single Borrowing shall reduce the outstanding Loans made pursuant to such Borrowing to an amount less than the Minimum Borrowing Amount, such Borrowing shall immediately be converted into Base Rate Loans, in the case of Loans denominated in Dollars, or into Loans with one month Interest Periods, in the case of Loans denominated in an Alternative Currency. All prepayments shall include payment of accrued interest on the principal amount so prepaid, shall be applied to the payment of interest before application to principal and shall include amounts payable, if any, under Section 3.5. All payments received in Dollars which are required to be applied in Euros and/or Sterling shall be converted to Euros or Sterling, as the case may be, at the Spot Rate on the date of such prepayment.

B. <u>Payments</u>. All Scheduled Term Repayments, all mandatory prepayments and unless otherwise specified by the Borrower, all voluntary prepayments shall be applied (i) first to the payment of Base Rate Loans, if any, and second to the payment of Eurocurrency Loans and (ii) with respect to Eurocurrency Loans, in such order as the Borrower shall request (and in the absence of such request, as the Administrative Agent shall determine). All payments

87

shall include payment of accrued interest on the principal amount so paid, shall be applied to the payment of interest before application to principal and shall include amounts payable, if any, under <u>Section 3.5</u>. Each payment applied to the Term Loans shall, to the extent permitted by applicable laws, be deemed to apply to the portion thereof that was used to repay and replace the loans originally incurred by the Borrower to purchase the Capital Stock of TG.

- Call Protection. In the event that prior to the first anniversary of the Effective Date, the Borrower makes any prepayment of all or part of the Term B Dollar Loans in connection with any replacement or refinancing of the Term B Dollar Loans with or from the proceeds of a new term loan having an "Applicable Eurocurrency Margin" (or other equivalent term) that, at any time prior to the first anniversary of the Effective Date is, or could upon satisfaction of certain conditions be, less than the Applicable Eurocurrency Margin for the Term B Dollar Loans, the Borrower shall pay a prepayment fee to Administrative Agent for the benefit of the Term B Dollar Lenders equal to 1% of the principal amount of the Term B Dollar Loans so refinanced, provided that the foregoing prepayment fee shall not be applicable if such prepayment is made in connection with the termination of all of the Revolving Commitments and a repayment of all of the Loans hereunder. Solely for purposes of this Section 4.5(c), any amendment, restatement or other modification to this Agreement prior to the first anniversary of the Effective Date that reduces the Applicable Eurocurrency Margin applicable to the Term B Dollar Loans shall be treated as if the Term B Dollar Loans were refinanced in full.
- D. <u>Prepayment of Term C Dollar Loans</u>. Notwithstanding anything to the contrary contained in this Agreement, including without limitation, any provision of <u>Article IV</u> hereof, the Borrower shall have the right to prepay in whole (but not in part) the outstanding Term B Dollar Loans without any obligation to prepay any portion of the Term C Dollar Loans.

12.6. Method and Place of Payment

A. Except as otherwise specifically provided herein, all payments under this Agreement shall be made to the Administrative Agent, for the ratable account of the Lenders entitled thereto, not later than 1:00 p.m. (New York City time) on the date when due and shall be made in immediately available funds and in each case to the account specified therefor for the Administrative Agent or if no account has been so specified at the Payment Office. The Administrative Agent will thereafter cause to be distributed on the same day (if payment was actually received by the Administrative Agent prior to 1:00 p.m. (New York City time) on such day) like funds relating to the payment of principal or interest or fees ratably to the Lenders entitled to receive any such payment in accordance with the terms of this Agreement. If and to the extent that any such distribution shall not be so made by the Administrative Agent in full on the same day (if payment was actually received by the Administrative Agent prior to 1:00 p.m. (New York City time) on such day), the Administrative Agent shall pay to each Lender its ratable amount thereof and each such Lender shall be entitled to receive from the Administrative Agent, upon demand, interest on such amount at the overnight Federal Funds Rate (or the applicable cost of funds with respect to amounts denominated in Euros or Sterling) for each day from the date such amount is paid to the Administrative Agent until the date the Administrative Agent pays such amount to such Lender.

88

B. Any payments under this Agreement which are made by the Borrower later than 1:00 p.m. (New York City time) shall, for the purpose of calculation of interest, be deemed to have been made on the next succeeding Business Day. Whenever any payment to be made hereunder shall be stated to be due on a day which is not a Business Day, the due date thereof shall be extended to the next succeeding Business Day and, with respect to payments of principal, interest shall be payable during such extension at the applicable rate in effect immediately prior to such extension, except that with respect to Eurocurrency Loans, if such next succeeding applicable Business Day is not in the same month as the date on which such payment would otherwise be due hereunder or under any Note, the due date with respect thereto shall be the next preceding Business Day.

12.7. Net Payments

A. All payments made by the Borrower hereunder or under any Loan Document will be made without setoff, counterclaim or other defense. All payments hereunder and under any of the Loan Documents (including, without limitation, payments on account of principal and interest and fees) shall be made by the Borrower free and clear of and without deduction or withholding for or on account of any present or future tax, duty, levy, impost, assessment or other charge of whatever nature now or hereafter imposed by any Governmental Authority, but excluding therefrom (i) any tax imposed on or measured by the overall net income (including franchise taxes imposed in lieu of net income taxes) of the Lender or the lending office of the Lender in respect of which the payment is made by the United States or by the jurisdiction (or any political subdivision or taxing authority thereof) in which its lending office is located, (ii) any branch profits tax imposed by the United States or any similar tax imposed by the jurisdiction in which the Borrower is located, (iii) in the case of any Lender organized under the laws of any jurisdiction other than the United States or any state thereof (including the District of Columbia), any taxes imposed by the United States by means of withholding at the source unless such withholding results from a change in applicable law, treaty or regulations or the interpretation or administration thereof (including, without limitation, any guideline or policy not having the force of law) by any authority charged with the administration thereof subsequent to the date such Lender becomes a Lender hereunder, (iv) any taxes to which the Lender is subject (to the extent of the tax rate then in effect) on the date this Agreement is executed

or to which such Lender would be subject on such date if a payment hereunder had been received by the Lender on such date and with respect to any Lender that becomes a party hereto after the date hereof, any taxes to which such Lender is subject on the date it becomes a party hereto (other than taxes which each of the other Lenders is entitled to reimbursements for pursuant to the terms of this Agreement), (v) taxes to which the Lender becomes subject subsequent to the date referred to in clause (iv) above as a result of a change in the residence, place of incorporation, or principal place of business of the Lender, a change in the branch or lending office of the Lender participating in the transactions set forth herein or other similar circumstances or as a result of the recognition by the Lender of gain on the sale, assignment or participation by the Lender of the participating interests in its creditor positions hereunder and (vi) any withholding tax that is imposed as a result of a Lender's failure to comply with the provisions of Section 4.7(d) (such tax or taxes, other than the tax or taxes described in Sections 4.7(a)(i) through (vi), being herein referred to as "Tax" or "Taxes"). If the Borrower is required by law to make any deduction or withholding of any Taxes from any payment due hereunder or

89

under any of the Loan Documents, then the amount payable will be increased to such amount which, after deduction from such increased amount of all such Taxes required to be withheld or deducted therefrom, will not be less than the amount due and payable hereunder had no such deduction or withholding been required. A certificate as to any additional amounts payable to a Lender under this Section 4.7 submitted to the Borrower by such Lender shall show in reasonable detail the amount payable and the calculations used to determine in good faith such amount and shall, absent manifest error, be final, conclusive and binding upon all parties hereto.

- B. If the Borrower makes any payment hereunder or under any of the Loan Documents in respect of which it is required by law to make any deduction or withholding of any Taxes, it shall pay the full amount to be deducted or withheld to the relevant taxation or other authority within the time allowed for such payment under applicable law and shall deliver to the Lenders within 30 days after it has made such payment to the applicable authority an original or certified copy of a receipt issued by the relevant Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to the Lender or Administrative Agent.
- C. Without prejudice to the other provisions of Section 4.7, if any Lender, or the Administrative Agent on its behalf, is required by law to make any payment on account of Taxes on or in relation to any amount received or receivable hereunder or under any of the Loan Documents by such Lender, or the Administrative Agent on its behalf, or any liability for Tax in respect of any such payment is imposed, levied or assessed against any Lender or the Administrative Agent on its behalf, the Borrower will promptly, following receipt of the certificate described in the immediately following sentence, indemnify such person against such Tax payment or liability, together with any interest, penalties and expenses (including reasonable counsel fees and expenses) payable or incurred in connection therewith, including any tax of any Lender or the Administrative Agent arising by virtue of payments under this Section 4.7(c), computed in a manner consistent with this Section 4.7(c). A certificate as to the amount of such payment by such Lender, or the Administrative Agent on its behalf, showing calculations thereof in reasonable detail, absent manifest error, shall be final, conclusive and binding upon all parties hereto for all purposes.
- D. (a) To the extent permitted by applicable law, each Lender that is a Non-U.S. Participant shall deliver to Borrower and Administrative Agent on or prior to the Initial Borrowing date (or in the case of a Lender that is an Assignee, on the date of such assignment to such Lender) two accurate and complete original signed copies of IRS Form W-8BEN, W-8ECI, or W-8IMY (or any successor or other applicable form prescribed by the IRS) certifying to such Lender's entitlement to a complete exemption from, or a reduced rate of, United States withholding tax on interest payments to be made under this Agreement or any Note. If a Lender that is a Non-U.S. Participant is claiming a complete exemption from withholding on interest pursuant to Section 881(c) of the Code, the Lender shall deliver (along with two accurate and complete original signed copies of IRS Form W-8BEN) a certificate substantially in the form of Exhibit 4.7(d) (any such certificate, a Section 4.7(d)(i) Certificate"). In addition, each Lender that is a Non-U.S. Participant agrees that from time to time after the Initial Borrowing date, (or in the case of a Lender that is an Assignee, after the date of the assignment to such Lender), when a lapse in time (or change in circumstances occurs) renders the prior certificates hereunder obsolete or inaccurate in any material respect, such Lender shall, to the extent permitted under

90

applicable law, deliver to the Borrower and Administrative Agent two new and accurate and complete original signed copies of an IRS Form W-8BEN, W-8ECI, or W-8IMY (or any successor or other applicable forms prescribed by the IRS), and if applicable, a new Section 4.7(d)(i) Certificate, to confirm or establish the entitlement of such Lender or Agent to an exemption from, or reduction in, United States withholding tax on interest payments to be made under this Agreement or any Note.

- (b) Each Lender that is not a Non-U.S. Participant (other than any such Lender which is taxed as a corporation for U.S. federal income tax purposes) shall provide two properly completed and duly executed copies of IRS Form W-9 (or any successor or other applicable form) to Borrower and Administrative Agent certifying that such Lender is exempt from United States backup withholding tax. To the extent that a form provided pursuant to this Section 4.7(d) (ii) is rendered obsolete or inaccurate in any material respects as result of change in circumstances with respect to the status of a Lender, such Lender or Agent shall, to the extent permitted by applicable law, deliver to Borrower and Administrative Agent revised forms necessary to confirm or establish the entitlement to such Lender's exemption from United States backup withholding tax.
- E. Each Lender agrees that, as promptly as practicable after it becomes aware of the occurrence of any event or the existence of any condition that would cause the Borrower to make a payment in respect of any Taxes to such Lender pursuant to Section 4.7(a) or a payment in indemnification for any Taxes pursuant to Section 4.7(c), it will use reasonable efforts to make, fund or

maintain the Loan (or portion thereof) of such Lender with respect to which the aforementioned payment is or would be made through another lending office of such Lender if as a result thereof the additional amounts which would otherwise be required to be paid by such the Borrower in respect of such Loans (or portions thereof) or participation in Letters of Credit pursuant to Section 4.7(a) or Section 4.7(c) would be materially reduced, and if, as determined by such Lender, in its reasonable discretion, the making, funding or maintaining of such Loans or participation in Letters of Credit (or portions thereof) through such other lending office would not otherwise materially adversely affect such Loans or such Lender. The Borrower agrees to pay all reasonable expenses incurred by any Lender in utilizing another lending office of such Lender pursuant to this Section 4.7(e).

F. If the Administrative Agent or any Lender (or Participant) receives any refund with respect to any Taxes as to which it has been indemnified by the Borrower, or with respect to which the Borrower has paid additional amounts pursuant to this Section 4.7, it shall pay over to the Borrower an amount equal to such refund (but only to the extent of indemnity payments made, or additional amounts paid, by the Borrower under this Section 4.7 with respect to the Taxes giving rise to such refund), net of all out-of-pocket expenses of the Administrative Agent or such Lender (or Participant) and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund); provided that the Borrower, upon the request of the Administrative Agent or such Lender (or Participant), agrees to repay the amount paid over to the Borrower, to the Administrative Agent or such Lender (or Participant), together with any interest, penalties and additions to tax, in the event the Administrative Agent or such Lender (or Participant) is required to repay such refund to such Governmental Authority. This paragraph shall not be construed to require the Administrative Agent or any Lender (or

91

Participant) to make available its tax returns (or any other information relating to its taxes which it deems confidential) to the Borrower or any other Person.

SECTION 13

CONDITIONS OF CREDIT

13.1. Conditions Precedent to the Closing Date

The obligation of the Lenders to make the Initial Loan and the obligation of the Facing Agent to issue and the Lenders to participate in Letters of Credit under this Agreement were subject to the fulfillment, on or prior to the Closing Date, of each of the following conditions (capitalized terms used in this Article but not defined herein have the meanings ascribed to such terms in this Agreement prior to giving effect to the Third Amendment):

- A. <u>Credit Agreement and Notes.</u> The Borrower, the Administrative Agent and each Lender shall have duly executed and delivered to the Administrative Agent, with a signed counterpart for each Lender, this Agreement (including all schedules and exhibits), and the Borrower shall have duly executed and delivered to the Administrative Agent the Notes payable to the order of each applicable Lender which has requested a Note in the amount of their respective Commitments and all other Loan Documents shall have been duly executed and delivered by the appropriate Credit Party to Agent, all of which shall be in full force and effect;
- B. <u>Collateral Security Agreement: UK Debenture.</u> The Borrower and each Subsidiary Guarantor shall have duly authorized, executed and delivered a Collateral Security Agreement in form and substance satisfactory to the Administrative Agent (as modified, supplemented or amended from time to time, the "<u>Collateral Security Agreement</u>") and shall have delivered to Collateral Agent all the Pledged Securities and Pledged Intercompany Notes referred to therein then owned, if any, by such Credit Party, (y) endorsed in blank in the case of promissory notes constituting Pledged Securities referred to therein then owned, if any, by such Credit Party, and (z) together with executed and undated stock powers, in the case of capital stock constituting Pledged Securities and the other documents and instruments required to be delivered under the Collateral Security Agreement and TG shall have duly authorized, executed and delivered an English law governed Guarantee and Debenture in favor of the UK Security Trustee in form and substance satisfactory to the Administrative Agent (as modified, supplemented or amended from time to time, the "<u>UK Debenture</u>") together with:
 - (a) proper financing statements (Form UCC-1 or such other financing statements or similar notices as shall be required by local law) fully executed for filing under the UCC or other appropriate filing offices of each jurisdiction as may be necessary or, in the reasonable opinion of the Administrative Agent, desirable to perfect the security interests purported to be created by the Collateral Security Agreement;
 - (b) certified copies of Requests for Information or Copies (Form UCC-7), or equivalent reports, listing all effective financing statements or similar notices that name the Borrower or its Subsidiaries (by its actual name or any trade

92

name, fictitious name or similar name), or any division or other operating unit thereof, as debtor and that are filed in the jurisdiction referred to in said clause (i), together with copies of such other financing statements (none of which shall cover the Collateral except to the extent evidencing Permitted Liens or for which the Administrative Agent shall have received satisfactory evidence of release);

(c) evidence of the completion (or arrangements acceptable to the Administrative Agent for the

completion) of all other recordings and filings of, or with respect to, the Collateral Security Agreement and the UK Debenture and all other actions as may be necessary or, in the opinion of the Administrative Agent, desirable to perfect the security interests intended to be created by the Collateral Security Agreement, the UK Debenture or any other Security Document;

- (d) such amendments, modifications or supplements to the Pledged Intercompany Notes as may be reasonably requested by the Administrative Agent, each such amendment, modification or supplement to be in a form reasonably satisfactory to the Administrative Agent; and
- (e) evidence that all other actions necessary, or in the reasonable opinion of the Administrative Agent, desirable to perfect the security interests purported to be taken by the Collateral Security Agreement have been taken;
- C. <u>Pledge Agreement; Charges Over Shares.</u> (i) The Borrower and each Subsidiary Guarantor shall have duly executed and delivered a Pledge Agreement in the form of <u>Exhibit 5.1(c)</u> (as modified, supplemented or amended from time to time, the "<u>Pledge Agreement</u>") and (ii) the Borrower or the applicable Subsidiary Guarantors shall have duly executed and delivered English law governed charges over shares in form and substance satisfactory to the Administrative Agent (as modified, supplemented or amended from time to time, the "<u>UK Pledge Agreements</u>") in respect to any shares of TG or UK Holdco 1 held by the Borrower or a Subsidiary Guarantor;

D. <u>Subsidiary Guaranty Agreements</u>.

- (a) <u>Subsidiary Guaranty</u>. Each Subsidiary Guarantor shall have duly executed and delivered the Subsidiary Guaranty substantially in the form of <u>Exhibit 5.1(d)(i)</u>;
- (b) <u>Headquarters Subsidiary Guaranty Agreement</u>. Huntsman Headquarters Corporation shall have duly executed and delivered the Headquarters Subsidiary Guaranty Agreement substantially in the form of <u>Exhibit 5.1(d)(ii)</u>;
- E. <u>Mortgages; Mortgage Policies; Surveys</u>. The Administrative Agent shall have received:
- (a) fully executed counterparts of a deed of trust, all in form and substance satisfactory to the Administrative Agent (the "Mortgages"), which Mortgage shall cover the Mortgaged Property of the Borrower or a Domestic

93

Subsidiary, together with a recording instruction letter from Vinson & Elkins L.L.P., addressed to and accepted by the relevant title insurance company under which such title insurance company accepts delivery of executed counterparts of the applicable Mortgage to be promptly delivered to the appropriate recorder's office for recording in all places to the extent necessary or desirable, in the reasonable judgment of the Administrative Agent, to create a valid and enforceable first priority lien (subject to Permitted Real Property Encumbrances) on the applicable Mortgaged Property, subject only to Permitted Liens, in favor of Collateral Agent (or such other trustee as may be required or desired under local law) for the benefit of the Secured Parties;

- (b) mortgagee title insurance policies issued by title insurance companies satisfactory to the Administrative Agent (the "Mortgage Policies") with respect to the Mortgaged Properties in amounts satisfactory to the Administrative Agent assuring the Administrative Agent that the Mortgages with respect to such Mortgaged Properties are valid and enforceable first priority mortgage liens on the respective Mortgaged Properties, free and clear of all defects, encumbrances and other Liens except Permitted Liens, and the Mortgage Policies shall be in form and substance satisfactory to the Administrative Agent and shall include, as appropriate, an endorsement for future advances under this Agreement and the Notes and for any other matter that the Administrative Agent in its discretion may request, shall not include an exception for mechanics' liens, and shall provide for affirmative insurance and such reinsurance as the Administrative Agent in its discretion may request; and
- (c) to the extent requested by the Administrative Agent, a survey, in form and substance satisfactory to the Administrative Agent, of each Mortgaged Property dated a recent date acceptable to the Administrative Agent, certified by a licensed professional surveyor satisfactory to the Administrative Agent, provided, however, in the event that any survey delivered pursuant to this provision is dated on a date which is more than six (6) months prior to the Closing Date, but less than one (1) year prior to the Closing Date, such survey shall be acceptable so long as such survey otherwise complies with the ALTA/ACSM standards required by the Administrative Agent, and the owner and/or lessee of the Mortgaged Property delivers a "no change survey affidavit" in a form which is acceptable to the title insurance company issuing the Mortgage Policy, so that the title insurance company will delete any general survey exception in such Mortgage Policy;
- F. <u>Perfection</u>. Each Credit Party shall have delivered to the Administrative Agent true and correct copies of Perfection Certificates in the form of <u>Exhibit 5.1(f)</u> (the "<u>Perfection Certificates</u>"), each of which shall be in full force and effect and in form and substance satisfactory to the Administrative Agent as of the Closing Date;
- G. <u>Termination of Prior Credit Agreements</u>. On the Closing Date, the total commitments under each of the Prior Credit Agreements shall have been terminated, all loans thereunder shall have been repaid in full, together with interest thereon, and all other amounts

owing pursuant to such agreements shall have been repaid in full and such agreements shall have been terminated on terms and conditions satisfactory to the Administrative Agent and the Required Lenders and be of no further force or effect and the creditors thereunder shall have terminated, released or modified all security interests and Liens on the assets owned by the Borrower and its Subsidiaries in a manner satisfactory to the Administrative Agent, it being understood and agreed that for all purposes under this Agreement, such repayment shall be deemed to occur simultaneously with the effectiveness of this Agreement;

- H. <u>Intercreditor Agreement and Public Note Document Deliveries</u>. The Administrative Agent shall have received (i) a fully executed copy of the Intercreditor Agreement and (ii) copies of all opinions and certificates delivered pursuant to the terms of any Public Note Document in connection with the Transactions;
 - I. <u>Documents</u>. The Administrative Agent shall have received certified copies of:
 - (a) the UK Holdco Note executed by UK Holdco 1 payable to Huntsman Finco;
 - (b) Foreign Intercompany Notes executed by each Foreign Subsidiary that received an Intercompany Loan from UK Holdco 1 in connection with the Prior HI Credit Agreement;
 - (c) Foreign Intercompany Loan Security Documents, in form and substance acceptable to the Administrative Agent, executed by each Foreign Subsidiary listed on Schedule 5.1(i)(iii); and
 - (d) the UK Petrochem Holdings Note, along with the guaranty of the UK Petrochem Holdings Note by TG;
- J. <u>Corporate Proceedings.</u> The Administrative Agent shall have received from each Credit Party a certificate, dated the Closing Date, signed by a Responsible Officer of such Person, and attested to by the secretary or any assistant secretary, or equivalent officer, or any manager (in the case of a limited liability company) of such Person with appropriate insertions, together with copies of such Person's Organizational Documents and the consents of the members of such Person referred to in such certificate and all of the foregoing (including each such Organizational Document and consent) shall be satisfactory to the Administrative Agent; and
 - (a) All corporate and/or limited liability company and legal proceedings and all instruments and agreements to be executed by each Credit Party in connection with the transactions contemplated by this Agreement and the Loan Documents shall be reasonably satisfactory in form and substance to the Administrative Agent, and the Administrative Agent shall have received all information and copies of all certificates, documents and papers, including good standing certificates, bring-down certificates and any other records of corporate and/or limited liability company proceedings and governmental approvals, if any, which the Administrative Agent reasonably may have requested in connection

95

therewith, such documents and papers, where appropriate, to be certified by proper corporate or governmental authorities;

- (b) The ownership and capital structure (including without limitation, the terms of any capital stock, options, warrants or other securities issued by Borrower or any of its Subsidiaries) of the Borrower and its Subsidiaries shall be in form and substance reasonably satisfactory to the Administrative Agent and the Lenders;
- K. <u>Foreign Intercompany Loan Corporate Proceedings</u>. The Administrative Agent shall have received from each Foreign Subsidiary of the Borrower party to a Foreign Intercompany Loan Security Document, a copy (certified as being a true, complete and up to date copy by the secretary of such Person) of such Person's Organizational Documents;
- L. <u>Incumbency</u>. The Administrative Agent shall have received a certificate of the secretary or assistant secretary, or equivalent officer, or any manager (in the case of a limited liability company) of each Credit Party, dated the Closing Date, as to the incumbency and signature of the officers of each such Person executing any document (in form and substance satisfactory to the Administrative Agent) and any certificate or other document or instrument to be delivered pursuant hereto or thereto by or on behalf of such Person, together with evidence of the incumbency of such secretary, assistant secretary, or equivalent officer or any manager (in the case of a limited liability company);
- M. <u>Financial Statements</u>. The Borrower shall have delivered to the Administrative Agent and each Lender the financial statements as provided in <u>Section 6.5(a)</u> in form and substance satisfactory to the Administrative Agent and the Required Lenders:
- N. <u>Approvals</u>. All necessary governmental (domestic and foreign) and third party approvals in connection with this Agreement and the transactions contemplated hereby and otherwise referred to herein shall have been obtained and remain in effect, and all applicable waiting periods shall have expired without any action being taken by any competent authority which restrains, prevents or imposes materially adverse conditions upon the consummation of all or any part of this Agreement or the transactions contemplated hereby and otherwise referred to herein except for those approvals of non-Governmental Authorities under contracts which are not

material and which are not required to be delivered at the closing thereof. Additionally, there shall not exist any judgment, order, injunction or other restraint issued or filed or a hearing seeking injunctive relief or other restraint pending or notified prohibiting or imposing material adverse conditions upon all or any part of this Agreement or the transactions contemplated hereby, or the making of the Loans or the issuance of Letters of Credit;

O. <u>Litigation</u>. No litigation by any entity (private or governmental) shall be pending or, to the best knowledge of the Borrower, threatened with respect to this Agreement, any other Loan Document or any documentation executed in connection herewith or the transactions contemplated hereby, or which the Administrative Agent or the Required Lenders shall determine could reasonably be expected to have a Material Adverse Effect;

96

- P. <u>Public Notes</u>. HLLC shall have delivered to the Administrative Agent certified copies of all material documents executed in connection with the Transaction pursuant to any Public Note Documents, including any certificates and legal opinions relating thereto, each in form and substance acceptable to the Administrative Agent;
- Q. Merger. The Merger shall have been consummated in accordance with the Merger Agreement and applicable law. The Administrative Agent shall have received copies of the Merger Agreement and all certificates and other documents delivered thereunder. The Administrative Agent shall be reasonably satisfied with the material terms and conditions of the Merger;
- R. <u>Pro Forma Balance Sheet</u>. The Administrative Agent shall have received the Pro Forma Balance Sheet in form and substance satisfactory to the Administrative Agent and the Required Lenders;
- S. Opinions of Counsel. The Administrative Agent shall have received from (i) Vinson & Elkins L.L.P., special counsel to the Borrower, an opinion addressed to the Administrative Agent and each of the Lenders and dated the Closing Date, which shall be in substantially the form of Exhibit 5.1(s)(i), (ii) Stoel Rives LLP, special Utah counsel to the Borrower, an opinion addressed to the Administrative Agent and each of the Lenders and dated the Closing Date, which shall be in substantially the form of Exhibit 5.1(s) (iii), (iii) Alvord and Alvord, special Surface Transportation Board Counsel, an opinion addressed to the Administrative Agent and each of the Lenders and dated the Closing Date, which shall be in substantially the form of Exhibit 5.1(s)(iii) or shall have made arrangements satisfactory to the Administrative Agent for receipt of such opinion within thirty (30) days after the Closing Date and (iv) local counsel to the Borrower (in the United States and in England), an opinion addressed to the Administrative Agent and each of the Lenders and dated the Closing Date, in form and substance satisfactory to the Administrative Agent, covering the perfection of the security interests granted pursuant to the Security Documents;
- T. <u>Fees.</u> The Borrower shall have paid to the Agents and the Lenders all costs, fees and expenses (including, without limitation, legal fees and expenses) payable to the Agents and the Lenders to the extent then due including all breakage, if any, and other fees, interest and expenses due and owing under the Prior Credit Agreements as if paid in full;
- U. <u>Solvency.</u> The Administrative Agent shall have received a solvency certificate, in form and substance reasonably satisfactory to the Administrative Agent, executed by a Responsible Officer on behalf of the Borrower with respect to the solvency of the Borrower;
- V. <u>Tax Sharing Agreement</u>. The Administrative Agent shall have received a certified copy of the fully executed Tax Sharing Agreement;
- W. <u>Environmental Report</u>. The Administrative Agent shall have received access to copies of the most recent environmental risk assessment reports in the possession of the Borrower or its Subsidiaries or performed at the request of the Borrower or its Subsidiaries;

97

- X. <u>Insurance</u>. The Administrative Agent shall be satisfied with the insurance coverage in effect on the Closing Date pertaining to the assets of the Borrower and the Subsidiary Guarantors, and shall have received evidence satisfactory to it that the Administrative Agent shall have been named as a loss payee, mortgagee and additional insured on all such policies of insurance, as appropriate;
- Y. <u>Whitewash Procedures</u>. The Administrative Agent shall be satisfied that each element of the whitewash procedures with respect to TG has been completed;
- Z. <u>Officer's Certificate.</u> The Administrative Agent shall have received a certificate executed by a Responsible Officer on behalf of the Borrower, dated the Closing Date and in form and substance satisfactory to the Administrative Agent;
- AA. <u>Existing Indebtedness</u>. After giving effect to this Agreement and the other transactions contemplated hereby, Borrower and its Subsidiaries shall not have any Indebtedness outstanding except for the Loans, the Public Notes and the Indebtedness listed on <u>Schedule 8.2(b)</u> (to this Agreement prior to giving effect to the Third Amendment) and other Indebtedness permitted by <u>Section 8.2</u>;
 - BB. <u>Debt Ratings</u>. The Borrower shall have received a prospective senior secured debt rating with the respect to

CC. Other Matters. All corporate and other proceedings taken in connection with this Agreement at or prior to the date of this Agreement, and all documents incident thereto will be reasonably satisfactory in form and substance to the Administrative Agent; and the Administrative Agent shall have received such other instruments and documents as the Administrative Agent shall reasonably request in connection with the execution of this Agreement, and all such instruments and documents shall be reasonably satisfactory in form and substance to the Administrative Agent.

13.2. Conditions Precedent to All Credit Events

The obligation of each Lender to make Loans (including Loans made on the Effective Date) and the obligation of any Facing Agent to issue or any Lender to participate in any Letter of Credit hereunder in each case shall be subject to the fulfillment at or prior to the time of each such Credit Event of each of the following conditions:

- A. <u>Representations and Warranties.</u> The representations and warranties contained in this Agreement and the other Loan Documents shall each be true and correct in all material respects at and as of such time, as though made on and as of such time, except to the extent such representations and warranties are expressly made as of a specified date in which event such representation and warranties shall be true and correct as of such specified date;
- B. <u>No Default.</u> No Event of Default or Unmatured Event of Default shall have occurred and shall then be continuing on such date or will occur after giving effect to such Credit Event;
 - C. Notice of Borrowing; Notice of Issuance.

98

- (a) Prior to the making of each Loan, the Administrative Agent shall have received a Notice of Borrowing meeting the requirements of <u>Section 2.5</u>.
- (b) Prior to the issuance of each Letter of Credit, the Administrative Agent and the respective Facing Agent shall have received a Notice of Issuance meeting the requirements of Section 2.9(b);
- D. <u>Other Information.</u> The Administrative Agent shall have received such other instruments, documents and opinions as it may reasonably request in connection with such Credit Event, and all such instruments and documents shall be reasonably satisfactory in form and substance to the Administrative Agent.

The acceptance of the benefits of each such Credit Event by the Borrower shall be deemed to constitute a representation and warranty by it to the effect of paragraphs (a), (b), (c) and (d) of this <u>Section 5.2</u> (except that no opinion need be expressed as to any Agent's or Required Lenders' satisfaction with any document, instrument or other matter).

Each Lender hereby agrees that by its execution and delivery of its signature page hereto and by the funding of its Loan to be made on the Closing Date, such Lender approves of and consents to each of the matters set forth in Section 5.1, and Section 5.2 which must be approved by, or which must be satisfactory to, the Agents or the Required Lenders or Lenders, as the case may be; provided that, in the case of any agreement or document which must be approved by, or which must be satisfactory to, the Required Lenders, the Administrative Agent or the Borrower shall have delivered a copy of such agreement or document to such Lender on or prior to the Closing Date if requested.

13.3. Additional Conditions to All Credit Events

In addition to the conditions precedent set forth in Section 5.2, so long as any Revolving Lender is an Impaired Lender, no Facing Agent will be required to issue any Letter of Credit or to amend any outstanding Letter of Credit to increase the Stated Amount thereof, alter the drawing terms thereunder or extend the expiry date thereof, and the Swing Line Lender will not be required to make any Swing Line Loan, unless the Administrative Agent, the Swing Line Lender or such Facing Agent, as the case may be, is reasonably satisfied that any exposure that would result therefrom is eliminated or fully covered by the Revolving Commitments of the Non-Impaired Lenders or by Cash Collateralization or other arrangement or a combination thereof satisfactory to the Borrower, the Administrative Agent and the Swing Line Lender or such Facing Agent, as the case may be.

13.4. Additional Conditions to Usage of Revolving Commitments

In addition to the conditions precedent set forth in <u>Sections 5.2</u> and <u>5.3</u>:

A. if at the end of any Fiscal Quarter, (i) the Assigned Dollar Value of all outstanding Revolving Loans and Swing Line Loans is \$0, (ii) outstanding LC Obligations have been Cash Collateralized in an amount not less than 105% of the Stated Amount thereof and (iii) the Borrower would not have been in compliance with Section 9.1 if there had been a Revolving Loan of at least \$1 outstanding on such date, then (A) the Borrower shall not request a

Stated Amount thereof, request an issuance, increase or extension of a Letter of Credit, until the date of delivery (the "Certificate Delivery Date") to the Administrative Agent of a certificate demonstrating compliance with <u>Section 9.1</u> as of the last day of the next Fiscal Quarter for which the Borrower is in compliance therewith and (B) the Borrower agrees that all such Cash Collateralization shall remain in place until the Certificate Delivery Date; and

B. at any time from and after the Certificate Delivery Date until the date of delivery to the Administrative Agent of a certificate demonstrating compliance with Section 9.1 as of the last day of the next Fiscal Quarter for which the Borrower is then in compliance with Section 9.1, no Revolving Loan or Swing Line Loan may be made or requested and no Letter of Credit may be issued, extended or increased or requested (except to the extent that the same shall be Cash Collateralized in an amount not less than 105% of the Stated Amount thereof), on any date, if the Senior Secured Leverage Ratio as of such date, after giving effect to such Loan or issuance, extension or increase, on a Pro Forma Basis, would exceed 3.75 to 1.00.

SECTION 14

REPRESENTATIONS AND WARRANTIES

In order to induce the Lenders to enter into this Agreement and to make the Loans, and issue (or participate in) the Letters of Credit as provided herein, the Borrower makes the following representations and warranties as of the Effective Date and as of the date of each subsequent Credit Event, all of which shall survive the execution and delivery of this Agreement and the Notes and the making of the Loans and issuance of the Letters of Credit:

14.1. Corporate Status

The Borrower and each of its Subsidiaries (i) is a duly organized and validly existing corporation, partnership or limited liability company or other entity in good standing under the laws of the jurisdiction of its organization (or the equivalent thereof in the case of Foreign Subsidiaries), (ii) has the requisite power and authority to own its property and assets and to transact the business in which it is engaged and presently proposed to engage in and (iii) is duly qualified and is authorized to do business and is in good standing ((where relevant) in (y) its jurisdiction of organization and (z) each other jurisdiction where the ownership, leasing or operation of property or the conduct of its business requires such qualification, authorization or good standing, except (1) for such failure to be so qualified, authorized or in good standing which, in the aggregate, would not have a Material Adverse Effect and (2) as a result of any transaction expressly permitted under Section 8.3 hereof.

14.2. Corporate Power and Authority

The Borrower and each of its Subsidiaries has the applicable power and authority to execute, deliver and perform the terms and provisions of each of the Documents to which it is a party and has taken all necessary corporate or other appropriate action to authorize the execution, delivery and performance by it of each of such Documents. As of the Effective Date

100

(or such later date as a Document is to be executed and delivered in accordance with the terms hereof) the Borrower and each of its Subsidiaries has duly executed and delivered each of the Documents to which it is a party, and each of such Documents constitutes its legal, valid and binding obligation enforceable in accordance with its terms, except to the extent that the enforceability thereof may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws generally affecting creditors' rights and by equitable principles (regardless of whether enforcement is sought in equity or at law).

14.3. No Violation

Neither the execution, delivery or performance by the Borrower and each of its Subsidiaries of the Documents to which it is a party (including, without limitation, the granting of Liens pursuant to the Security Documents, nor compliance by it with the terms and provisions thereof, nor the consummation of the transactions contemplated therein (i) will contravene any provision of any Requirement of Law applicable to the Borrower or any of its Subsidiaries, (ii) will conflict with or result in any breach of or constitute a tortious interference with any of the terms, covenants, conditions or provisions of, or constitute a default under, or result in the creation or imposition of (or the obligation to create or impose) any Lien (except pursuant to the Security Documents) upon any of the property or assets of the Borrower or any of its Subsidiaries pursuant to the terms of any material Contractual Obligation to which the Borrower or any of its Subsidiaries is a party or by which it or any of its property or assets is bound or to which it may be subject, (iii) will violate any provision of any Organizational Document of the Borrower or any of its Subsidiaries or (iv) require any approval of stockholders or any approval or consent of any Person (other than a Governmental Authority) except as have been obtained on or prior to the Effective Date.

14.4. Governmental and Other Approvals

Except for the recording of the Mortgages and filings (in respect of certain Security Documents) and actions with appropriate Governmental Authorities which shall be recorded and filed, respectively, on, or as soon as practicable after, the Closing Date, no order, consent, approval, license, authorization or validation of, or filing, recording or registration with (except as have been obtained or made on or prior to the Closing Date), or exemption by, any Governmental Authority, is required to authorize, or is required in connection with, (i) the execution, delivery and performance of any Document or (ii) the legality, validity, binding effect or enforceability of any such Document.

14.5. Financial Statements; Financial Condition; Undisclosed Liabilities; Projections; etc.

A. <u>Financial Statements</u> The balance sheet of the Borrower at December 31, 2005 and 2006 and the statements of income, cash flows and shareholders' equity of Borrower for the Fiscal Years ended December 31, 2004, 2005 and 2006 or other period ended on such dates, as the case may be, fairly present in all material respects the financial condition and results of operation and cash flows of Borrower and its consolidated subsidiaries as of such dates and for such periods. Copies of such statements have been furnished to the Lenders prior to the

101

Effective Date and have been examined by Deloitte & Touche LLP, independent certified public accountants, who delivered an unqualified opinion in respect thereto.

- B. <u>Solvency.</u> On and as of the Effective Date, after giving effect to this Agreement and to all Indebtedness (including the Loans) being incurred, and to be incurred (and the use of proceeds thereof), and Liens created, and to be created, by the Borrower and its Subsidiaries in connection with the transactions contemplated hereby, the Borrower and each of its Material Subsidiaries are Solvent.
- C. No Undisclosed Liabilities. Except as fully reflected in the financial statements and the notes related thereto delivered pursuant to Section 6.5(a) and set forth on Schedule 6.5(c) there were, to the best of Borrower's knowledge, as of the Effective Date no liabilities or obligations other than in the ordinary course of business consistent with past practices (with respect to the Borrower and its Subsidiaries of any nature whatsoever (whether absolute, accrued, contingent or otherwise and whether or not due) which, either individually or in aggregate, would be material to the Borrower and its Subsidiaries, taken as a whole. As of the Effective Date, the Borrower does not know of any basis for the assertion against the Borrower or any of its Subsidiaries of any such liability or obligation of any nature whatsoever that is not fully reflected in the financial statements or the notes related thereto delivered pursuant to Section 6.5(a) or set forth on Schedule 6.5(c) which, either individually or in the aggregate, could be material to the Borrower and its Subsidiaries, taken as a whole.
- D. <u>Indebtedness</u>. <u>Schedule 8.2(b)(ii)</u> sets forth a true and complete list of all Indebtedness (other than the Loans, the Letters of Credit and the Public Notes of the Borrower and its Subsidiaries) as of the Effective Date, to the extent that, in each case, such Indebtedness is in excess of \$5,000,000 (provided, that the aggregate principal amount of Indebtedness not so listed does not exceed the Dollar Equivalent (as determined on the Effective Date) of \$20,000,000), in each case showing the aggregate principal amount thereof (and the aggregate amount of any undrawn commitments with respect thereto) and the name of the respective obligor and any other entity which directly or indirectly guaranteed such debt. The Borrower has delivered or caused to be delivered to the Administrative Agent a true and complete copy of the form of each instrument evidencing Indebtedness for money borrowed listed on Schedule 8.2(b)(ii) and of each instrument pursuant to which such Indebtedness for money borrowed was issued, in each case, other than Indebtedness of the type described in Section 8.2(b)(xiii).
- E. <u>Projections/Budgets</u>. On and as of the Effective Date, the financial projections, attached hereto as <u>Schedule 6.5(e)</u> and each of the budgets delivered after the Effective Date pursuant to <u>Section 7.2(e)</u> (collectively, the "<u>Projections</u>") are, or will be at the time made, based on good faith estimates and assumptions made by the management of the Borrower, and there are no statements or conclusions in any of the Projections which, at the time made, are based upon or include information known to the Borrower to be misleading in any material respect or which fail to take into account material information known to Borrower regarding the matters reported therein. On and as of the Effective Date, the Borrower believes that the Projections are reasonable and attainable, it being understood that uncertainty is inherent in any forecast or projection and that no assurance can be given that the results set forth in the Projections will actually be attained or that the projections will be suitable or sufficient for any purpose relevant to the Lenders.

102

F. <u>No Material Adverse Change</u>. As of the Effective Date and at any time thereafter, there has been no material adverse change in the business, condition (financial or otherwise), assets, liabilities or operations of the Borrower and its Subsidiaries (taken as a whole) since December 31, 2006 based on the financial statements delivered pursuant to <u>Section 6.5(a)</u>.

14.6. Litigation

There are no actions, suits or proceedings pending or, to the best knowledge of the Borrower or any of its Subsidiaries, threatened in writing against the Borrower or any of its Subsidiaries (i) with respect to any Loan Document or (ii) that are reasonably likely to have a Material Adverse Effect.

14.7. Disclosure

All factual information (taken as a whole) heretofore or contemporaneously furnished by or on behalf of the Borrower or any of its Subsidiaries in writing to any Lender (including, without limitation, all information contained in the Documents) (other than the Projections as to which Section 6.5(e) applies, which fairly discloses the matters therein in good faith) for purposes of or in connection with this Agreement or any transaction contemplated herein is, and all other such factual information (taken as a whole) hereafter furnished by or on behalf of the Borrower or any of its Subsidiaries in writing to any Lender for purposes of or in connection with this Agreement or any transaction contemplated herein are and will be true and accurate in all material respects on the date as of which such information is dated or certified and not incomplete by omitting to state any fact necessary to make such information (taken as a whole) not misleading in any material respect at such time in light of the circumstances under which such information was provided. As of the Effective Date, the Borrower has disclosed to the Lenders on or before the Effective Date, all contractual, corporate or other restrictions

to which the Borrower or any of its Subsidiaries is or will be subject as of the Effective Date, and all other matters known to any of them, that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Effect.

14.8. Use of Proceeds; Margin Regulations

- A. <u>Term Loan Proceeds</u>. All proceeds of the Additional Term Loans (as defined in the Third Amendment) incurred on the Effective Date shall be used by the Borrower to repay in full all Existing Term Loans (as defined in the Third Amendment) other than Converting Term Loans (as defined in the Third Amendment).
- B. <u>Revolving Loan Proceeds</u>. All proceeds of the Revolving Loans incurred hereunder shall be used by the Borrower for ongoing working capital needs and general corporate purposes (other than to voluntarily prepay Term Loans).
- C. <u>Swing Line Loans</u>. All proceeds of the Swing Line Loans incurred hereunder shall be used by the Borrower for ongoing working capital needs and general corporate purposes (other than to voluntarily prepay Term Loans); <u>provided</u>, <u>however</u>, that no Swing Line Loans may be requested by the Borrower on the Closing Date.

103

D. <u>Margin Regulations</u>. No part of the proceeds of any Loan will be used to purchase or carry any margin stock (as defined in Regulation U of the Board), directly or indirectly, or to extend credit for the purpose of purchasing or carrying any such margin stock for the purpose of reducing or retiring any indebtedness which was originally incurred to purchase or carry any margin security or for any other purpose which might cause any of the loans or extensions of credit under this Agreement to be considered a "purpose credit" within the meaning of Regulation T, U or X of the Board.

14.9. Tax Returns and Payments

The Borrower and each of the its Subsidiaries have timely filed or caused to be filed all tax returns which are required to be filed, except where failure to file any such returns would not reasonably be expected to have a Material Adverse Effect, and have paid or caused to be paid all taxes shown to be due and payable on said returns or on any assessments made against them or any of their respective material properties and all other material taxes, fees or other charges imposed on them or any of their respective properties by any Governmental Authority (other than those the amount or validity of which is contested in good faith by appropriate proceedings and with respect to which reserves in conformity with GAAP have been provided on the books of the Borrower or any such Subsidiary, as the case may be), except where failure to take any such action could not reasonably be expected to have a Material Adverse Effect; and no tax liens have been filed and no claims are being asserted with respect to any such taxes, fees or other charges (other than such liens or claims, the amount or validity of which is currently being contested in good faith by appropriate proceedings and with respect to which reserves in conformity with GAAP (or prior to the Effective Date, applicable accounting practice) have been provided) which could be reasonably expected to have a Material Adverse Effect.

14.10. Compliance With ERISA

A. Each Plan has been operated and administered in a manner so as not to result in any liability of the Borrower or any of its Subsidiaries for failure to comply with the applicable provisions of ERISA and the Code in excess of \$50,000,000; no Reportable Event which could reasonably be expected to result in the termination of any Plan has occurred with respect to a Plan; to the best knowledge of the Borrower, no Multiemployer Plan is insolvent or in reorganization; no Plan has an accumulated or waived funding deficiency, has permitted decreases in its funding standard account or has applied for an extension of any amortization period within the meaning of Section 412 of the Code; neither the Borrower nor any of its Subsidiaries nor any ERISA Affiliate has incurred any liability to or on account of a Plan pursuant to Section 409, 502(i), 502(l), 515, 4062, 4063, 4064, 4069, 4201 or 4204 of ERISA or Section 4971 or 4975 of the Code in excess of \$50,000,000; no proceedings have been instituted to terminate any Plan; using actuarial assumptions and computation methods consistent with subpart 1 of Subtitle E of Title IV of ERISA, the Borrower and its Subsidiaries and its ERISA Affiliates would not have any liability to all Multiemployer Plans in the event of a complete withdrawal therefrom, as of the close of the most recent Fiscal Year of each such Plan ending prior to the date of any Credit Event in excess of \$50,000,000; no Lien imposed under the Code or ERISA on the assets of the Borrower or any of its Subsidiaries or any ERISA Affiliate exists or is likely to arise on account of any Plan; and the Borrower and its Subsidiaries do not maintain

104

or contribute to any employee welfare benefit plan (as defined in Section 3(1) of ERISA) which provides benefits to retired employees (other than as required by Section 601 of ERISA) or any employee pension benefit plan (as defined in Section 3(2) of ERISA), the ongoing annual obligations with respect to either of which could reasonably be expected to have a Material Adverse Effect.

B. (i) Each Foreign Pension Plan is in compliance and in good standing (to the extent such concept exists in the relevant jurisdiction) in all respects with all laws, regulations and rules applicable thereto, including all funding requirements, and the respective requirements of the governing documents for such Foreign Pension Plan, except for such failures that individually or in the aggregate could neither (A) result in liabilities in excess of \$50,000,000 nor (B) reasonably be expected to result in a Material Adverse Effect; (ii) with respect to each Foreign Pension Plan maintained or contributed to by the Borrower or any Subsidiary, (x) that is required by applicable law to be funded in a trust or other funding vehicle is in material compliance with applicable law regarding funding requirements, except for such failures that individually or in the aggregate could neither (A) result in liabilities in excess of \$50,000,000 nor (B) reasonably be expected to result in a Material Adverse Effect, and (y) that is not required by applicable law to be funded in a trust

or other funding vehicle, reasonable book reserves have been established in accordance with prudent business practice or where required by ordinary accounting practices in the jurisdiction in which such Foreign Pension Plan is maintained; (iii) all material contributions required to have been made by the Borrower or any Subsidiary to any Foreign Pension Plan, except for such failures that individually or in the aggregate could neither (A) result in liabilities in excess of \$50,000,000 nor (B) reasonably be expected to result in a Material Adverse Effect have been made within the time required by law or by the terms of such Foreign Pension Plan; and (iv) to the knowledge of the Borrower and its Subsidiaries, no actions or proceedings have been taken or instituted to terminate or wind-up a Foreign Pension Plan with respect to which the Borrower and its Subsidiaries taken as a whole could have any liability, except for such failures that individually or in the aggregate could neither (A) result in liabilities in excess of \$50,000,000 nor (B) reasonably be expected to result in a Material Adverse Effect.

14.11. Ownership of Property

The Borrower and each of its Subsidiaries has good and marketable title or, with respect to real property, valid fee simple title (or in each case, the relevant foreign equivalent, if any) to, or a subsisting leasehold interest in, or a valid contractual agreement or other valid right to use, all such Person's material real property, and good title (or relevant foreign equivalent) to, a valid leasehold interest in, or valid contractual rights or other valid right to (or an agreement for the acquisition of same) use all such Person's other material property (but excluding Intellectual Property), and, in each case, none of such property is subject to any Lien except for Permitted Liens. Neither this Agreement nor any other Documents, nor any transaction contemplated under any such agreement, will affect any right, title or interest of the Borrower or any of its Subsidiaries in and to any of the assets of the Borrower or any such Subsidiary in a manner that would have or is reasonably likely to have a Material Adverse Effect. As of the Closing Date, the Borrower and its Domestic Subsidiaries have granted Mortgages to secure the Obligations on all parcels of real estate identified on Schedule 6.21(c) as Mortgaged Properties.

105

14.12. Capitalization of the Borrower

On the Effective Date, the capitalization of the Borrower will be as set forth on Schedule 6.12(a) hereto. The Capital Stock of the Borrower has been duly authorized and validly issued. Except as set forth on Schedule 6.12(a), no authorized but unissued or treasury shares of Capital Stock of the Borrower are subject to any option, warrant, right to call or commitment of any kind or character. A complete and correct copy of the limited liability company agreement of the Borrower in effect on the Effective Date has been delivered to the Administrative Agent. Except as set forth on Schedule 6.12(a), the Borrower does not have any outstanding stock or securities convertible into or exchangeable for any shares of its Capital Stock, or any rights issued to any Person (either preemptive or other) to subscribe for or to purchase, or any options for the purchase of, or any agreements providing for the issuance (contingent or otherwise) of, or any calls, commitments or claims or any character relating to any of its Capital Stock or any stock or securities convertible into or exchangeable for any of its Capital Stock. Neither the Borrower nor any of its Subsidiaries is subject to any obligation (contingent or otherwise) to repurchase or otherwise acquire or retire any shares of its Capital Stock or any convertible securities, rights or options of the type described in the preceding sentence. As of the Effective Date, all of the issued and outstanding shares of Capital Stock of the Borrower are owned of record by the stockholders as set forth on Schedule 6.12(a) hereto.

14.13. Subsidiaries

- A. <u>Organization.</u> <u>Schedule 6.13</u> sets forth as of the Effective Date a true, complete and correct list of each Subsidiary of the Borrower and indicates for each such Subsidiary (i) its jurisdiction of organization and (ii) its ownership (by holder and percentage interest). The Borrower has no Subsidiaries except for Subsidiaries permitted to be created pursuant to this Agreement, and those Subsidiaries listed as on Schedule 6.13.
- B. <u>Capitalization</u>. As of the Effective Date, all of the issued and outstanding Capital Stock of each Subsidiary of the Borrower has been duly authorized and validly issued, and, to the extent applicable in the case of Foreign Subsidiaries, is fully paid and non-assessable and is owned as set forth on <u>Schedule 6.13</u>, free and clear of all Liens except for Permitted Liens. No authorized but unissued or treasury shares of Capital Stock of any Subsidiary of the Borrower are subject to any option, warrant, right to call or commitment of any kind or character except as set forth on <u>Schedule 6.13</u>. On and after the relevant date of formation, the Borrower directly owns 100% of the Capital Stock of each Receivables Subsidiary owned directly by the Borrower, and the Borrower has pledged (and delivered for pledge) the Capital Stock of each such Receivables Subsidiary (and any promissory notes received by the Borrower or any other Credit Party from such Receivables Subsidiary) to the Collateral Agent pursuant to the Collateral Security Agreement.
- C. <u>Restrictions on or Relating to Subsidiaries.</u> There does not exist any consensual encumbrance or restriction on the ability of (i) any Subsidiary of the Borrower to pay dividends or make any other distributions on its Capital Stock or any other interest or participation in its profits owned by the Borrower or any Subsidiary of the Borrower, or to pay any Indebtedness owed to the Borrower or a Subsidiary of the Borrower or any of the Borrower or any of the Borrower's Subsidiaries or

106

(iii) the Borrower or any of its Subsidiaries to transfer any of its properties or assets to the Borrower or any of its Subsidiaries, except, in each case, for such encumbrances or restrictions permitted under Section 8.5.

Neither the Borrower nor any of its Subsidiaries is in default under or in violation of any Requirement of Law or material Contractual Obligation or under its Organizational Documents, as the case may be, in each case the consequences of which default or violation, either in any one case or in the aggregate, would have a Material Adverse Effect.

14.15. Investment Company Act

Neither the Borrower nor any of its Subsidiaries is an "investment company" or a company "controlled" by an "investment company", within the meaning of the Investment Company Act of 1940, as amended.

14.16. Subordination Provisions

The subordination provisions contained in the Senior Subordinated Note (2013) Documents, the Senior Subordinated Note (2014) Documents and the Senior Subordinated Note (2015) Documents are enforceable against the issuer of the respective security and the holders thereof, and the Loans and all other Obligations entitled to the benefits of any Loan Document and any related guaranty are within the definitions of "Senior Indebtedness" included in such provisions.

14.17. Environmental Matters

(i) The operations of and the real property owned or operated by the Borrower and each of its Subsidiaries are in compliance with all applicable Environmental Laws except where the failure to be in compliance, individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect; (ii) the Borrower and each of its Subsidiaries has obtained and will continue to maintain all Environmental Permits, and all such Environmental Permits are in good standing and the Borrower and its Subsidiaries are in compliance with all terms and conditions of such Environmental Permits, except where failure to so obtain, maintain or comply, individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect; (iii) neither the Borrower nor any of its Subsidiaries nor any of their present or past properties or operations (whether owned or leased) is subject to: (A) any Environmental Claim or other written claim, request for information, judgment, order, decree or agreement from or with any Governmental Authority or private party related to any material violation of or material non-compliance with Environmental Laws or Environmental Permits to the extent any of the foregoing could reasonably be expected to have a Material Adverse Effect, (B) any pending or, to the knowledge of the Borrower, threatened judicial or administrative proceeding, action, suit or investigation related to any Environmental Laws or Environmental Permits which could reasonably be expected to have a Material Adverse Effect, (C) any Remedial Action which if not taken could reasonably be expected to have a Material Adverse Effect or (D) any liabilities, obligations or costs arising from the Release or substantial threat of a material Release of a Contaminant into the environment where such Release or substantial threat of a material

107

Release could reasonably be expected to have a Material Adverse Effect; (iv) neither the Borrower nor any of its Subsidiaries has received any written notice or claim to the effect that the Borrower or any of its Subsidiaries is or may be liable to any Person as a result of the Release or substantial threat of a material Release of a Contaminant into the environment, which notice or claim could reasonably be expected to result in a Material Adverse Effect, and (v) no Environmental Lien has attached to any property (whether owned or leased) of the Borrower or of any of its Subsidiaries which could, if determined adversely to Borrower or any of its Subsidiaries, reasonably be expected to have a Material Adverse Effect, nor are there any facts or circumstances currently known to the Borrower or any of its Subsidiaries that may reasonably be expected to give rise to such an Environmental Lien.

14.18. Labor Relations

Neither the Borrower nor any of its Material Subsidiaries is engaged in any unfair labor practice that could reasonably be expected to have a Material Adverse Effect. Except to the extent that the same could not reasonably be expected to result in a Material Adverse Effect, there is (i) no significant unfair labor practice complaint pending against the Borrower or any of its Subsidiaries or, to the best knowledge of the Borrower, threatened against any of them before the National Labor Relations Board or appropriate national court or other forum, and no significant grievance or significant arbitration proceeding arising out of or under any collective bargaining agreement is so pending against the Borrower or any of its Subsidiaries or, to the best knowledge of the Borrower, threatened against any of them and (ii) no significant strike, labor dispute, slowdown or stoppage is pending against the Borrower or any of its Subsidiaries or, to the best knowledge of the Borrower, threatened against the Borrower or any of its Subsidiaries.

14.19. Intellectual Property, Licenses, Franchises and Formulas

Each of the Borrower and its Subsidiaries owns or holds licenses or other rights to or under all of the patents, patent applications, trademarks, service marks, trademark and service mark registrations and applications therefor, trade names, copyrights, copyright registrations and applications therefor, trade secrets, proprietary information, computer programs or data bases (collectively, "Intellectual Property") except where the failure to own or hold such Intellectual Property could not reasonably be expected to result in a Material Adverse Effect, and has obtained assignments of all franchises, licenses and other rights of whatever nature, regarding Intellectual Property necessary for the present conduct of its business, without any known conflict with the rights of others, except such conflicts which could not reasonably be expected to have a Material Adverse Effect. Neither the Borrower nor any of its Subsidiaries has knowledge of any existing or threatened claim by any Person contesting the validity, enforceability, use or ownership of the Intellectual Property which could reasonably be expected to have a Material Adverse Effect, or of any existing state of facts that would support a claim that use by the Borrower or any of its Subsidiaries of any such Intellectual Property has infringed or otherwise violated any proprietary rights of any other Person which could reasonably be expected to have a Material Adverse Effect.

14.20. Certain Fees

No broker's or finder's fees or commissions or any similar fees or commissions will be payable by the Borrower, the Borrower or any of its Subsidiaries with respect to the incurrence and maintenance of the Obligations, any other transaction contemplated by the Documents or any services rendered in connection with such transactions.

14.21. Security Documents

- Security Agreement Collateral. The provisions of the Security Documents upon execution and delivery thereof are effective to create in favor of the Collateral Agent or, as the case may be, the UK Security Trustee for the benefit of the Secured Parties, a legal, valid and enforceable security interest in all right, title and interest of the applicable Credit Party in the Collateral (other than the Collateral described in the Mortgages) owned by such Credit Party, and the Collateral Security Agreement, the Intercreditor Agreement, the Pledge Agreement and the UK Security Documents, together with the filings of Form UCC-1 (or other similar filing, if any) in all relevant jurisdictions and delivery of all possessory collateral create a first lien on, and security interest in (or similar interest in respect of), all right, title and interest of the Borrower and such Credit Parties in all of the Collateral described therein, subject to no other Liens other than Permitted Liens. Except for titled vehicles, vessels and other collateral which may not be perfected through the filing of financing statements under the Uniform Commercial Code (or similar applicable law) of the appropriate jurisdiction (or similar filings in each relevant jurisdiction) and which have an aggregate fair market value of less than \$5,000,000, and except for patents, trademarks, trade names and copyrights to the extent perfection would require filing in any foreign jurisdiction, all such Liens are perfected Liens (or similar legal status). The recordation in the United States Patent and Trademark Office and in the United States Copyright Office of assignments for security made pursuant to the Collateral Security Agreement will be effective, under Federal law, to perfect the security interest granted to the Collateral Agent for the benefit of the Secured Parties in the trademarks, patents and copyrights covered by such the Collateral Security Agreement. The recordation with the United States Surface Transportation Board of assignments for security made pursuant to the Security Agreement will be effective under Federal law, to create a valid first lien in favor of the Collateral Agent in the railcars covered by the Collateral Security Agreement
- B. <u>Pledged Securities</u>. The security interests created in favor of the Collateral Agent, as pledgee for the benefit of the Secured Parties under the Collateral Security Agreement and the Pledge Agreement, constitute perfected security interests in the Pledged Securities, if any, subject to no security interests of any other Person except for the Liens granted under or pursuant to the Collateral Security Agreement and except for Liens of the types described in clauses (i) and (vi) of the definition of "<u>Customary Permitted Liens</u>". No filings or recordings are required in order to perfect the security interests created in the Pledged Securities under the Collateral Security Agreement other than with respect to filings required by applicable foreign law and UCC financing statements with respect to uncertificated Pledged Securities.
- C. <u>Real Estate Collateral</u>. The Mortgages create, as security for the obligations purported to be secured thereby, a valid and enforceable (and upon the due recording thereof under applicable law) perfected security interest in and Lien on all of the Mortgaged

109

Property (including, without limitation, all fixtures and improvements relating to such Mortgaged Property and affixed or added thereto on or after the Closing Date) in favor of the Collateral Agent (or such other agent or trustee as may be named therein) for the benefit of the Secured Parties, superior to and prior to the rights of all third Persons (except that the security interest created in the Mortgaged Property may be subject to the Permitted Liens related thereto). As of the Effective Date, Schedule 6.21(c) contains a true and complete list of each parcel of real property owned or leased by the Borrower and its Subsidiaries in the United States, the United Kingdom or other jurisdiction in which a material plant is located and the type of interest therein held by the Borrower or such Subsidiary and indicates for each such parcel whether it is a Mortgaged Property. The Borrower or a Subsidiary of the Borrower has good and marketable title to all Mortgaged Property free and clear of all Liens except those described in the first sentence of this Section 6.21(c).

14.22. Anti-Terrorism Law

- A. Neither Borrower nor, to the knowledge of Borrower, any of its Affiliates is in violation of any laws relating to terrorism or money laundering ("Anti-Terrorism Laws"), including Executive Order No. 13224 on Terrorist Financing, effective September 24, 2001 (the "Executive Order"), and the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, Public Law 107-56 (as amended) (the "Patriot Act").
- B. Neither Borrower nor, to the actual knowledge of a Responsible Officer of the Borrower, any Affiliate or broker or other agent of Borrower is, acting or benefiting in any capacity in connection with any Loans hereunder is any of the following:
 - (a) a person that is listed in the annex to, or is otherwise subject to the provisions of, the Executive Order;
 - (b) a person owned or controlled by, or acting for or on behalf of, any person that is listed in the annex to, or is otherwise subject to the provisions of, the Executive Order;
 - (c) a person with which any Lender is prohibited from dealing or otherwise engaging in any transaction by any Anti-Terrorism Law;

- (d) a person that commits, threatens or conspires to commit or supports "terrorism" as defined in the Executive Order; or
- (e) a person that is named as a "specially designated national and blocked person" on the most current list published by the USA Treasury Department Office of Foreign Assets Control ("OFAC") at its official website or any replacement website or other replacement official publication of such list.

110

SECTION 15

AFFIRMATIVE COVENANTS

The Borrower hereby agrees that, so long as any of the Commitments remain in effect, or any Loan or LC Obligation remains outstanding and unpaid or any other amount is owing to any Lender or the Administrative Agent hereunder, the Borrower shall:

15.1. Financial Statements

Furnish, or cause to be furnished, to each Lender:

- A. Quarterly Financial Statements. As soon as available, but in any event not later than 45 days (or in the event that a request for an extension of the required filing date for the Form 10-Q with the SEC of any Person whose consolidated financial statements include the financial results of the Borrower has been timely filed, the last day of such requested extension period, but in no event later than 55 days) after the end of each of the first three quarterly periods of each Fiscal Year of Borrower, (i) the unaudited consolidated balance sheet of Borrower and its consolidated Subsidiaries as at the end of such quarter setting forth in comparative form the audited balance sheet of the Borrower and its consolidated Subsidiaries for the prior Fiscal Year, (ii) the related unaudited consolidated statement of income of the Borrower and its consolidated Subsidiaries as at the end of such quarter and for the portion of the Fiscal Year through the end of such quarter setting forth in comparative form the figures for the related periods in the prior Fiscal Year and (iii) the related unaudited consolidated statements of cash flow of Borrower and its consolidated Subsidiaries for the portion of the Fiscal Year through the end of such quarter, and setting forth in comparative form figures for the related period in the prior Fiscal Year, all of which shall be certified by a Responsible Financial Officer of Borrower, subject to normal year-end audit adjustments; and
- B. Annual Financial Statements. As soon as available, but in any event within 90 days (or in the event that a request for an extension of the required filing date for the Form 10-K with the SEC of any Person whose consolidated financial statements include the financial results of the Borrower has been timely filed, the last day of such requested extension period, but in no event later than 105 days) after the end of each Fiscal Year of Borrower, a copy of the consolidating and consolidated balance sheet of Borrower and its consolidated Subsidiaries as at the end of such year and the related consolidating and consolidated statements of income and of cash flows for such year, and setting forth in each case in comparative form the figures for the previous year and such consolidated statements shall be accompanied by a balance sheet as of such date, and a statement of income and cash flows for such period, reflecting on a combined basis, for Subsidiaries and on a combined basis for Unrestricted Subsidiaries, the consolidating entries for each of such types of Subsidiaries; all such financial statements shall be complete and correct in all material respects and shall be prepared in accordance with GAAP applied consistently throughout the periods reflected therein and with prior periods (except as approved by the accountants preparing such statements or the Responsible Financial Officer, as the case may be, and disclosed therein) and, in the case of the consolidated financial statements referred to in this Section 7.1(b), accompanied by a report thereon of Deloitte & Touche LLP or such other independent certified public accountants of recognized national standing, which report

111

shall contain no "going concern" or like qualification or exception or any qualification and shall state that such financial statements present fairly the financial position of Borrower and its Subsidiaries as at the dates indicated and the results of their operations and cash flow for the periods indicated in conformity with GAAP.

C. <u>Delivery.</u> Information or documents required to be delivered pursuant to <u>Section 7.1(a)</u>, <u>Section 7.1(b)</u> or <u>Section 7.2(f)</u> shall be deemed to have been delivered on the date on which such information or document is actually available for review by the Lenders and either (i) has been posted by the Borrower on the Borrower's website at http://www.huntsman.com or at http://www.sec.gov or (ii) has been posted on the Borrower's behalf on Intralinks/IntraAgency. At the request of the Administrative Agent, the Borrower will provide by electronic mail electronic versions (i.e., soft copies) of all documents containing such information.

15.2. Certificates; Other Information

Furnish to the Administrative Agent for distribution to each Lender (or, if specified below, to the Administrative Agent):

A. <u>Accountant's Certificates</u>. Concurrently with the delivery of the financial statements referred to in <u>Section 7.1(b)</u>, to the extent not contrary to the then current recommendations of the American Institute of Certified Public Accountants, a certificate from Deloitte & Touche or other independent certified public accountants of nationally recognized standing, stating that, in the course of their annual audit of the books and records of the Borrower, no Event of Default or Unmatured Event of Default with respect to <u>Articles VII, VIII</u> and <u>IX</u>, insofar as they relate to accounting and financial matters, has come to their attention which was continuing at the end of such Fiscal Year or on the date of their certificate, or if such an Event of Default or Unmatured Event of Default has come to

their attention, the certificate shall indicate the nature of such Event of Default or Unmatured Event of Default;

- B. Officer's Certificates. Concurrently with the delivery of the financial statements referred to in Sections 7.1(a) and 7.1(b), a certificate of a Responsible Financial Officer substantially in the form of Exhibit 7.2(b) (a "Compliance Certificate") stating that, to the best of such officer's knowledge, (i) such financial statements present fairly, in accordance with GAAP, the financial condition and results of operations of the Borrower and its Subsidiaries for the period referred to therein (subject, in the case of interim statements, to normal recurring adjustments) and (ii) no Event of Default or Unmatured Event of Default has occurred, except as specified in such certificate and, if so specified, the action which the Borrower proposes to take with respect thereto, which certificate shall set forth detailed computations to the extent necessary to establish the Borrower's compliance with the covenants set forth in Article IX of this Agreement;
- C. <u>Audit Reports and Statements.</u> Promptly following the Borrower's receipt thereof, copies of all consolidated financial or other consolidated reports or statements, if any, submitted to the Borrower or any of its Subsidiaries by independent public accountants relating to any annual or interim audit of the books of the Borrower or any of its Subsidiaries;

112

- D. <u>Management Letters</u>. Promptly after receipt thereof, a copy of any "management letter" received by Huntsman Corporation (to the extent such "management letter" pertains to the Borrower), the Borrower or any of its Subsidiaries from its certified public accountants;
- E. <u>Budgets</u>. As soon as available and in any event within sixty (60) days following the first day of each Fiscal Year of the Borrower (i) an annual budget in form satisfactory to the Administrative Agent (including budgeted statements of earnings and cash flows but not including segment data) prepared by the Borrower for such Fiscal Year, which shall be accompanied by the statement of a Responsible Financial Officer of the Borrower to the effect that, to the best of his knowledge at the time made, such budget is a reasonable estimate for the periods covered thereby;
- F. <u>Public Filings</u>. Within 10 days after the same become public, copies of all financial statements and reports which Huntsman Corporation or the Borrower may make to, or file with the SEC or any successor or analogous Governmental Authority;
- G. <u>Insurance Information</u>. The Borrower shall deliver to the Administrative Agent information concerning insurance at the times and in the manner specified in <u>Section 7.8</u>;
- H. <u>USA Patriot Act</u>. Each Lender subject to the Patriot Act hereby notifies the Borrower that pursuant to the requirements of the Patriot Act, it is required to obtain, verify and record information that identifies the Borrower, which information includes the name and address of the Borrower and other information that will allow such Lender to identify the Borrower in accordance with the Patriot Act; and
- I. <u>Other Requested Information</u>. Such other information respecting the respective properties, business affairs, financial condition and/or operations of the Borrower or any of its Subsidiaries as the Administrative Agent or any Lender (through the Administrative Agent) may from time to time reasonably request.

15.3. Notices

Promptly and in any event within five Business Days in the case of <u>clauses (a)</u>, (<u>d</u>) and (<u>e</u>) below, 30 days in the case of <u>clauses (b)</u> and (<u>c</u>) below, or one Business Day in the case of <u>clause (f)</u> below after a Responsible Officer of the Borrower or of any of its Subsidiaries obtains knowledge thereof, give written notice to the Administrative Agent (which shall promptly provide a copy of such notice to each Lender) of:

- A. <u>Event of Default or Unmatured Event of Default</u>. The occurrence of any Event of Default or Unmatured Event of Default, accompanied by a statement of a Responsible Financial Officer setting forth details of the occurrence referred to therein and stating what action the Borrower proposes to take with respect thereto.
- B. <u>Litigation and Related Matters</u>. The commencement of, or any material development in, any action, suit, proceeding or investigation affecting the Borrower or any of its Subsidiaries or any of their respective properties before any arbitrator or Governmental Authority, (i) in which the amount involved that the Borrower reasonably determines is not

113

covered by insurance or other indemnity arrangement is \$50,000,000 or more, (ii) with respect to any Document or any material Indebtedness or preferred stock of the Borrower or any of its Subsidiaries or (iii) which, if determined adversely to the Borrower or any of its Subsidiaries, could reasonably be expected to have a Material Adverse Effect.

C. <u>Environmental</u>.

(a) The occurrence of one or more of the following, to the extent that any of the following, if adversely determined, would have a Material Adverse Effect or, in any event, could reasonably be expected to result in liability to the Borrower or any of its Subsidiaries in excess of \$50,000,000: (A) written notice, claim or request for information to the effect that the Borrower or

any of its Subsidiaries is or may be liable in any material respect to any Person as a result of the presence of or the Release or substantial threat of a material Release of any Contaminant into the environment; (B) written notice that the Borrower or any of its Subsidiaries is subject to investigation by any Governmental Authority evaluating whether any Remedial Action is needed to respond to the presence or to the Release or substantial threat of a material Release of any Contaminant into the environment; (C) written notice that any property, whether owned or leased by, or operated on behalf of, the Borrower or its Subsidiaries is subject to a material Environmental Lien; (D) written notice of violation to the Borrower or any of its Subsidiaries of any Environmental Laws or Environmental Permits, or (E) commencement or written threat of any judicial or administrative proceeding alleging a violation of any Environmental Laws or Environmental Permits; provided, however, that the provisions of this clause (i) shall not require the Borrower to violate or breach any confidentiality covenants to which it is bound.

- (b) Upon written request by the Administrative Agent, the Borrower shall promptly submit to the Administrative Agent and the Lenders a report providing an update of the status of each environmental, health or safety compliance, hazard or liability issue identified in any notice or report required pursuant to <u>clause (i)</u> above and any other environmental, health and safety compliance obligation, remedial obligation or liability that could reasonably be expected to have a Material Adverse Effect. All such notices shall describe in reasonable detail the nature of the claim, investigation, condition, occurrence or Remedial Action and the Borrower's or such Subsidiary's response thereto.
- D. <u>Notice of Change of Control</u>. Each occasion that any Change of Control shall occur and such notice shall set forth in reasonable detail the particulars of each such occasion.
- E. <u>Notices under Transaction Documents</u>. Promptly following the receipt or delivery thereof, copies of any material demands, notices or documents received or delivered by the Borrower or any Subsidiary of the Borrower outside of the ordinary course of business under or pursuant to any Public Note Document, any Organizational Document of the Borrower, any material joint venture agreement and any other material agreement from time to time identified by the Administrative Agent (provided, that the foregoing shall apply to material demands, notices or documents under the Limited Liability Company Agreement of the Borrower, only to the extent required under applicable law to be delivered to the members of the Borrower, as the case may be, in their capacity as members).

114

F. <u>UK Insolvency Proceedings</u>. A meaningful threat of or notice in respect of any insolvency proceeding involving any Foreign Subsidiary incorporated under the laws of England and Wales.

15.4. Conduct of Business and Maintenance of Existence

Continue to engage in business of the same general type as now conducted by it and preserve, renew and keep in full force and effect its and each Subsidiary's organizational existence and take all reasonable action to maintain all rights, privileges and franchises material to its and those of each of its Subsidiaries' businesses except to the extent that failure to take any such action could not in the aggregate reasonably be expected to have a Material Adverse Effect or as otherwise permitted pursuant to Sections 8.3 and comply and cause each of its Subsidiaries to comply with all Requirements of Law except to the extent that failure to comply therewith would not in the aggregate reasonably be expected to have a Material Adverse Effect.

15.5. Payment of Obligations

Pay or discharge or otherwise satisfy at maturity or, to the extent permitted hereby, prior to maturity or before they become delinquent, as the case may be, and cause each of its Subsidiaries to pay or discharge or otherwise satisfy at or before maturity or before they become delinquent, as the case may be:

- (a) all material taxes, assessments and governmental charges or levies imposed upon any of them or upon any of their income or profits or any of their respective properties or assets prior to the date on which penalties attach thereto; and
- (b) all lawful claims prior to the time they become a Lien (other than Permitted Liens) upon any of their respective properties or assets; provided, however, that, in the case of both clauses (i) and (ii), neither the Borrower nor any of its Subsidiaries shall be required to pay or discharge any such material tax, assessment, charge, levy or claim (A) while the same is being contested by it in good faith and by appropriate proceedings diligently pursued so long as the Borrower or such Subsidiary, as the case may be, shall have set aside on its books adequate reserves in accordance with GAAP (segregated to the extent required by GAAP) with respect thereto and title to any material properties or assets is not jeopardized in any material respect or (B) which could not reasonably be expected to have Material Adverse Effect.

15.6. Inspection of Property, Books and Records

Keep, or cause to be kept, and cause each of its Subsidiaries to keep or cause to be kept, adequate records and books of account, in which complete entries are to be made reflecting its and their business and financial transactions, such entries to be made in accordance with sound accounting principles consistently applied and permit, and cause each of its Subsidiaries to permit, any Lender or its respective representatives, at any reasonable time, and from time to time, upon reasonable request made to the Borrower by such Lender and upon reasonable notice during normal business hours, to visit and inspect its and their respective properties, to examine and make copies of and take abstracts from its and their respective records and books of account, and to discuss its and their respective affairs, finances and accounts with its and their respective principal officers, directors and with the written consent of the Borrower (which consent shall

not be required if any Event of Default has occurred and is continuing), independent public accountants, provided that the Borrower may attend any such meetings (and by this provision the Borrower authorizes such accountants to discuss with the Lenders and such representatives the affairs, finances and accounts of the Borrower and its Subsidiaries).

15.7. ERISA

(a) (i) As soon as practicable and in any event within ten (10) days after the Borrower or any of its Subsidiaries or ERISA Affiliates knows or has reason to know that a Reportable Event has occurred with respect to any Plan, deliver, or cause such Subsidiary or ERISA Affiliate to deliver, to the Administrative Agent a certificate of a Responsible Officer of the Borrower or such Subsidiary or ERISA Affiliate, as the case may be, setting forth the details of such Reportable Event and the action, if any, which the Borrower or such Subsidiary or ERISA Affiliate is required or proposes to take, together with any notices required or proposed to be given; (ii) upon the request of any Lender made from time to time, deliver, or cause each Subsidiary or ERISA Affiliate to deliver, to each Lender a copy of the most recent actuarial report and annual report completed with respect to any Plan; (iii) as soon as possible and in any event within ten (10) days after the Borrower or any of its Subsidiaries or ERISA Affiliates knows or has reason to know that any of the following have occurred or is reasonably likely to occur with respect to any Plan: (A) such Plan has been terminated, reorganized, petitioned or declared insolvent under Title IV of ERISA, (B) the Plan Sponsor intends to terminate such Plan under Section 4041(b) or (c), (C) the PBGC has instituted or will institute proceedings under Section 515 of ERISA to collect a delinquent contribution to such Plan or under Section 4042 of ERISA to terminate such Plan, (D) that an accumulated funding deficiency has been incurred or that an application has been made to the Secretary of the Treasury for a waiver or modification of the minimum funding standard (including any required installment payments) or an extension of any amortization period under Section 412 of the Code, or (E) that the Borrower, or any Subsidiary of the Borrower or any ERISA Affiliate will incur any material liability (including, but not limited to, contingent or secondary liability) to or on account of the termination of or withdrawal from a Plan under Section 401(a)(29), 4971 or 4975 of the Code or Section 409 or 502(1) of ERISA, deliver, or cause such Subsidiary or ERISA Affiliate to deliver, to the Administrative Agent a written notice thereof; and (iv) as soon as possible and in any event within thirty (30) days after the Borrower or any of its Subsidiaries or ERISA Affiliates knows or has reason to know that any of them has caused a complete withdrawal or partial withdrawal (within the meaning of Sections 4203 and 4205, respectively, of ERISA) from any Multiemployer Plan, deliver, or cause such Subsidiary or ERISA Affiliate to deliver, to the Administrative Agent a written notice thereof. For purposes of this Section 7.7, the Borrower shall be deemed to have knowledge of all facts known by the Plan Administrator of any Plan of which the Borrower is the Plan Sponsor, and each Subsidiary and ERISA Affiliate of the Borrower shall be deemed to have knowledge of all facts known by the Plan Administrator of any Plan of which such Subsidiary or ERISA Affiliate, respectively, is a Plan Sponsor. In addition to its other obligations set forth in this Article VII, the Borrower shall, and shall cause each of its Subsidiaries and ERISA Affiliates to:

(i) provide the Administrative Agent with prompt written notice, with respect to any Plan, of any failure to satisfy the minimum funding standard requirements of Section 412 of the Code;

116

- (ii) furnish to the Administrative Agent, promptly after delivery of the same to the PBGC, a copy of any delinquency notice pursuant to Section 412(n)(4) of the Code;
- (iii) correct any such failure to satisfy funding requirements or delinquency referred to in the foregoing clauses (A) and (B) within ninety (90) days after the occurrence thereof, except where the failure to so satisfy would not reasonably be expected to have a Material Adverse Effect;
- (iv) comply in good faith in all material respects with the requirements set forth in Section 4980B of the Code and with Sections 601(a) and 606 of ERISA;
- (v) at the request of any Lender, deliver to such Lender (and a copy to the Administrative Agent) a complete copy of the most recent annual report (Form 5500) of each Plan required to be filed with the Internal Revenue Service; and
- (vi) at the request of any Lender, deliver to such Lender (and a copy to the Administrative Agent) copies of the most recent annual reports received by the Borrower or any Subsidiary of the Borrower or any ERISA Affiliate with respect to any Plan or Foreign Pension Plan no later than ten (10) days after the date of such request.
- B. The Borrower shall, and shall cause each of its Subsidiaries to, establish, maintain and operate all Foreign Pension Plans in compliance in all respects with all laws, regulations and rules applicable thereto and the respective requirements of the governing documents for such Plans, except for such failures that individually or in the aggregate could neither (i) result in liabilities in excess of \$50,000,000 nor (ii) reasonably be expected to result in a Material Adverse Effect.

15.8. Maintenance of Property, Insurance

(i) Except to the extent that the failure to do so could not, in any case, reasonably be expected to result in a Material Adverse Effect, keep, and cause each of its Subsidiaries to keep, all property (including, but not limited to, equipment) useful and necessary for its business in good working order and condition, normal wear and tear and damage by casualty excepted, subject to Section 8.3, (ii) maintain, and shall cause each of its Subsidiaries to maintain, with financially sound and reputable insurers, insurance with respect to its material properties and business against loss or damage of the kinds customarily insured against by Persons engaged in

insurance policies to the extent requested by the Administrative Agent (A) shall be endorsed to the Administrative Agent's reasonable satisfaction for the benefit of the Collateral Agent for the benefit of the Secured Parties (including, without limitation, by naming the Collateral Agent as loss payee or additional insured, as appropriate); and (B) shall state that such insurance policy shall not be canceled or revised without thirty days' prior written notice thereof by the insurer to the Administrative Agent and (iii) furnish to the Administrative Agent, on the Closing Date and on the date of delivery of each annual financial statement, full information as to the insurance carried. At any time that insurance at levels described in Schedule 7.8 is not being maintained by or on behalf of the Borrower or any of its Subsidiaries, the Borrower will notify the Lenders in writing within two Business Days thereof and, if thereafter notified by the Administrative Agent or the Required Lenders to do so, the Borrower or any such Subsidiary, as the case may be, shall use commercially reasonable efforts to obtain insurance at such levels at least equal to those set forth on Schedule 7.8. If any portion of any Mortgaged Property is at any time located in an area identified by the Federal Emergency Management Agency (or any successor agency) as a Special Flood Hazard Area with respect to which flood insurance has been made available under the National Flood Insurance Act of 1968 (as now or hereafter in effect or successor act thereto), then the Borrower or the applicable Credit Party shall maintain, or cause to be maintained, with IRIC or another financially sound and reputable insurer, flood insurance in amounts and otherwise sufficient to comply with all applicable Flood Insurance Laws and shall, upon the reasonable request of the Administrative Agent, deliver to the Administrative Agent evidence of such compliance in form and substance reasonably acceptable to the Administrative Agent.

15.9. Environmental Laws

The Borrower shall, and shall cause each of its Subsidiaries, in the exercise of its reasonable business judgment, to take prompt and appropriate action to respond to any material non-compliance with Environmental Laws or Environmental Permits or to any material Release or a substantial threat of a material Release of a Contaminant, and upon request from the Administrative Agent, shall report to the Administrative Agent on such response. Without limiting the generality of the foregoing, whenever the Administrative Agent or any Lender has a reasonable basis to believe that the Borrower is not in material compliance with Environmental Laws or Environmental Permits or that any property of the Borrower or its Subsidiaries, or any property to which Contaminants generated by the Borrower or its Subsidiaries have come to be located ("Offsite Property") has or may become contaminated or subject to an order or decree such that any non-compliance, contamination or order or decree could reasonably be expected to have a Material Adverse Effect, then, to the extent the Borrower has the legal right to do so, the Borrower agrees to, at the Administrative Agent's request and the Borrower's expense: (i) cause an independent environmental engineer reasonably acceptable to the Administrative Agent to conduct such tests of the site where the alleged or actual non-compliance or contamination has occurred and prepare and deliver to the Administrative Agent, the Lenders and the Borrower a report(s) reasonably acceptable to the Administrative Agent setting forth the results of such tests, the Borrower's proposed plan and schedule for responding to any environmental problems described therein, and the Borrower's estimate of the costs thereof, and (ii) provide the Administrative Agent, the Lenders and the Borrower a supplemental report(s) of such engineer whenever the scope of the environmental problems or the Borrower's response thereto or the estimated costs thereof, shall materially change. Notwithstanding the

118

above, the Borrower shall not be obligated (other than as required by applicable law) to undertake any tests or remediation at any Offsite Property (a) that is not owned or operated by the Borrower or any of its Subsidiaries and (b) where Contaminants generated by persons other than the Borrower or any of its Subsidiaries have also come to be located.

B. Defend, indemnify and hold harmless the Administrative Agent and the Lenders, and their respective employees, the Administrative Agents, officers and directors, from and against any and all claims, demands, penalties, fines, liabilities, settlements, damages, costs and expenses of whatever kind or nature known or unknown, contingent or otherwise, arising out of, or in any way relating to the violation of, noncompliance with or liability under, any Environmental Law applicable to the operations of the Borrower, any of its Subsidiaries or their respective properties, or any orders, requirements or demands of Governmental Authorities related thereto, including, without limitation, reasonable attorneys' and consultants' fees, investigation and laboratory fees, costs arising from any Remedial Actions, court costs and litigation expenses, except to the extent that any of the foregoing arise out of the gross negligence or willful misconduct of the party seeking indemnification therefor. The agreements in this Section 7.9(b) shall survive repayment of the Notes and all other Obligations.

15.10. Use of Proceeds

Use all proceeds of the Loans as provided in Section 6.8.

15.11. Additional Security; Further Assurances

A. <u>Agreement to Grant Additional Security.</u> Promptly, and in any event within 30 days (unless otherwise extended at the discretion of the Administration Agent) after the acquisition by the Borrower or any Domestic Subsidiary of assets or real or personal property or leasehold interests of the type that would have constituted Collateral on the date hereof, in each case in which the

Collateral Agent or the Administrative Agent does not have a perfected security interest under the Security Documents (other than (u) Capital Stock subject to Section 7.11(c), (v) all assets owned by any Receivables Subsidiary, any of the Thai Holding Companies, or IRIC, (w) copyrights, patents and trademarks to the extent perfection would require filing in any foreign jurisdiction, (x) assets or real or personal property subject to Liens permitted under Section 8.1(c) under agreements which prohibit the creation of additional Liens on such assets, (y) any parcel of real estate or leasehold interest acquired after the Closing Date with a fair market value of less than \$10,000,000 or (z) any other asset with a fair market value of less than \$100,000 individually (provided that all such other assets collectively have a fair market value of less than \$10,000,000)) and within 30 days (unless otherwise extended at the discretion of the Administration Agent) after request by the Administrative Agent or Collateral Agent with respect to any other after acquired collateral deemed material by the Administrative Agent or Required Lenders, the Borrower will, and will cause each of their respective Domestic Subsidiaries to, take all necessary action, including (i) the filing of appropriate financing statements under the provisions of the UCC, applicable foreign, domestic or local laws, rules or regulations in each of the offices where such filing is necessary or appropriate to grant the Collateral Agent or the Administrative Agent for the benefit of the Secured Parties pursuant to the Collateral Security Agreement a perfected Lien (subject only to Permitted Liens) in such Collateral pursuant to and to the full extent required by the Security Documents and this

119

Agreement and (ii) with respect to real estate, the execution of a Mortgage, the obtaining of title insurance policies or indemnification agreements satisfactory to the Administrative Agent, title surveys and real estate appraisals satisfying the Requirements of Law.

- B. <u>Subsidiary Guaranties</u>. The Borrower agrees to cause (i) each Subsidiary (other than an Immaterial Subsidiary, a Receivables Subsidiary, any of the Thai Holding Companies, or IRIC) that is organized under the laws of a state of the United States of America or the District of Columbia and (ii) each other Subsidiary (other than an Immaterial Subsidiary) that is wholly owned by a corporation organized under the laws of a state of the United States or the District of Columbia and is disregarded as an entity separate from that owner under Treasury Regulation section 301.7701-3, to execute and deliver the Subsidiary Guaranty (or a supplement thereto) promptly, and in any event, within 30 days of such Person's having become a Subsidiary.
- C. <u>Pledge of New Subsidiary Stock</u>. The Borrower agrees to pledge (or cause its Subsidiaries to pledge) all of the Capital Stock of each new Domestic Subsidiary and, to the extent such pledge would not result in adverse tax consequences to the Borrower, 65% of the Capital Stock of each new first-tier Foreign Subsidiary established, acquired or created after the Closing Date to the Collateral Agent for the benefit of the Secured Parties pursuant to the Collateral Security Agreement and the other Security Documents promptly, and in any event, within 30 days of the establishment, acquisition or creation of such new Subsidiary; provided, that any filings or recordations with respect to Foreign Subsidiaries may be made after such 30-day period to the extent approved by the Administrative Agent.
- D. Grant of Security by New Subsidiaries. Subject to the provisions of Sections 7.11(a) and 7.11(c), the Borrower will promptly and, in any event, within 30 days of the establishment, acquisition or creation of a Domestic Subsidiary, cause each Domestic Subsidiary established or created in accordance with Section 8.7 to grant to the Collateral Agent for the benefit of the Secured Parties pursuant to the Collateral Security Agreement and the Pledge Agreement a first priority Lien (subject to Permitted Liens) on all property (tangible and intangible of the type that constitutes Collateral under the Security Documents) of such Domestic Subsidiary by executing and delivering an agreement substantially in the form of Exhibit A to the Collateral Security Agreement and an agreement substantially in the form of Exhibit A to the Pledge Agreement, or such other security agreements on other terms satisfactory in form and substance to the Administrative Agent. The Borrower shall cause each Domestic Subsidiary, at its own expense, to execute, acknowledge and deliver, or cause the execution, acknowledgment and delivery of, and thereafter register, file or record in any appropriate governmental office, any document or instrument reasonably deemed by the Administrative Agent to be necessary or desirable for the creation and perfection of the foregoing Liens. The Borrower will cause each of its Domestic Subsidiaries to take all actions reasonably requested by the Administrative Agent or the Required Lenders (including, without limitation, the filing of UCC-1's) in connection with the granting of such security interests.
- E. <u>Pledge of Equity in Unrestricted Subsidiaries</u>. The Borrower agrees to pledge (or cause its Domestic Subsidiaries to pledge) all of the Capital Stock owned directly by the Borrower or a Domestic Subsidiary of each domestic Unrestricted Subsidiary (and to the extent such pledge would not result in adverse tax consequences to the Borrower, 65% of the

120

Capital Stock of each first-tier Foreign Subsidiary) to the Collateral Agent for the benefit of the Secured Parties pursuant to the Pledge Agreement. The Borrower agrees to pledge or cause its Subsidiaries to pledge, to the Collateral Agent for the benefit of the Secured Parties pursuant to the Collateral Security Agreement all instruments evidencing indebtedness owed by any Unrestricted Subsidiary to the Borrower or any Domestic Subsidiary.

- F. <u>Receivables Financing Security.</u> No later than the time that any Receivables Documents are entered into, and no later than the time any capital is contributed or funds are advanced by the Borrower to the Receivables Subsidiary, the Borrower and each Participating Subsidiary shall execute and deliver to Collateral Agent for the benefit of the Secured Parties, the Receivables Subsidiary Pledge Agreement, accompanied, to the extent such Pledged Securities are certificated, by certificates representing the Pledged Securities.
- G. <u>Documentation for Additional Security</u>. The security interests required to be granted pursuant to this <u>Section 7.11</u> shall be granted pursuant to the Annexes to the Security Documents or such other security documentation satisfactory in

form and substance to the Administrative Agent and the Required Lenders and shall constitute valid and enforceable perfected security interests prior to the rights of all third Persons (other than any such rights arising in connection with Permitted Liens) and subject to no other Liens except Permitted Liens. The Additional Security Documents and other instruments related thereto shall be duly recorded or filed in such manner and in such places and at such times as are required by law to establish, perfect, preserve and protect the Liens, in favor of the Administrative Agent for the benefit of the Lenders, required to be granted pursuant to the Additional Security Document and, all taxes, fees and other charges payable in connection therewith shall be paid in full by the Borrower or its Subsidiaries. At the time of the execution and delivery of the Additional Security Documents, the Borrower shall cause to be delivered to the Administrative Agent such agreements, opinions of counsel and other related documents as may be reasonably requested by the Administrative Agent or the Required Lenders to assure themselves that this Section 7.11 has been complied with.

H. If, following a change in the relevant sections of the Code, the regulations and rules promulgated thereunder and any rulings issued thereunder and at the reasonable request of the Administrative Agent or the Required Lenders, counsel for the Borrower acceptable to the Administrative Agent and the Required Lenders does not within 60 days after such request deliver evidence reasonably satisfactory to the Administrative Agent with respect to any Foreign Subsidiary that is a Wholly-Owned Subsidiary of the Borrower that any of (i) a pledge of 66-2/3% or more of the total combined voting power of all classes of Capital Stock of such Foreign Subsidiary entitled to vote, (ii) the entering into by such Foreign Subsidiary of a guaranty in substantially the form of the Subsidiary Guaranty or (iii) the entering into by such Foreign Subsidiary of a security agreement in substantially the form of the Security Agreement, in any case could cause all or a portion of the earnings of such Foreign Subsidiary to be treated as a deemed dividend to such Foreign Subsidiary's United States parent or would otherwise violate applicable law or result in adverse tax consequences to the Borrower or its Subsidiaries (including, without limitation, in the form of distributions payable to any Parent Company pursuant to the Limited Liability Company Agreement of the Borrower), then in the case of a failure to deliver the evidence described in clause (i) above, that portion of such Foreign Subsidiary's outstanding Capital Stock not theretofore pledged pursuant to the Security

121

Documents shall be pledged to the Administrative Agent for the benefit of the Lenders pursuant to the Security Documents (or another pledge agreement in substantially similar form, if needed), (ii) in the case of a failure to deliver the evidence described in clause (ii) above, such Foreign Subsidiary shall execute and deliver a guaranty of the Obligations of the Borrower under the Loan Documents (subject to compliance with financial assistance laws or similar laws applicable to such Foreign Subsidiary), and (iii) in the case of a failure to deliver the evidence described in clause (iii) above, such Foreign Subsidiary (subject to compliance with financial assistance laws or similar laws applicable to such Foreign Subsidiary) shall execute and deliver a security agreement granting the Administrative Agent for the benefit of the Lenders a security interest in all of such Foreign Subsidiary's assets (to the extent that such Foreign Subsidiary would be required to grant a security interest in such assets under the provisions of Section 7.11(a) hereof if such assets had been acquired by a Domestic Subsidiary), in each case with all documents delivered pursuant to this Section 7.11 to be in form and substance reasonably satisfactory to the Administrative Agent and the Required Lenders, but in each case, only to the extent permitted without violating applicable law or resulting in adverse tax consequences. The deliveries set forth in this clause (h) shall also not be required to the extent that the Administrative Agent determines in its discretion that the cost of obtaining the security interests set forth above outweigh the benefits that such security interests provide to the Lenders.

15.12. End of Fiscal Years; Fiscal Quarters

The Borrower will, and will cause each of its Subsidiaries' annual accounting periods to end on December 31 of each year (each a "<u>Fiscal Year</u>", with quarterly accounting periods ending on March 31, June 30, September 30, December 31 of each Fiscal Year (each a "<u>Fiscal Quarter</u>"), unless otherwise required by applicable law.

15.13. Maintenance of Ratings

The Borrower will use commercially reasonable efforts to have a corporate family rating and a rating with respect to its senior secured debt issued by Moody's and a corporate rating and a rating with respect to its senior secured debt issued by S&P.

15.14. Certain Fees Indemnity

The Borrower covenants that it will indemnify the Administrative Agent and each Lender against and hold the Administrative Agent and each Lender harmless from any claim, demand or liability for broker's or finder's fees or similar fees or commissions alleged to have been incurred in connection with any of the transactions contemplated hereby.

15.15. Successor Agency Transfer

The Borrower covenants that upon or after the Fifth Amendment Effective Date, it will, and will cause each of its Subsidiaries to, take all necessary action reasonably requested by JPMCB, as successor Administrative Agent, Collateral Agent and UK Security Trustee to DB upon the Fifth Amendment Effective Date, to effect the assignment of and transfer to each of the Liens and security interests currently held by DB, for the benefit of the Secured Parties, to

122

NEGATIVE COVENANTS

The Borrower hereby covenants and agrees that, so long as any of the Commitments remain in effect or any Loan or LC Obligation remains outstanding and unpaid or any other amount is owing to any Lender or the Administrative Agent hereunder:

16.1. Liens

The Borrower will not, and will not permit any of its Subsidiaries to create, incur, assume or suffer to exist or agree to create, incur or assume any Lien in, upon or with respect to any of its properties or assets (including, without limitation, any securities or debt instruments of any of its Subsidiaries), whether now owned or hereafter acquired, or assign or otherwise convey any right to receive income to secure any obligation, except for the following Liens (herein referred to as "Permitted Liens"):

A. Liens created under the Security Documents (including, without limitation, Liens securing the Senior Secured Notes Obligations on a pari passu basis with the Obligations, but only to the extent that such Indebtedness is permitted by <u>Section 8.2(b)</u> (vi) and Liens on Cash, Cash Equivalents and Foreign Cash Equivalents securing LC Obligations;

B. Customary Permitted Liens;

- C. Liens on any property securing Indebtedness incurred or assumed for the purpose of financing all or any part of the acquisition, construction, repair or improvement cost of such property, or securing a Sale and Leaseback Transaction permitted hereunder, and any Lien securing Permitted Refinancing Indebtedness of any Indebtedness secured by any Lien permitted by this clause (c); provided, that (A) any such Lien does not extend to any other property (other than accessions and additions to the property covered thereby), (B) such Lien either exists on the date hereof or is created in connection with the acquisition, construction, repair or improvement of such property as permitted by this Agreement, (C) the indebtedness secured by any such Lien (or the Capitalized Lease Obligation with respect to any Capitalized Lease) when incurred, does not exceed 100% of the fair market value of such assets; and (D) the Indebtedness secured thereby is permitted to be incurred pursuant to Section 8.2(b)(iv), provided that any such Permitted Refinancing Indebtedness is not increased and is not secured by any additional assets;
- D. additional Liens incurred by the Borrower and its Subsidiaries which do not secure Indebtedness for money borrowed so long as the value of the property subject to such Liens, and the obligations secured thereby, do not exceed \$50,000,000 in the aggregate at any one time outstanding;
- E. Liens consisting of an agreement to sell, transfer or dispose of any asset (to the extent such sale, transfer or disposition is permitted hereby);

123

- F. Liens in favor of the Borrower or any of its Subsidiaries securing intercompany Indebtedness among the Borrower and its Subsidiaries permitted to be incurred in accordance with Section 8.2(b)(iii);
- G. Liens securing Indebtedness of Foreign Subsidiaries; <u>provided</u>, that the amount of such Indebtedness on the date that such Person incurs (as defined in <u>Section 8.2(a)</u>) such Indebtedness, after giving Pro Forma Effect to such incurrence, does not exceed 5% of the Borrower's Consolidated Net Tangible Assets as of the end of the most recent Fiscal Quarter for which the Borrower has delivered financial statements as required by Section 7.1 in the aggregate at any one time outstanding;
- H. Liens (1) existing on the Effective Date listed on Schedule 8.1(h) hereof and any extension, renewal or replacement thereof but only if the principal amount of the Indebtedness (including, for purposes of this Section 8.1(h), any additional Indebtedness incurred pursuant to revolving commitments in an amount not in excess of the available commitment as set forth on Schedule 8.2(b)(ii) secured thereby) is not increased and such Liens do not extend to or cover any other property or assets, (2) on property of Airstar Corporation incurred pursuant to the Airstar Aircraft Financing Documents and (3) on the assets of Nitrail Vegyipari Termeló Fejlesztó Résvénytár-ság (Nitrail Chemical Engineering and Production Co., Plc) which secure not more than \$2,000,000 of Indebtedness;
- I. Liens on Receivables Facility Assets transferred, directly or indirectly, (a) to a Receivables Subsidiary or (b) by a Receivables Subsidiary to the purchasers of such receivables (and the filing of financing statements in connection therewith) created by, and as set forth in, the Receivables Documents pursuant to a Permitted Accounts Receivable Securitization;
- J. Liens securing Acquired Debt permitted pursuant to the second proviso of Section 8.2(a) or pursuant to Section 8.2(b)(xi), provided, that any such Lien does not extend to any property other than the property of the newly acquired Subsidiary (and proceeds and accessions and additions to such property) that is subject to a Lien securing such Indebtedness as of the closing of the Acquisition of such Subsidiary;
- K. Liens on unearned insurance premiums securing Indebtedness incurred by Borrower and/or its Subsidiaries to finance such insurance premiums in a principal amount not to exceed at any time the amount of such insurance premiums to be paid by Borrower and/or its Subsidiaries for a three year period;
- L. Liens securing obligations arising in the ordinary course pursuant to standard documentation evidencing any Foreign Factoring Transaction; and
 - M. Liens on Cash, Cash Equivalents and Foreign Cash Equivalents securing obligations in respect of letters of

In connection with the granting of Liens of the type described in clause (c) of this <u>Section 8.1</u> by the Borrower or any of its Subsidiaries, at the reasonable request of the Borrower, and at the Borrower's expense, the Administrative Agent or the Collateral Agent shall take (and is hereby authorized to take) any actions reasonably requested by the Borrower in connection therewith (including, without limitation, by executing appropriate lien releases in favor of the

124

holder or holders of such Liens, in either case solely with respect to the item or items of equipment or other assets subject to such Liens).

16.2. Indebtedness

A. The Borrower will not, and will not permit any Subsidiary to, directly or indirectly, create, incur, issue, assume, guarantee or otherwise become directly or indirectly liable, contingently or otherwise (collectively, "incur" and collectively, an "incurrence"), with respect to any Indebtedness; provided that so long as no Event of Default has occurred and is continuing the Borrower may incur Indebtedness, and any Subsidiary may incur Indebtedness, if the Borrower's Interest Coverage Ratio for the Borrower's most recently ended four full fiscal quarters for which financial statements have been delivered pursuant to Section 7.1 would have been at least 2.00 to 1.00 determined on a Pro Forma Basis; provided, further, that in the case of an incurrence by a Subsidiary other than a Subsidiary Guarantor, the proceeds of such Indebtedness are used for a Permitted Acquisition, capital expenditures or such incurrence represents Acquired Debt.

- B. The limitations set forth in clause (a) of this <u>Section 8.2</u> shall not apply to any of the following items:
 - (a) Indebtedness incurred pursuant to this Agreement and the other Loan Documents;
- (b) Indebtedness described on <u>Schedule 8.2(b)(ii)</u> which Indebtedness was outstanding on the Effective Date and Permitted Refinancing Indebtedness in respect thereof; <u>provided</u>, <u>however</u>, that notwithstanding anything else in this <u>Section 8.2(b)(ii)</u> to the contrary, Indebtedness of Tioxide Southern Africa (Proprietary) Ltd. may be refinanced in an amount not in excess of the Dollar Equivalent of \$15,000,000 provided that such Indebtedness is in no way guaranteed by the Borrower or any Subsidiary of the Borrower;
- (c) Indebtedness of the Borrower and its Subsidiaries consisting of Intercompany Loans and guarantees thereof owing to the Borrower or any of its Subsidiaries; provided, that any such Indebtedness of the Borrower or a Subsidiary Guarantor owing to a Subsidiary that is not a Subsidiary Guarantor is subordinated in right of payment to the Obligations as set forth in Section 8.7(g); provided further that if any Subsidiary that incurred Indebtedness under this clause ceases to be a Subsidiary, such entity shall be deemed to have incurred on such date such Indebtedness as is outstanding on such date and such incurrence shall not be permitted under this clause (iii);
- (d) Indebtedness of the Borrower and its Subsidiaries secured by purchase money Liens or constituting Capitalized Lease Obligations or an Operating Financing Lease and Permitted Refinancing Indebtedness in respect thereof; <u>provided</u>, that, at the time such Indebtedness is incurred, after giving Pro Forma Effect to such incurrence, the sum of (1) the aggregate outstanding Capitalized Lease Obligations <u>plus</u> (2) the aggregate outstanding Attributable

125

Debt with respect to Operating Financing Leases <u>plus</u> (3) the aggregate outstanding principal amount of such purchase money Indebtedness <u>plus</u> (4) the aggregate outstanding amount of Indebtedness permitted by <u>Section 8.2(b)(xi)</u> does not exceed an amount equal to 5% of the Borrower's Consolidated Net Tangible Assets as of the end of the most recent Fiscal Quarter for which the Borrower has delivered financial statements as required by <u>Section 7.1</u>;

- (e) Indebtedness of the Borrower or its Subsidiaries under Interest Rate Agreements providing protection against fluctuations in interest rates so long as management of the Borrower or such Subsidiary, as the case may be, has determined that entering into such Interest Rate Agreements are bona fide hedging activities and under Other Hedging Agreements providing protection against fluctuations in currency or commodity values (in the case of commodity values, for a period not to exceed 36 months) in connection with the Borrower's or any of its Subsidiaries' operations so long as management of the Borrower or such Subsidiary, as the case may be, has determined that the entering into of such Other Hedging Agreements are bona fide hedging activities;
- (f) Indebtedness of the Borrower in respect of the Public Notes and guarantees thereof by the Borrower's Subsidiaries; provided, that any guarantees of Public Notes that are subordinated to the Obligations shall be subordinated to the Obligations in the same fashion as such Public Notes are subordinated to the Obligations;
- (g) Indebtedness of the Borrower and its Subsidiaries consisting of take-or-pay obligations contained in supply agreements entered into in the ordinary course of business;
- (h) Indebtedness of the Borrower to a Huntsman Affiliate that is subordinated to the Obligations in a manner reasonably satisfactory to the Administrative Agent;

- (i) Indebtedness of the Borrower and its Subsidiaries constituting Permitted Refinancing Indebtedness of (i) Term Loans hereunder and (ii) Indebtedness incurred under <u>Section 8.2(a)</u> above;
- (j) Indebtedness consisting of (i) Guarantee Obligations of any Subsidiary of the Borrower of the Obligations under any Loan Document and (ii) a guarantee by the Borrower or a Domestic Subsidiary of obligations of a Subsidiary or a guarantee by any Foreign Subsidiary of obligations of a Foreign Subsidiary, in each case, of Indebtedness permitted to be incurred under clause (a) or any other clause of this clause (b); provided, that any Guarantee Obligations of the Borrower or any Subsidiary with respect to intercompany Indebtedness of any Subsidiary is also permitted pursuant to clause (b)(iii);
- (k) Indebtedness of a Subsidiary of the Borrower issued and outstanding on or prior to the date on which such Subsidiary was acquired by the

126

Borrower or a Subsidiary of the Borrower in a transaction constituting an Acquisition (other than Indebtedness issued as consideration in, or to provide all or any portion of the funds utilized to consummate such Acquisition) ("<u>Acquired Debt</u>") and any Permitted Refinancing Indebtedness in respect thereof; <u>provided</u>, that the aggregate amount of such Indebtedness at the time such Indebtedness is incurred, after giving Pro Forma Effect to such incurrence, together with Indebtedness outstanding and permitted by <u>Section 8.2(b)(iv)</u> (without double counting and without giving effect to <u>Section 8.1(c)(C)</u>) does not exceed an amount equal to 5% of the Borrower's Consolidated Net Tangible Assets as of the end of the most recent Fiscal Quarter for which the Borrower has delivered financial statements as required by <u>Section 7.1</u>;

- (l) Indebtedness (including intraday cash management lines relating thereto) of the Borrower and of its Subsidiaries pursuant to over-draft or similar lines of credit (including, without limitation, treasury management arrangements, depository or other cash management services and commercial credit card and merchant card services) in existence on the Fifth Amendment Effective Date or designated by the Borrower to the Administrative Agent as "Overdraft Facilities" thereafter (including unsecured back-to-back lines of credit relating thereto among Foreign Subsidiaries, an "Overdraft Facility") such that the aggregate amount of such Indebtedness (other than intraday cash management lines relating thereto) permitted thereunder or outstanding under this clause (xii) at any one time does not exceed (without duplication) \$100,000,000 (or the Dollar Equivalent thereof) for more than one (1) consecutive Business Day, with respect to such Indebtedness (other than intraday cash management lines relating thereto), provided, that the aggregate principal amount of Indebtedness (other than intraday cash management lines relating thereto) outstanding under each such line shall be reduced to the Dollar Equivalent of \$25,000,000 during at least one day during each calendar month;
- (m) other unsecured Indebtedness of the Borrower and of its Subsidiaries not to exceed \$100,000,000 at any time;
- (n) (i) Receivables Facility Attributed Indebtedness and (ii) intercompany indebtedness of a Receivables Subsidiary owed to the Borrower or its Participating Subsidiaries to the extent such Indebtedness constitutes a permitted Investment pursuant to Section 8.7(n);
- (o) Indebtedness of Foreign Subsidiaries consisting of limited recourse obligations incurred in the ordinary course pursuant to standard documentation evidencing Foreign Factoring Transactions;
- (p) Indebtedness of Foreign Subsidiaries; <u>provided</u>, that the aggregate amount of such Indebtedness at the time such Indebtedness is incurred, together with the aggregate amount of all other Indebtedness outstanding under this clause (xvi) at such time, does not exceed 5% of the Borrower's Consolidated Net

127

Tangible Assets as of the end of the most recent Fiscal Quarter for which the Borrower has delivered financial statements as required by Section 7.1; and

- (q) Indebtedness consisting of obligations in respect of any Fifth Amendment Existing Letter of Credit.
- **16.3.** Consolidation, Merger, Purchase or Sale of Assets, etc.

The Borrower will not, and will not permit any of its Subsidiaries to, wind up, liquidate or dissolve any of their affairs or enter into any transaction of merger or consolidation, or convey, sell, lease or otherwise dispose of any of its properties or assets (or, with respect to any such transaction involving all or substantially all of the assets of the Borrower, enter into an agreement to do any of the foregoing at any future time without the Administrative Agent's prior written consent unless the effectiveness of such agreement is conditional upon the consent of the Administrative Agent), or enter into any Sale and Leaseback Transaction, except that:

A. [Reserved];

- B. Investments may be made to the extent permitted by <u>Section 8.7</u>;
- C. each of the Borrower and its Subsidiaries may lease (as lessor) real or personal property in the ordinary course of business other than to a Receivables Subsidiary;
- D. each of the Borrower and its Subsidiaries may make sales or transfers of inventory, Cash, Cash Equivalents and Foreign Cash Equivalents in the ordinary course of business other than to a Receivables Subsidiary;
- E. the Borrower and its Subsidiaries may sell or discount, in each case without recourse and in the ordinary course of business, Accounts Receivable arising in the ordinary course of business (x) which are overdue, or (y) which the Borrower or such Subsidiary may reasonably determine are difficult to collect but only in connection with the compromise or collection thereof consistent with customary industry practice (and not as part of any bulk sale or financing of receivables);
- F. the Borrower and its Subsidiaries may license its patents, trade secrets, know-how and other intellectual property relating to the manufacture of chemical products and by-products (the "<u>Technology</u>") provided that such license shall be assignable to the Administrative Agent or any assignee of the Administrative Agent without the consent of the licensee and no such license shall (i) transfer ownership of such Technology to any other Person or (ii) require the Borrower to pay any fees for any such use (such licenses permitted by this Section 8.3(f), hereafter "Permitted Technology Licenses");
- G. any Subsidiary of the Borrower (other than a Receivables Subsidiary) may be merged or consolidated (x) with or into the Borrower so long as the Borrower is the surviving entity, (y) with or into any one or more Wholly-Owned Subsidiaries of the Borrower (other than an Unrestricted Subsidiary, Airstar Corporation, Huntsman Headquarters Corporation or IRIC); provided, however, that a Wholly-Owned Subsidiary or Subsidiaries shall be the surviving entity or (z) with or into any Person in connection with the consummation of an Acquisition; provided,

128

however, that after giving effect to such merger or consolidation the surviving Subsidiary shall be a Wholly-Owned Subsidiary;

- H. the Borrower and its Subsidiaries may sell, transfer or otherwise dispose of any asset in connection with any Sale and Leaseback Transaction involving Indebtedness, Capitalized Lease Obligations or an Operating Financing Lease otherwise permitted hereunder;
- I. in any Fiscal Year, the Borrower or any Subsidiary may dispose of any of its assets (including in connection with Sale and Leaseback Transactions not involving Indebtedness, Capitalized Lease Obligations or an Operating Financing Lease) if the aggregate net book value (at the time of disposition thereof) of all assets disposed of by the Borrower and its Subsidiaries in such Fiscal Year pursuant to this clause (i) <u>plus</u> the aggregate book value of all the assets then proposed to be disposed of does not exceed 12.5% of the Consolidated Net Tangible Assets the Borrower and its Subsidiaries as of the end of the immediately preceding Fiscal Quarter for which the Borrower has delivered financial statements as required by <u>Section 7.1</u>; provided, however, that if (A) concurrently with any disposition of assets or within 360 days of receipt of proceeds in connection with such disposition, all or a portion of an amount equal to the net proceeds of such disposition are used by the Borrower or a Subsidiary to acquire other property used or to be used in the business referred to in <u>Section 8.9</u> and (B) the Borrower or such Subsidiary has complied with the provisions of <u>Section 7.11</u> with respect to such property, then such dispositions (or, to the extent that less than all of the net proceeds of any such disposition are used to acquire such other property, then dispositions in an amount equal to the net proceeds used to acquire such other property) shall be disregarded for purposes of calculations pursuant to this <u>Section 8.3(i)</u> (and shall otherwise be deemed to be permitted under this <u>Section 8.3)</u> from and after the date such proceeds are so used to acquire such property with respect to the acquisition of such other property;
- J. the Borrower or any Subsidiary of the Borrower may sell, lease, transfer or otherwise dispose of any or all of its assets to the Borrower or any other Wholly-Owned Subsidiary of the Borrower (other than (I) from the Borrower or a Domestic Subsidiary to a Foreign Subsidiary or (II) to a Receivables Subsidiary);
- K. any Subsidiary of the Borrower (other than a Receivables Subsidiary) may voluntarily liquidate, wind-up or dissolve;
- L. the Borrower and its Subsidiaries may, directly or indirectly, sell, contribute and make other transfers of Receivables Facility Assets to a Receivables Subsidiary and such Receivables Subsidiary may sell and make other transfers of Receivables Facility Assets to the Issuer, in each case pursuant to the Receivables Documents under a Permitted Accounts Receivable Securitization;
 - M. Foreign Subsidiaries may enter into Foreign Factoring Transactions; and
- N. the Borrower and its Subsidiaries may consummate the US Commodity Business Sale provided that not less than 75% of the Net Sale Proceeds therefrom are used within 90 days to (i) repay Senior Secured Notes (2010); (ii) repay Senior Notes (2012); (iii)

16.4. Dividends or Other Distributions

- A. Neither the Borrower nor any of its Subsidiaries will: (i) declare or pay any dividend or make any distribution on or in respect of its Capital Stock or to the direct or indirect holders of its Capital Stock, other than (x) dividends or distributions (A) payable solely in such Capital Stock or in options, warrants or other rights to purchase such Capital Stock, (B) dividends and distributions payable to the Borrower or a Wholly-Owned Subsidiary of the Borrower or payable to holders of minority interests in any Subsidiary so long as the Borrower or any other Subsidiary having an interest in such Subsidiary shall receive its proportionate share of such dividend or distribution ("Dividends"), and (y) cash distributions to members of the Borrower from time to time in accordance with the terms of the Tax Sharing Agreement ("Tax Distributions"), (ii) purchase, redeem or otherwise acquire or retire for value any Capital Stock of the Borrower, (iii) make any principal payment on or purchase, defease, redeem, prepay, or otherwise acquire or retire for value, prior to any scheduled final maturity or applicable redemption date, any Public Notes (or any Permitted Refinancing Indebtedness thereof) except to the extent set forth in Section 8.11(i) or (iv) make any Investment not specifically permitted by clauses (a) through (o) of Section 8.7 ("Unrestricted Investments") (any of the foregoing described in clauses (i) (iv) being hereafter referred to as a "Restricted Payment");
- B. Notwithstanding the limitations on Restricted Payments set forth in Section 8.4(a), so long as (x) there is no Default or Event of Default then outstanding or that would result therefrom, (y) the Borrower is able to incur \$1 of additional Indebtedness under Section 8.2(a) both before and after giving effect to such Restricted Payment on a Pro Forma Basis and (z) if the Assigned Dollar Value of all outstanding Revolving Loans and Swing Line Loans exceeds \$0 or any outstanding LC Obligations are not Cash Collateralized in an amount not less than 105% of the amount of such outstanding LC Obligations and, both before and after giving effect to such Restricted Payment on a Pro Forma Basis, the Senior Secured Leverage Ratio is not in excess of 3.75 to 1.00, the Borrower or any Subsidiary of the Borrower may make any Restricted Payment which together with all other Restricted Payments made pursuant to this Section 8.4 since the Effective Date would not exceed the sum of:
 - (a) \$400,000,000, plus
 - (b) 50% of the consolidated net income (as defined in the Senior Subordinated Note (2014) Indenture) (or, in the case of a Consolidated Net Loss, minus 100% of such Consolidated Net Loss) of the Borrower and its Subsidiaries accrued during the period (treated as one accounting period) from June 30, 2006 to the end of the most recent Fiscal Quarter for which financial statements have been delivered pursuant to Section 7.1, plus
 - (c) Available Equity Proceeds.
- C. Notwithstanding the foregoing, the Borrower may pay Dividends within 60 days after the date of declaration thereof if at such date of declaration such Dividend would

130

have complied with this <u>Section 8.4</u>; <u>provided</u>, <u>however</u>, that such Dividend shall be included (without duplication) in the calculation of Restricted Payments for <u>purposes</u> of <u>Section 8.4(b)</u>.

D. In addition to Restricted Payments permitted by clauses (b) and (c) above, the Borrower may pay Dividends to any Huntsman Parent Company which are contemporaneously applied to pay dividends on common stock of Huntsman Corporation at a rate not to exceed \$0.40 per share per annum (such amount to be appropriately adjusted to reflect any stock split, reverse stock split, stock dividend, stock issuance or similar transactions made after the Effective Date so that the aggregate amount of dividends payable after such transaction is the same as the amount payable immediately prior to such transaction).

16.5. Certain Restrictions on Subsidiaries

The Borrower will not, and will not permit any of its Subsidiaries to create or otherwise cause or permit to exist, or to become effective, any consensual encumbrance or restriction (other than pursuant to the Loan Documents) on the ability of any Subsidiary of the Borrower to (i) pay dividends or make any other distributions on its Capital Stock, (ii) pay any Indebtedness or other obligation owed to the Borrower or any of its other Subsidiaries, (iii) make any loans or advances to the Borrower or any of its other Subsidiaries, or (iv) transfer any of its property or assets to the Borrower or any of its other Subsidiaries, except:

- A. any encumbrance or restriction pursuant to an agreement in effect at or entered into on the Effective Date and reflected on <u>Schedule 8.5(a)</u> hereto or any extension, replacement or refinancing thereof not prohibited herein;
- B. any such encumbrance or restriction consisting of customary non-assignment provisions in any Contractual Obligation entered into in the ordinary course of business to the extent such provisions restrict the transfer or assignment of any agreement evidencing or securing such Contractual Obligation;
- C. in the case of clause (iv) above, Permitted Liens or other restrictions contained in security agreements securing Indebtedness or operating leases permitted hereby, to the extent such restrictions restrict the transfer of the property subject to such security agreements or operating leases;
- D. any restrictions on transfer of an asset (including Capital Stock) pursuant to an agreement to sell, transfer or dispose of such asset, to the extent such sale would be permitted hereby;

- E. any encumbrance or restriction on a Receivables Subsidiary as set forth in the Receivables Documents, or any encumbrance or restriction on a Participating Subsidiary with respect to Receivables Facility Assets as set forth in Receivables Documents;
 - F. restrictions on Foreign Subsidiaries in Overdraft Facilities;
- G. any restrictions on a Person (other than an Unrestricted Subsidiary) at the time such Person becomes a Subsidiary, so long as such restrictions were not entered into in contemplation of such Person becoming a Subsidiary; and

131

H. any restrictions or encumbrances on (i) Huntsman Chemical Company of Canada, Inc. or on any Foreign Subsidiary of Huntsman Petrochemical Corporation, (ii) Huntsman Advanced Materials (Netherlands) BV or any of its Foreign Subsidiaries or (iii) any Foreign Subsidiary organized under the laws of any Middle Eastern or Asian jurisdiction, which are set forth in any agreement governing Indebtedness permitted pursuant to Section 8.2(b)(xvi).

16.6. Issuance of Stock

- A. The Borrower will not, and will not permit any of its Subsidiaries to, directly or indirectly, issue (except, with respect to the Borrower, as permitted pursuant to clause (b) below), sell, assign, pledge or otherwise encumber (other than with Liens of the type described in clauses (i) and (vi) of the definition of "Customary Permitted Liens") or dispose of any shares of Capital Stock of any Subsidiary of the Borrower, except (i) to the Borrower, (ii) to another Wholly-Owned Subsidiary of the Borrower, (iii) to qualifying directors or to satisfy other similar requirements, in each case, pursuant to Requirements of Law, (iv) pursuant to the Loan Documents or the Foreign Intercompany Loan Security Documents or (v) pledges constituting Permitted Liens described in clauses (a), (d) or (e) of Section 8.1. Notwithstanding the foregoing, the Borrower or its Subsidiaries shall be permitted to sell all of the outstanding Capital Stock of a Subsidiary to the extent permitted pursuant to Section 8.3.
- B. The Borrower shall not issue any Capital Stock, except nonredeemable Capital Stock and except for such issuances of Capital Stock (including private placements) (x) where after giving effect to such issuance, no Event of Default will exist under Section 10.1(m) and (y) where the Administrative Agent and the Required Lenders have consented (such consent not to be unreasonably withheld) to the terms and conditions of such offering.

16.7. Loans and Investments

The Borrower will not, and will not permit any of its Subsidiaries to, make any Investments except:

- A. the Borrower and its Domestic Subsidiaries may acquire and hold Cash and Cash Equivalents;
- B. the Borrower and its Subsidiaries may hold the Investments as set forth on Schedule 8.7(b) hereto;
- C. the Borrower and its Subsidiaries may make or maintain advances (i) for relocation and related expenses and other advances to their employees in the ordinary course of business and (ii) for any other advances to their employees in the ordinary course of business in an aggregate principal amount not exceeding \$10,000,000 (or the Dollar Equivalent thereof) at any one time outstanding;
- D. the Borrower and its Subsidiaries may acquire and hold (i) Investments consisting of extensions of credit in the nature of accounts receivable arising from the granting of trade credit in the ordinary course of business, and (ii) Investments (including debt obligations) received in connection with the bankruptcy or reorganization of suppliers and customers and

132

other Persons and in settlement of delinquent obligations of, and other disputes with, customers and suppliers and other Persons arising in the ordinary course of business;

- E. the Borrower and its Subsidiaries may make deposits in a customary fashion in the ordinary course of business:
- F. the Borrower and its Subsidiaries may acquire and hold debt securities as partial consideration for a sale of assets pursuant to Section 8.3 or 4.4(c) to the extent permitted by any such Section;
- G. (1) the Borrower may make or maintain intercompany loans and advances to any of its Wholly-Owned Subsidiaries, (2) any Subsidiary of the Borrower may make or maintain intercompany loans and advances to the Borrower and (3) any Subsidiary of the Borrower may make or maintain intercompany loans and advances to any Wholly-Owned Subsidiary of the Borrower (including, without limitation, pursuant to Permitted Entrustment Loan Arrangements) (collectively, "Intercompany Loans"), provided, that each Intercompany Loan made by a Foreign Subsidiary or a non-Wholly-Owned Domestic Subsidiary, on the one hand, to the Borrower or a Wholly-Owned Domestic Subsidiary of the Borrower, on the other hand, shall contain the subordination provisions set forth on Exhibit 8.7(g);

- H. (i) the Borrower and its Subsidiaries may make Investments after the Fifth Amendment Effective Date in the Capital Stock of Persons that are Foreign Subsidiaries and may capitalize or forgive any Indebtedness owed to them by a Foreign Subsidiary (treating such capitalization or forgiveness as an Investment for purposes of this clause (i)); provided, that the aggregate outstanding amount of such Investments pursuant to this subclause (i) (excluding Investments consisting solely of the contribution of the Capital Stock of a Foreign Subsidiary to a Foreign Subsidiary organized in a jurisdiction acceptable to Administrative Agent and the Investment described on Schedule 8.7(h)) shall not exceed an aggregate outstanding amount equal to the sum of \$300,000,000 plus the aggregate amount contributed to Foreign Subsidiaries for Acquisitions permitted pursuant to Section 8.7(m), (ii) the Borrower and its Domestic Subsidiaries may make Investments in the Capital Stock of a Person that is a Domestic Subsidiary; provided, that the requirements of Section 7.11 are satisfied and (iii) Foreign Subsidiaries of the Borrower may make Investments in the Capital Stock of other Foreign Subsidiaries of the Borrower and may capitalize or forgive any Indebtedness owed to them by a Foreign Subsidiary (treating such capitalization or forgiveness as an Investment for purposes of this clause (iii));
 - I. Foreign Subsidiaries of the Borrower may invest in cash, Cash Equivalents and Foreign Cash Equivalents;
- J. so long as (1) no Unmatured Event of Default or Event of Default exists either before or after giving effect thereto and (2) the Borrower is in compliance with Section 9.1 both before and after giving effect thereto on a Pro Forma Basis (whether or not such Section 9.1 would otherwise be applicable), the Borrower and its Subsidiaries may (i) make any Investment in any Permitted Unconsolidated Venture or in any Unrestricted Subsidiary (provided, that the Borrower shall have complied with Section 7.11(e) in connection with such Investment) consisting of an amount not in excess of the Available Unrestricted Subsidiary Investment

133

Basket; and (ii) solely to the extent such Investment is used by any of HF II Australia Holdings Company LLC, Huntsman Australia Holdings Corp., HCPH Holdings Pty Limited, Huntsman Chemical Australia Unit Trust or their Subsidiaries, make Investments in such entities to permanently prepay Indebtedness in an aggregate amount not to exceed \$50,000,000.

- K. the Borrower may make intercompany loans to Huntsman Corporation, the proceeds of which shall be utilized by Huntsman Corporation to pay legal, franchise tax, audit, and other expenses directly relating to the administration or legal existence of the Borrower; <u>provided</u>, that the aggregate outstanding principal amount of such intercompany loans shall not exceed \$3,000,000 at any time outstanding (without giving effect to any write-downs or write-offs thereof) and which amount shall not include any intercompany loans or advances made or deemed to have been made for any reason in respect of accrued but unpaid interest on any intercompany loans previously made to Huntsman Corporation, including the capitalization thereof);
- L. the Borrower may make Investments in Rubicon and LPC, so long as: (i) the Administrative Agent possesses a valid, perfected Lien on the applicable Credit Party's interests in such Joint Venture, (ii) such Joint Venture does not have any Indebtedness for borrowed money at any time on or after the date of such Investment other than to the partners in such Joint Venture and (iii) the documentation governing such Joint Venture does not contain a restriction on distributions or loan repayments as applicable, to the Borrower or to the applicable Subsidiary holding the interest in such Joint Venture;
- M. the Borrower or any of its Subsidiaries may purchase all or a significant part of the assets of a business conducted by another Person, make any Investment in any Person which, after the Effective Date as a result of such Investment becomes a Wholly-Owned Subsidiary of the Borrower which is not an Unrestricted Subsidiary or, to the extent permitted under Section 8.3, enter into any merger, consolidation or amalgamation with any other Person (any such purchase, Investment or merger, an "Acquisition"); provided, however, that such Acquisition shall not be permitted unless, (i) after giving effect thereto on a Pro Forma Basis, (a) no Event of Default or Unmatured Event of Default would exist hereunder and (b) the Borrower is able to borrow \$1 of additional Indebtedness pursuant to Section 8.2(a); (ii) if the total consideration given and Indebtedness assumed in connection with such Acquisition exceeds \$20,000,000, after giving effect to such Acquisition, the Borrower's Available Liquidity, minus the lesser of (A) the aggregate amount of utilized Overdraft Facilities of the Borrower and its Subsidiaries or (B) \$60,000,000 shall equal or exceed \$450,000,000; (iii) the Borrower and its Subsidiaries have complied to the extent applicable with the requirements of Section 7.11 hereof with respect to any required additional Security Documents and (iv) such Acquisition has been approved by the board of directors of the Person to be acquired; provided, further, however, notwithstanding anything else herein to the contrary, no Acquisition (other than Acquisitions in which the total consideration paid (including for purposes hereof any assumed Indebtedness) does not exceed \$20,000,000 for any single Acquisition or \$80,000,000 for all such Acquisitions in the aggregate since the Effective Date) shall be permitted unless the Borrower is in compliance with Section 9.1 on a Pro Forma Basis;
- N. the Borrower or any of its Subsidiaries may make Investments in the Receivables Subsidiary and any Participating Subsidiaries prior to the occurrence and

134

continuance of an Event of Default under <u>Section 10.1(n)</u> which in the judgment of the Borrower are reasonably necessary in connection with any Permitted Accounts Receivable Securitization;

O. in addition to Investments permitted pursuant to clauses (a) through (n) above, the Borrower or any of its Subsidiaries may make other Investments (A) in an outstanding amount not to exceed \$50,000,000 in the aggregate or (B) with Available Equity Proceeds; provided, that in each case the Borrower shall have complied to the extent applicable with Section 7.11 in connection with such Investment; and provided further, that the Borrower may not make or own any investment in margin stock;

- P. the Borrower or any of its Subsidiaries may make Unrestricted Investments constituting Restricted Payments that are permitted by <u>Section 8.4(b)</u>; <u>provided</u>, <u>that</u>, the Borrower shall have complied to the extent applicable with <u>Section 7.11</u> in connection with such Unrestricted Investments; and <u>provided</u> <u>further</u>, that the Borrower may not make or own any investment in margin stock; and
- Q. the Borrower or any of its Subsidiaries may make loans and advances to their respective customers in an aggregate amount not to exceed at any time \$10,000,000.

16.8. Transactions with Affiliates

The Borrower will not, and the Borrower will not cause or permit any of its Subsidiaries to, directly or indirectly, enter into or permit to exist any transaction or series of related transactions (including, without limitation, the purchase, sale, lease or exchange of any property or the rendering of any service) with or for the benefit of any of the Borrower's Affiliates other than (x) the entry by the Borrower and its Subsidiaries into the transactions contemplated by a Permitted Accounts Receivable Securitization, (y) transactions that are on terms that are fair and reasonable to the Borrower or to any such Subsidiary and that are on terms that are no less favorable to the Borrower or to such Subsidiary than those that might reasonably have been obtained in a comparable transaction on an arm's-length basis from a Person that is not an Affiliate, or (z) any transaction arising in the ordinary course of business of the Borrower or of such Subsidiary: provided, however, that with respect to transactions between the Borrower or any of its Subsidiaries and any of their respective Affiliates arising in the ordinary course of business (including, without limitation, purchase or supply contracts relating to products or raw materials) a Responsible Officer of the Borrower shall, not later than the date of delivery of the financial statements referred to in Section 7.1(b), have reviewed the aggregate of such transactions and determined that, in the aggregate, such transactions are on terms that are fair and reasonable to the Borrower or to such Subsidiary and are no less favorable to the Borrower or to such Subsidiary than those that might reasonably have been obtained in a comparable transactions on an arm's-length basis from a Person that is not an Affiliate. The foregoing restrictions will not apply to (1) reasonable and customary directors' fees, indemnification and similar arrangements and payments thereunder; (2) any transaction between the Borrower and any Wholly-Owned Subsidiary (other than an Unrestricted Subsidiary) of the Borrower or between Wholly-Owned Subsidiaries (other than an Unrestricted Subsidiary) to the extent that any such transaction is otherwise in compliance with the terms of this Agreement; (3) loans or advances to officers of the Borrower and of its Subsidiaries for bona fide business purposes of the Borrower or of such Subsidiary not to exceed \$10,000,000 in the aggregate at any one time

135

outstanding for the Borrower and its Subsidiaries; (4) Investments permitted under <u>Section 8.7(k)</u>; (5) Restricted Payments permitted under <u>Section 8.4</u>; or (6) issuances by the Borrower of its Capital Stock to the extent such issuance is otherwise in compliance with the terms of this Agreement. The restriction set forth in this <u>Section 8.8</u> will not apply to the execution and delivery of or payments made under the Tax Sharing Agreement.

16.9. Lines of Business

The Borrower will not, and will not permit any Subsidiary (other than a Receivables Subsidiary) to enter into or acquire any line of business which is not reasonably related to the chemical or petrochemical business, provided, that none of Huntsman Finco or any Thai Holding Company will engage in any business other than (a) holding Capital Stock of its Subsidiaries and (b) in the case of Huntsman Finco, the borrowing and lending funds pursuant to the Intercompany Loans. IRIC shall only engage in the business of serving as a captive insurance company for the Borrower and its Subsidiaries and engaging in such necessary activities related thereto as may be permitted to be engaged in by a Utah captive insurance company pursuant to applicable Utah captive insurance company rules and regulations; provided, that IRIC shall not hold cash or other Investments except in a manner consistent with Schedule 8.9.

16.10. Fiscal Year

The Borrower shall not change its Fiscal Year.

16.11. Limitation on Voluntary Payments and Modifications of Indebtedness; Modifications of Certificate of Incorporation, By-Laws and Certain Other Agreements; Etc.

The Borrower will not, and will not permit any of its Subsidiaries to:

- make (or give any notice in respect of) any voluntary or optional payment or prepayment on or redemption or acquisition for value of (including, without limitation, by way of depositing with the trustee with respect thereto or any other Person money or securities before due for the purpose of paying when due) any obligations under any Public Notes (other than with the proceeds of Permitted Refinancing Indebtedness or with Available Equity Proceeds); provided that if there is no Default or Event of Default then outstanding or that would result therefrom, the Borrower or any of its Subsidiaries may make payments or prepayments on, or redemption or acquisition of: (A) any obligations under the Senior Secured Notes (2010) or the Senior Notes (2012) (including with Net Sale Proceeds from Asset Dispositions and Net Recovery Proceeds from Recovery Events not required by the terms of Section 4.4(c)(ii) to be used to prepay the Loans); and (B) any obligations under any Public Notes with Restricted Payments permitted under Section 8.4(b);
- (b) amend, modify or terminate, or permit the amendment, modification, or termination of any provision in any way materially adverse to the interests of the Lenders (as determined by the Administrative Agent in its sole reasonable discretion after reasonable advance notice of such proposed change) of

any Public Note Document that would require the consent of any holder of such Public Note (including, without limitation, by the execution of any supplemental indenture);

- (c) amend, modify or change in any way adverse, in any material respect, to the interests of the Lenders, its Organizational Documents (including, without limitation, by filing or modification of any certificate of designation) or by-laws, or any agreement entered into by it, with respect to its Capital Stock, the Tax Sharing Agreement, or enter into any new agreement with respect to its Capital Stock or any new tax sharing agreement which in any way could reasonably be expected to be adverse to the interests of the Lenders; or
- if the Assigned Dollar Value of all outstanding Revolving Loans and Swing Line Loans exceeds \$0 or any outstanding LC Obligations are not Cash Collateralized in an amount not less than 105% of the amount of such outstanding LC Obligations and, after giving effect to any payment referred to below on a Pro Forma Basis, the Senior Secured Leverage Ratio would be in excess of 3.75 to 1.00, make any payment of interest or principal or other amount in respect of any Indebtedness of the Borrower or any Subsidiary held by any Affiliate of the Borrower that is not the Borrower or a Subsidiary of the Borrower, including but not limited to that certain promissory note with an effective date of May 30, 2008, as amended and restated effective as of January 6, 2009, issued by the Borrower to Huntsman Corporation, on any date unless, on a Pro Forma Basis both before and after giving effect to such payment, the Senior Secured Leverage Ratio as of such date would not exceed 3.75 to 1.00.

The Administrative Agent agrees that, with respect to any matters required to be reasonably satisfactory or acceptable to it, it shall exercise its reasonable judgment in making, and shall not unreasonably withhold or delay, such determination.

16.12. Accounting Changes

The Borrower shall not, nor shall it permit any of its Subsidiaries to make or permit to be made any change in accounting policies affecting the presentation of financial statements or reporting practices from those employed by it on the date hereof, unless (i) such change is required by GAAP, (ii) such change is disclosed to the Lenders through the Administrative Agent or otherwise and (iii) relevant prior financial statements that are affected by such change are restated (in form and detail satisfactory to the Administrative Agent) as may be required by GAAP to show comparative results. If any changes in GAAP or the application thereof from that used in the preparation of the financial statements referred to in Section 6.5(a) hereof occur after the Closing Date and such changes result in, in the sole judgment of the Administrative Agent, a meaningful change in the calculation of any financial covenants or restrictions set forth in this Agreement, then the parties hereto agree to enter into and diligently pursue negotiations in order to amend such financial covenants and restrictions so as to equitably reflect such changes, with the desired result that the criteria for evaluating the financial condition and results of operations of the Borrower and its Subsidiaries shall be the same after such changes as if such changes had not been made.

137

16.13. Permitted Accounts Receivable Securitization and Foreign Factoring Transactions

The Borrower shall not, nor shall it permit any of its Subsidiaries to, enter into any Receivables Documents other than in connection with a Permitted Accounts Receivable Securitization or a Foreign Factoring Transaction permitted by Sections 8.1(1) and 8.2(b)(xv) (unless such Receivables Documents have been approved by the Administrative Agent or are non-material documentation entered into pursuant to such approved Receivables Documents and/or represent additional documentation in customary form that is contemplated by Receivables Documents for a Permitted Accounts Receivable Securitization and are not adverse to the Lenders) or amend or modify in any material respect which is adverse to the Lenders any of such Receivables Documents unless such amendment or modification has been approved by the Administrative Agent; provided, however, that if the Receivables Documents, after giving effect to such amendment or modification, would constitute a Permitted Accounts Receivable Securitization, then such approval of the Administrative Agent shall not be required. No Unrestricted Subsidiary may be a Participating Subsidiary in a Permitted Accounts Receivable Securitization.

SECTION 17

FINANCIAL COVENANTS

The Borrower hereby agrees that, at any time when Revolving Loans or LC Obligations are outstanding (and for this purpose, the term "LC Obligations" shall exclude the Stated Amount of any outstanding and undrawn Letter of Credit for which the Borrower or any Affiliate thereof has Cash Collateralized such LC Obligations to the Administrative Agent (or the applicable Facing Agent) in an amount not less than 105% of the Stated Amount of such Letter of Credit):

17.1. Senior Secured Leverage Ratio

Unless compliance herewith is waived by the Majority Revolving Facility Lenders, the Borrower will not permit for any Test Period ending on the last date of any Fiscal Quarter, the Senior Secured Leverage Ratio to exceed 3.75 to 1.00.

SECTION 18

EVENTS OF DEFAULT

18.1. Events of Default

Any of the following events, acts, occurrences or states of facts shall constitute an "Event of Default" for purposes of this Agreement:

A. <u>Failure to Make Payments When Due.</u> There shall occur a default in the payment of (i) principal on any of the Loans or any reimbursement obligation with respect to any Letter of Credit; or (ii) interest on any of the Loans or any fee or any other amount owing

138

hereunder or under any other Loan Document when due and such default in payment shall continue for five (5) Business Days; or

- B. <u>Representations and Warranties</u>. Any representation or warranty made by or on the part of the Borrower or any Credit Party, as the case may be, contained in any Loan Document or any document, instrument or certificate delivered pursuant hereto or thereto shall have been incorrect or misleading in any material respect when made or deemed made; or
- C. <u>Covenants</u>. The Borrower shall (i) default in the performance or observance of any term, covenant, condition or agreement on its part to be performed or observed under <u>Article VIII</u> or <u>Article IX</u> hereof or <u>Sections 3.8, 7.3(a), 7.9, 7.10</u> or <u>7.11</u> or (ii) default in the due performance or observance by it of any other term, covenant or agreement contained in this Agreement and such default shall continue unremedied for a period of thirty (30) days after written notice to the Borrower by the Administrative Agent or any Lender; <u>provided</u> that any Revolver Event of Default shall not constitute an Event of Default with respect to any of the Term Facilities until the earlier of (x) the date that is forty-five (45) days after the date such Event of Default arises with respect to the Revolving Facility and (y) the date on which the Administrative Agent exercises any remedies with respect to the Revolving Facilities in accordance with the provisions of this <u>Section 10.1</u>; and <u>provided</u>, <u>further</u>, that any Revolver Event of Default may be waived from time to time by the Majority Revolving Facility Lenders pursuant to <u>Section 9.1</u>; or
- D. <u>Default Under Other Loan Documents</u>. Any Credit Party shall default in the performance or observance of any term, covenant, condition or agreement on its part to be performed or observed hereunder or under any Loan Document (and not constituting an Event of Default under any other clause of this <u>Section 10.1</u>) and such default shall continue unremedied for a period of thirty (30) days after written or telephonic (immediately confirmed in writing) notice thereof has been given to the Borrower by the Administrative Agent; or
- E. <u>Voluntary Insolvency, Etc.</u> The Borrower or any of its Material Subsidiaries shall become insolvent or generally fail to pay, or admit in writing its inability to pay, its debts as they become due, or shall voluntarily commence any proceeding or file any petition under any bankruptcy, insolvency or similar law or seeking dissolution or reorganization or the appointment of a receiver, trustee, administrator, custodian, liquidator or similar officer for it or a substantial portion of its property, assets or business or to effect a plan or other arrangement with its creditors, or shall file any answer admitting the material allegations of an involuntary petition filed against it in any bankruptcy, insolvency or similar proceeding, or shall be adjudicated bankrupt, or shall make a general assignment for the benefit of creditors, or shall consent to, or acquiesce in the appointment of, a receiver, trustee, custodian, administrator, liquidator or similar officer for a substantial portion of its property, assets or business, shall call a meeting of its creditors with a view to arranging a composition or adjustment of its debts or shall take any corporate action authorizing any of the foregoing; or
- F. <u>Involuntary Insolvency, Etc.</u> Involuntary proceedings or an involuntary petition shall be commenced or filed against the Borrower or any of its Material Subsidiaries under any bankruptcy, insolvency or similar law or seeking the dissolution or reorganization of it or the appointment of a receiver, trustee, custodian, administrator, liquidator or similar officer for

139

it or of a substantial part of its property, assets or business, or any similar writ, judgment, warrant of attachment, execution or process shall be issued or levied against a substantial part of its property, assets or business, and (other than a petition for administration) such proceedings or petition shall not be dismissed, or such writ, judgment, warrant of attachment, execution or similar process shall not be released, vacated or fully bonded, within sixty (60) days after commencement, filing or levy, as the case may be, or any order for relief shall be entered in any such proceeding; or

G. <u>Default Under Other Agreements.</u> (i) The Borrower or any of its Subsidiaries shall default in the payment when due, whether at stated maturity or otherwise, of any amount pursuant to any Indebtedness (other than Indebtedness owed to the Lenders under the Loan Documents) in excess of \$50,000,000 in the aggregate beyond the period of grace if any, provided in the instrument or agreement under which such Indebtedness was created, or (ii) a default shall occur in the performance or observance of any agreement under any such Indebtedness or contained in any instrument or agreement evidencing, securing or relating thereto, or any other event shall occur or condition exist, the effect of which default or other event or condition is to cause, or to permit the holder or holders of such Indebtedness (or a trustee or agent on behalf of such holder or holders) to cause (determined without regard to whether any notice of acceleration or similar notice is required), any such Indebtedness to become due or be repaid prior to its stated maturity or (iii) any such Indebtedness of the Borrower or any of its Subsidiaries shall be declared to be due and payable, or required to be prepaid (other than such Indebtedness that is required to be prepaid upon a "Change of Control" under a Public Note Document that would not cause a Change of Control hereunder) other than by a regularly scheduled required prepayment (other than with proceeds of the event giving rise to such

prepayment), prior to the stated maturity thereof; or

- H. <u>Invalidity of Subordination Provisions</u>. The subordination provisions of any agreement or instrument governing the Senior Subordinated Note (2013) Documents, the Senior Subordinated Note (2014) Documents, the Senior Subordinated Note (2015) Documents or any other subordinated Indebtedness in excess of \$50,000,000 is for any reason revoked or invalidated, or otherwise cease to be in full force and effect, any Person contests in any manner the validity or enforceability thereof or denies that it has any further liability or obligation thereunder, or the Loans and the other Obligations hereunder entitled to receive the benefits of any Loan Document is for any reason subordinated or does not have the priority contemplated by this Agreement or such subordination provisions; or
- I. <u>Judgments</u>. One or more judgments or decrees shall be entered against the Borrower or any of its Subsidiaries involving, individually or in the aggregate, a liability (to the extent not paid or covered by a reputable insurance company or indemnitor as to which coverage or indemnification, as the case may be, has not been disclaimed) of \$50,000,000 or more and all such judgments or decrees shall not have been vacated, discharged, satisfied, stayed or bonded pending appeal within thirty (30) days from the entry thereof; or
- J. <u>Security Documents</u>. At any time after the execution and delivery thereof, any of the Security Documents shall cease to be in full force and (other than as permitted pursuant to the provisions thereof or hereof) cease to create a valid and perfected lien on and

140

security interest in, any material portion of the Collateral having the lien priority required by this Agreement and the Security Documents; or

- K. <u>Guaranties</u>. Any Guaranty or any provision thereof shall (other than as a result of the actions taken by the Administrative Agent or the Lenders to release such Guaranty) cease to be in full force and effect in accordance with its terms (other than as permitted pursuant to the terms hereof and thereof), or any Credit Party or any Person acting by or on behalf of such Guarantor shall deny or disaffirm such Credit Party's obligations under any Guaranty; or
- ERISA. (a) Either (i) any Reportable Event which the Required Lenders determine constitutes reasonable grounds for the termination of any Plan by the PBGC or of any Multiemployer Plan or for the appointment by the appropriate United States District Court of a trustee to administer or liquidate any Plan or Multiemployer Plan shall have occurred, (ii) a trustee shall be appointed by a United States District Court to administer any Plan or Multiemployer Plan, (iii) the PBGC shall institute proceedings to terminate any Plan or Multiemployer Plan or to appoint a trustee to administer any Plan; (iv) the Borrower or any of its Subsidiaries or any of their ERISA Affiliates shall become liable to the PBGC or any other party under Section 4062, 4063 or 4064 of ERISA with respect to any Plan; or (v) the Borrower or any of its Subsidiaries or any of their ERISA Affiliates shall become liable to make a current payment with respect to any Multiemployer Plan under Section 4201 et seq. of ERISA; if as of the date thereof or any subsequent date, the sum of each of the Borrower's and its Subsidiaries' and their ERISA Affiliates' various liabilities (such liabilities to include, without limitation, any liability to the PBGC or to any other party under Section 4062, 4063 or 4064 of ERISA with respect to any Plan, or to any Multiemployer Plan under Section 4201 et seq. of ERISA) as a result of such events listed in subclauses (i) through (v) above exceeds \$50,000,000; or (b) Either (i) a foreign governmental authority has instituted proceedings to terminate a Foreign Pension Plan or a foreign governmental authority has appointed a trustee to administer any Foreign Pension Plan in place of the existing administrator, in each case by reason of a distress termination within the meaning of Section 4041(c) of ERISA, treating such Foreign Pension Plan as if it were subject to ERISA; or (ii) any Foreign Pension Plan that is required by applicable law to be funded in a trust or other funding vehicle has failed to comply with such funding requirements; if, as of the date thereof or as of any subsequent date, the sum of each of the Borrower's and its Subsidiaries' various liabilities to any Foreign Pension Plan solely as a result of such events listed in subclauses (i) and (ii) of this clause (b) exceeds the Dollar Equivalent of \$50,000,000; or
 - M. Change of Control. A Change of Control shall occur; or
- N. <u>Receivables Facility</u>. Any event (after the expiration of any applicable grace periods) as specified in the Receivables Documents for any Permitted Accounts Receivable Securitization related to Receivables Facility Attributed Indebtedness at such time of \$50,000,000 or more shall entitle the Persons (other than a Receivable Subsidiary) financing Receivables Facility Assets pursuant to such Permitted Accounts Receivable Securitization prior to the scheduled or mutually agreed upon (at a time when no default exists thereunder) termination thereof to terminate or permanently cease funding the financing of Receivables Facility Assets pursuant to such Permitted Accounts Receivable Securitization.

141

If any of the foregoing Events of Default shall have occurred and be continuing, the Administrative Agent, at the written direction of the Required Lenders, shall take one or more of the following actions (provided that, in the case of an Event of Default described in clause (c) above arising solely from a breach of Section 9.1, prior to the earlier of (x) the date that is forty-five (45) days after such Event of Default and (y) the date the Administrative Agent exercises any remedies pursuant to this proviso, the Administrative Agent shall take such actions (x) at the request of the Majority Revolving Facility Lenders rather than the Required Lenders and (y) only with respect to the Revolving Facility): (i) by written or oral or telephonic notice (in the case of oral or telephonic notice confirmed in writing immediately thereafter) to the Borrower declare the Total Commitments to be terminated whereupon the Total Commitments shall forthwith terminate, (ii) by written or oral or telephonic notice (in the case of oral or telephonic notice confirmed in

writing immediately thereafter) to the Borrower declare all sums then owing by the Borrower hereunder and under the Loan Documents to be forthwith due and payable, whereupon all such sums shall become and be immediately due and payable without presentment, demand, protest or notice of any kind, all of which are hereby expressly waived by the Borrower, (iii) terminate any Letter of Credit in accordance with its terms, (iv) direct the Borrower to pay (and the Borrower agrees that upon receipt of such notice, or upon the occurrence of any Event of Default specified in Section 10.1(e) or Section 10.1(f) with respect to the Borrower it will pay) to the Administrative Agent at the Payment Office such additional amount of cash, to be held as security by the Administrative Agent, as is equal to the Assigned Dollar Value of the aggregate Stated Amount of all Letters of Credit issued for the account of the Borrower and its Subsidiaries and then outstanding, and (v) enforce, as the Administrative Agent (to the extent permitted under the applicable Security Documents), or direct the Collateral Agent to enforce pursuant to the Security Documents, as the case may be, all of the Liens and security interests created pursuant to the Security Documents. In cases of any occurrence of any Event of Default described in Section 10.1(e) or Section 10.1(f) with respect to the Borrower, the Loans, together with accrued interest thereon, shall become due and payable forthwith without the requirement of any such acceleration or request, and without presentment, demand, protest or other notice of any kind, all of which are expressly waived by the Borrower, any provision of this Agreement or any other Loan Document to the contrary notwithstanding, and other amounts payable by the Borrower hereunder shall also become immediately due and payable all without notice of any kind.

Notwithstanding anything to the contrary contained in this Agreement, upon the occurrence and during the continuance of an Event of Default, (a) the Borrower irrevocably waives the right to direct the application of any and all payments at any time or times thereafter received by Agent from or on behalf of the Borrower, and the Administrative Agent shall have the continuing and exclusive right to apply and to reapply any and all payments received at any time or times after the occurrence and during the continuance of an Event of Default. Notwithstanding anything to the contrary contained in this Agreement (including, without limitation, Article IV hereof), all payments (including the proceeds of any Asset Disposition or other sale of, or other realization upon, all or any part of the Collateral) received after acceleration of the Obligations shall be applied: first, to all fees, costs and expenses incurred by or owing to the Administrative Agent and any Lender with respect to this Agreement, the other Loan Documents or the Collateral; second, to accrued and unpaid interest on the Obligations (including any interest which but for the provisions of the Bankruptcy Code, would have accrued on such amounts); third, to the principal amount of the Obligations outstanding and to cash

142

collateralize outstanding Letters of Credit (pro rata among all such Obligations based upon the principal amount thereof or the outstanding face amount of such Letters of Credit, as applicable, and with respect to amounts applied to Term Loans, pro rata among all remaining Scheduled Term Repayments thereof). Any balance remaining shall be delivered to the Borrower or to whomever may be lawfully entitled to receive such balance or as a court of competent jurisdiction may direct.

Anything in this <u>Section 10.1</u> to the contrary notwithstanding, the Administrative Agent shall, at the request of the Required Lenders, rescind and annul any acceleration of the Loans by written instrument filed with the Borrower; provided that at the time such acceleration is so rescinded and annulled: (A) all past due interest and principal (other than principal due solely as a result of such acceleration), if any, on the Loans and all other sums payable under this Agreement and the other Loan Documents shall have been duly paid, and (B) no other Event of Default shall have occurred and be continuing which shall not have been waived in accordance with the provisions of <u>Section 12.1</u> hereof. Upon any such rescission and annulment, the Administrative Agent shall return to the Borrower any cash collateral delivered pursuant to the preceding paragraph.

18.2. Rights Not Exclusive

The rights provided for in this Agreement and the other Loan Documents are cumulative and are not exclusive of any other rights, powers, privileges or remedies provided by law or in equity, or under any other instrument, document or agreement now existing or hereafter arising.

SECTION 19

THE ADMINISTRATIVE AGENT

In this Article XI, the Lenders agree among themselves as follows:

19.1. Appointment

The Lenders hereby appoint JPMCB as the Administrative Agent (for purposes of this Article XI, the term "Administrative Agent" shall, except for purposes of Section 11.9, include JPMCB in its capacity as the Collateral Agent pursuant to the Security Documents) to act as specified herein and in the other Loan Documents. Each Lender hereby irrevocably authorizes and each holder of any Note by the acceptance of such Note shall be deemed to irrevocably authorize the Administrative Agent to take such action on its behalf under the provisions hereof, the other Loan Documents (including, without limitation, to give notices and take such actions on behalf of the Required Lenders as are consented to in writing by the Required Lenders or all Lenders, as the case may be) and any other instruments, documents and agreements referred to herein or therein and to exercise such powers hereunder and thereunder as are specifically delegated to the Administrative Agent by the terms hereof and thereof and such other powers as are reasonably incidental thereto. The Administrative Agent may perform any

of its duties hereunder and under the other Loan Documents, by or through its officers, directors, agents, employees or affiliates.

19.2. Nature of Duties

A. The Administrative Agent shall not have any duties or responsibilities except those expressly set forth in this Agreement. The duties of the Administrative Agent shall be mechanical and administrative in nature. EACH LENDER HEREBY ACKNOWLEDGES AND AGREES THAT, SUBJECT TO SECTION 11.2(b), THE ADMINISTRATIVE AGENT SHALL NOT HAVE, BY REASON OF THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT, A FIDUCIARY RELATIONSHIP TO OR IN RESPECT OF ANY LENDER. Nothing in any of the Loan Documents, expressed or implied, is intended to or shall be so construed as to impose upon the Administrative Agent any obligations in respect of any of the Loan Documents except as expressly set forth herein or therein. Each Lender shall make its own independent investigation of the financial condition and affairs of the Borrower in connection with the making and the continuance of the Loans hereunder and shall make its own appraisal of the credit worthiness of the Borrower, and the Administrative Agent shall not have any duty or responsibility, either initially or on a continuing basis, to provide any Lender with any credit or other information with respect thereto, whether coming into its possession before making of the Loans or at any time or times thereafter. Except as expressly set forth in the Loan Documents, the Administrative Agent shall not have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Subsidiaries that is communicated to or obtained by the financial institution serving as the Administrative Agent or any of its Affiliates in any capacity. The Administrative Agent will promptly notify each Lender at any time that the Required Lenders have instructed it to act or refrain from acting pursuant to Article X.

B. The Administrative Agent hereby declares that it, including in its capacity as Collateral Agent, holds and shall

- (a) all rights, title and interest that may now or hereafter be mortgaged, charged or assigned or otherwise secured in favor of the Administrative Agent and/or the Collateral Agent by or pursuant to the Loan Documents governed by English law and all proceeds of enforcement of such security; and
- (b) the benefit of all representations, covenants, guarantees, indemnities and other contractual provisions governed by English law given in favor of the Administrative Agent and/or the Collateral Agent (other than any such benefits given to the Administrative Agent and/or the Collateral Agent solely for its own benefit), on trust (for which the perpetuity period shall be 80 years) for itself and the other Lenders from time to time.

19.3. Exculpation, Rights Etc.

hold:

Neither the Administrative Agent nor any of its officers, directors, agents, employees or affiliates shall be liable to any Lender for any action taken or omitted by them

144

hereunder or under any of the other Loan Documents, or in connection herewith or therewith, unless caused by its or their gross negligence or willful misconduct. The Administrative Agent shall not be responsible to any Lender for any recitals, statements, representations or warranties herein or for the execution, effectiveness, genuineness, validity, enforceability, collectability, or sufficiency of any of the Loan Documents or any other document or the financial condition of the Borrower. The Administrative Agent shall not be required to make any inquiry concerning either the performance or observance of any of the terms, provisions or conditions of this Agreement or any of the Loan Documents or any other document or the financial condition of the Borrower, or the existence or possible existence of any Unmatured Event of Default or Event of Default unless requested to do so by the Required Lenders. The Administrative Agent may at any time request instructions from the Lenders with respect to any actions or approvals (including the failure to act or approve) which by the terms of any of the Loan Documents, the Administrative Agent is permitted or required to take or to grant, and if such instructions are requested, the Administrative Agent shall be absolutely entitled to refrain from taking any action or to withhold any approval and shall not be under any liability whatsoever to any Person for refraining from any action or withholding any approval under any of the Loan Documents until it shall have received such instructions from the Required Lenders. Without limiting the foregoing, no Lender shall have any right of action whatsoever against the Administrative Agent as a result of the Administrative Agent acting, approving or refraining from acting or approving under any of the Loan Documents in accordance with the instructions of the Required Lenders or, to the extent required by Section 12.1, all of the Lenders.

19.4. Reliance

The Administrative Agent shall be entitled to rely, and shall be fully protected in relying, upon any notice, writing, resolution notice, statement, certificate, order or other document or any telephone, telex, teletype, telecopier or electronic message reasonably believed by it to be genuine and correct and to have been signed, sent or made by the proper Person, and, with respect to all matters pertaining herein or to any of the other Loan Documents and its duties hereunder or thereunder, upon advice of counsel selected by the Administrative Agent.

19.5. Indemnification

To the extent the Administrative Agent is not reimbursed and indemnified by the Borrower, the Lenders will reimburse and indemnify the Administrative Agent for and against any and all liabilities, obligations, losses, damages, claims, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever which may be imposed on, incurred by, or asserted against the Administrative Agent, acting pursuant hereto in such capacity, in any way relating to or arising out of this Agreement or any of the other Loan Documents or any action taken or omitted by the Administrative Agent under this Agreement or any of the other Loan

For purposes of this <u>Section 11.5</u>, "Aggregate Pro Rata Share" means, when used with reference to any Lender and any described aggregate or total amount, an amount equal to the result obtained by multiplying such desired aggregate or total amount by a fraction the numerator of which shall be the aggregate principal amount of such Lender's Loans and the denominator of which shall be aggregate of all of the Loans outstanding hereunder.

19.6. The Administrative Agent In Its Individual Capacity

With respect to its Loans and Commitments (and its Pro Rata Share of each Facility thereof), the Administrative Agent shall have and may exercise the same rights and powers hereunder and is subject to the same obligations and liabilities as and to the extent set forth herein for any other Lender or holder of Obligations. The terms "Lenders", "holder of Obligations" or "Required Lenders" or any similar terms shall, unless the context clearly otherwise indicates, include the Administrative Agent in its individual capacity as a Lender, one of the Required Lenders or a holder of Obligations. The Administrative Agent may accept deposits from, lend money to, and generally engage in any kind of banking, trust or other business with the Borrower or any Subsidiary or affiliate of the Borrower as if it were not acting as the Administrative Agent hereunder or under any other Loan Document, including, without limitation, the acceptance of fees or other consideration for services without having to account for the same to any of the Lenders.

19.7. Notice of Default

The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of any Event of Default or Unmatured Event of Default hereunder unless the Administrative Agent has received written notice from a Lender or the Borrower referring to this Agreement describing such Event of Default or Unmatured Event of Default and stating that such notice is a "notice of default". In the event that the Administrative Agent receives such a notice, the Administrative Agent shall give prompt notice thereof to the Lenders.

19.8. Holders of Obligations

The Administrative Agent may deem and treat the payee of any Obligation as reflected on the books and records of the Administrative Agent as the owner thereof for all purposes hereof unless and until a written notice of the assignment or transfer thereof shall have been filed with the Administrative Agent pursuant to Section 12.8(c). Any request, authority or consent of any Person who, at the time of making such request or giving such authority or consent, is the holder of any Obligation shall be conclusive and binding on any subsequent holder, transferee or assignee of such Obligation or of any Obligation or Obligations granted in exchange therefor.

19.9. Resignation by the Administrative Agent

A. The Administrative Agent may resign from the performance of all its functions and duties hereunder at any time by giving fifteen (15) Business Days' prior written notice to the Borrower and the Lenders. Such resignation shall take effect upon the acceptance by a successor Administrative Agent of appointment pursuant to clauses (b) and (c) below or as otherwise provided below.

146

- B. Upon any such notice of resignation, the Required Lenders shall appoint a successor Administrative Agent who shall be satisfactory to the Borrower and shall be an incorporated bank or trust company.
- C. If a successor Administrative Agent shall not have been so appointed within said fifteen (15) Business Day period, the Administrative Agent, with the consent of the Borrower, shall then appoint a successor who shall serve as the Administrative Agent until such time, if any, as the Required Lenders, with the consent of the Borrower, appoint a successor the Administrative Agent as provided above.
- D. If no successor Administrative Agent has been appointed pursuant to clause (b) or (c) by the twentieth (20th) Business Day after the date such notice of resignation was given by the Administrative Agent, the Administrative Agent's resignation shall become effective and the Required Lenders shall thereafter perform all the duties of the Administrative Agent hereunder until such time, if any, as the Required Lenders, with the consent of the Borrower, appoint a successor Administrative Agent as provided above.
- E. Notwithstanding anything herein to the contrary, if at any time the Required Lenders determine that the Person serving as Administrative Agent is (without taking into account any provision in the definition of "Defaulting Lender" requiring notice from or a determination by the Administrative Agent or any other party) a Defaulting Lender, the Required Lenders (determined after giving effect to Section 12.1) may, with the consent of the Borrower and by notice to such Person, remove such Person as Administrative Agent and, with the consent of the Borrower, appoint a replacement Administrative Agent hereunder. Such removal will, to the fullest extent permitted by applicable law, be effective on the date a replacement Administrative Agent is appointed in accordance with this paragraph.
 - F. In addition to the foregoing, if (i) a Revolving Lender becomes, and during the period it remains, an Impaired

Lender, and (ii) the Borrower has failed to comply with its obligations pursuant to Section 3.8(a)(ii), any Facing Agent and/or the Swing Line Lender may, upon prior written notice to the Borrower and the Administrative Agent, resign as Facing Agent or Swing Line Lender, respectively, effective at the close of business New York time on a date specified in such notice (which date may not be less than 30 days after the date of such notice); provided that such resignation by a Facing Agent will have no effect on the validity or enforceability of any Letter of Credit then outstanding or on the obligations of the Borrower or any Lender under this Agreement with respect to any such outstanding Letter of Credit or otherwise to the Facing Agent; and provided, further, that such resignation by the Swing Line Lender will have no effect on its rights in respect of any outstanding Swing Line Loans or on the obligations of the Borrower or any Lender under this Agreement with respect to any such outstanding Swing Line Loan.

G. Any resignation by the Administrative Agent pursuant to Section 11.9(a) shall, unless the Administrative Agent gives notice to the Borrower otherwise, also constitute its resignation as a Facing Agent and the Swing Line Lender, and such resignation as a Facing Agent and the Swing Line Lender shall become effective simultaneously with the discharge of the Administrative Agent from its duties and obligations as set forth in Section 11.9(a) (except as to already outstanding Letters of Credit and LC Obligations issued by it and Swing Line Loans

147

made by it, as to which such Facing Agent and the Swing Line Lender shall continue in such capacities until the LC Obligations relating thereto shall be reduced to zero and such Swing Line Loans shall have been repaid, as applicable, or until the successor Administrative Agent shall succeed to the roles of a Facing Agent and the Swing Line Lender in accordance with the next sentence and perform the actions required by the next sentence). Upon the acceptance of a successor's appointment as Administrative Agent hereunder, unless the retiring Administrative Agent and such successor gives notice to Borrower otherwise, (i) such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring Facing Agent and Swing Line Lender and (ii) the successor Facing Agent shall issue letters of credit in substitution for the Letters of Credit, if any, outstanding at the time of such succession and issued by the retiring Facing Agent or make other arrangements satisfactory to the retiring Facing Agent to effectively assume the obligations of the retiring Facing Agent with respect to such Letters of Credit. At the time any such resignation of a Facing Agent shall become effective, the Borrower shall pay all unpaid fees accrued for the account of the retiring Facing Agent pursuant to Section 2.9(e).

H. Notwithstanding the resignation of DB as Administrative Agent and Collateral Agent (the "Resigning Agent") as of the Fifth Amendment Effective Date, the Borrower and each Lender agree that the provisions of this Article XI and any other provisions of this Agreement or any other Loan Document regarding payment of costs and expenses and indemnification of the Administrative Agent or the Collateral Agent, together with any provision of any Loan Document that shall accrue to the benefit of any retiring or resigning Agent, shall continue to inure to the benefit of the Resigning Agent on and following the Fifth Amendment Effective Date with respect to actions of the Resigning Agent taken (i) on or prior to the Fifth Amendment Effective Date, and (ii) subsequent to the Fifth Amendment Effective Date pursuant to the Loan Documents or that certain Successor Agency Agreement, dated as of the Fifth Amendment Effective Date, by and among JPMorgan Chase Bank, N.A., the Resigning Agent and the Borrower. DB shall be a third party beneficiary for purposes of this Section 11.9(h).

19.10. Administrative Agent or the Collateral Agent as UK Security Trustee

A. In this Agreement, any rights and remedies exercisable by, any documents to be delivered to, or any other indemnities or obligations in favor of the Administrative Agent or the Collateral Agent shall be, as the case may be, exercisable by, delivered to, or be indemnities or other obligations in favor of, the Administrative Agent or the Collateral Agent (or any other Person acting in such capacity) in its capacity as the UK Security Trustee to the extent that the rights, deliveries, indemnities or other obligations relate to the UK Security Documents or the security thereby created. Any obligations of the Administrative Agent or the Collateral Agent (or any other Person acting in such capacity) in this Agreement shall be obligations of the Administrative Agent or the Collateral Agent in its capacity as UK Security Trustee to the extent that the obligations relate to the UK Security Documents or the security thereby created. Additionally, in its capacity as UK Security Trustee, the Administrative Agent or the Collateral Agent (or any other Person acting in such capacity) shall have (i) all the rights, remedies and benefits in favor of the Administrative Agent or the Collateral Agent contained in the provisions of the whole of this Section 11.10; (ii) all the powers of an absolute owner of the security constituted by the UK Security Documents and (iii) all the rights, remedies and powers granted

148

to it and be subject to all the obligations and duties owed by it under the UK Security Documents and/or any of the Loan Documents.

- B. Each Lender, the Administrative Agent and the Collateral Agent hereby appoint the UK Security Trustee to act as its trustee under and in relation to the UK Security Documents and to hold the assets subject to the security thereby created as trustee for itself and other Secured Parties on the trusts and other terms contained in the UK Security Documents and the Administrative Agent and each Secured Party hereby irrevocably authorize the UK Security Trustee to exercise such rights, remedies, powers and discretions as are specifically delegated to the UK Security Trustee by the terms of the UK Security Documents together with all such rights, remedies, powers and discretions as are reasonably incidental thereto.
- C. Any reference in this Agreement to Liens stated to be in favor of the Administrative Agent or the Collateral Agent shall be construed so as to include a reference to Liens granted in favor of the UK Security Trustee.
- D. The Lenders agree that at any time that the UK Security Trustee shall be a Person other than the Administrative Agent or the Collateral Agent, such other Person shall have the rights, remedies, benefits and powers granted to the

Administrative Agent or the Collateral Agent in its capacity as the UK Security Trustee in this Agreement.

- E. Nothing in this <u>Section 11.10</u> shall require the UK Security Trustee to act as a trustee at common law or to be holding any property on trust, in any jurisdiction outside the United States or the United Kingdom which may not operate under principles of trust or where such trust would not be recognized or its effects would not be enforceable.
- **19.11.** The Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents.

Notwithstanding any other provision of this Agreement or any provision of any other Loan Document, each of the Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents are named as such for recognition purposes only, and in their respective capacities as such shall have no powers, duties, responsibilities or liabilities with respect to this Agreement or the other Loan Documents or the transactions contemplated hereby and thereby; it being understood and agreed that the Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents shall be entitled to all indemnification and reimbursement rights in favor of "Agents" as provided for under Section 11.5. Without limitation of the foregoing, none of Joint Lead Arrangers, Joint Book Runners, Co-Syndication Agents, Co-Documentation Agents and Senior Managing Agents shall, solely by reason of this Agreement or any other Loan Documents, have any fiduciary relationship in respect of any Lender or any other Person.

149

SECTION 20

MISCELLANEOUS

20.1. No Waiver; Modifications in Writing

Except as expressly provided in this Agreement, no failure or delay on the part of the Administrative Agent or any Lender in exercising any right, power or remedy hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any such right, power or remedy preclude any other or further exercise thereof or the exercise of any other right, power or remedy. The remedies provided for herein are cumulative and are not exclusive of any remedies that may be available to the Administrative Agent or any Lender at law or in equity or otherwise. Neither this Agreement nor any terms hereof may be amended, modified, supplemented, waived, discharged, terminated or otherwise changed unless such amendment, modification, supplement, waiver, discharge, termination or other change is in writing signed by the Borrower and the Required Lenders; provided that no such amendment, modification, supplement, waiver, discharge, termination or other change shall, without the consent of each Lender (other than a Defaulting Lender) with Obligations directly affected thereby in the case of the following clause (i), (i) extend the final scheduled maturity of any Loan or Note (including, without limitation, by amending or modifying the proviso contained in the definition of Revolver Termination Date or Term B Loan Maturity Date), or extend the stated maturity of any Letter of Credit beyond the Revolver Termination Date, or reduce the rate or extend the time of payment of interest (except for waivers of Default Rate interest) or fees thereon, or reduce or forgive the principal amount thereof, (ii) release all or substantially all of the Collateral (except as expressly provided in the Security Documents) or release any Guarantor (other than (x) a Guarantor that is not a Material Subsidiary or (y) in connection with a transaction permitted by Section 8.3), (iii) amend, modify or waive any provision of this Section 12.1, (iv) reduce the percentage specified in the definition of Required Lenders (it being understood that, with the consent of the Required Lenders (or the Lenders providing Additional Term Loans in the case of an amendment pursuant to Section 2.1(a)(ii)), the definition of "Required Lenders" shall include lenders with respect to additional revolving loans or term loans pursuant to this Agreement so long as such additional revolving loans or term loans are on substantially the same basis as the Revolving Loans or Term Loans, as the case may be, are included on the date hereof) or (v) consent to the assignment or transfer by the Borrower of any of its rights and obligations under this Agreement; provided, further, that no such amendment, modification, supplement, waiver, discharge, termination or other change shall (1) increase the Commitments of any Lender over the amount thereof then in effect without the consent of such Lender (it being understood that waivers or modifications of conditions precedent, covenants, Events of Default or Unmatured Events of Default shall not constitute an increase of the Commitment of any Lender, and that an increase in the available portion of any Commitment of any Lender shall not constitute an increase in the Commitment of such Lender), (2) without the consent of the applicable Facing Agent, amend, modify or waive any provision of Section 2.9 or alter such Facing Agent's rights or obligations with respect to Letters of Credit that it has issued or may be required to issue, (3) without the consent of the Administrative Agent, amend, modify or waive any provision of Article XI as same applies to the Administrative Agent or any other provisions as same relates to the rights or obligations of the Administrative Agent, (4) without the consent of the Administrative Agent, amend, modify or waive any provisions relating to the rights or obligations of the Administrative

150

Agent under the other Loan Documents, (5) without the consent of the Majority Lenders of each Facility, amend the definition of Majority Lenders, (6) without the consent of the Majority Lenders of the applicable Facility, amend the Scheduled Term Repayments for such Facility, or (7) without the consent of the Majority Lenders of each Facility which is being allocated a lesser prepayment, repayment or commitment reduction, alter the required application of any prepayments or repayments (or commitment reduction), as between the various Facilities pursuant to Section 4.5(a) (although the Required Lenders may waive in whole or in part, any such prepayment, repayment (other than Scheduled Term Repayments) or commitment reduction so long as the application, as amongst the various Facilities, of any such prepayment, repayment or commitment reduction which is still required to be made is not altered).

Agreement as contemplated by clauses (i) through (v), inclusive, of the first proviso to the third sentence of Section 12.1(a), the consent of the Required Lenders is obtained but the consent of one or more of such other Lenders whose consent is required is not obtained, then the Borrower shall have the right, so long as all non-consenting Lenders whose individual consent is required are treated as described in either clause (A) or (B) below, to either (A) replace each such non-consenting Lender or Lenders (or, at the option of the Borrower if the respective Lender's consent is required with respect to less than all Loans, to replace only the respective Loans of the respective nonconsenting Lender which gave rise to the need to obtain such Lender's individual consent) with one or more Replacement Lenders pursuant to Section 3.7 so long as at the time of such replacement, each such Replacement Lender consents to the proposed amendment, modification, supplement. waiver, discharge, termination or other change or (B) terminate such non-consenting Lender's Revolving Commitment and repay all outstanding Loans of such Lender which gave rise to the need to obtain such Lender's consent, in accordance with Section 4.1(b) and 4.3; provided that, unless the Revolving Commitment terminated and Loans repaid pursuant to the preceding clause (B) are immediately replaced in full at such time through the addition of new Lenders or the increase of the Commitments and/or outstanding Loans of existing Lenders (who in each case must specifically consent thereto), then in the case of any action pursuant to preceding clause (B), the specific consent of the Required Lenders (determined before giving effect to the proposed action) thereto shall be required; provided, further, that in any event the Borrower shall not have the right to replace a Lender, terminate its Revolving Commitment or repay its Loans solely as a result of the exercise of such Lender's rights (and the withholding of any required consent by such Lender) contemplated by the second proviso to the third sentence of Section 12.1(a).

C. In addition, notwithstanding the foregoing, this Agreement may be amended with the written consent of Administrative Agent, Borrower and the Lenders providing the relevant Replacement Term Loans (as defined below) to permit the refinancing or modification of all outstanding Term Loans of any Facility ("Refinanced Term Loans") with one or more replacement or modified Term Facilities hereunder ("Replacement Term Loans"), provided that (a) any Lender that does not consent to the amendment and that holds Refinanced Term Loans receives payment in full of the principal amount of and interest accrued on each Refinanced Term Loan made by it or is replaced as provided in Section 3.7, (b) the aggregate principal amount of such Replacement Term Loans shall not exceed the aggregate principal amount of such Refinanced Term Loans of any Lender that does not provide Replacement Term Loans as having been paid in full

151

immediately prior to the amendment) shall approve such increase, (c) the Applicable Margin for Eurocurrency Loans and the Applicable Margin for Base Rate Loans for the Replacement Term Loans shall not be higher than such applicable margins for the relevant Term Facility of Refinanced Term Loans unless the Required Lenders (treating the Refinanced Loans of any Lender that does not provide Replacement Term Loans as having been paid in full immediately prior to the amendment) shall approve such increase, (d) the Weighted Average Life to Maturity of such Replacement Term Loans shall not be shorter than the Weighted Average Life to Maturity of the Refinanced Term Loans at the time of such amendment and (e) all other terms applicable to such Replacement Term Loans shall be substantially identical to, or less favorable to the Lenders providing such Replacement Term Loans than those applicable to such Refinanced Term Loans except to the extent necessary to provide for covenants and other terms applicable to any period after the latest final maturity of Term Loans in effect immediately prior to such amendment unless the Required Lenders (treating the Refinanced Loans of any Lender that does not provide Replacement Term Loans as having been paid in full immediately prior to the amendment) shall approve such terms.

D. Notwithstanding the foregoing, upon the execution and delivery of all documentation required by Administrative Agent to be delivered pursuant to Section 2.1(a)(ii) in connection with an Additional Term Loan, this Agreement shall be deemed amended without further action by any Lender to reflect, as applicable, the new Lenders and the terms of such Additional Term Loan.

20.2. Further Assurances

The Borrower agrees to do such further acts and things and to execute and deliver to the Administrative Agent such additional assignments, agreements, powers and instruments, as the Administrative Agent may reasonably require or deem advisable to carry into effect the purposes of this Agreement or any of the Loan Documents or to better assure and confirm unto the Administrative Agent its rights, powers and remedies hereunder.

20.3. Notices, Etc

A. Except where oral or telephonic instructions or notices are authorized herein to be given, all notices, demands, instructions and other communications required or permitted to be given to or made upon any party hereto or any other Person shall be in writing and shall be personally delivered or sent by registered or certified mail, postage prepaid, return receipt requested, or by a reputable overnight or courier delivery service, or by telecopier, and shall be deemed to be given for purposes of this Agreement on the third day after deposit in registered or certified mail, postage prepaid, and otherwise on the date that such writing is delivered or sent to the intended recipient thereof, or in the case of notice delivered by telecopy, upon completion of transmission with a copy of such notice also being delivered under any of the other methods provided above, all in accordance with the provisions of this Section 12.3. Unless otherwise specified in a notice sent or delivered in accordance with the foregoing provisions of this Section 12.3, notices, demands, instructions and other communications in writing shall be given to or made upon the respective parties hereto at their respective addresses (or to their respective telecopier numbers) indicated (i) in the case of any Lender, in such Lender's latest administrative questionnaire submitted to the Administrative Agent, (ii) in the case of any

Assignee, in the applicable Assignment and Assumption Agreement or (iii) in the case of any other party hereto, on <u>Schedule 12.3</u>, and, in the case of telephonic instructions or notices, by calling the telephone number or numbers indicated for such party on such administrative questionnaire, such Assignment and Assumption Agreement or <u>Schedule 12.3</u>, as the case may be.

- B. Notices and other communications to or by the Administrative Agent, the Collateral Agent, the UK Security Trustee and the Lenders hereunder may be delivered or furnished by electronic communication (including e-mail and Internet or intranet websites (the "Platform") pursuant to procedures approved by the Administrative Agent, provided that the foregoing shall not apply to notices pursuant to Article II, Section 4.1, Section 4.3 and Section 7.3 unless otherwise agreed by the Administrative Agent (or, as the case may be, the Collateral Agent or UK Security Trustee) and the applicable Lender. The Administrative Agent, the Collateral Agent, the UK Security Trustee or Borrower may, in its discretion, agree to accept notices and other communications to it hereunder by electronic communications pursuant to procedures approved by it, provided that approval of such procedures may be limited to particular notices or communications.
- C. Unless the Administrative Agent otherwise prescribes, (i) notices and other communications sent to an e-mail address shall be deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail or other written acknowledgement), provided that if such notice or other communication is sent after 5:00 p.m. (New York City time), such notice or communication shall be deemed to have been sent at the opening of business on the next Business Day for the recipient, and (ii) notices or communications posted to an Internet or intranet website shall be deemed received upon the deemed receipt by the intended recipient at its e-mail address as described in the foregoing clause (i) of notification that such notice or communication is available and identifying the website address therefor.
- D. THE PLATFORM IS PROVIDED "AS IS" AND "AS AVAILABLE". THE AGENT PARTIES (AS DEFINED BELOW) DO NOT WARRANT THE ACCURACY OR COMPLETENESS OF THE COMMUNICATIONS, OR THE ADEQUACY OF THE PLATFORM AND EXPRESSLY DISCLAIM LIABILITY FOR ERRORS OR OMISSIONS IN THE COMMUNICATIONS. NO WARRANTY OF ANY KIND, EXPRESS, IMPLIED OR STATUTORY, INCLUDING, WITHOUT LIMITATION, ANY WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NON-INFRINGEMENT OF THIRD PARTY RIGHTS OR FREEDOM FROM VIRUSES OR OTHER CODE DEFECTS, IS MADE BY THE AGENT PARTIES IN CONNECTION WITH THE COMMUNICATIONS OR THE PLATFORM. IN NO EVENT SHALL THE ADMINISTRATIVE AGENT OR ANY OF ITS AFFILIATES OR ANY OF THEIR RESPECTIVE OFFICERS, DIRECTORS, EMPLOYEES, AGENTS, ADVISORS OR REPRESENTATIVES (COLLECTIVELY, "AGENT PARTIES") HAVE ANY LIABILITY TO THE BORROWER, ANY LENDER OR ANY OTHER PERSON OR ENTITY FOR DAMAGES OF ANY KIND, INCLUDING, WITHOUT LIMITATION, DIRECT OR INDIRECT, SPECIAL, INCIDENTAL OR CONSEQUENTIAL DAMAGES, LOSSES OR EXPENSES (WHETHER IN TORT, CONTRACT OR OTHERWISE) ARISING OUT OF THE BORROWER'S OR THE ADMINISTRATIVE AGENT'S TRANSMISSION OF

153

COMMUNICATIONS THROUGH THE INTERNET, EXCEPT TO THE EXTENT THE LIABILITY OF ANY AGENT PARTY IS FOUND IN A FINAL NON-APPEALABLE JUDGMENT BY A COURT OF COMPETENT JURISDICTION TO HAVE RESULTED PRIMARILY FROM SUCH AGENT PARTY'S GROSS NEGLIGENCE OR WILLFUL MISCONDUCT.

20.4. Costs, Expenses and Taxes

Generally. The Borrower agrees (without duplication) to pay all reasonable costs and expenses of the A. Administrative Agent in connection with the negotiation, preparation, printing, typing, reproduction, execution and delivery of this Agreement and the other Loan Documents and the documents and instruments referred to herein and therein and any amendment, waiver, consent relating hereto or thereto or other modifications of (or supplements to) any of the foregoing and any and all other documents and instruments furnished pursuant hereto or thereto or in connection herewith or therewith, including without limitation, the reasonable fees and out-of-pocket expenses of Winston & Strawn LLP, special counsel to the Administrative Agent and the UK Security Trustee, and any local counsel retained by the Administrative Agent relative thereto or the reasonable allocated costs of staff counsel as well as the fees and out-of-pocket expenses of counsel, independent public accountants and other outside experts retained by the Administrative Agent or the UK Security Trustee in connection with the administration of this Agreement and the other Loan Documents, and all search fees, appraisal fees and expenses, title insurance policy fees, costs and expenses and filing and recording fees and all costs and expenses (including, without limitation, Attorney Costs), if any, of the Administrative Agent, the UK Security Trustee and the Lenders in connection with the enforcement of this Agreement, any of the Loan Documents or any other agreement furnished pursuant hereto or thereto or in connection herewith or therewith. In addition, the Borrower shall pay any and all present and future stamp, documentary transfer, excise, property, and other similar taxes payable or determined to be payable in connection with the execution, delivery or enforcement of this Agreement, any Loan Document, or otherwise with respect to or in connection with any Loan Document, or the making of any Loan (other than taxes based on the net income of the Lenders), and agrees to save and hold the Administrative Agent, the UK Security Trustee and each Lender harmless from and against any and all liabilities with respect to or resulting from any delay by the Borrower in paying, or omission by the Borrower to pay, such taxes. Any portion of the foregoing fees, costs and expenses which remains unpaid more than thirty (30) days following the Administrative Agent's, the UK Security Trustee's, any Agents' or any Lender's statement and request for payment thereof shall bear interest from the date that the Borrower receives such statement and request to the date of payment, for the first 10 days at the Base Rate, and thereafter at the Default Rate. Subject to Section 4.7, the Borrower will indemnify and hold harmless the Administrative Agent, the UK Security Trustee, each Agent and each Lender and each director, officer, employee, partner, advisor, agent, attorney, trustee and Affiliate of the Administrative Agent, the UK Security Trustee, each Agent and each Lender (each such Person an "Indemnified Party") from and against all losses, claims, damages, penalties, obligations (including removal or remedial actions), expenses or liabilities which arise out of, in any way relate to, or result from the transactions contemplated by this Agreement or any of the other Loan Documents and to reimburse each Indemnified Party upon their demand, for any Attorney Costs incurred in connection with investigating, preparing to defend or defending any such loss, claim, damage, liability, action or claim; provided, however, (a) that no Indemnified Party shall have the right to be so indemnified hereunder for any loss, claim, damage, penalties, obligations, expense or liability to the extent it arises or results from the gross negligence or willful misconduct or bad faith of such Indemnified Party as finally determined by a court of competent jurisdiction and (b) that nothing contained herein shall affect the obligations and liabilities of the Lenders to the Borrower contained herein. If any action, suit or proceeding arising from any of the foregoing is brought against the Administrative Agent, the UK Security Trustee, any Agent, any Lender or any other Indemnified Party, the Borrower will, if requested by the Administrative Agent, the UK Security Trustee, any Agent, any Lender or any such Indemnified Party, resist and defend such action, suit or proceeding or cause the same to be resisted and defended by counsel reasonably satisfactory to the Person or Persons indemnified or intended to be indemnified. Each Indemnified Party shall, unless the Administrative Agent, the UK Security Trustee, an Agent, a Lender or other Indemnified Party has made the request described in the preceding sentence and such request has been complied with, have the right to employ its own counsel (or (but not as well as) staff counsel) to investigate and control the defense of any matter covered by such indemnity and the reasonable fees and expenses of such counsel shall be at the expense of the indemnifying party. Excluding any liability to the extent arising out of the gross negligence, willful misconduct or bad faith of any Indemnified Party as finally determined by a court of competent jurisdiction, the Borrower further agrees to indemnify and hold each Indemnified Party harmless from all loss, cost (including Attorney Costs), liability and damage whatsoever incurred by any Indemnified Party by reason of any violation of any Environmental Laws or Environmental Permits or for the Release or threatened Release of any Contaminants into the environment for which the Borrower or any of its Subsidiaries has any liability or which occurs upon the Mortgaged Property or which is related to any property currently or formerly owned, leased or operated by or on behalf of the Borrower or any of its Subsidiaries, or by reason of the imposition of any Environmental Lien in respect of the Borrower or its Subsidiaries or which occurs by a breach of any of the representations, warranties or covenants relating to environmental matters contained herein, provided that, with respect to any liabilities arising from acts or failure to act for which the Borrower or any of its Subsidiaries is strictly liable under any Environmental Law or Environmental Permit, the Borrower's obligation to each Indemnified Party under this indemnity shall likewise be without regard to fault on the part of the Borrower or any such Subsidiary. If the Borrower shall fail to do any act or thing which it has covenanted to do hereunder or any representation or warranty on the part of the Borrower or any Subsidiary contained herein or in any other Loan Document shall be breached, the Administrative Agent may (but shall not be obligated to) do the same or cause it to be done or remedy any such breach, and may expend its funds for such purpose, and will use its best efforts to give prompt written notice to the Borrower that it proposes to take such action. Any and all amounts so expended by the Administrative Agent shall be repaid to it by the Borrower promptly upon the Administrative Agent's demand therefor, with interest at the Default Rate in effect from time to time during the period including the date so expended by the Administrative Agent to the date of repayment. To the extent that the undertaking to indemnify, pay or hold harmless the Administrative Agent, the UK Security Trustee or any Lender as set forth in this Section 12.4 may be unenforceable because it is violative of any law or public policy, the Borrower shall make the maximum contribution to the payment and satisfaction of each of the indemnified liabilities which is permissible under applicable law. The obligations of the Borrower under this Section 12.4 shall survive the termination of this Agreement, the assignment by any Lender of all or any part of its Credit Exposure hereunder and the discharge

155

of the Borrower's other Obligations hereunder. Except as specifically provided for in this Agreement, no party hereto shall be entitled to recover from any other party hereto any amount in respect of exemplary, punitive, special, indirect, remote, or speculative damages, including lost profits.

- B. Foreign Exchange Indemnity. If any sum due from the Borrower under this Agreement or any order or judgment given or made in relation hereto has to be converted from the currency (the "first currency") in which the same is payable hereunder or under such order or judgment into another currency (the "second currency") for the purpose of (i) making or filing a claim or proof against the Borrower with any Governmental Authority or in any court or tribunal, or (ii) enforcing any order or judgment given or made in relation hereto, the Borrower shall indemnify and hold harmless each of the Persons to whom such sum is due from and against any loss actually suffered as a result of any discrepancy between (a) the rate of exchange used to convert the amount in question from the first currency into the second currency, and (b) the rate or rates of exchange at which such Person, acting in good faith in a commercially reasonable manner, purchased the first currency with the second currency after receipt of a sum paid to it in the second currency in satisfaction, in whole or in part, of any such order, judgment, claim or proof. The foregoing indemnity shall constitute a separate obligation of the Borrower distinct from its other obligations hereunder and shall survive the giving or making of any judgment or order in relation to all or any of such other obligations. Notwithstanding the foregoing, payments of principal and interest on Loans denominated in Euros, Sterling or an Alternative Currency, as the case may be, shall be made in Euros, Sterling or such Alternative Currency, as the case may be.
- C. Notwithstanding the resignation of the Resigning Agent as of the Fifth Amendment Effective Date, the Borrower agrees that the provisions of this Section 12.4 and any other provisions of this Agreement or any other Loan Document regarding payment of costs and expenses and indemnification of the Administrative Agent or the Collateral Agent, together with any provision of any Loan Document that shall accrue to the benefit of any retiring or resigning Agent, shall continue to inure to the benefit of the Resigning Agent on and following the Fifth Amendment Effective Date with respect to actions of the Resigning Agent taken (i) on or prior to the Fifth Amendment Effective Date, and (ii) subsequent to the Fifth Amendment Effective Date pursuant to the Loan Documents or that certain Successor Agency Agreement, dated as of the Fifth Amendment Effective Date, by and among JPMorgan Chase Bank, N.A., the Resigning Agent and the Borrower. DB shall be a third party beneficiary for purposes of this Section 12.4(c).

20.6. Adjustment; Setoff

- A. If any lender (a "Benefited Lender") shall at any time receive any payment of all or part of its Loans, or interest thereon, or receive any collateral in respect thereof (whether voluntarily or involuntarily, by setoff, pursuant to events or proceedings of the nature referred to in Section 10.1(e) or Section 10.1(f) hereof, or otherwise) in a greater proportion than any such payment to and collateral received by any other Lender in respect of such other Lender's Loans or interest thereon not expressly provided hereby, such Benefited Lender shall purchase for cash from the other Lenders such portion of each such other Lender's Loans, or shall provide such other Lenders with the benefits of any such collateral, or the proceeds thereof, as shall be necessary to cause such Benefited Lender to share the excess payment or benefits of such collateral or proceeds ratably with each Lender except to the extent expressly provided hereby; provided, however, that if all or any portion of such excess payment or benefits is thereafter recovered from such Benefited Lender, such purchase shall be rescinded, and the purchase price and benefits returned, to the extent of such recovery, but without interest unless the Benefited Lender from which such excess payment is recovered is required by court order to pay interest thereon, in which case each Lender returning funds to such Benefited Lender shall pay its pro rata share of such interest. The Borrower agrees that each Lender so purchasing a portion of another Lender's Loans may exercise all rights of payment (including, without limitation, rights of setoff) with respect to such portion as fully as if such Lender were the direct holder of such portion.
- B. In addition to any rights and remedies of the Lenders provided by law, each Lender shall have the right, without prior notice to the Borrower, any such notice being expressly waived by the Borrower, upon the occurrence and during the continuance of an Event of Default, to setoff and apply against any Obligations, whether matured or unmatured, of the Borrower to such Lender, any amount owing from such Lender to the Borrower, at or at any time after, the happening of any of the above-mentioned events, and the aforesaid right of setoff may be exercised by such Lender against the Borrower or against any trustee in bankruptcy, debtor in possession, assignee for the benefit of creditors, receivers, or execution, judgment or attachment creditor of the Borrower, or against anyone else claiming through or against, the Borrower or such trustee in bankruptcy, debtor in possession, assignee for the benefit of creditors, receivers, or execution, judgment or attachment creditor, notwithstanding the fact that such right of setoff shall not have been exercised by such Lender prior to the making, filing or issuance, or service upon such Lender of, or of notice of, any such petition, assignment for the benefit of creditors, appointment or application for the appointment of a receiver, or issuance of execution, subpoena, order or warrant. Each Lender agrees promptly to notify the Borrower and the Administrative Agent after any such setoff and application made by such Lender, provided that the failure to give such notice shall not affect the validity of such setoff and application.
- C. The Borrower expressly agrees that to the extent the Borrower makes a payment or payments and such payment or payments, or any part thereof, are subsequently invalidated, declared to be fraudulent or preferential, set aside or are required to be repaid to a trustee, receiver, or any other party under any bankruptcy act, state or federal law, common law or equitable cause, then to the extent of such payment or repayment, the Indebtedness to the Lenders or part thereof intended to be satisfied shall be revived and continued in full force and effect as if said payment or payments had not been made.

157

20.7. Execution in Counterparts

- A. This Agreement may be executed in any number of counterparts and by different parties hereto on separate counterparts, each of which counterparts, when so executed and delivered, shall be deemed to be an original and all of which counterparts, taken together, shall constitute but one and the same Agreement. Delivery of an executed counterpart of a signature page of this Agreement by telecopy shall be effective as delivery of a manually executed counterpart of this Agreement.
- B. <u>Electronic Execution of Assignments</u>. The words "execution," "signed," "signature," and words of like import in any Assignment and Assumption shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act.

20.8. Binding Effect; Assignment; Addition and Substitution of Lenders

- A. This Agreement shall be binding upon, and inure to the benefit of, the Borrower, the Administrative Agent, the Resigning Agent, the Lenders, all future holders of the Notes and their respective successors and assigns; provided, however, that the Borrower may not assign its rights or obligations hereunder or in connection herewith or any interest herein (voluntarily, by operation of law or otherwise) without the prior written consent of the Administrative Agent and all of the Lenders.
- B. Each Lender may at any time sell to one or more banks or other entities ("Participants") participating interests in all or any portion of its Commitment and Loans or participation in Letters of Credit or any other interest of such Lender hereunder (in respect of any Lender, its "Credit Exposure"). In the event of any such sale by a Lender of participating interests to a Participant, such Lender's obligations under this Agreement shall remain unchanged, such Lender shall remain solely responsible for the performance thereof, and the Borrower and the Administrative Agent shall continue to deal solely and directly with such Lender in connection with

such Lender's rights and obligations under this Agreement. At the time of the sale of a participating interest, the Lender transferring the interest (i) shall cause the Participant to provide the forms required under Section 4.7(d) as if such Participant became a Lender on the date of the sale and (ii) shall, if required under applicable law, deliver revised forms in accordance Section 4.7(d) reflecting the portion of the interest sold and the portion of the interest retained. Further, the Participant shall be subject to the obligations of Section 3.6 and Section 4.7 as if such Participant was a Lender. The Borrower agrees that if amounts outstanding under this Agreement or any of the Loan Documents are due or unpaid, or shall have been declared or shall have become due and payable upon the occurrence and during the continuance of an Event of Default, each Participant shall be deemed to have the right of setoff in respect of its participating interest in amounts owing under this Agreement and the Loan Documents to the same extent as if the amount of its participating interest were owing directly to it as a Lender under this Agreement or any other Loan Document; provided, however, that such

158

right of setoff shall be subject to the obligation of such Participant to share with the Lenders, and the Lenders agree to share with such Participant, as provided in Section 12.6. The Borrower also agrees that each Participant shall be entitled to the benefits of Section 3.6 and Section 4.7 with respect to its participation in the Loans outstanding from time to time, as if such Participant becomes a Lender on the date it acquired an interest pursuant to this Section 12.8(b); provided that, no participation shall be made to any Person under this section if, at the time of such participation, the Participant's benefits under Section 3.6 or Section 4.7 would be greater than the benefits that the participating Lender was entitled to under Section 3.6 or Section 4.7 (and if any participation is made in violation of the foregoing, the Participant will not be entitled to the incremental amounts). Each Lender agrees that any agreement between such Lender and any such Participant in respect of such participating interest shall not restrict such Lender's right to approve or agree to any amendment, restatement, supplement or other modification to, waiver of, or consent under, this Agreement or any of the Loan Documents except to the extent that any of the foregoing would (i) extend the final scheduled maturity of any Loan or Note in which such Participant is participating (it being understood that amending the definition of any Scheduled Term B Dollar Repayments (other than the Term B Loan Maturity Date), shall not constitute an extension of the final scheduled maturity of any Loan or Note) or extend the stated maturity of any Letter of Credit in which such Participant is participating beyond the Revolver Termination Date, or reduce the rate or extend the time of payment of interest or fees on any such Loan, Note or Letter of Credit (except in connection with a waiver of applicability of any postdefault increase in interest rates) or reduce the principal amount thereof, or increase the amount of the Participant's participation over the amount thereof then in effect (it being understood that waivers or modifications of conditions precedent, covenants, representations, warranties, Events of Default or Unmatured Events of Default or of a mandatory reduction in Commitments shall not constitute a change in the terms of such participation, and that an increase in any Commitment or Loan shall be permitted without the consent of any Participant if the Participant's participation is not increased as a result thereof), (ii) consent to the assignment or transfer by the Borrower of any of its rights and obligations under this Agreement or (iii) release all or substantially all of the Collateral under all of the Security Documents (except as expressly provided in the Loan Documents) supporting the Loans and/or Letters of Credit hereunder in which such Participant is participating.

C. Any Lender may at any time assign to one or more Eligible Assignees, including an Affiliate thereof (each an "Assignee"), all or any part of its Credit Exposure pursuant to an Assignment and Assumption Agreement, provided that any assignment of all or any portion of any Lender's Credit Exposure to an Assignee other than an Affiliate of such Lender or another Lender, or in the case of a Lender that is a Fund, any Related Fund of any Lender (i) shall be an assignment of its Credit Exposure in an amount not less than the Dollar Equivalent of \$1,000,000 (treating any Fund and its Related Funds as a single Eligible Assignee) (or if less the entire amount of Lender's Credit Exposure with respect to such Facility, provided, that, if such Lender and its Affiliates (or in the case of a Fund and its Related Funds) collectively hold Credit Exposure at least equal to such minimum amounts, any one or more of such Affiliates and/or Related Funds must simultaneously assign Credit Exposure such that the aggregate Credit Exposure assigned satisfies such minimum amount) and (ii) shall require the prior written consent of the Administrative Agent (not to be unreasonably withheld) and, other than with respect to assignments by the Joint Lead Arrangers, and their Affiliates during the primary syndication of this Agreement, provided no Event of Default then exists and is

159

continuing, the Borrower (the consent of the Borrower not to be unreasonably withheld or delayed), and; provided, further, that notwithstanding the foregoing limitations, any Lender may at any time assign all or any part of its Credit Exposure to any Affiliate of such Lender or to any other Lender (or in the case of a Lender which is a Fund, to any Related Fund of such Lender), except that no Revolving Lender may assign all or any part of its Credit Exposure relating to the Revolving Facility to any Person other than another Revolving Lender without the prior written consent (in each case not to be unreasonably withheld or delayed) of the Administrative Agent, each Facing Agent, the Swing Line Lender and, unless an Event of Default has occurred and is continuing, the Borrower. Upon execution of an Assignment and Assumption Agreement and the payment of a nonrefundable assignment fee of \$3,500 (provided that no such fee shall be payable upon assignments by any Lender which is a Fund to one or more Related Funds and only one such fee shall be payable for contemporaneous assignments by a Lender to Related Funds) in immediately available funds to the Administrative Agent at its Payment Office in connection with each such assignment, written notice thereof by such transferor Lender to the Administrative Agent and the recording by the Administrative Agent of such assignment and the resulting effect upon the Loans, Revolving Commitment of the assigning Lender and the Assignee, the Assignee shall have, to the extent of such assignment, the same rights, benefits and obligations as it would have if it were a Lender hereunder and the holder of the Obligations (provided that the Borrower and the Administrative Agent shall be entitled to continue to deal solely and directly with the assignor Lender in connection with the interests so assigned to the Assignee until written notice of such assignment, together with payment instructions, addresses and related information with respect to the Assignee, shall have been given to the Borrower and the Administrative Agent by the assignor Lender and the Assignee) and, if the Assignee has expressly assumed, for the benefit of the Borrower, some or all of the transferor Lender's obligations hereunder, such transferor Lender shall be relieved of its obligations hereunder to the extent of such assignment and assumption, and except as described above, no further consent or action by the Borrower, the Lenders, or the Administrative Agent shall be required. At the time of each

assignment pursuant to this Section 12.8(c) to a Person which is not already a Lender hereunder, the respective Assignee shall provide to the Borrower and the Administrative Agent the appropriate forms and certificates as provided in Section 4.7(d), if applicable. Each Assignee shall take such Credit Exposure subject to the provisions of this Agreement and to any request made, waiver or consent given or other action taken hereunder, prior to the receipt by the Administrative Agent and the Borrower of written notice of such transfer, by each previous holder of such Credit Exposure. Such Assignment and Assumption Agreement shall be deemed to amend this Agreement and Schedule 1.1(a) hereto, to the extent, and only to the extent, necessary to reflect the addition of such Assignee as a Lender and the resulting adjustment of all or a portion of the rights and obligations of such transferor Lender under this Agreement, the Maximum Commitment, the determination of its Pro Rata Share (rounded to twelve decimal places), the Loans, any outstanding Letters of Credit and any new Notes, if requested, to be issued, at the Borrower's expense, to such Assignee, and no further consent or action by the Borrower or the Lenders shall be required to effect such amendments.

D. The Borrower authorizes each Lender to disclose to any Participant or Assignee (each, a "<u>Transferee</u>") and any prospective Transferee any and all financial information in such Lender's possession concerning the Borrower and any Subsidiary of the Borrower which has been delivered to such Lender by the Borrower pursuant to this Agreement or which has been delivered to such Lender by the Borrower in connection with such Lender's credit

160

evaluation of the Borrower prior to entering into this Agreement, <u>provided</u> that, such Transferee or prospective Transferee agrees to treat any such information which is not public as confidential in accordance with the terms of <u>Section 12.14</u> hereof.

- E. Notwithstanding any other provision set forth in this Agreement, any Lender may at any time pledge or assign all or any portion of its rights under this Agreement and the other Loan Documents (including, without limitation, the Notes held by it) to any Federal Reserve Bank in accordance with Regulation A of the Federal Reserve Board without notice to, or the consent of, the Borrower, provided that, no such pledge or assignment of a security interest under this Section 12.8(e) shall release a Lender from any obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto. Any Lender which is a fund may pledge all or any portion of its Notes or Loans to its trustee or its security holders in support of its obligations to its trustee. No such pledge or assignment shall release the transferor Lender from its obligations hereunder.
- F. In the event that the holder of any Note (including any Lender) shall transfer such Note, it shall immediately advise Administrative Agent and Borrower of such transfer, and Administrative Agent and Borrower shall be entitled conclusively to assume that no transfer of any Note has been made by any holder (including any Lender) unless and until Administrative Agent and Borrower shall have received written notice to the contrary. Except as otherwise provided in this Agreement or as otherwise expressly agreed in writing by all of the other parties hereto, no Lender shall, by reason of the transfer of a Note or otherwise, be relieved of any of its obligations hereunder. Each transferee of any Note shall take such Note subject to the provisions of this Agreement and to any request made, waiver or consent given or other action taken hereunder, prior to the receipt by Administrative Agent and Borrower of written notice of such transfer, by each previous holder of such Note, and, except as expressly otherwise provided in such transfer, Administrative Agent and Borrower shall be entitled conclusively to assume that the transferee named in such notice shall hereafter be vested with all rights and powers under this Agreement with respect to the Pro Rata Share of the Lender named as the payee of the Note which is the subject of such transfer.

20.9. CONSENT TO JURISDICTION; MUTUAL WAIVER OF JURY TRIAL

(A) EACH PARTY HERETO HEREBY IRREVOCABLY SUBMITS TO THE NONEXCLUSIVE JURISDICTION OF ANY UNITED STATES FEDERAL OR NEW YORK STATE COURT SITTING IN THE CITY OF NEW YORK IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT, AND HEREBY IRREVOCABLY AGREES THAT ALL CLAIMS IN RESPECT TO SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN ANY SUCH UNITED STATES FEDERAL OR NEW YORK STATE COURT AND THE BORROWER, THE ADMINISTRATIVE AGENT, THE UK SECURITY TRUSTEE AND EACH LENDER IRREVOCABLY WAIVE ANY OBJECTION, INCLUDING, WITHOUT LIMITATION, ANY OBJECTION TO THE LAYING OF VENUE OR BASED ON THE GROUNDS OF FORUM NON CONVENIENS WHICH ANY OF THEM MAY NOW OR HEREAFTER HAVE TO THE BRINGING OF ANY SUCH ACTION OR PROCEEDING IN SUCH RESPECTIVE JURISDICTIONS.

161

- (B) AS A METHOD OF SERVICE, THE BORROWER, THE ADMINISTRATIVE AGENT, THE UK SECURITY TRUSTEE AND EACH LENDER IRREVOCABLY CONSENT TO THE SERVICE OF ANY AND ALL PROCESS IN ANY SUCH ACTION OR PROCEEDING, BROUGHT IN ANY SUCH UNITED STATES FEDERAL OR NEW YORK STATE COURT BY THE DELIVERY OF COPIES OF SUCH PROCESS TO THE BORROWER, THE ADMINISTRATIVE AGENT, THE UK SECURITY TRUSTEE OR EACH RESPECTIVE LENDER, AS THE CASE MAY BE, AT THE ADDRESSES SPECIFIED ON THEIR RESPECTIVE SIGNATURE PAGES TO THIS AGREEMENT OR BY CERTIFIED MAIL DIRECT TO SUCH RESPECTIVE ADDRESSES.
- (C) EACH PARTY HERETO HEREBY EXPRESSLY WAIVES ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION OR PROCEEDING TO ENFORCE OR DEFEND ANY RIGHT, POWER OR REMEDY UNDER OR IN CONNECTION WITH THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT. THE TERMS AND THE PROVISIONS OF THIS SECTION CONSTITUTE A MATERIAL INDUCEMENT TO LENDERS ENTERING INTO THIS AGREEMENT.

20.10. GOVERNING LAW

THIS AGREEMENT AND EACH NOTE SHALL BE DEEMED TO BE A CONTRACT MADE UNDER THE LAWS OF THE STATE OF NEW YORK, AND FOR ALL PURPOSES SHALL BE CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS AND DECISIONS OF SAID STATE.

20.11. Severability of Provisions

Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof or affecting the validity or enforceability of such provision in any other jurisdiction.

20.12. Headings

The Table of Contents and Article and Section headings used in this Agreement are for convenience of reference only and shall not affect the construction of this Agreement.

20.13. Termination of Agreement

This Agreement shall terminate when the Commitment of each Lender has terminated and all outstanding Obligations and Loans have been paid in full and all Letters of Credit have expired or been terminated; <u>provided</u>, <u>however</u>, that the rights and remedies of the Administrative Agent and each Lender with respect to any representation and warranty made by the Borrower pursuant to this Agreement or any other Loan Document, and the indemnification provisions contained in this Agreement and any other Loan Document, shall be continuing and shall survive any termination of this Agreement or any other Loan Document.

162

20.14. Confidentiality

Each of the Lenders severally agrees to keep confidential all non-public information pertaining to the Borrower and its Subsidiaries and their respective predecessors in interest which is provided to it by any such parties in accordance with such Lender's customary procedures for handling confidential information of this nature and in a prudent fashion, and shall not disclose such information to any Person except (i) to the extent such information is public when received by such Lender or becomes public thereafter due to the act or omission of any party other than a Lender, (ii) to the extent such information is independently obtained from a source other than the Borrower or its Subsidiaries and such information from such source is not, to such Lender's knowledge, subject to an obligation of confidentiality or, if such information is subject to an obligation of confidentiality, that disclosure of such information is permitted, (iii) to an Affiliate of such Lender (or its investment advisor), counsel, auditors, ratings agencies, examiners of any regulatory authority having or asserting jurisdiction over such Lender, accountants and other consultants retained by the Administrative Agent or any Lender or to any Affiliate of a Lender which is a direct or indirect contractual counterparty in swap agreements with the Borrower or a Subsidiary of the Borrower or such contractual counterparty's professional advisor (so long as such contractual counterparty or professional advisor to such contractual counterparty agrees to be bound by the provisions of this Section 12.14) or to the National Association of Insurance Commissioners or any similar organization or any nationally recognized rating agency that requires access to information about a Lender's investment portfolio in connection with rating issued with respect to such Lender, (iv) in connection with any litigation or the enforcement of the rights of any Lender or the Administrative Agent under this Agreement or any other Loan Document, (v) to the extent (x) required by any applicable statute, rule or regulation or court order (including, without limitation, by way of subpoena) or pursuant to the request of any Governmental Authority having or asserting jurisdiction over any Lender or the Administrative Agent or any of their respective affiliates; provided, however, that in such event, if the Lender(s) are not legally prohibited from doing so, the Lender shall provide the Borrower with prompt notice of such requested disclosure so that the Borrower may seek a protective order or other appropriate remedy, and, in any event, the Lenders will endeavor in good faith to provide only that portion of such information which, in the reasonable judgment of the Lender(s), is relevant and legally required to be provided (y) requested by any nationally recognized rating agency that requires access to information about a Lender's investment portfolio in connection with rating issued with respect to such Lender or (z) requested by any pledgee referred to in Section 12.8(e) or a direct or indirect contractual counterparty in swap agreements with a Lender or a Person that such Lender is a direct or indirect counterparty in a swap agreement or such contractual counterparty's professional advisor (so long as such pledgee or contractual counterparty or professional advisor to such contractual counterparty agrees to be bound by the provisions of this Section 12.14) or (vi) to the extent disclosure to other entities is appropriate in connection with any proposed or actual assignment or grant of a participation by any of the Lenders of interests in this Agreement and/or any of the other Loan Documents to such other entities (who will in turn be required to maintain confidentiality as if they were Lenders parties to this Agreement). In no event shall the Administrative Agent or any Lender be obligated or required to return any such information or other materials furnished by the Borrower.

163

20.15. Concerning the Collateral and the Loan Documents

A. <u>Authority</u>. Each Lender authorizes and directs JPMCB and/or its designated affiliate to act as Collateral Agent under the Collateral Security Agreement and or UK Security Trustee under the UK Security Documents and to enter into the Loan Documents relating to the Collateral (including, without limitation, the Collateral Security Agreement) for the benefit of the Lenders and

the other Secured Parties. Each Lender agrees that any action taken by the Administrative Agent or the Required Lenders (or, where required by the express terms hereof, a different proportion of the Lenders) in accordance with the provisions hereof or of the other Loan Documents, and the exercise by the Administrative Agent, the Collateral Agent, the UK Security Trustee or the Required Lenders (or, where so required, such different proportion) of the powers set forth herein or therein, together with such other powers as are reasonably incidental thereto, shall be authorized and binding upon all of the Lenders. Without limiting the generality of the foregoing, the Administrative Agent or the Collateral Agent, as the case may be, shall have the sole and exclusive right and authority to (i) act as the disbursing and collecting agent for the Lenders with respect to all payments and collections arising in connection herewith and with the Loan Documents relating to the Collateral; (ii) execute and deliver each Loan Document relating to the Collateral and accept delivery of each such agreement delivered by the Borrower or any of its Subsidiaries, (iii) act as Collateral Agent for the Lenders and certain other Secured Parties for purposes stated in the Security Documents to the extent such perfection is required under the Loan Documents, provided, however, the Collateral Agent hereby appoints, authorizes and directs each Lender to act as collateral sub-agent for the Collateral Agent and the Lenders for purposes of the perfection of all security interests and Liens with respect to the Borrower's and its Subsidiaries' respective deposit accounts maintained with, and cash and Cash Equivalents held by, such Lender; (iv) manage, supervise and otherwise deal with the Collateral; (v) take such action as is necessary or desirable to maintain the perfection and priority of the security interests and liens created or purported to be created by the Loan Documents, and (vi) except as may be otherwise specifically restricted by the terms hereof or of any other Loan Document, exercise all remedies given to the Administrative Agent or the Lenders with respect to the Collateral under the Loan Documents relating thereto, applicable law or otherwise.

B. Release of Collateral.

- (a) The Administrative Agent and the Lenders hereby direct the Administrative Agent, the Collateral Agent or the UK Security Trustee, as the case may be, to release, in accordance with the terms hereof, any Lien held by the Administrative Agent, the Collateral Agent or the UK Security Trustee, as the case may be, for the benefit of the Secured Parties (and in the case of a sale of all of the assets or Capital Stock of a Subsidiary under clause (B) below, to release the affected Subsidiary from its guaranty):
 - (i) against all of the Collateral, upon final and indefeasible payment in full of the Loans and Obligations and termination hereof;
 - (ii) against any part of the Collateral sold, transferred or disposed of by the Borrower or any of its Subsidiaries to the extent such sale, transfer or

164

disposition is permitted hereby (or permitted pursuant to a waiver or consent of a transaction otherwise prohibited hereby);

- (iii) against any Collateral acquired by the Borrower or any of its Subsidiaries after the Closing Date and at least 70% of the purchase price therefor is within 120 days of the acquisition thereof financed with Indebtedness secured by a Lien permitted by Section 8.1(c);
- (iv) so long as no Default or Event of Default has occurred and is continuing, in the sole discretion of the Administrative Agent upon the request of the Borrower, against any part of the Collateral with a fair market value of less than \$10,000,000 in the aggregate during the term of this Agreement as such fair market value may be certified to the Administrative Agent, the Collateral Agent and the UK Security Trustee by the Borrower in an officer's certificate acceptable in form and substance to the Administrative Agent, the Collateral Agent and the UK Security Trustee;
- (v) against a part of the Collateral which release does not require the consent of all of the Lenders as set forth in Section 12.1(a)(ii), if such release is consented to by the Required Lenders;
- (vi) against the Collateral consisting of Receivables Facility Assets upon the entry by the Borrower and/or its Subsidiaries into a Permitted Account Receivable Securitization; provided, however, that (y) neither the Administrative Agent nor the Collateral Agent nor the UK Security Trustee shall be required to execute any such document on terms which, in its opinion, would expose it to liability or create any obligation or entail any consequence other than the release of such Liens without recourse or warranty, and (z) such release shall not in any manner discharge, affect or impair the Obligations or any Liens upon (or obligations of the Borrower or any of its Subsidiaries in respect of) all interests retained by the Borrower and/or any of its Subsidiaries, including (without limitation) the proceeds of any sale, all of which shall continue to constitute part of the Collateral; and
- (vii) against any cash collateral securing LC Obligations as contemplated by <u>Section 5.4</u> and <u>Article IX</u>, upon the written request of the Borrower to the Administrative Agent or the applicable Facing Agent, as the case may be, if at the time of the release of the Lien the Senior Secured Leverage Ratio, on a Pro Forma Basis, for the Test Period for the most recently ended Fiscal Quarter was not in excess of 3.75 to 1.00.
- (b) Each of the Lenders hereby directs the Administrative Agent to (or to cause the Administrative Agent to) execute and deliver or file such termination and partial release statements and such other things as are necessary to release Liens to be released pursuant to this Section 12.15 promptly upon the effectiveness of any such release or enter into intercreditor agreements contemplated or permitted herein.

- C. No Obligation. Neither the Administrative Agent nor the Collateral Agent nor the UK Security Trustee shall have any obligation whatsoever to any Lender or to any other Person to assure that the Collateral exists or is owned by the Borrower or any of its Subsidiaries or is cared for, protected or insured or has been encumbered or that the Liens granted to the Administrative Agent, the Collateral Agent or the UK Security Trustee herein or pursuant to the Loan Documents have been properly or sufficiently or lawfully created, perfected, protected or enforced or are entitled to any particular priority, or to exercise at all or in any particular manner or under any duty of care, disclosure or fidelity, or to continue exercising, any of the rights, authorities and powers granted or available to the Administrative Agent, the Collateral Agent or the UK Security Trustee in any of the Loan Documents, it being understood and agreed that in respect of the Collateral, or any act, omission or event related thereto, the Administrative Agent the Collateral Agent or the UK Security Trustee may act in any manner it may deem appropriate, in its sole discretion, given the Administrative Agent's, the Collateral Agent's and the UK Security Trustee's own interests in the Collateral as one of the Lenders and that neither the Administrative Agent nor the Collateral Agent nor the UK Security Trustee shall have any duty or liability whatsoever to any Lender, provided, that, notwithstanding the foregoing, the Administrative Agent, the Collateral Agent and the UK Security Trustee shall be responsible for their respective grossly negligent actions or actions constituting intentional misconduct.
- D. <u>Senior Secured Note Liens</u>. Each Lender hereby instructs Administrative Agent and the Collateral Agent to enter into the Collateral Security Agreement and the Intercreditor Agreement and the UK Security Trustee to enter into the UK Security Documents and the Intercreditor Agreement and such amendments or modifications thereto and to the other Security Documents consistent herewith and as Administrative Agent or the Collateral Agent or the UK Security Trustee reasonably determines to be necessary to cause the Liens on the Collateral securing the Obligations (other than the Capital Stock Collateral) to be *pari passu* with the Liens on the Senior Secured Note Obligations (as defined in the Intercreditor Agreement). Each Lender agrees that it shall not challenge or question in any proceeding the validity or enforceability of this <u>Section 12.15(d)</u> or any corresponding provisions with respect thereto in the Collateral Security Agreement or the Intercreditor Agreement.

20.16. Intentionally Omitted

20.17. Registry

The Borrower hereby designates the Administrative Agent to serve as the Borrower's agent, solely for purposes of this Section 12.17 to maintain a register (the "Register") on which it will record the Commitments from time to time of each of the Lenders, the Loans made by each of the Lenders and each repayment in respect of the principal amount of the Loans of each Lender. Failure to make any such recordation, or any error in such recordation shall not affect the Borrower's obligations in respect of such Loans. With respect to any Lender, the transfer of the Commitments of such Lender and the rights to the principal of, and interest on, any Loan made pursuant to such Commitments shall not be effective until such transfer is recorded on the Register maintained by the Administrative Agent with respect to ownership of such Commitments and Loans and prior to such recordation all amounts owing to the transferor with respect to such Commitments and Loans shall remain owing to the transferor. The registration of assignment or transfer of all or part of any Commitments and Loans shall be

166

recorded by the Administrative Agent on the Register only upon the acceptance by the Administrative Agent of a properly executed and delivered Assignment and Assumption Agreement pursuant to Section 12.8(c). Coincident with the delivery of such an Assignment and Assumption Agreement to the Administrative Agent for acceptance and registration of assignment or transfer of all or part of a Loan, or as soon thereafter as practicable, the assigning or transferor Lender shall surrender the Note evidencing such Loan, and thereupon one or more new Notes in the same aggregate principal amount then owing to such assignor or transferor Lender shall be issued to the assigning or transferor Lender and/or the new Lender. The Borrower agrees to indemnify the Administrative Agent from and against any and all losses, claims, damages and liabilities of whatsoever nature which may be imposed on, asserted against or incurred by the Administrative Agent in performing its duties under this Section 12.17.

20.18. Accounts Receivable Securitization

- A. By its execution of this Agreement, each Lender agrees, for the benefit of the holders from time to time of interests in trade receivables under the Permitted Accounts Receivables Securitization not to:
 - (a) challenge the "true sale" characterization of the sales and transfers of Accounts Receivables by the Borrower or any Participating Subsidiary to a Receivables Subsidiary pursuant to a Permitted Accounts Receivable Securitization;
 - (b) join in any proceeding in whole or in part to commence or consent to the commencement of a case against a Receivables Subsidiary under the Federal Bankruptcy Code or any other applicable bankruptcy, insolvency or similar federal or state law or file a petition seeking or consenting to reorganization or relief under any applicable federal or state law relating to bankruptcy, or seek or consent to the appointment of a receiver, liquidator, assignee, trustee, sequestrator (or other similar official) of a Receivables Subsidiary or any substantial part of its assets; or
 - (c) assert or consent to any attempt by any person to assert that a Receivables Subsidiary should be substantively consolidated with the Borrower or any other Subsidiary.
- B. By its execution of this Agreement, each Lender further authorizes the Administrative Agent, the Collateral Agent and the UK Security trustee, in each case, with the approval of the Administrative Agent, to enter into an intercreditor agreement

with the Persons providing a Permitted Accounts Receivables Securitization as long as the provisions of any such agreement are not materially more burdensome to the Lenders than are typical for like receivables transactions.

20.19. Certain Guarantee Obligations.

The Borrower hereby guarantees all obligations of each of its Subsidiaries (for so long as such Subsidiary remains a Subsidiary) under all Interest Rate Agreements and Other Hedging Agreements entered into by such Subsidiary with any Lender or any Affiliate of a

167

Lender (even if such Person subsequently ceases to be a Lender hereunder for any reason), which obligations are pursuant to the terms of such Interest Rate Agreements and Other Hedging Agreements expressly secured by the security interests granted under the Collateral Security Agreement. The provisions of Sections 4 through 9 of the Subsidiary Guaranty are hereby incorporated herein by reference mutatis mutandis as if all references to "Guarantor" and "Guaranteed Obligations" were references to the Borrower and the obligations guaranteed by this <u>Section 12.19</u>, respectively.

20.20. Redesignation of Unrestricted Subsidiaries

Any Unrestricted Subsidiary may be redesignated as a Subsidiary provided that (i) the Borrower shall have delivered to the Administrative Agent (not less than 30 days prior to the date the Borrower desires such redesignation to be effective) a notice signed by a Responsible Financial Officer identifying the Unrestricted Subsidiary to be so redesignated and providing such other information as the Administrative Agent may request, (ii) immediately before and immediately after the effectiveness of such redesignation, no Default or Event of Default exists or will exist (including, without limitation, the permissibility of any Investment, Indebtedness, Liens or other obligations existing at such Subsidiary) and, if the Unrestricted Subsidiary is a Foreign Subsidiary, the Borrower shall be in compliance with the provisions of Section 8.7(m) as if the designation of such Unrestricted Subsidiary as a Subsidiary were an Acquisition, (iii) Borrower has complied, to the extent applicable, with the provisions of Section 7.11 and the applicable Subsidiary, on the effective date of such redesignation is in compliance with the terms and conditions of all applicable Security Documents, (iv) such Unrestricted Subsidiary is the subsidiary of either the Borrower or a Subsidiary, (v) the Administrative Agent has received such other documents (including without limitation any additional security documents whether or not required by Section 7.11), instruments and opinions as it may reasonably request in connection with such redesignation, and all such instruments, documents and opinions shall be reasonably satisfactory in form and substance to the Administrative Agent and (vi) on the desired effective date of such redesignation, the Borrower shall deliver a certificate from a Responsible Officer confirming clauses (ii) through (v) above and that the representations and warranties contained in this Agreement and the other Loan Documents are true and correct in all material respects on the date of, and after giving effect to, such redesignation as though made on such date (except to the extent such representations and warranties are expressly made of a specified date in which event they shall be true as of such date). Effective at the time of delivery of the certificate required pursuant to clause (vi) above, the Unrestricted Subsidiary Investment Basket shall be increased by (A) if such Unrestricted Subsidiary was so designated after the Effective Date, the fair market value of such Subsidiary or (B) if such Unrestricted Subsidiary was so designated on the Effective Date, the lesser of (y) the aggregate amount of outstanding Investments made after the Effective Date by the Borrower or any Subsidiary in such Unrestricted Subsidiary or (z) the fair market value of such Unrestricted Subsidiary immediately prior to the effective date of such redesignation. The Borrower agrees that any merger or consolidation of any Unrestricted Subsidiary with or into Borrower or any Subsidiary shall be required to satisfy the conditions of this Section 12.20 prior to completing any such transaction.

168

20.21. Administrative Agent and Collateral Agent as Joint Creditors

Each of the Credit Parties and each Lender and Agent and the Collateral Agent agrees that each of the Administrative Agent and Collateral Agent shall be a joint and several creditor (in Dutch: *hoofdelijk schuldeiser*) (together with the relevant Lender or Agent) of the Obligations and Guaranteed Obligations owed to each Lender or Agent under or in connection with all Loan Documents and that accordingly each of the Administrative Agent and Collateral Agent will have its own independent right to demand payment and performance by the obligors of such obligations. However, any discharge of a Credit Party of any such obligation to the Administrative Agent, Collateral Agent or any other relevant Lender, Agent or creditor referred to above, shall, to the same extent, discharge such Credit Party vis-à-vis the others in respect of such obligation, and a Lender, Agent and the Administrative Agent and Collateral Agent shall not by virtue of this Section be entitled to pursue a Credit Party concurrently for the same obligation.

20.22. Amendment With Respect to Revolver Events of Default.

No amendment, modification or waiver of <u>Section 9.1</u>, any of the financial definitions used in determining compliance with <u>Section 9.1</u> or this <u>Section 12.22</u> or waiver of any Revolver Event of Default, may be effected without the consent of the Majority Revolving Facility Lenders.

20.23. Term C Dollar Lenders.

Each Term C Dollar Lender agrees that upon the earlier to occur of (x) the Term B Loan Maturity Date and (y) the date that all Term B Dollar Loans are prepaid or repaid in full (for any reason) (such earlier date, the "<u>Determination Date</u>"):

- (a) Without limitation of any of the terms of the Loan Documents or any provisions thereof, the Borrower shall have the right in its sole discretion at any time to refinance or replace all of the Facilities outstanding immediately prior to the Determination Date under this Agreement (other than the Term C Dollar Facility) (collectively, the "Refinanced Facility Debt") pursuant to separate credit documentation. In connection therewith:
 - (i) all collateral, guarantees and other credit support that secures, guarantees or otherwise supports the Term C Dollar Facility pursuant to the Loan Documents (including, without limitation, the Security Documents) shall be available (on a <u>pari passu</u> basis in payment and lien priority, and in any event on the same basis as the Facilities are afforded under the Loan Documents as of the Fourth Amendment Effective Date) to secure, guarantee or otherwise support any and all such Refinanced Facility Debt, together with any other senior secured indebtedness of the Borrower (including, without limitation, one or more revolving and/or term credit facilities), in an aggregate principal amount not to exceed the greater of (A) the principal amount of Refinanced Facility Debt outstanding on the Determination Date (or its equivalent in any other currency) and (B) \$2,100,000,000 (or its equivalent in any other currency) (such greater amount, the "Minimum Floor Amount"), as the Minimum Floor Amount may be

169

reduced after the Determination Date pursuant to the last sentence of the definition of "Permitted Refinancing Indebtedness"; and

- (ii) the Loan Documents shall be amended and other documents and agreements will be entered into (such amendments, documents and agreements solely requiring the signatures of the Administrative Agent and the Borrower to be effective) in order to effectuate the foregoing and to make any necessary conforming changes.
- (b) Notwithstanding anything to the contrary contained in Section 4.4(c)(i), no prepayment of proceeds from a Recovery Event shall be required pursuant to such Section to the extent that (x) no Event of Default or Unmatured Event of Default then exists and (y) the Borrower delivers a certificate to the Administrative Agent on or prior to such date stating that an amount equal to the proceeds of such Recovery Event is expected to be used to purchase assets used or to be used in the businesses referred to in Section 8.9 within 360 days following the date of receipt of such proceeds (which certificate shall set forth the estimates of the proceeds to be so expended); provided that (1) if all or any portion of such proceeds not so applied to such prepayment are not so used (or contractually committed to be used) within such 360 day period as provided above, such remaining portion shall be applied on the last day of the period or such earlier date as the Borrower is obligated to make an offer to purchase Senior Secured Notes (2010) due to such Recovery Event as a mandatory repayment of principal of outstanding Loans as provided in Section 4.4(c)(i) and (2) if all or any portion of such proceeds result from a Recovery Event involving Collateral owned by the Borrower or a Domestic Subsidiary (other than the Capital Stock of a Foreign Subsidiary), then such proceeds shall be required to be reinvested in assets located in the United States constituting Collateral (to the extent not used to repay Loans pursuant to Section 4.4(c)(i)).
- (c) <u>Article IX</u> shall no longer apply for any purpose under this Agreement (including, without limitation, for the purposes of <u>Section 2.1(a)(ii)</u>, <u>Section 7.2(b)</u>, <u>Section 8.7(j)</u> and <u>Section 8.7(m)</u>).
- (d) "Permitted Refinancing Indebtedness" shall mean, with respect to any Indebtedness, any Indebtedness refinancing, extending, renewing or refunding such Indebtedness; provided, however, that any such refinancing Indebtedness shall (i) be issued by the same obligor as the Indebtedness being so refinanced (or by Huntsman Corporation or a Parent Company) and be on terms, taken as a whole, not more restrictive than the terms of the documents governing the Indebtedness being so refinanced; (ii) if the Indebtedness being so refinanced is subordinated to the Obligations, be subordinated to the Obligations on substantially the same terms (or on terms at least as favorable to the Lenders) as Indebtedness being so refinanced; (iii) be in a principal amount (as determined as of the date of the incurrence of such refinancing Indebtedness in accordance with GAAP) not exceeding the principal amount of the Indebtedness being refinanced on such date plus any call premiums, prepayment fees, costs and expenses paid in connection with such refinancing; (iv) not have a Weighted Average Life to Maturity less than the Indebtedness being refinanced; (v) if the Indebtedness being refinanced is Public Notes, be unsecured Indebtedness maturing no earlier than the then latest Term Maturity Date; and (vi) be upon terms and subject to documentation which is in form and substance reasonably satisfactory in all material respects to the Administrative Agent. Notwithstanding

170

clauses (ii) and (v) above, after the Determination Date, any such Indebtedness refinancing, extending, renewing or refunding the Senior Subordinated Notes (2013), the Senior Subordinated Notes (2014), the Senior Subordinated Notes (2015), any similar senior subordinated notes, any senior unsecured notes and any senior secured notes may be senior second-lien notes, senior unsecured notes and, to the extent capacity exists pursuant to the Minimum Floor Amount, senior first-lien secured notes (provided that the Minimum Floor Amount shall be reduced after the Determination Date by the aggregate principal amount of any senior first-lien secured Indebtedness incurred pursuant to this sentence, unless such Permitted Refinancing Indebtedness refinances, extends, renews or refunds senior first-lien secured Indebtedness).

[signature pages follow]

Exhibit 2.l(c)

FORM OF SWING LINE LOAN PARTICIPATION CERTIFICATE

[Name of Revolving I	Lender]	
Dear Sir or Madam:		
supplemented or other Administrative Agent	rwise modified from time to time), among	Agreement, dated as of August 16, 2005 (as amended, amended and restated, g Huntsman International LLC, JPMorgan Chase Bank, N.A., as tions signatory thereto, the undersigned hereby acknowledges receipt from t in the following Swing Line Loan:
Date of Swing Line L	oan:	
Principal Amount of S	Swing Line Loan:	
		Very truly yours,
		JPMORGAN CHASE BANK, N.A.
		Ву:
		Name:
		Title:
(1) This an	nount is equal to such Revolving Lender	's Pro Rata Share of the Dollar Equivalent of the Swing Line Loan.

EXHIBIT D TO FIFTH AMENDMENT TO THE HUNTSMAN INTERNATIONAL LLC CREDIT AGREEMENT

Exhibit 2.2(a)(2)

FORM OF REVOLVING NOTE

New York, New York

FOR VALUE RECEIVED, the undersigned, HUNTSMAN INTERNATIONAL LLC, a Delaware limited liability company ("Borrower"), hereby unconditionally promises to pay to the order of (the "Lender") at the office of JPMorgan Chase Bank, N.A., located at 1111 Fannin, 10th Floor, Houston, Texas 77002, in the Applicable Currency in which such Revolving Loan was made, in funds customary for the settlement of international transactions in such Applicable Currency and in immediately available funds, the Dollar Equivalent principal amount of (a) (\$), or, if less, (b) the aggregate unpaid principal amount of all Revolving Loans made by the Lender to Borrower pursuant to Section 2.1(b) of the Credit Agreement hereinafter referred to. The principal amount of each Revolving Loan evidenced hereby shall be payable as set forth in the Credit Agreement hereinafter referred to, with any then outstanding principal amount of the Revolving Loans made by the Lender being payable on the Revolver Termination Date. Borrower further agrees to pay interest in like money at such office on the unpaid principal amount hereof from time to time outstanding at the applicable interest rate per annum determined as provided in, and payable as specified in, Articles III and IV of the Credit Agreement.

The holder of this Revolving Note is authorized to record the date, Type, currency and amount of each Revolving Loan made by the Lender pursuant to Section 2.1 of the Credit Agreement, each continuation thereof, the date of each interest rate conversion pursuant to Section 2.6 of the Credit Agreement and the Assigned Dollar Value of the principal amount subject thereto, the date and

amount of each payment or prepayment of principal hereof, and in the case of each Eurocurrency Loan, the length of the Interest Period with respect thereto on the records of the Lender, and any such recordation shall (in the absence of manifest error) constitute <u>prima facie</u> evidence of the accuracy of the information recorded; provided, however, that the failure to make any such recordation shall not affect the obligations of Borrower in respect of such Revolving Loan.

This Revolving Note is one of the Revolving Notes referred to in the Credit Agreement dated as of August 16, 2005, as amended through the Fifth Amendment (and as further amended, supplemented or otherwise modified from time to time, the "Credit Agreement"), among the Borrower, JPMorgan Chase Bank, N.A., as a Lender and Administrative Agent for the Lenders, and the financial institutions signatory thereto, and is subject to the provisions thereof, and is subject to optional and mandatory prepayment in whole or in part as provided therein. Terms defined in the Credit Agreement are used herein with their defined meanings unless otherwise defined herein.

Upon the occurrence and during the continuance of anyone or more of the Events of Default specified in the Credit Agreement, all amounts then remaining unpaid on this Revolving Note may become, or may be declared to be, immediately due and payable, all as provided therein.

1

All parties now and hereafter liable with respect to this Revolving Note, whether maker, principal, surety, guarantor, endorser or otherwise, hereby waive presentment, demand, protest and all other notices of any kind.

THIS REVOLVING NOTE SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE INTERNAL LAWS OF THE STATE OF NEW YORK.

By:		
Name:		
Title:		
2		

HUNTSMAN INTERNATIONAL LLC

EXHIBIT E TO FIFTH AMENDMENT TO THE HUNTSMAN INTERNATIONAL LLC CREDIT AGREEMENT

Exhibit 2.2(a)(3)

FORM OF SWING LINE NOTE

New York, New York, 20

FOR VALUE RECEIVED, the undersigned, Huntsman International LLC, a Delaware limited liability company ("Borrower"), unconditionally promises to pay to the order of JPMorgan Chase Bank, N.A. ("Swing Line Lender") or its registered assigns, at the office of Swing Line Lender, located at 1111 Fannin, 10th Floor, Houston, Texas 77002 or such other office as Swing Line Lender may request, in the Applicable Currency in which such Swing Line Loan was made, in funds customary for the settlement of international transactions in such Applicable Currency and in immediately available funds, the Dollar Equivalent principal amount of the aggregate unpaid principal amount of all Swing Line Loans made by Swing Line Lender to the undersigned pursuant to Section 2.1(c) of the Credit Agreement. The principal amount of each Swing Line Loan evidenced hereby shall be payable as set forth in the Credit Agreement hereinafter referred to, with any outstanding principal amount of the Swing Line Loans made by Swing Line Lender being payable on the Revolver Termination Date. Borrower further agrees to pay interest on the unpaid principal amount hereof in like money from time to time from the date hereof at the rates and on the dates specified in Articles III and IV of the Credit Agreement. Swing Line Lender, and any such recordation shall (in the absence of manifest error) constitute prima facie evidence of the accuracy of the information so recorded; provided, however, that the failure of Swing Line Lender to make such recordation (or any error in such recordation) shall not affect the obligations of Borrower hereunder or under the Credit Agreement.

This Swing Line Note is a Swing Line Note referred to in the Credit Agreement dated as of August 16, 2005, as amended through the Fifth Amendment (and as further amended, supplemented or otherwise modified from time to time, the "Credit Agreement"), among Borrower, JPMorgan Chase Bank, N.A., as Administrative Agent for the Lenders, and the financial institutions signatory thereto, and is subject to the provisions thereof, and is subject to optional and mandatory prepayment in whole or in part as

provided therein. Terms defined in the Credit Agreement are used herein with their defined meanings unless otherwise defined herein.

Upon the occurrence and during the continuance of anyone or more of the Events of Default specified in the Credit Agreement all amounts then remaining unpaid on this Swing Line Note may become, or may be declared to be, immediately due and payable as provided in the Credit Agreement.

All parties now and hereafter liable with respect to this Swing Line Note, whether maker, principal, surety, guarantor, endorser or otherwise, hereby waive presentment, demand, protest and all other notices of any kind.

1

THIS SWING LINE NOTE SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE INTERNAL LAWS OF THE STATE OF NEW YORK.

HUNTSMAN INTERNATIONAL LLC

By:

Name:

Title:

EXHIBIT F TO FIFTH AMENDMENT TO THE HUNTSMAN INTERNATIONAL LLC CREDIT AGREEMENT

Exhibit 2.5

FORM OF NOTICE OF BORROWING(2)

Date:

JPMorgan Chase Bank, N.A. 1111 Fannin, 10th Floor Houston, Texas 77002 as Administrative Agent

Attention: Monica M Espitia Telecopy: 713-427-6307

Dear Sir or Madam:

Reference is made to that certain Credit Agreement, dated as of August 16, 2005, as amended through the Fifth Amendment (and as further amended, supplemented or otherwise modified from time to time, the "Credit Agreement"), among Huntsman International LLC, a Delaware limited liability company (the "Borrower"), JPMorgan Chase Bank, N.A., as Administrative Agent for the Lenders, and the financial institutions signatory thereto. Capitalized terms used herein and not otherwise defined herein shall have the meanings assigned to such terms in the Credit Agreement. The undersigned hereby gives notice pursuant to Section 2.5 of the Credit Agreement of its request for the Lenders to make a Borrowing as follows:

1. Aggregate Principal Amount to be Borrowed(3)

Such irrevocable written notice must be given (x) not later than 1:00 p.m., New York City time at the Notice Office, (A) at least three Business Days prior to the requested borrowing date, if all or any part of the requested Revolving Loans are to be Dollar-denominated Eurocurrency Loans or (B) at least four Business days prior to the requested borrowing date, if all or any part of the requested Revolving Loans are to be Eurocurrency Loans denominated in an Alternative Currency; (y) not later than 1:00 p.m., New York City time, at least one Business Day prior to the requested borrowing date, with respect to Borrowings of Base Rate Loans; and (z) no later than 1:00 p.m., New York City time, (A) on the Business Day of the requested borrowing date, with respect to Dollar-denominated Swing Line Loans or (B) at least one Business Day prior to the requested borrowing date, with respect to Swing Line Loans denominated in an Alternative Currency.

Dollars (\$3,000,000) and, if greater, in integral multiples of One Million Dollars (\$1,000,000) (or, if less, the then Total Available Revolving Commitment); (y) in the case of Eurocurrency Loans, Five Million Dollars (\$5,000,000), in the case of a Borrowing in Dollars, Two Million Pounds (£2,000,000), in the case of a Borrowing in Sterling, and Five Million Euros (€5,000,000), in the case of a Borrowing in Euros and, if greater, in integral multiples (i) in the case of a Borrowing in Dollars, One Million Dollars (\$1,000,000), (ii) in the case of a Borrowing in Sterling, Seven Hundred Fifty Thousand Pounds (£750,000) or (iii) in the case of a Borrowing in

1

			1		
	2.	Borrowing Date			
	3.	Type of Loan or combination thereo	of(4)		
	4.	If Borrowing is to include Eurocurr	ency Loans indicate	:	Eurocurrency Loan
		Amount			\$
		Initial Interest Period			
	5.	If Borrowing is to include Revolvin	g Loans indicate:		
		Overdraft Reserve Amoun	t		\$
The undersigned represents and warrants that the borrowing requested hereby complies with the requirements of Section 2.1, if applicable, and Section 5.2 of the Credit Agreement.					
			HUNTSMAN INT	ERNATIONAL	LLC
			By:		
			Name:		
			Title:		

Euros, One Million Euros (€1,000,000) in excess thereof (or, if less, the then Total Available Revolving Commitment); and (z) in the case of Swing Line Loans, Five Hundred Thousand Dollars (\$500,000) (or the Dollar Equivalent thereof in an Alternative Currency) and in any amount greater than such amount (or, if less, the then Total Available Revolving Commitment or Swing Line Commitment).

(4) Specify whether Borrowing is to be Eurocurrency Loans, Base Rate Loans, a combination thereof, or a Swing Line Loan at the applicable Interest Rate.

2

EXHIBIT G TO FIFTH AMENDMENT TO THE HUNTSMAN INTERNATIONAL LLC CREDIT AGREEMENT

Exhibit 2.6

FORM OF NOTICE OF CONVERSION OR CONTINUATION(5)

JPMorgan Chase Bank, N.A. 1111 Fannin, 10th Floor Houston, Texas 77002

Attention: Monica M. Espitia Telecopy: 713-427-6307

Dear Sir or Madam:

Reference is made to that certain Credit Agreement, dated as of August 16, 2005, as amended through the Fifth Amendment (and as further amended, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>"), among Huntsman International LLC, a Delaware limited liability company (the "<u>Borrower</u>"), JPMorgan Chase Bank, N.A., as the administrative agent (the "<u>Administrative Agent</u>"), and the financial institutions signatory thereto. Capitalized terms used herein and not otherwise defined herein shall have the meanings assigned to such terms in the Credit Agreement. The undersigned hereby gives notice pursuant to Section 2.6 of

the Credit Agreement that it (a) [elects to convert Base Rate Loans into Eurocurrency Loans]; (b) [elects to convert Eurocurrency Loans into Base Rate Loans]; or (c) [elects to continue Eurocurrency Loans for an additional Interest Period] under the Credit Agreement, and sets forth below the terms on which such [conversion] [continuation] is requested to be made:

[Date of conversion or continuation (which is a Business Day)(6)]

This notice must be received by Agent at its Notice Office not later than 1:00 p.m. (New York City time) at least (i) three Business Days in advance of the conversion date or continuation date, if the Loans are to be converted into or continued as Eurocurrency Loans, (ii) one Business Day in advance of the conversion date, if the Loans are to be converted into Base Rate Loans or (iii) four Business Days in advance of the continuation date, if the Loans are to be continued are non-Dollar denominated Revolving Loans.

No (i) conversion in whole or in part of Base Rate Loans to Eurocurrency Loans, and (ii) no continuation in whole or in part of Dollar denominated Eurocurrency Loans shall be permitted at any time at which an Unmatured Event of Default or an Event of Default shall have occurred and be continuing.

(6) If it is a conversion from Eurocurrency Loans, the date must be the last date of the Interest Period.

1

Aggregate Amount of Eurocurrency Loans or Base Rate Loans to be converted or continued(7)

[Interest Period (if the Loans are to be converted into or continued as Eurocurrency Loans)](8)

Very truly yours,

HUNTSMAN INTERNATIONAL LLC

By:		
Name:		
Title:		

(7) For Eurocurrency Loans, the aggregate principal amount of each Borrowing by the Borrower hereunder shall be not less than: \$5,000,000, in the case of a Borrowing in Dollars, £2,000,000, in the case of a Borrowing in Sterling, and €5,000,000, in the case of a Borrowing in Euros and, if greater, shall be in integral multiples of (i) in the case of a Borrowing in Dollars, \$1,000,000, (ii) in the case of a Borrowing in Sterling, £750,000, or (iii) in the case of a Borrowing in Euros, €1,000,000, above such minimum (or, if less, the then Total Available Revolving Commitment).

(8) Which shall be subject to the definition of "Interest Period" set forth in the Credit Agreement and shall, in any event, end on or before (i) in case of a Term Loan, the Term Maturity Date and (ii) in case of a Revolving Loan, the Revolver Termination Date. At any time when an Unmatured Event of Default or an Event of Default has occurred and is continuing, no Interest Period of more than one month may be selected with respect to any non-Dollar denominated Revolving Loan.

2

EXHIBIT H TO FIFTH AMENDMENT TO THE HUNTSMAN INTERNATIONAL LLC CREDIT AGREEMENT

Exhibit 2.9(b)

FORM OF NOTICE OF ISSUANCE

[Huntsman International LLC Letterhead]

Dated (9)

TO: [The applicable Facing Agent](10)

the Fifth Amendment (and as further amended, modified or sup	rative Agent for the Lenders under the Credit Agreement, amended through oplemented from time to time, the " <u>Credit Agreement</u> "), dated as of <u>orrower</u> "), JPMorgan Chase Bank, N.A., as Administrative Agent for the me to time.
JPMorgan Chase Bank, N.A. 1111 Fannin, 10th Floor Houston, Texas 77002	
Dear Sirs:	
We hereby request that [Name of Proposed F Guarantee] for the account of the undersigned and the followin of Issuance") in the aggregate stated amount of (11) in the fo	
For purposes of this Notice of Issuance, unle respective meanings provided in the Credit Agreement.	ss otherwise defined herein, all capitalized terms used herein shall have the
The beneficiary of the requested Letter of Cr following terms and conditions (13) and will have a st	edit will be (12), and such Letter of Credit will include the ated expiration date of (14).
least 3 Business Days prior written notice before the proposed	Business Day. The Administrative Agent and Facing Agent must receive at Date of Issuance, unless a shorter period is agreed to between the The Notice of Issuance must be received before 12:00 p.m. (New York
(10) If the Facing Agent is JPMorgan Chase Bank, N	I.A., the address is 1111 Fannin, 10th Floor, Houston, Texas 77002.
(11) Stated Amount of Letter of Credit.	
(12) Insert name and address of beneficiary.	
	1
We hereby certify that:	
respects on the Date of Issuance, both before and imm	contained in the Loan Documents will be true and correct in all material nediately after giving effect to the issuance of the Letter of Credit requested sentation or warranty which by its terms is made as of a specified date shall cts only as of such specified date).
	Event of Default has occurred and is continuing nor, after giving effect to will such a Event of Default or Unmatured Event of Default occur.
Copies of all documentation with respect to t	he supported transaction are attached hereto.
	HUNTSMAN INTERNATIONAL LLC
	By:
	Name:
	Title:
	[SUBSIDIARY ACCOUNT]
	By:

Name:

Title:

⁽¹³⁾ Insert description of the terms and conditions of the Letter of Credit.

(14) Insert last date upon which drafts may be presented. All Letters of Credit shall have an expiration date of twelve (12) months or less from the Date of Issuance. No Standby Letter of Credit or extension shall expire later than the date five (5) days prior to the Revolver Termination Date and no Commercial Letter of Credit or extension shall expire later than the day thirty (30) days prior to the Revolver Termination Date.

2

EXHIBIT I TO FIFTH AMENDMENT TO THE HUNTSMAN INTERNATIONAL LLC CREDIT AGREEMENT

Exhibit 12.8(c)

FORM OF ASSIGNMENT AND ASSUMPTION AGREEMENT

This Assignment and Assumption Agreement ("<u>Assignment</u>") is dated as of the Effective Date set forth below and is entered into by and between the Assignor identified in item 1 below ("<u>Assignor</u>") and the Assignee identified in item 2 below ("<u>Assignee</u>"). Capitalized terms used but not defined herein shall have the meanings given to them in the Credit Agreement identified below (as amended, restated, supplemented or otherwise modified from time to time the "<u>Credit Agreement</u>"), receipt of a copy of which is hereby acknowledged by the Assignee. The Standard Terms and Conditions set forth in Annex 1 to <u>Exhibit 12.8(c)</u> to the Credit Agreement are hereby agreed to and incorporated herein by reference and made a part of this Assignment as if set forth herein in full.

For an agreed consideration, Assignor hereby irrevocably sells and assigns to Assignee, and Assignee hereby irrevocably purchases and assumes from Assignor, subject to and in accordance with the Standard Terms and Conditions and the Credit Agreement, as of the Effective Date inserted by Administrative Agent as contemplated below, the interest in and to all of Assignor's rights and obligations under the Credit Agreement and any other documents or instruments delivered pursuant thereto that represents the amount and percentage interest identified below of all of Assignor's outstanding rights and obligations under the respective facilities identified below (including, to the extent included in any such facilities, Letters of Credit) (the "Assigned Interest"). Such sale and assignment is without recourse to Assignor and, except as expressly provided in this Assignment, without representation or warranty by Assignor.

below (including, to the extent in	cluded in any such fac	nor's outstanding rights and cilities, Letters of Credit) (th provided in this Assignment,	e "Assigned Interest"). Sue	ch sale and assignment is
1.	Assignor:				
2.	Assignee:				
3.	Credit Agreement: The Credit Agreement dated as of August 16, 2005, as amended through the Fifth Amendment, among Huntsman International LLC (the "Borrower") the financial institutions signatory thereto, and JPMorgan Chase Bank, N.A., as Administrative Agent.				
			1		
4.	Assigned Interest:				
	Assigned		Aggregate Amount of Commitment/Loans for all Lenders	Amount of Commitment/Loans Assigned	Percentage Assigned of Commitment/Loans(15)
[Revolv	ving Commitment/ Term C	Commitment]	\$	\$	\$
(15)	Set forth, to at least 9 dec	imals, as a percentage	of the Commitment/Loans/	LC Obligations of all Lend	ers thereunder.
Effectiv	ve Date: , 2	20			
	The terms set for	orth in this Assignmen	nt are hereby agreed to:		
	ASSIGNOR ASSIGNEE [NAME OF ASSIGNOR] [NAME OF ASSIGNEE]				
Ву:			Ву:		
	ame: itle:		Name Title:		

	Payment Instructions:
	Attention: Reference:
	Address for Notices:
	Relationship Contact: Telephone:
	Fax:
Consented to and](16)	
PMorgan Chase Bank, N.A., as Administrative Agent	
By: Jame:	
itle:	
(16) To be added only if the consent of the Administrative A	Agent is required by the terms of <u>Section 12.8(c)</u> of the Credit Agreement.
	3
Consented to:](17)	
IUNTSMAN INTERNATIONAL LLC	
By: Jame: Title:	
(17) To be added only in the circumstances (if any) set forth	h in Section 12.8(c) of the Credit Agreement.
	4
	

ANNEX FOR ASSIGNMENT AND ASSUMPTION AGREEMENT

ANNEX I

HUNTSMAN INTERNATIONAL LLC

CREDIT AGREEMENT

STANDARD TERMS AND CONDITIONS FOR ASSIGNMENT AND ASSUMPTION AGREEMENT

1. <u>Representations and Warranties</u>.

Assignor. [Each] [The] Assignor (a) represents and warrants that (i) it is the legal and beneficial owner of the Assigned Interest, (ii) the Assigned Interest is free and clear of any lien, encumbrance or other adverse claim and (iii) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and to consummate the transactions contemplated hereby; and (b) assumes no responsibility with respect to (i) any statements, warranties or representations made in or in connection with any Loan Document, (ii) the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Credit Agreement, any other Loan Document or any other instrument or document delivered pursuant thereto, other than this Assignment, or any collateral thereunder, (iii) the financial condition of the Borrower or any of its Subsidiaries or Affiliates or any other Person obligated in respect of any Loan Document or (iv) the performance or observance by the Borrower or any of its Subsidiaries or Affiliates or any other Person of any of their respective obligations under any Loan Documents.

Assignee. [Each] [The] Assignee (a) represents and warrants that (i) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and to consummate the transactions contemplated hereby and to become a Lender under the Credit Agreement, (ii) it meets all requirements of an Eligible Assignee under the Credit Agreement, (iii) from and after the Effective Date, it shall be bound by the provisions of the Credit Agreement and, to the extent of the Assigned Interest, shall have the obligations of a Lender thereunder, (iv) it has received a copy of the Credit Agreement, together with copies of the most recent financial statements delivered pursuant to Section 7.1 thereof, as applicable, and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into this Assignment and to purchase the Assigned Interest on the basis of which it has made such analysis and decision and (v) has sent to Borrower if required to be delivered to Borrower or attached to this Assignment if required to be delivered to Administrative Agent any documentation required to be delivered by it to Borrower and/or Administrative Agent pursuant to the terms of the Credit Agreement, duly completed and executed by [the] [each such] Assignee; and (b) agrees that (i) it will, independently and without reliance on the Administrative Agent, [the] [each such] Assignor or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Loan Documents, (ii) appoints and authorizes each of the Administrative Agent and the Collateral Agent to take such action as agent on its behalf and to exercise such powers under the Credit Agreement and the other Loan Documents as are delegated to or otherwise conferred upon the Administrative Agent or the Collateral Agent, as the case may be, by the terms thereof, together with such powers as are reasonably incidental thereto; and (iii) it will perform in accordance with their terms all of the obligations which by the terms of the Loan Documents are required to be performed by it as a Lender.

5

- 2. <u>Payment.</u> Subject to the terms of the Credit Agreement, from and after the Effective Date, the Administrative Agent shall make all payment in respect to the Assigned Interest (including payments of principal, interest, fees and other amounts) to **[the]** [each such] Assignor for amounts which have accrued to but excluding the Effective Date and to [the] [each] Assignee for amounts which have accrued from and after the Effective Date.
- 3. <u>General Provisions</u>. This Assignment shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and assigns. This Assignment may be executed in any number of counterparts, which together shall constitute one instrument. Delivery of an executed counterpart of a signature page of this Assignment by telecopy shall be effective as delivery of a manually executed counterpart of the Assignment. THIS ASSIGNMENT SHALL BE DEEMED TO BE A CONTRACT MADE UNDER THE LAWS OF THE STATE OF NEW YORK, AND FOR ALL PURPOSES SHALL BE CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS AND DECISIONS OF SAID STATE, INCLUDING SECTIONS 5-1401 AND 5-1402 OF THE NEW YORK GENERAL OBLIGATIONS LAW BUT EXCLUDING ALL OTHER CHOICE OF LAW AND CONFLICTS OF LAWS RULES.

6

ANNEX A TO FIFTH AMENDMENT TO THE HUNTSMAN INTERNATIONAL LLC CREDIT AGREEMENT

EXTENDING REVOLVING LENDER CONSENT

Reference is made to (i) the Credit Agreement, dated as of August 16, 2005 (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement"), entered into by and among Huntsman International LLC, a Delaware limited liability company (the "Borrower"), Deutsche Bank AG New York Branch, as administrative agent (the "Existing Agent") and the Lenders party thereto and (ii) the Fifth Amendment to the Credit Agreement (the "Amendment") to which this Extending Revolving Lender Consent is attached. Unless otherwise defined herein, capitalized terms used herein and defined in the Credit Agreement or the Amendment, as applicable, are used herein as therein defined.

Pursuant to <u>Section 12.1(a)</u> of the Credit Agreement, the undersigned Revolving Lender hereby (i) waives, solely for the purpose of the Amendment, (A) the requirement that pursuant to <u>Section 4.1(a)</u> of the Credit Agreement each reduction or adjustment of the Revolving Commitments shall apply proportionately to the Revolving Commitments of each Revolving Lender and (B) the notice requirement pursuant to <u>Section 4.1(a)</u> of the Credit Agreement with respect to a voluntary reduction in Revolving Commitments and (ii) consents to the extension of the Revolving Termination Date to March 9, 2014 (as defined in the Amended Credit Agreement) with respect to the amount of its Revolving Commitment specified below:

Amount of Revolving Commitment to be extended as Revolving Commitments under the Amended Credit Agreement

\$

Consented to and agreed as of the Amendment Effective Date:

Name:			
Title:			

ANNEX B
FIFTH AMENDMENT TO THE
HUNTSMAN INTERNATIONAL LLC
CREDIT AGREEMENT

CONSENT AND REAFFIRMATION

Reference is made to (i) the Credit Agreement, dated as of August 16, 2005 (as heretofore amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement"), entered into by and among Huntsman International LLC, a Delaware limited liability company (the "Borrower"), Deutsche Bank AG New York Branch, as Administrative Agent, the Agents party thereto and the Lenders party thereto, (ii) the Fifth Amendment to Credit Agreement (the "Amendment") dated as of even date herewith among the Borrower and the Lenders party thereto and (iii) the Successor Agency Agreement dated as of even date herewith (the "Successor Agency Agreement") among Deutsche Bank AG New York Branch in the capacities therein stated, JPMorgan Chase Bank, N.A. (the "Successor Administrative Agent"), JPMorgan Chase Bank, N.A. (the "Successor Collateral Agent"), JPMorgan Chase Bank, N.A. or an affiliate thereof designated by JPMorgan Chase Bank, N.A. (the "Successor UK Security Trustee"), and the Borrower (the Amendment and the Successor Agency Agreement being, collectively, the "Fifth Amendment Documents"). Unless otherwise defined herein, capitalized terms used herein and defined in the Credit Agreement or the Amendment, as applicable, are used herein as therein defined.

- 1. Each of the undersigned hereby (a) acknowledges receipt of a copy of each Fifth Amendment Document, (b) consents to and approves the execution, delivery and performance of the Fifth Amendment Documents and the performance of the Credit Agreement as amended by the Fifth Amendment.
- 2. After giving effect to each Fifth Amendment Document and the amendments and modifications to the Loan Documents (including, without limitation, waivers of provisions of any Loan Documents) effectuated by the Fifth Amendment Documents (collectively, the "Modifications"), each of the undersigned ratifies, reaffirms and agrees to perform all of its obligations under each Loan Document to which it is a party (whether as original signatory thereto, by supplement thereto, by operation of law or otherwise), and agrees that all such obligations remain in full force and effect including, without limitation, all of its obligations under each of the following Loan Documents to which it is a party:
 - (a) the Amended Credit Agreement;
 - (b) each Note;
 - (c) each Security Document, including without limitation: (i) the Collateral Security Agreement dated as of August 16, 2005, as heretofore amended, modified or supplemented (including by Supplement No. 1 to Collateral Security Agreement dated as of December 20, 2005), (ii) the Pledge Agreement dated as of August 16, 2005, as heretofore amended, modified or supplemented (including by Supplement No. 1 to Pledge Agreement dated as of December 20, 2005), (iii) the UK Pledge Agreements, (iv) the UK Debenture, and (v) the Mortgages;
 - (d) each Guaranty, including, without limitation, the Subsidiary Guaranty dated as of August 16, 2005, as heretofore amended, modified or supplemented (including by Supplement No. 1 to Subsidiary Guaranty dated as December 20, 2005, Supplement No. 2 to Subsidiary Guaranty dated as of December 20, 2005, and Supplement No. 3 to Subsidiary Guaranty dated as of December 20, 2005); and
- 3. After giving effect to each Fifth Amendment Document and the Modifications effectuated thereby, each of the undersigned, with respect to each Security Document to which it is a party (a)

1

reaffirms and ratifies the Liens granted by the undersigned under such Security Document, (b) confirms and acknowledges that the Liens granted by the undersigned under such Security Document remain in full force and effect, and, (c) without limitation to the foregoing, confirms and acknowledges that:

- (i) such Liens, if granted to or in favor of Deutsche Bank AG New York Branch as Collateral Agent, shall continue without interruption in favor of the Successor Collateral Agent for the benefit of the persons secured by such Security Document (whether defined in such Security Document as "Secured Parties" or otherwise) to secure the obligations secured by such Security Document (whether defined in such Security Document as "Secured Obligations," "Secured Liabilities" or otherwise); and
- (ii) such Liens, if granted to or in favor of Deutsche Bank AG New York Branch as UK Security Trustee, shall continue without interruption in favor of the Successor UK Security Trustee for the benefit of the persons secured by such Security Document (whether defined in such Security Document as "Secured Parties" or otherwise) to secure the obligations secured by such Security Document (whether defined in such Security Document as "Secured Obligations," "Secured Liabilities" or otherwise).

- 4. After giving effect to each Fifth Amendment Document and the Modifications effectuated thereby, each of the undersigned agrees that from and after the Amendment Effective Date, (a) each reference to the Credit Agreement in the Loan Documents shall be deemed to be a reference to the Amended Credit Agreement, (b) each reference in the Loan Documents to the Administrative Agent shall be deemed a reference to the Successor Administrative Agent (except as expressly provided in the Successor Agency Agreement), (c) each reference in the Loan Documents to the Collateral Agent shall be deemed a reference to the Successor Collateral Agent (except as expressly provided in the Successor Agency Agreement), and (d) each reference in the Loan Documents to the UK Security Trustee shall be deemed a reference to the Successor Agency Agreement).
- 5. Each of the undersigned agrees that this Consent and Reaffirmation is made for the benefit of the Successor Administrative Agent, the Successor Collateral Agent, the Successor UK Security Trustee, the Lenders from time to time party to the Credit Agreement and the other persons secured by any of the Security Documents (whether defined in such Security Documents as "Secured Parties" or otherwise).
- 6. EACH OF THE UNDERSIGNED AGREES THAT THIS CONSENT AND REAFFIRMATION SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK.

[SIGNATURE PAGES FOLLOW]

2

IN WITNESS WHEREOF, the undersigned has caused this Consent and Reaffirmation to be duly executed and delivered as of March 9, 2010.

3

HUNTSMAN INTERNATIONAL LLC
Зу:
Name: Title:
EXECUTED as a deed by FIOXIDE AMERICAS INC.
By:
Name: Fitle:
Witnessed by:
Executed and delivered as a deed on behalf of TIOXIDE GROUP acting by:
Name:
Name:

AIRSTAR CORPORATION

HUNTSMAN ADVANCED MATERIALS AMERICAS LLC

HUNTSMAN ADVANCED MATERIALS LLC

HUNTSMAN AUSTRALIA INC.

HUNTSMAN CHEMICAL PURCHASING CORPORATION

HUNTSMAN ENTERPRISES, INC.

HUNTSMAN ETHYLENEAMINES LLC

HUNTSMAN FUELS LLC

HUNTSMAN INTERNATIONAL FINANCIAL LLC

HUNTSMAN INTERNATIONAL FUELS LLC

HUNTSMAN INTERNATIONAL TRADING CORPORATION

HUNTSMAN MA INVESTMENT CORPORATION

HUNTSMAN MA SERVICES CORPORATION

HUNTSMAN PETROCHEMICAL LLC

HUNTSMAN PETROCHEMICAL PURCHASING CORPORATION

HUNTSMAN PROCUREMENT CORPORATION

HUNTSMAN PROPYLENE OXIDE LLC

110111011111	(1010111011(0, 212)	
By:	Huntsman Procurement Corporation, its General Partne	er e
POLYMER N	MATERIALS INC.	
Ву:		
Name:		
Title:		
	4	
	4	

HUNTSMAN PURCHASING, LTD.

CERTIFICATION

I, Peter R. Huntsman, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Huntsman Corporation and Huntsman International LLC;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrants as of, and for, the periods presented in this report;
- 4. The registrants' other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrants, including their consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrants' disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrants' internal control over financial reporting that occurred during the registrants' most recent fiscal quarter (the registrants' fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrants' internal control over financial reporting; and
- 5. The registrants' other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrants' auditors and the audit committee of registrants' board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrants' ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrants' internal control over financial reporting.

Date: May 7, 2010

/s/ PETER R. HUNTSMAN

Peter R. Huntsman Chief Executive Officer QuickLinks

Exhibit 31.1

CERTIFICATION

CERTIFICATION

I, J. Kimo Esplin, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Huntsman Corporation and Huntsman International LLC;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrants as of, and for, the periods presented in this report;
- 4. The registrants' other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrants, including their consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrants' disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrants' internal control over financial reporting that occurred during the registrants' most recent fiscal quarter (the registrants' fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrants' internal control over financial reporting; and
- 5. The registrants' other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrants' auditors and the audit committee of registrants' board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrants' ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrants' internal control over financial reporting.

Date: May 7, 2010

/s/ J. KIMO ESPLIN

J. Kimo Esplin Chief Financial Officer QuickLinks

Exhibit 31.2

CERTIFICATION

Exhibit 32.1

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report on Form 10-Q of Huntsman Corporation and Huntsman International LLC (the "Companies") for the period ended March 31, 2010, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Peter R. Huntsman, Chief Executive Officer of the Companies, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that, to my knowledge:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Companies.

/s/ PETER R. HUNTSMAN

Peter R. Huntsman

Chief Executive Officer

May 7, 2010

QuickLinks

Exhibit 32.1

 $\frac{\text{CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002}{\text{CONTRACT OF 2002}}$

Exhibit 32.2

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report on Form 10-Q of Huntsman Corporation and Huntsman International LLC (the "Companies") for the period ended March 31, 2010, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, J. Kimo Esplin, Chief Financial Officer of the Companies, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that, to my knowledge:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Companies.

/s/ J. KIMO ESPLIN

J. Kimo Esplin Chief Financial Officer

May 7, 2010

QuickLinks

Exhibit 32.2

 $\frac{\text{CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002}{\text{CONTRACT OF 2002}}$